

# Glyphosate -Human Health and Ecological Risk Assessment Final Report

Prepared for:

# **USDA**, Forest Service



**Forest Health Protection** 

GSA Contract No. **GS-10F-0082F** USDA Forest Service BPA: **WO-01-3187-0150** USDA Purchase Order No.: **43-1387-2-0238** Task No. **9** 

Submitted to: **Dave Thomas**, COTR Forest Health Protection Staff USDA Forest Service Rosslyn Plaza Building C, Room 7129C 1601 North Kent Street Arlington, VA 22209



Submitted by: Patrick R. Durkin Syracuse Environmental Research Associates, Inc. 5100 Highbridge St., 42C Fayetteville, New York 13066-0950 Telephone: (315) 637-9560 Fax: (315) 637-0445 E-Mail: <u>SERA\_INC@msn.com</u> Home Page: www.sera-inc.com

March 1, 2003

# TABLE OF CONTENTS

LI	ST OF WORK	SHEETS v
LI	ST OF ATTA	CHMENTS v
LI	ST OF TABLI	ES v
LI	ST OF FIGUR	vi
AC	CRONYMS, A	BBREVIATIONS, AND SYMBOLS vii
CC	OMMON UNI	T CONVERSIONS AND ABBREVIATIONS ix
CC	ONVERSION	OF SCIENTIFIC NOTATION x
EX	ECUTIVE SU	JMMARY xi
1.	INTRODUC	TION
2.	PROGRAM 2.1 OVERV	DESCRIPTION
	2.1. OVERV	CAL DESCRIPTION AND COMMERCIAL FORMULATIONS 2-2
	2.2. OTEM	ATION METHODS 2-3
	2.3. HITER 2.4 LISES A	ND APPI ICATION RATES 2-4
	2.5. USE ST	ATISTICS
3.	HUMAN HE	CALTH RISK ASSESSMENT
	3.1. HAZAR	D IDENTIFICATION
	3.1.1.	Overview
	3.1.2.	Mechanisms of Action
	3.1.3.	Kinetics and Metabolism 3-4
	3.1.4.	Acute Systemic Toxicity 3-6
	3.1.5.	Subchronic and Chronic Systemic Toxicity 3-7
	3.1.6.	Effects on Nervous System 3-8
	3.1.7.	Effects on Immune System 3-11
	3.1.8.	Effects on Endocrine Function 3-12
	3.1.9.	Reproductive and Teratogenic Effects 3-12
	3.1.10.	Carcinogenicity and Mutagenicity 3-16
	3.1.11.	Irritation and Sensitization 3-17
	3.1.12.	Systemic Toxic Effects from Dermal Exposure
	3.1.13.	Inhalation Exposure 3-18

		3.1.14.	Adjuvants, Role of Surfactants	. 3-19
		5.1.15.		. 3-23
	3.2.	EXPOS	URE ASSESSMENT	. 3-26
		3.2.1.	Overview	. 3-26
		3.2.2.	Workers	. 3-27
		3.2.3.	General Public	. 3-31
	3.3.	DOSE	E-RESPONSE ASSESSMENT	. 3-38
		3.3.1.	Overview	. 3-38
		3.3.2.	Existing Guidelines	. 3-38
		3.3.3	Dose-Response and Dose-Severity Relationships	. 3-39
		3.3.4.	RfD Values Used in Risk Assessment	. 3-42
	2.4	DICK		2 4 4
	3.4.	RISK 2 4 1		. 3-44
		3.4.1.	Overview	. 3-44
		3.4.2.	Workers	. 3-45
		3.4.3.		. 3-47
		3.4.4.	Sensitive Subgroups	. 3-4/
		3.4.5.		. 3-48
		3.4.6.	Cumulative Effects	. 3-48
4.	ECO	LOGIC	AL RISK ASSESSMENT	4-1
	4.1.	HAZA	ARD IDENTIFICATION	4-1
		4.1.1.	Overview	4-1
		4.1.2.	Toxicity to Terrestrial Organisms	4-3
			4.1.2.1. Mammals	4-3
			4.1.2.2. Birds	4-4
			4.1.2.3. Invertebrates	4-5
			4.1.2.4. Soil Microorganisms	4-7
			4.1.2.5. Terrestrial Plants	4-8
			4.1.2.6. Field Studies	. 4-10
		4.1.3.	Aquatic Organisms	. 4-11
			4.1.3.1. Fish	. 4-11
			4.1.3.1.1. Overview	. 4-11
			4.1.3.1.2. Standard Toxicity Studies	. 4-12
			4.1.3.1.3. Effect of Surfactants	. 4-13
			4.1.3.1.4. Sub-lethal Effects	. 4-16
			4.1.3.1.5. Field Studies	. 4-19
			4.1.3.2. Amphibians	. 4-19
			4.1.3.3. Aquatic Invertebrates	. 4-20
			4.1.3.5. Aquatic Plants	. 4-21

#### TABLE OF CONTENTS (continued)

4.2.	EXPOSURE ASSESSMENT
	4.2.1. Overview       4-23         4.2.2. Terrestrial Animals       4-23
	4.2.3. Terrestrial Plants
	4.2.4. Aquatic Organisms 4-31
4.3.	DOSE-RESPONSE ASSESSMENT 4-31
	4.3.1. Overview
	4.3.2. Toxicity to Terrestrial Organisms 4-32
	4.3.3. Aquatic Organisms 4-36
4.4.	RISK CHARACTERIZATION 4-40
	4.4.1. Overview
	4.4.2. Terrestrial Organisms 4-41
	4.4.3. Aquatic Organism 4-43
LIST	OF WORKS CONSULTED

#### APPENDICES

Appendix 1:	Commercial	formulations	of glyphosate
-------------	------------	--------------	---------------

5.

- Appendix 2: Glyphosate used by the Forest Service
- Appendix 3a: Summary of standard toxicity values in mammals for various formulations
- Appendix 3b: Summary of standard toxicity values in wildlife species for various formulations
- Appendix 3c: Summary of standard toxicity values in aquatic species for various formulations
- Appendix 4: Case reports of poisoning by glyphosate formulations
- Appendix 5: Effects on mammals of long-term exposure to glyphosate
- Appendix 6: Assays for reproductive/teratogenic effects in mammals after exposure to glyphosate
- Appendix 7: Studies assessing the mutagenicity of glyphosate
- Appendix 8: Summary of field or field simulation studies on glyphosate formulations
- Appendix 9: Toxicity of glyphosate and glyphosate formulations to fish
- Appendix 10: Acute toxicity of glyphosate to aquatic invertebrates
- Appendix 11: Toxicity of glyphosate and glyphosate formulations to aquatic plants

#### LIST OF WORKSHEETS

Supplement 1: Glyphosate -WordPerfect Worksheets for Human Health and Ecological Risk Assessments, SERA WPWS 01-43-09-02a, Version 2.04, March 1, 2003.

Supplement 2: Glyphosate -EXCEL Worksheets for Human Health and Ecological Risk Assessments, SERA EXWS 01-43-09-02a, Version 2.04, March 1, 2003.

#### LIST OF ATTACHMENTS

Attachment 1:	Bibliography of All Citations Encountered on Glyphosate.	
Attachment 2:	2: Documentation for Worksheets Version 2.02 - Human Health and Ecologi	
	Risk Assessments.	
Attachment 3:	Documentation for the Use of GLEAMS and Auxiliary Programs,	
	SERA AT 2000-01b, dated September 24, 2000.	

#### LIST OF TABLES

Table 2-1:	Selected Physical and Chemical properties 2-4
Table 2-2:	Use of glyphosate by USDA Forest Service in 2001 by Type of Use 2-5
Table 2-3:	Use of glyphosate by USDA Forest Service in 2001 by Region 2-6
Table 3-1:	Summary of the exposure assessments for workers
Table 3-2:	Exposure rates used in general worker exposure assessments 3-50
Table 3-3:	Summary of the exposure assessments for the general public 3-51
Table 3-4:	Glyphosate specific pesticide parameters used in GLEAMS modeling
	and estimation of concentrations in ambient water
Table 3-5:	Estimated concentrations of glyphosate in a small (0.25 acre, 1 meter deep) pond
	adjacent to a 10 acre plot based on GLEAMS modeling with different soil types
	and annual rainfall rates and using a normalized application rate of 1 lb/acre 3-53
Table 3-6:	Estimated concentrations of glyphosate in a small stream (4,420 m <sup>3</sup> /day)
	adjacent to a 10 acre plot based on GLEAMS modeling with different soil
	types and annual rainfall rates and using a normalized application rate of
	1 lb/acre
Table 3-7:	Summary of the risk characterization for workers
Table 3-8:	Summary of risk characterization for the general public 3-56
Table 4-1:	Estimates of relative potency and toxicological interaction of
14010 1 11	glyphosate and POEA 4-45
Table 4-2:	Estimates of relative potency and toxicological interaction of
	glyphosate and POEA in five species of salmonids
Table 4-3:	Summary of exposure scenarios for terrestrial Animals
Table 4-4:	Summary of quantitative risk characterization for terrestrial animals
Table 4-5:	Ouantitative risk characterization for aquatic species

# LIST OF FIGURES

Figure 2-1: Structure and dissociation constants (pK <sub>a</sub> ) for the various forms of glyphosate 2	2-6
Figure 2-2: Use of glyphosate by the USDA Forest Service in various regions of the United	
States	2-7
Figure 2-3: Agricultural use of glyphosate in the United States for 1992 2	2-8
Figure 3-1: Residues of glyphosate on cowberries and bilberries after the application of	
glyphosate at 0.67 lb/acre 3-	-57
Figure 3-2: Dose-severity relationships for glyphosate 3-	-58

# ACRONYMS, ABBREVIATIONS, AND SYMBOLS

acid equivalents
active ingredient
adverse-effect level
American Conference of Governmental Industrial Hygienists
acetylcholinesterase
aminomethylphosphonate
Agency for Toxic Substances and Disease Registry
bioconcentration factor
body weight
cholinesterase
centimeter
dichlorophenoxyacetic acid
concentration causing 50% inhibition of a process
concentration causing complete inhibition of a process
Endocrine Disrupter Screening and Testing Advisory Committee
environmental impact statement
female
first filial generation
frank effect level
Forest Health
Federal Insecticide, Fungicide and Rodenticide Act
Forest Service
Food Quality Protection Act
gram
gas chromatography
generally recognized as safe
hazard quotient
International Agency for Research on Cancer
isopropylamine
Integrated Risk Information System
kilogram
organic carbon partition coefficient
octanol-water partition coefficient
skin permeability coefficient
liter
pound
lethal concentration, 50% mortality
lethal dose, 50% mortality
lethal dose, 95% mortality
lowest-observed-adverse-effect level

# ACRONYMS, ABBREVIATIONS, AND SYMBOLS (continued)

m	meter
Μ	male
MATC	maximum allowable toxicant concentration
mg	milligram
mg/kg/day	milligrams of agent per kilogram of body weight per day
mL	milliliter
MRID	master record identification number
MS	mass spectrometry
MSDS	material safety data sheet
MW	molecular weight
MOS	margin of safety
MSDS	material safety data sheet
NCI	National Cancer Institute
NNG	N-nitrosoglyphosate
NOAEL	no-observed-adverse-effect level
NOEC	no-observed-effect concentration
NOEL	no-observed-effect level
NPE	nonylphenol polyethoxylate
NRC	National Research Council
OPP	Office of Pesticide Programs
рКа	dissociation constant
POEA	polyoxyethyleneamine
ppm	parts per million
RBC	red blood cells
RED	reregistration eligibility decision
RfD	reference dose
RTU	ready to use
UF	uncertainty factor
U.S.	United States
U.S. EPA	U.S. Environmental Protection Agency
USDA	United States Department of Agriculture
USGS	United States Geological Survey
>	greater than
≥	greater than or equal to
<	less than
$\leq$	less than or equal to
=	equal to
~	approximately equal to

To convert	Into	Multiply by
acres	hectares (ha)	0.4047
acres	square meters (m <sup>2</sup> )	4,047
atmospheres	millimeters of mercury	760
centigrade	Fahrenheit	1.8C°+32
centimeters	inches	0.3937
cubic meters (m <sup>3</sup> )	liters (L)	1,000
Fahrenheit	centigrade	0.556F°-17.8
feet per second (ft/sec)	miles/hour (mi/hr)	0.6818
gallons (gal)	liters (L)	3.785
gallons per acre (gal/acre)	liters per hectare (L/ha)	9.34
grams (g)	ounces, (oz)	0.03527
grams (g)	pounds, (oz)	0.002205
hectares (ha)	acres	2.471
hectares (ha)	square meters	10,000
inches (in)	centimeters (cm)	2.540
kilograms (kg)	ounces, (oz)	35.274
kilograms (kg)	pounds, (lb)	2.2046
kilograms per hectare (hg/ha)	pounds per acre (lb/acre)	0.892
kilometers (km)	miles (mi)	0.6214
liters (L)	cubic centimeters (cm <sup>3</sup> )	1,000
liters (L)	gallons (gal)	0.2642
liters (L)	ounces, fluid (oz)	33.814
miles (mi)	kilometers (km)	1.609
miles per hour (mi/hr)	cm/sec	44.70
milligrams (mg)	ounces (oz)	0.000035
meters (m)	feet	3.281
ounces (oz)	grams (g)	28.3495
ounces per acre (oz/acre)	grams per hectare (g/ha)	70.1
ounces per acre (oz/acre)	kilograms per hectare (kg/ha)	0.0701
ounces fluid	cubic centimeters (cm <sup>3</sup> )	29.5735
pounds (lb)	grams (g)	453.6
pounds (lb)	kilograms (kg)	0.4536
pounds per acre (lb/acre)	kilograms per hectare (kg/ha)	1.121
pounds per acre (lb/acre)	mg/square meter (mg/m <sup>2</sup> )	112.1
pounds per acre (lb/acre)	$\mu$ g/square centimeter ( $\mu$ g/cm <sup>2</sup> )	11.21
pounds per gallon (lb/gal)	grams per liter (g/L)	119.8
square centimeters (cm <sup>2</sup> )	square inches (in <sup>2</sup> )	0.155
square centimeters (cm <sup>2</sup> )	square meters (m <sup>2</sup> )	0.0001
square meters (m <sup>2</sup> )	square centimeters (cm <sup>2</sup> )	10,000
yards	meters	0.9144

# COMMON UNIT CONVERSIONS AND ABBREVIATIONS

Note: All references to pounds and ounces refer to avoirdupois weights unless otherwise specified.

Scientific	Decimal	Verbal
Notation	Equivalent	Expression
$1 \cdot 10^{-10}$	0.000000001	One in ten billion
$1 \cdot 10^{-9}$	0.00000001	One in one billion
$1 \cdot 10^{-8}$	0.00000001	One in one hundred million
$1 \cdot 10^{-7}$	0.0000001	One in ten million
$1 \cdot 10^{-6}$	0.000001	One in one million
$1 \cdot 10^{-5}$	0.00001	One in one hundred thousand
$1 \cdot 10^{-4}$	0.0001	One in ten thousand
$1 \cdot 10^{-3}$	0.001	One in one thousand
$1 \cdot 10^{-2}$	0.01	One in one hundred
$1 \cdot 10^{-1}$	0.1	One in ten
$1 \cdot 10^{0}$	1	One
$1 \cdot 10^{1}$	10	Ten
$1 \cdot 10^2$	100	One hundred
$1 \cdot 10^{3}$	1,000	One thousand
$1 \cdot 10^4$	10,000	Ten thousand
$1 \cdot 10^{5}$	100,000	One hundred thousand
$1 \cdot 10^{6}$	1,000,000	One million
$1 \cdot 10^{7}$	10,000,000	Ten million
$1 \cdot 10^{8}$	100,000,000	One hundred million
$1 \cdot 10^{9}$	1,000,000,000	One billion
$1 \cdot 10^{10}$	10,000,000,000	Ten billion

# **CONVERSION OF SCIENTIFIC NOTATION**

#### **EXECUTIVE SUMMARY**

#### **INTRODUCTION**

This document provides risk assessments for human health effects and ecological effects to support an assessment of the environmental consequences of using glyphosate in Forest Service vegetation management programs. This document has four chapters, including the introduction, program description, risk assessment for human health effects, and risk assessment for ecological effects or effects on wildlife species. Each of the two risk assessment chapters has four major sections, including an identification of the hazards associated with glyphosate, an assessment of potential exposure to this compound, an assessment of the dose-response relationships, and a characterization of the risks associated with plausible levels of exposure.

In the preparation of this risk assessment, literature searches of glyphosate were conducted in the open literature using PubMed, TOXLINE as well as the U.S. EPA CBI files. Several reviews and risk assessments on glyphosate conducted by the U.S. EPA and others were also consulted. The search of U.S. EPA's FIFRA/CBI files indicated that there are 5829 submissions on glyphosate and glyphosate formulations. While many of these studies were conducted to support the initial registration and reregistration of glyphosate, a substantial number of studies were conducted and submitted to U.S. EPA after 1993, the date of the U.S. EPA Reregistration Eligibility Decision document on glyphosate. Because of the extensive published literature on glyphosate, the reregistration document for glyphosate was used where possible to summarize information for the earlier CBI studies. Although full copies of some key studies were obtained from the earlier CBI submissions, the acquisition of the CBI studies focused on the post-1993 period.

The human health and ecological risk assessments presented in this document are not, and are not intended to be, comprehensive summaries of all of the available information and these risk assessment do not cite all of the available literature. To review each of these studies would far exceed the resources available to the Forest Service and, more importantly, would make the document very difficult to read and review. In some respects, an all inclusive and detailed review of each study would tend to obscure rather than inform. As an alternative, this document focuses on information that is likely to impact the risk assessments.

#### **PROGRAM DESCRIPTION**

Glyphosate is a herbicide that is used in Forest Service programs primarily in conifer release, noxious weed control, and site preparation. There are currently 35 commercial formulations of glyphosate that are registered for forestry applications. All commercial formulations of glyphosate that are used in forestry applications contain the isopropylamine salt of glyphosate. Some formulations contain only this salt of glyphosate as an aqueous solution. Other formulations contain surfactants and some of these surfactants are toxic, particularly to aquatic organisms. Technical grade glyphosate has been classified as toxicologically insignificant by the U.S. EPA. A surfactant used in at least one major commercial formulation contains 1,4-dioxane, and the toxicity of this impurity is specifically considered in this risk assessment. The most

common method of application for glyphosate in Forest Service programs involves is backpackapplied directed foliar sprays. Other application methods that are use occasionally are broadcast foliar ground applications, cut stem applications, and direct application to the emergent aquatic vegetation. Based on recent Forest Service use reports, the typical application rate is about 2 lb a.e./acre, with most application rates occurring over a range of 0.5 lbs a.e./acre to 7 lbs a.e./acre. The total annual use of glyphosate by the Forest Service is only about 0.275 percent of the agricultural use. Thus, there is no basis for asserting that Forest Service programs will substantially contribute to general concentrations of glyphosate nationally. The potential for local contamination of environmental media by the use of glyphosate in Forest Service programs is discussed in the human health risk assessment and the ecological risk assessment.

#### HUMAN HEALTH RISK ASSESSMENT

*Hazard Identification* – The herbicidal activity of glyphosate is due primarily to the inhibition of the shikimate pathway which is involved in the synthesis of aromatic amino acids in plants and microorganisms. This metabolic pathway does not occur in humans or other animals and thus this mechanism of action is not directly relevant to the human health risk assessment. Two specific biochemical mechanisms of action have been identified or proposed for glyphosate: uncoupling of oxidative phosphorylation and inhibition of hepatic mixed function oxidases. Both glyphosate and the polyethoxylated tallow amine (POEA) surfactant used in Roundup will damage mucosal tissue, although the mechanism of this damage is likely to differ for these two agents. Many of the effects of acute oral exposure to high doses of glyphosate or Roundup are consistent with corrosive effects on the mucosa.

The available experimental studies indicate that glyphosate is not completely absorbed after oral administration and is poorly absorbed after dermal applications. Two dermal absorption studies have been published on glyphosate and both of these studies indicate that glyphosate is very poorly absorbed across the skin.

Like all chemicals, glyphosate as well as commercial formulations of glyphosate may be toxic at sufficiently high exposure levels. In rats and mice, acute oral  $LD_{50}$  values of glyphosate range from approximately 2,000 to 6,000 mg/kg. Formulations of glyphosate with a POEA surfactant have been used in many suicides and attempted suicides. Gastrointestinal effects (vomiting, abdominal pain, diarrhea), irritation, congestion, or other forms of damage to the respiratory tract, pulmonary edema, decreased urinary output sometimes accompanied by acute renal tubular necrosis, hypotension, metabolic acidosis, and electrolyte imbalances, probably secondary to the gastrointestinal and renal effects, are seen in human cases of glyphosate/surfactant exposure.

One of the more consistent signs of subchronic or chronic exposure to glyphosate is loss of body weight. This effect has been noted in mice, rats, dogs, and rabbits. This observation is consistent with experimental data indicating that glyphosate may be an uncoupler of oxidative phosphorylation. Other signs of toxicity seem general and non-specific. A few studies report changes in liver weight, blood chemistry that would suggest mild liver toxicity, or liver pathology. Changes in pituitary weight have also been observed. Signs of kidney toxicity, which might be

expected based on the acute toxicity of glyphosate, have not been reported consistently and are not severe. Various hematological changes have been observed that may be secondary to mild dehydration.

Glyphosate has been specifically tested for neurotoxicity in rats after both acute and subchronic exposures and has been tested for delayed neurotoxicity in hens. In both the animal data as well as the clinical literature involving suicide attempts, there is no clear pattern suggestive of a specific neurotoxic action for glyphosate or its commercial formulations. The weight of evidence suggests that any neurologic symptoms associated with glyphosate exposures are secondary to other toxic effects. No studies are reported that indicate morphologic abnormalities in lymphoid tissues which could be suggestive of an effect on the immune system. As discussed in the ecological risk assessment, one study has asserted that glyphosate causes immune suppression in a species of fish. This study, however, is deficient in several respects and does not provide a basis for impacting the hazard identification for effects on the immune system.

Only three specific tests on the potential effects of glyphosate on the endocrine system have been conducted and all of these tests reported no effects. All of these assays are *in vitro* – i.e., not conducted in whole animals. Thus, such studies are used qualitatively in the hazard identification to assess whether there is a plausible biologic mechanism for asserting that endocrine disruption is plausible. Because they are *in vitro* assays, measures of *dose* and quantitative use of the information in dose/response assessment is not appropriate. For glyphosate, these studies to not indicate a basis for suggesting that glyphosate is an endocrine disruptor. Nonetheless, glyphosate has not undergone an extensive evaluation for its potential to interact or interfere with the estrogen, androgen, or thyroid hormone systems. Thus, the assessment of the potential endocrine effects of glyphosate cannot be overly interpreted.

Glyphosate has been subject to multi-generation reproduction studies which measure overall effects on reproductive capacity as well as teratology studies which assay for a compounds ability to cause birth defects. Signs of teratogenic activity have not been observed in standard assays in both rats and rabbits. In a multi-generation reproduction study in rats, effects on the kidney were observed in male offspring. This effect is consistent with the acute systemic toxicity of glyphosate, rather than a specific reproductive effect. Several other subchronic and chronic studies of glyphosate have been conducted with no mention of treatment-related effects on endocrine glands or reproductive organs. A single study has reported substantial decreases in libido, ejaculate volume, sperm concentrations, semen initial fructose and semen osmolality as well as increases in abnormal and dead sperm in rabbits after acute oral exposures to glyphosate. This study is inconsistent with other studies reported on glyphosate and is poorly documented – i.e., specific doses administered to the animals are not specified. In addition, the use of gelatin capsules, as in this study results, in a high spike in body burden that is not typical or particularly relevant to potential human exposures - other than attempted suicides. Numerous epidemiological studies have examined relationships between pesticide exposures or assumed pesticide exposures in agricultural workers and reproductive outcomes. Of those studies that have specifically addressed potential risks from glyphosate exposures, adverse reproductive

effects have not been noted.

Based on standard animal bioassays for carcinogenic activity *in vivo*, there is no basis for asserting that glyphosate is likely to pose a substantial risk. The Re-registration Eligibility Decision document on glyphosate prepared by the U.S. EPA indicates that glyphosate is classified as Group E: Evidence of non-carcinogenicity for humans. This classification is also indicated in U.S. EPA's most recent publication of tolerances for glyphosate and is consistent with an assessment by the World Health Organization. This assessment has been challenged based on some studies that indicate marginal carcinogenic activity. As with any compound that has been studied for a long period of time and tested in a large number of different systems, some equivocal evidence of carcinogenic potential is apparent and may remain a cause of concern, at least in terms of risk perception. While these concerns are understandable, there is no compelling basis for challenging the position taken by the U.S. EPA and no quantitative risk assessment for cancer is conducted as part of the current analysis.

Glyphosate formulations used by the Forest Service are classified as either non-irritating or only slightly irritating to the skin and eyes in standard assays required for product registration. Based on a total of 1513 calls to a poison control center reporting ocular effects associated with the use of Roundup, 21% were associated with no injury, 70% with transient minor injury, 2% with some temporary injury. The most frequently noted symptoms included blurred vision, a stinging or burning sensation, lacrimation. No cases of permanent damage were reported.

Various glyphosate formulations contain a POEA surfactant at a level of up to about 20%. Other formulations of glyphosate recommend the use of a surfactant to improve the efficacy of glyphosate. While surfactants are typically classified as "inert" ingredients in herbicides, these compounds are not toxicologically inert and some surfactants may be more toxic than the herbicides with which they are used. Although surfactants may play a substantial role in the interpretation of a large number of suicides and attempted suicides involving the ingestion of glyphosate formulations, primarily Roundup, the acute mammalian toxicity of different glyphosate formulations do not appear to differ substantially. This is in contrast to the available data on the toxicity of various formulations to aquatic species, as detailed in the ecological risk assessment.

*Exposure Assessment* – Exposure assessments are developed for both workers and members of the general public. Two types of work exposure assessments are considered: general and accidental/incidental. The term *general* exposure assessment is used to designate those exposures that involve estimates of absorbed dose based on the handling of a specified amount of a chemical during specific types of applications. The accidental/incidental exposure scenarios involve specific types of events that could occur during any type of application. For general exposures in workers, exposure rates are expressed in units of mg of absorbed dose per kilogram of body weight per pound of chemical handled. For glyphosate, there are several worker exposure studies involving backpack applications that can be used to assess the quality general estimates used in many Forest Service risk assessments. These studies indicate that these general methods may be extremely conservative. Nonetheless, for this risk assessment, the standard worker exposure rates

are used, recognizing that the upper range of exposures may overestimate risk. This conservative approach has little impact on the interpretation of risk because none of the worker exposures exceed a hazard quotient of unity. Central estimates of worker exposures span a very narrow range from 0.026 mg/kg/day to about 0.045 mg/kg/day. The upper range of exposures for the different application methods are about a factor of 10 higher, spanning a range from about 0.1 mg/kg/day to 0.3 mg/kg/day.

Under normal circumstances, members of the general public should not be exposed to substantial levels of glyphosate as a result of Forest Service activities. Nonetheless, any number of exposure scenarios can be constructed for the general public, depending on various assumptions regarding application rates, dispersion, canopy interception, and human activity. Several highly conservative scenarios are developed for this risk assessment. The two types of exposure scenarios developed for the general public include acute exposure and longer-term or chronic exposure. All of the acute exposure scenarios are primarily accidental. They assume that an individual is exposed to the compound either during or shortly after its application. Specific scenarios are developed for direct spray, dermal contact with contaminated vegetation, as well as the consumption of contaminated fruit, water, and fish. Most of these scenarios should be regarded as extreme, some to the point of limited plausibility. The longer-term or chronic exposure scenarios parallel the acute exposure scenarios for the consumption of contaminated fruit, water, and fish but are based on estimated levels of exposure for longer periods after application. Most acute accidental exposure scenarios for members of the general public are less than or similar to the general exposure scenarios in workers. The major exception is the scenario for an accidental spill of 200 gallons of a field solution into a small pond. This leads to modeled estimates of exposure in the range of 0.3 to about 4 mg/kg/day. This is an extraordinarily extreme and conservative scenario that is used in all Forest Service risk assessments. Most longer term estimates of exposure for members of the general public are much lower than exposure estimates for workers. The one exception involves the longer term consumption of contaminated fruit, which leads to timeweighted average estimated doses of 0.003 to 0.08 mg/kg/day.

**Dose-response Assessment** – Generally, the dose-response assessments used in Forest Service risk assessments adopt RfDs proposed by the U.S. EPA as indices of 'acceptable' exposure. An RfD is basically defined as a level of exposure that will not result in any adverse effects in any individual. The U.S. EPA RfDs are used because they generally provide a level of analysis, review, and resources that far exceed those that are or can be conducted in the support of most Forest Service risk assessments. In addition, it is desirable for different agencies and organizations within the federal government to use concordant risk assessment values.

The most recent RfD on glyphosate is that proposed by the U.S. EPA Office of Pesticide Programs. This RfD of 2 mg/kg/day was proposed originally in the RED for glyphosate and was also used in the recent glyphosate pesticide tolerances. This RfD is based on teratogenicity study in rabbits (Rodwell et al. 1980b) in which no effects observed in offspring at any dose levels and maternal toxicity was observed at 350 mg/kg/day with a NOAEL of 175 mg/kg/day. Using an uncertainty factor of 100 - 10 for sensitive individuals and 10 for species-to-species extrapolation

– U.S. EPA/OPP derived the RfD of 2 mg/kg/day, rounding the value of 1.75 mg/kg/day to one significant digit.

For the current risk assessment, the RfD 2 mg/kg/day derived by U.S. EPA/OPP is used as the basis for characterizing risk from longer-term exposures in this risk assessment. For short-term exposures, the value of 2 mg/kg/day recommended by U.S. EPA's Office of Drinking Water is used. Since this is identical to the chronic RfD, this approach is equivalent to applying the same RfD to be short-term and long-term exposures. Given the lack of a significant dose-duration relationship for glyphosate, this approach seems appropriate.

*Risk Characterization* – The risk characterization for both workers and members of the general public are reasonably consistent in unambiguous. For both groups, there is very little indication of any potential risk at the typical application rate of 2 lbs a.e./acre. Even at the upper range of plausible exposures in workers, most hazard quotients are below the level of concern.

For workers, the highest hazard quotient – i.e., 0.2, the upper range for workers involved in broadcast ground spray – is below the level of concern by a factor of about 5. The highest hazard quotient for any accidental exposure scenario for workers - i.e., 0.006 for the upper range of the hazard quotient for spill over the lower legs for one hour - is lower than the level of concern by a factor of over 150. Confidence in these assessments is reasonably high because of the availability of dermal absorption data in human as well as worker exposure studies. The Forest Service may apply glyphosate at a maximum rate of 7 lbs a.e./acre, a factor of 3.5 higher than the typical application rate of 2 lbs a.e./acre. This has essentially no impact of the risk characterization for workers. The highest hazard quotient for the typical application rate is 0.2. For an application rate of 7 lbs a.e./acre, the corresponding hazard quotient would be higher by a factor of 3.5 or 0.7, which is still below the level of concern.

From a practical perspective, the most likely accidental exposure for workers that might require medical attention involves accidental contamination of the eyes. Glyphosate and glyphosate formulations are skin and eye irritants. Quantitative risk assessments for irritation are not normally derived, and, for glyphosate specifically, there is no indication that such a derivation is warranted. Glyphosate with the POEA surfactant is about as irritating as standard dish washing detergents, all purpose cleaners, and baby shampoos. As with the handling of any chemical, including a variety of common household products, reasonable care should be taken to avoid contact of skin and eyes.

The only area of remarkable uncertainty involving worker exposures concerns the potential health effects during brown-and-burn operations. The combustion of wood and wood by-products may produce a number of toxic compounds. This is a concern with brown-and-burn operations but does not pertain to the use of glyphosate or any other herbicide. The potential effects of combustion products is common to all risk assessments of materials that might be subject to burning. With the exception of some plastics, the combustion products of which are known to pose a risk to fire fighters, the combustion products of most chemicals have not been examined in

detail. The necessity of addressing this data gap must be weighed against the need to address other data gaps on glyphosate and other chemicals. The combustion products of burning wood and vegetation are respiratory irritants as well as carcinogens, and exposure to these combustion products should be avoided. There is no basis for believing that the presence of low or even high levels of glyphosate residues will have a significant impact on this hazard.

For members of the general public, none of the longer-term exposure scenarios exceed or even approach a level of concern. Although there are several uncertainties in the longer-term exposure assessments for the general public, the upper limits for hazard indices are below a level of concern by factors of about 25 (longer term consumption of contaminated fruit) to over two million (2,500,000 for longer-term consumption of fish by the general population). The risk characterization is thus relatively unambiguous: based on the available information and under the foreseeable conditions of application and exposure, there is no route of exposure or exposure scenario suggesting that the general public will be at risk from longer-term exposure to glyphosate. As with the hazard characterization for workers, an application rate of 7.5 lbs a.e./acre makes no difference in the assessment of potential risks. At this application rate, the highest hazard quotient would be about  $0.14 [0.04 \times 3.5]$ , which is still below a level of concern by a factor of about 7.

One acute exposure scenario does exceed the level of concern at the upper range at the typical application rate of 2 lbs a.e./acre. The exposure scenario for the consumption of contaminated water after an accidental spill into a small pond results in an excursion above the RfD at the upper limit of exposure – i.e., a hazard quotient of 2. This exposure scenario is extreme to the point of limited plausibility. This sort of scenario is routinely used in Forest Service risk assessments as an index of the measures that should be taken to limit exposure in the event of a relatively large spill into a relatively small body of water. For glyphosate, as well as for most other chemicals, this exposure assessment indicates that such an event would require measures to ensure that members of the general public do not consume contaminated water.

At the highest application rate that might be used in Forest Service programs, the accidental spill scenario is the only other scenario that results in a hazard quotient above unity. At this application rate, the associated dose is about 14 mg/kg, which is still below the dose of 184 mg/kg associated with no apparent overt effects in humans by a factor of over 10.

#### ECOLOGICAL RISK ASSESSMENT

*Hazard Identification* – There are several standard toxicity studies in experimental mammals that were conducted as part of the registration process and there is a large body of published information on the toxicity of glyphosate to mammals. Just as these studies are used in the human health risk assessment to identify the potential toxic hazards associated with exposures to glyphosate, they can also be used to identify potential toxic effects in wildlife mammalian species. Loss of body weight is the most commonly seen effect of glyphosate in mammals. Inhibition of oxidative phosphorylation has been implicated as a possible mechanism by which glyphosate causes weight loss in experimental mammals; however, there is not adequate information about

terrestrial wildlife from which to make a further assessment about the importance of this mechanism. As in the human health risk assessment, the potential significance of non-specific toxic effects can be assessed from the available toxicity studies in mammals. Because toxicity data in mammals are available in few species of experimental mammals, the use of these data to assess the potential hazards to large number of diverse mammalian wildlife species is an uncertain process. Nonetheless, there do not appear to be any systematic differences among mammalian species, including humans, when comparable toxicity values are expressed in units of mg/kg/day. While the available data are limited, this apparent consistency among species diminishes concern with the use of data based on a limited subset of species to characterize risk for terrestrial mammals in general.

In assessing potential effects in birds, the most relevant data for this risk assessment are the standard dietary and bird reproduction studies required for registration as well as the acute oral  $LD_{50}$  studies. The available toxicity studies do not suggest any specific or unique toxicity in birds compared to mammals. As in mammals, there is suggestive evidence that glyphosate may inhibit oxidative phosphorylation and consequently reduce food conversion efficiency. Also consistent with the data in experimental mammals is the apparent lack of teratogenic activity in birds.

The honey bee is the standard test organism for assessing the potential effects of pesticides on terrestrial invertebrates and there is a standard set of studies available on this species. In addition, studies are available on a relatively wide range of other terrestrial invertebrates including earthworms, isopods, snails, spiders, butterflies, and other terrestrial arthropods.

Glyphosate is readily metabolized by soil bacteria and many species of soil microorganisms can use glyphosate as sole carbon source. Nonetheless, microorganisms, like higher plants, do have the shikimate pathway for the production of aromatic amino acids. At the molecular level, glyphosate occupies the binding site of phosphoenol pyruvate, the second substrate of 5enolpyruvylshikimate 3-phosphate synthase, mimicking an intermediate state of the ternary enzyme-substrate complex. This inhibits the shikimic acid pathway in plants, effectively blocking the synthesis of certain phenolic compounds and the synthesis of aromatic amino acids. This, in turn, leads to a variety of toxic effects in plants, including the inhibition of photosynthesis in plants as well as inhibition of respiration and nucleic acid synthesis in plants and microorganisms.

Since glyphosate inhibits this pathway, toxicity to microorganisms may be expected and glyphosate has been considered as an antimicrobial agent for human pathogens. Nonetheless, there is very little information suggesting that glyphosate will be harmful to soil microorganisms under field conditions and a substantial body of information indicating that glyphosate is likely to enhance or have no effect on soil microorganisms. Most field studies involving microbial activity in soil after glyphosate exposures note an increase in soil microorganisms or microbial activity and the application of glyphosate may cause transient increases in soil fungi that may be detrimental to some plants. While the mechanism of this apparent enhancement is unclear, it is plausible that glyphosate treatment resulted in an increase in the population of pathogenic fungi in soil because glyphosate was used as a carbon source by the fungi and/or treatment with glyphosate resulted in

increased nutrients for fungi in the soil. There is no indication that the transient enhancement in populations of soil fungi or bacteria will result in any substantial or lasting damage to soil ecology.

In higher plants, inhibition of the shikimic acid pathway leads to an inhibition or cessation of growth, cellular disruption, and, at sufficiently high levels of exposure, plant death. The time course for these effects can be relatively slow, depending on the plant species, growth rate, climate, and application rate. Gross signs of toxicity include wilting and yellowing of the vegetation, followed by browning, breakdown of plant tissue, and, ultimately, root decomposition. Standard toxicity studies are available on seedling emergence and vegetative vigor in a number of different plant species. The drift studies are also highly relevant to the assessment of risk in that unintended drift is one of the more plausible exposure scenarios for nontarget terrestrial plant species. The lowest reported effect level in drift studies is 1/33 of an application rate of 1.121 kg/ha that was associated with transient damage in soybeans. This treatment corresponds to 0.034 kg/ha [1.121 kg/ha  $\div$  33] or about 0.03 lb/acre. At much higher concentrations – in the range of 0.7 lbs/acre – there is a plausible basis for concern that exposure to substantial glyphosate drift may have long term impacts on bryophyte and lichen communities.

In addition to the laboratory bioassays or field observations on single species, there are a number field studies that have assessed the effects of glyphosate on groups of terrestrial organisms, both animals and plants. These studies indicate that effects on terrestrial animals are likely to be secondary to effects on vegetation when glyphosate is applied at application rates comparable to or greater than those contemplated by the Forest Service. In some cases, the effects noted in filed studies appeared to be beneficial to some species under study. In most cases, the effects noted were changes in population density that reflected changes in food availability or suitable habitat.

In aquatic species, the acute lethal potency of glyphosate and glyphosate formulations has been relatively well-defined. These values are typically expressed as time-specific  $LC_x$  values where x is the estimate of the proportion of fish that die – e.g., 96 hour  $LC_{50}$ . A large number of acute  $LC_{50}$  values have been determined in various species of fish. As in the human health risk assessment, the formulation of glyphosate with surfactants, especially the POEA surfactant commonly used in glyphosate formulations, has a pronounced effect on the acute lethal potency of glyphosate.

The U.S. EPA typically uses  $LC_{50}$  values or fractions of  $LC_{50}$  values as the basis for characterizing risk, as in the U.S. EPA RED on glyphosate. A common concern with this approach is that more subtle non-lethal effects that may impact of the stability of fish populations in the field may not be properly assessed. The available information on the sub-lethal effects associated with glyphosate is summarized in this risk assessment and NOEC (no observable effect concentration) values form the basis of the risk characterization.

Lastly, field studies are available on the effects of glyphosate applications on fish populations. As with the risk characterization for terrestrial species, these studies have limitations in terms of their quantitative use in a risk assessment but are nonetheless highly relevant to the risk assessment and

may be used to further assess the quality of the risk characterization based on laboratory bioassay.

The toxicity of glyphosate to aquatic plants has been evaluated by U.S. EPA based on studies submitted for the registration of glyphosate. In addition, several studies are available from the open literature as well as more recent studies submitted to U.S. EPA. These studies are available for both algae and aquatic macrophytes. As would be expected from a herbicide, glyphosate is much more toxic to aquatic plants than animals.

*Exposure Assessment* – Terrestrial animals might be exposed to any applied herbicide from direct spray, the ingestion of contaminated media (vegetation, prey species, or water), grooming activities, or indirect contact with contaminated vegetation. The highest exposures for terrestrial vertebrates will occur after the consumption of contaminated vegetation or contaminated insects. In acute exposure scenarios, doses as high as 225 mg/kg are estimated. Other routes of exposure, like the consumption of contaminated water or direct spray, lead to lower levels of exposure. In chronic exposure scenarios, the estimated daily doses at the upper limits of exposure are in the range of about 50 to 80 mg/kg/day and are associated with highly conservative assumptions regarding the consumption of contaminated vegetation.

The primary hazards to non-target terrestrial plants are associated with unintended direct deposition or spray drift. Unintended direct spray will result in an exposure level equivalent to the application rate. At least some plants that are sprayed directly with glyphosate at or near the recommended range of application rates will be damaged. Based on the AgDRIFT model, no more than 0.0058 of the application rate would be expected to drift 100 m offsite after low boom ground applications. In order to encompass a wide range of field conditions, GLEAMS simulations were conducted for clay, loam, and sand at annual rainfall rates from 5 to 250 inches. Under arid conditions (i.e., annual rainfall of about 10 inches or less), there is no or very little runoff. Under these conditions, degradation, not dispersion, accounts for the decrease of glyphosate results in runoff losses that range from about negligible up to about 45% of the application rate, depending primarily on the amount of rainfall rater than differences in soil type.

The potential for effects on aquatic species are based on estimated concentrations of glyphosate in water that are identical to those used in the human health risk assessment without further elaboration. For an accidental spill, the central estimate for the concentration of glyphosate in a small pond is estimated at about 18.2 mg/L with a range from 1.8 to 127 mg/L. For longer term exposure scenarios, the expected concentrations of glyphosate in ambient water range from 0.0001 to 0.008 mg/L with a central value of 0.001 mg/L.

**Dose-response Assessment** – For mammals, the toxicity data used to characterize risk are identical to those used in the human health risk assessment – i.e., a NOAEL of 175 mg/kg with an associated LOAEL of 350 mg/kg. The 175 mg/kg NOAEL and 350 mg/kg LOAEL values are used for both the acute and chronic risk assessments. This approach is taken because of the lack of a substantial dose-duration or dose-severity relationship for glyphosate. For birds, a dose of

100 mg/kg is used as a NOAEL for characterizing chronic risks. It should be noted that this dose is very close to the NOAEL of 175 mg/kg used for mammals and is consistent with the apparent lack of variability in the toxicity of glyphosate among species. As in the assessment for mammals, this NOAEL is based on a repeated dose study for reproductive effects. The acute NOAEL is taken as 562 mg/kg from a five-day dietary studies in bobwhite quail and mallard ducks. Toxicity to terrestrial invertebrates is characterized using a standard set of studies in honey bees. The NOEC used in this risk assessment is taken as 50  $\mu$ g/bee.

The assessment of potential effects in plants is based on standard toxicity studies required for pesticide registration involving pre-emergence and post-emergence exposures. In seedling emergence assays, very high concentrations – i.e., 10 lb a.i./acre or about 7.5 lbs a.e./acre – will modestly inhibit seed germination in both monocots and dicots. The NOEC for seed germination is 4.5 lb a.e./acre in both monocots and dicots. This value is used to assess the consequences of off-site movement of glyphosate in runoff. Glyphosate appears to be more toxic in vegetative vigor assays – i.e., direct application to the foliage of growing plants. The lowest reported NOEC for growth in standard bioassays required for registration is 0.035 lb a.e./acre. The highest reported NOEC for growth is 0.56 lb a.e./acre. This range of values for sensitive and relatively insensitive species is used to assess the consequences of off-site drift of glyphosate.

The dose-response assessment for fish is substantially complicated by information indicating that some fish species such as salmonids are more sensitive to glyphosate than other species of fish and by information indicating that some surfactants are very toxic to fish and may substantially increase to the toxicity of glyphosate to fish. These factors are further complicated by gaps in the available data. Given the apparently high sensitivity of some salmonids to glyphosate, it would be desirable to have a life cycle toxicity study or at least an egg-and-fry study available on salmonids. In addition, given the apparently high toxicity of surfactant formulations compared to technical grade glyphosate, a life cycle toxicity study on at least one formulation containing a toxic surfactant would be desirable. Such studies, however, are not available. Consequently, an approximation method commonly used is mixtures risk assessment (the relative potency method) is employed to estimate a chronic NOEC of 2.57 mg/L for technical grade glyphosate in sensitive species of fish based on an observed NOEC value of 25.7 mg/L in tolerant species of fish. Similarly, NOEC values for glyphosate formulations containing toxic surfactants are estimated at 0.36 mg/L for sensitive species and 0.64 mg/L for tolerant species. A similar approach is used estimate the potential for acute effects based on 96-hour  $LC_{50}$  values.  $LC_{50}$  values rather than data on sublethal effects are used to characterize risks from acute exposures because most of the data on sublethal effects are based on very short-term exposures to concentrations in the range of 96-hour  $LC_{50}$  values. Most of the available toxicity data suggest that amphibians are no more sensitive to glyphosate than fish. Consequently, a separate dose-response assessment for amphibians is not conducted in this risk assessment.

The issues in the dose-response assessment for aquatic invertebrates are very similar to those encountered in the dose-response assessment for fish. There is sufficient data to assert that some glyphosate formulations that contain toxic surfactants may be much more toxic to aquatic invertebrates than technical grade glyphosate. There is only one chronic study on technical grade glyphosate and no chronic studies on glyphosate formulations. Similar to the approach used in the dose-response assessment for fish, a chronic NOEC of 50 mg/L for technical grade glyphosate is used to estimate a chronic NOEC of 0.7 mg/L for glyphosate formulations containing toxic surfactants. The potential for acute effects in aquatic invertebrates are based on  $LC_{50}$  values of 780 mg/L for technical grade glyphosate and 11mg/L for glyphosate formulations containing toxic surfactants.

Glyphosate appears to be about equally toxic to both algae and aquatic macrophytes. In terms of growth inhibition, the NOEC of 3 mg/L in duckweed is used to characterize risk due to inhibition. At lower concentrations – i.e., in the range of 0.002 mg/L to 0.3 mg/L or higher – stimulation of algal growth may be a more common response and has been noted in several studies.

*Risk Characterization* – The current risk assessment for glyphosate generally supports the conclusions reached by U.S. EPA: *Based on the current data, it has been determined that effects to birds, mammals, fish and invertebrates are minimal.* At the typical application rate of 2 lbs a.e./acre, none of the hazard quotients for acute or chronic scenarios reach a level of concern even at the upper ranges of exposure for terrestrial organisms. For the application rate of 7 lbs a.e./acre, central estimates of the hazard quotients somewhat exceed the level of concern for the direct spray of a honey bee. That the upper range of the hazard quotients, the level of concern is exceeded modestly in acute scenarios for a large mammal consuming contaminated vegetation and a small bird consuming insects. In the chronic exposure scenarios, the hazard quotient for a large bird concern by a factor of about 3. As with all longer term exposure scenarios involving the consumption of contaminated vegetation, the plausibility of this exposure scenario is limited because damage to the treated vegetation – i.e., vegetation directly sprayed at the highest application rate – would reduce and perhaps eliminate the possibility of any animal actually consuming this vegetation over a prolonged period.

For relatively tolerant nontarget species of plants, there is no indication that glyphosate is likely to result in damage at distances as close as 25 feet from the application site. For sensitive species at the upper range of application rates, there is a modest excursion about the NOEC at offsite distances of 100 feet or less. It should be noted, however, that all of these drift estimates are based on low-boom ground sprays. Many applications of glyphosate are conducted by directed foliar applications using backpacks. In such cases, little if any damage due to drift would be anticipated. Nontarget terrestrial plants are not likely to be affected by runoff of glyphosate under any conditions.

The primary hazards to fish appear to be from acute exposures to the more toxic formulations. At the typical application rate of 2 lbs a.e./acre, the hazard quotients for the more toxic formulations at the upper ranges of plausible exposure indicate that the  $LC_{50}$  values for these species will be not reached or exceeded under worst-case conditions. At an application rate of 7 lbs a.e./acre, the acute exposures are estimated to slightly exceed the  $LC_{50}$  value for typical species and exceed the

 $LC_{50}$  value for sensitive species by a factor of about 2. In these worst-case scenarios, the exposure estimates are based on a severe rainfall (about 7 inches over a 24 hour period) in an area where runoff is favored – a slope toward a stream immediately adjacent to the application site. This is a standard worst-case scenario used in Forest Service risk assessments to guide the Forest Service in the use of herbicides. This risk characterization strongly suggests that the use of the more toxic formulations near surface water is not prudent.

The use of less toxic formulations result in acute hazard quotients that do not approach a level of concern for any species. Nonetheless, the hazard quotient of 0.08 for sensitive species at an application rate of 2 lbs/acre is based on an  $LC_{50}$  value rather than a NOEC. Thus, the use of glyphosate near bodies of water where sensitive species of fish may be found (i.e., salmonids) should be conducted with substantial care to avoid contamination of surface water. Concern for potential effects on salmonids is augmented by the potential effects of low concentrations of glyphosate on algal populations.

The likelihood of direct acute toxic effects on aquatic invertebrates or longer term direct effects on any fish species seems extremely remote based on central estimates of the hazard quotient and unlikely base on upper ranges of the hazard quotient. The hazard quotient of 0.044 for longer term effects of the more toxic formulations on sensitive fish is based on an estimated NOEC and thus is not, in itself, of substantial concern. Aquatic plants appear to be somewhat less sensitive to glyphosate than the most sensitive aquatic animals. There is no indication that adverse effects on aquatic plants are plausible.

#### **1. INTRODUCTION**

This document provides risk assessments for human health effects and ecological effects to support an assessment of the environmental consequences of using glyphosate in Forest Service vegetation management programs. This risk assessment is an update to the previous USDA Forest Service risk assessment of glyphosate (SERA 1996). A large number of commercial formulations of glyphosate are available and all formulations that are currently registered for forestry applications are covered in this risk assessment.

This document has four chapters, including the introduction, program description, risk assessment for human health effects, and risk assessment for ecological effects or effects on wildlife species. Each of the two risk assessment chapters has four major sections, including an identification of the hazards associated with glyphosate, an assessment of potential exposure to this compound, an assessment of the dose-response relationships, and a characterization of the risks associated with plausible levels of exposure. These are the basic steps recommended by the National Research Council of the National Academy of Sciences (NRC 1983) for conducting and organizing risk assessments.

This is a technical support document and it addresses some specialized technical areas. Nevertheless an effort was made to ensure that the document can be understood by individuals who do not have specialized training in the chemical and biological sciences. Certain technical concepts, methods, and terms common to all parts of the risk assessment are described in plain language in a separate document (SERA 2001a). Some of the more complicated terms and concepts are defined, as necessary, in the text.

In the preparation of this risk assessment, literature searches of glyphosate were conducted in the open literature using PubMed, TOXLINE as well as the U.S. EPA CBI files. Several reviews and risk assessments on glyphosate conducted by the U.S. EPA were also consulted. These include the science chapters for human health (U.S. EPA/OPP 1993a) and ecological effects (U.S. EPA/OPP 1993b), the U.S. EPA Reregistration Eligibility Decision (RED) document on glyphosate (U.S. EPA/OPP 1993), pesticide tolerances for glyphosate (U.S. EPA/OPP 2002), the U.S. EPA Drinking Water Criteria Document on glyphosate (U.S. EPA/ODW 1992) and the IRIS entry for this compound (U.S. EPA/IRIS 1993). Additional reviews consulted in the preparation of this document include the EXTOXNET review of this compound (Extoxnet 1996), summaries prepared by the registrant of glyphosate (Monsanto The Agricultural Group 1995a,b,c,d), risk assessments conducted by the World Health Organization on glyphosate (WHO 1994) and AMPA, a metabolite of glyphosate (WHO 1998), public health goals for concentrations of glyphosate in drinking water prepared by CalEPA (CalEPA 1997; Howd et al. 2000), a reduced risk rationale for Roundup ULTRA (Wratten 1998), a human health risk assessment on glyphosate and glyphosate formulations published by Williams et al. (2000) as well as a review of environmental concerns with the use of glyphosate (Cox 1998a; Cox 1998b).

The search of U.S. EPA's FIFRA/CBI files indicated that there are 5829 submissions on glyphosate and glyphosate formulations. While many of these studies were conducted to support the initial registration and reregistration of glyphosate, a substantial number of studies (n=1288) were conducted and submitted to U.S. EPA after 1993, the date of the U.S. EPA Reregistration Eligibility Decision document on glyphosate (U.S. EPA/OPP 1993). Because of the extensive published literature on glyphosate, the reregistration document for glyphosate (U.S. EPA/OPP 1993) was used where possible to summarize information for the earlier CBI studies. Although full copies of some key studies (n=47) were obtained from the earlier CBI submissions, the acquisition of the CBI studies focused on the post-1993 period (n=138). Full text copies of the CBI studies [n=185] were kindly provided by the U.S. EPA Office of Pesticide Programs. The CBI studies were reviewed, and synopses of the information that can be disclosed from most relevant studies are included in the appendices to this document.

The human health and ecological risk assessments presented in this document are not, and are not intended to be, comprehensive summaries of all of the available information and these risk assessment do not cite all of the available literature. The level of detail presented in the appendices and the discussions in chapters 2, 3, and 4 of the risk assessment are intended to be sufficient to support a review of the risk analyses. As noted above, glyphosate has been extensively tested and nearly 6000 studies on glyphosate have been identified. To review each of these studies would far exceed the resources available to the Forest Service and, more importantly, would make the document very difficult to read and review. In some respects, an all inclusive and detailed review of each study would tend to obscure rather than inform.

As an alternative, this document focuses on information that is likely to impact the risk assessments. This information was identified from a screening of each of the identified citations using available abstracts, key words, and other available details. In addition, the relevance of studies was also assessed by consulting the available reviews, detailed above. Nonetheless, the selection of studies for inclusion into this risk assessment is an admittedly judgmental process. In order to maintain transparency, this risk assessment is accompanied by a complete bibliography of all studies encountered in the literature search. This bibliography is included as Attachment 1 and indicates which documents were retrieved.

For glyphosate specifically, there is a body of literature on the development, use, and safety of genetically-modified glyphosate tolerant crops such as soybeans (Shirai et al. 1998) and tobacco (Ye et al. 2001). This document, which concerns the non-agricultural use of glyphosate by the Forest Service, does not address any issues concerning crops that are genetically-modified for glyphosate tolerance.

The Forest Service will update this and other similar risk assessments on a periodic basis and welcomes input from the general public on the selection of studies included in the risk assessment. This input is helpful, however, only if recommendations for including additional studies in the body of these risk assessments specify why and/or how the new or not previously included information would be likely to alter the conclusions reached in the risk assessments.

For the most part, the risk assessment methods used in this document are similar to those used in risk assessments previously conducted for the Forest Service as well as risk assessments conducted by other government agencies. Details regarding the specific methods used to prepare the human health risk assessment are provided in SERA (2001a).

Risk assessments are usually expressed with numbers; however, the numbers are far from exact. *Variability* and *uncertainty* may be dominant factors in any risk assessment, and these factors should be expressed. Within the context of a risk assessment, the terms *variability* and *uncertainty* signify different conditions.

*Variability* reflects the knowledge of how things may change. Variability may take several forms. For this risk assessment, three types of variability are distinguished: statistical, situational, and arbitrary. Statistical variability reflects, at least, apparently random patterns in data. For example, various types of estimates used in this risk assessment involve relationships of certain physical properties to certain biological properties. In such cases, best or maximum likelihood estimates can be calculated as well as upper and lower confidence intervals that reflect the statistical variability in the relationships. Situational variability describes variations depending on known circumstances. For example, the application rate or the applied concentration of a herbicide will vary according to local conditions and goals. As discussed in the following section, the limits on this variability are known and there is some information to indicate what the variations are. In other words, situational variability is not random. Arbitrary variability, as the name implies, represents an attempt to describe changes that cannot be characterized statistically or by a given set of conditions that cannot be well defined. This type of variability dominates some spill scenarios involving either a spill of a chemical on to the surface of the skin or a spill of a chemical into water. In either case, exposure depends on the amount of chemical spilled and the area of skin or volume of water that is contaminated.

*Variability* reflects a knowledge or at least an explicit assumption about how things may change, while *uncertainty* reflects a lack of knowledge. For example, the focus of the human health dose-response assessment is an estimation of an "acceptable" or "no adverse effect" dose that will not be associated with adverse human health effects. For glyphosate and for most other chemicals, however, this estimation regarding human health must be based on data from experimental animal studies, which cover only a limited number of effects. Generally, judgment is the basis for the methods used to make the assessment. Although the judgments may reflect a consensus (i.e., be used by many groups in a reasonably consistent manner), the resulting estimations of risk cannot be proven analytically. In other words, the estimates regarding risk involve uncertainty. The primary functional distinction between variability and uncertainty is that variability is expressed quantitatively, while uncertainty is generally expressed qualitatively.

In considering different forms of variability, almost no risk estimate presented in this document is given as a single number. Usually, risk is expressed as a central estimate and a range, which is sometimes very large. Because of the need to encompass many different types of exposure as

well as the need to express the uncertainties in the assessment, this risk assessment involves numerous calculations.

Most of the calculations are relatively simple, and the very simple calculations are included in the body of the document. Some of the calculations, however, are cumbersome. For those calculations, a set of worksheets is included as an attachment to the risk assessment. The worksheets provide the detail for the estimates cited in the body of the document. The worksheets are divided into the following sections: general data and assumptions, chemical specific data and assumptions, exposure assessments for workers, exposure assessments for the general public, and exposure assessments for effects on nontarget organisms. The worksheets are included at the end of this risk assessment and further documentation for these worksheets are included as Attachment 2 (SERA 2001b). As detailed in Attachment 2, two versions of the worksheets that are in the spreadsheet format are used only as a check of the worksheets that are in the spreadsheet format are used only as a check of the worksheets that are in the spreadsheet format are used only as a check of the worksheets that are in the spreadsheet format are used only as a check of the worksheets that are in the spreadsheet format are used only as a check of the worksheets that are in the word processing format. Both sets of worksheets are provided with the hard-text copy of this risk assessment as well as with the electronic version of the risk assessment.

#### 2. PROGRAM DESCRIPTION

#### **2.1. OVERVIEW**

Glyphosate is a herbicide that is used in Forest Service programs primarily in conifer release, noxious weed control, and site preparation. There are currently 35 commercial formulations of glyphosate that are registered for forestry applications. All commercial formulations of glyphosate that are used in forestry applications contain the isopropylamine salt of glyphosate. Some formulations contain only this salt of glyphosate as an aqueous solution. Other formulations contain surfactants and some of these surfactants are toxic, particularly to aquatic organisms. Technical grade glyphosate contains an impurity, N-nitrosoglyphosate, but the amount of this impurity in glyphosate has been classified as toxicologically insignificant by the U.S. EPA. A surfactant used in at least one major commercial formulation contains 1,4-dioxane, and the toxicity of this impurity is specifically considered in this risk assessment. The most common method of application for glyphosate in Forest Service programs involves is backpackapplied directed foliar sprays. Other application methods that are use occasionally are broadcast foliar ground applications, cut stem applications, and direct application to the emergent aquatic vegetation. Based on recent Forest Service use reports, the typical application rate is about 2 lb a.e./acre, with most application rates occurring over a range of 0.5 lbs a.e./acre to 7 lbs a.e./acre. The total annual use of glyphosate by the Forest Service is only about 0.275 percent of the agricultural use. Thus, there is no basis for asserting that Forest Service programs will substantially contribute to general concentrations of glyphosate nationally. The potential for local contamination of environmental media by the use of glyphosate in Forest Service programs is discussed in the human health risk assessment and the ecological risk assessment.

#### 2.2. CHEMICAL DESCRIPTION AND COMMERCIAL FORMULATIONS

Glyphosate is the common name for N-(phosphonomethyl)glycine:

$$\begin{array}{c} O & O \\ \parallel \\ HO - C - CH_2 - NH_2 - CH_2 - P - OH \\ \downarrow \bigcirc \\ O \end{array}$$

Selected chemical and physical properties of glyphosate are summarized in Table 2-1. Additional information is presented in worksheet B03. At ambient temperatures, glyphosate is a white crystalline substance. In the crystalline form, glyphosate has both positive and negative regions of charge, indicated by the circled plus (+) and minus (-) signs in the schematic above. Such dipolar ion species are sometimes referred to as a *zwitterions*. In aqueous solutions, the hydrogen atoms of the carboxylic acid (COOH) and phosphate ( $PO_2H_2$ ) groups may be associated (e.g., -COOH) or dissociated (e.g., -COO<sup>-</sup> + H<sup>+</sup>) depending on the pH of the solution. The dissociation constants, or pK<sub>a</sub> values, for these reactions are illustrated in Figure 2-1. The pH of most

biological fluids range from approximately 5 to 9. Thus, within this range of pH, glyphosate has a net negative charge and is predominantly in form of  $H_2G^{-1}$  or  $HG^{-2}$ , as illustrated in Figure 2-1.

Technical grade glyphosate contains an impurity, N-nitrosoglyphosate, which is sometimes abbreviated as NNG. Specific information on nitrosamine concentrations in glyphosate formulations have been submitted to U.S. EPA (e.g., Hirsch and Augustin 1987). This information has been reviewed in the preparation of the current risk assessment but specific details may not be disclosed in this risk assessment except for information that has been made publically available by U.S. EPA. The U.S. EPA has determined that 92% of technical grade glyphosate contains NNG at less than one part per million (<1 mg/L) and that this amount is toxicologically insignificant.

Similarly, a surfactant used in Roundup contains 1,4-dioxane as an impurity. The upper limit of this compound in Roundup is about 0.03% (Monsanto 1990). The toxicologic significance of these compounds are discussed in Section 3.1.9.

The previous USDA Forest Service risk assessment of glyphosate (SERA 1996) had covered only four formulations: Roundup, Roundup Pro, Accord, and Rodeo. Since the preparation of the 1996 risk assessment, glyphosate has come off patent protection, resulting in a rapid increase in registrants and formulations. Thus, there are currently 35 commercial formulations of glyphosate that are registered for forestry applications. Each of the formulations are detailed in Appendix 1. All commercial formulations of glyphosate that are used in forestry applications contain the isopropylamine salt of glyphosate.

Several liquid formulations of glyphosate appear to contain either glyphosate alone in water (e.g., Aquamaster, Glyfos Aquatic, Roundup Custom, Rodeo). Other liquid formulations of glyphosate in a tallow amine surfactant (e.g., Credit, Glyfos, Glyfos Pro, Glyfos X-TRA, Honcho, Mirage, Razor, Roundup ORIGINAL, Roundup PRO). Some formulations of glyphosate do not indicate whether or not surfactants are present in the formulation.

The specific identity of the surfactants, other inerts, contaminants, and impurities has been disclosed to the U.S. EPA as part of the registration process and this information has been reviewed in the preparation of this risk assessment (e.g., Barclay 1987a,b,c; Benard 2002; Brakel 1999; Buchanan 1998; Fickel and Mahlburg 1999; Friis 1995a,b,c; Lystbaek 1994; Mierkowski 1999; Miller 2002a,b,c; Stevens 2000a,b,c; Taylor 1993; Wratten 1999). This information includes information on the manufacturing process, identity and quantity of the inerts/impurities in the formulations as well as additional information on the composition of some inerts that are themselves complex mixtures. This information may not be disclosed in this risk assessment because it is classified as *trade secret* under Sections 10(f) and 12(a)(2)(D) of the Federal Insecticide, Fungicide and Rodenticide Act (FIFRA). While this is a limitation in terms of the ability to communicate specific types of information to the general public and other interested parties, some information is also available in the open literature and is discussed in this risk assessment.

Inerts are classified by the U.S. EPA as ranging from inerts of toxicologic concern (List 1) to inerts of minimal concern (List 4) (U.S. EPA/OPP 1998b). Some inerts - i.e., those listed under SARA Title III, Section 313 - are specified on the product material safety data sheets, as specified in Appendix 1, and can be publicly disclosed. Most formulations of glyphosate that do not contain a surfactant indicate that a nonionic surfactant should be added to the field solution prior to application. The concentration of surfactants in glyphosate formulations tends to range from about 1% to 11% and with the most common class of surfactants being characterized as polyoxyethyleneamines or POEA (Acquavella et al. 1999a). As detailed in both the human health risk assessment (Section 3) and the ecological risk assessment (Section 4), the use of glyphosate with a surfactant may substantially impact risk.

#### 2.3. APPLICATION METHODS

Glyphosate formulations may be applied by directed foliar, broadcast foliar, or aerial methods. The most common method of application for glyphosate in Forest Service programs involves is backpack-applied directed foliar sprays. In directed foliar applications, the herbicide sprayer or container is carried by backpack and the herbicide is applied to selected target vegetation. Application crews may treat up to shoulder high brush, which means that chemical contact with the arms, hands, or face is plausible. To reduce the likelihood of significant exposure, application crews are directed not to walk through treated vegetation. Usually, a worker treats approximately 0.5 acre/hour with a plausible range of 0.25-1.0 acre/hour. Glyphosate may also be applied in hack and squirt applications, in which the bark and cambium of a standing tree is cut with a hatchet and the herbicide is then applied to the cut using a squirt bottle. This treatment is used to eliminate large trees during site preparation, conifer release operations, or rights-of-way maintenance. As with selective foliar applications, a worker usually will treat approximately 0.5 acre/hour with a plausible range of 0.25-1.0 acres/hour.

Broadcast foliar ground applications may occasionally be conducted and involve the use of a two to six nozzle boom mounted on a tractor or other heavy duty vehicle. With this equipment, workers will typically treat 11 to 21 acres per hour, with the low end of this range representative of a four-wheel drive vehicle in tall grass and the upper end of the range representative of a large bulldozer (USDA 1989b p 2-9 to 2-10).

Two additional application methods are also used occasionally in Forest Service programs. The first is a cut stem application. This is used in the treatment of some noxious weeds, notably *Arundo donax*. One of the more common methods for treating that plant is to cut the stem, and then spray or paint on some undiluted Foresters' Non-Selective Herbicide or Glyphos Pro to the cut stump surface. The other method of application is an aquatic application for aquatic noxious weeds. This can involve the application of glyphosate, often with a surfactant, to the emergent vegetation - i.e., above-surface plant parts. Glyphosate is not applied over open water in Forest Service programs.

As indicated in Appendix 1, some glyphosate formulations (e.g., Accord SP and Glypro) are labeled for aerial applications. Liquid formulations of glyphosate are applied through specially

designed spray nozzles and booms. The nozzles are designed to minimize turbulence and maintain a large droplet size, both of which contribute to a reduction in spray drift. Aerial applications may only be made under meteorological conditions that minimize the potential for spray drift. In aerial applications, approximately 40–100 acres may be treated per hour.

In some instances, areas treated with glyphosate may be subject to brown-and-burn operations. As indicated in USDA (1989b), these operations involve burning a treated area 45–180 days after treatment with the herbicide.

#### 2.4. USES AND APPLICATION RATES

The use of glyphosate in Forest Service Programs for fiscal year 2001, the most recent year for which data are available, is detailed in Appendix 2 and summarized in Table 2-2. Glyphosate is used in Forest Service Programs primarily in conifer release (58.2%), noxious weed control (15.1%), and site preparation (16.4%). Other minor uses (totaling 10.3% of use) include other weed control (agricultural, aquatic, and nursery), hardwood release, facilities maintenance, recreation improvement, right-of-way management, seed orchard protection, and wildlife habitat improvement. The application rates are about 3.31 lb/acre (SD 1.77) for conifer release, 1.06 lb/acre (SD 2.14) for noxious weed control, and 3.34 lb/acre (SD 1.76) for site preparation.

For this risk assessment, the average application rate will be taken as 2 lb a.e./acre. This is an essentially arbitrary selection but is reasonably close to the overall average application rate used in all types of program activities (Table 2-2). This application rate is used in the worksheets that accompany this risk assessment. The range of application rates will be taken as 0.5 lbs a.e./acre to 7 lbs a.e./acre to reflect plausible ranges that the Forest Service may use. While the upper range is somewhat below the highest reported application rate of 9.96 lb/acre (i.e., Forest 7 in Region 5), it is likely to be more representative of programs that the Forest Service is likely to conduct in the future. This range of application rates is not used in the worksheets but the consequences of varying application rates within this range is considered in the risk characterization for human health (Section 3.4) and ecological effects (Section 4.4).

For this risk assessment, the extent to which a formulation of glyphosate is diluted prior to application primarily influences dermal and direct spray scenarios, both of which are dependent on 'field dilution'(i.e., the concentration of glyphosate in the applied spray). In all cases, the higher the concentration of glyphosate - equivalent to the lower dilution of glyphosate - the greater the risk. For this risk assessment, the lowest dilution is taken as 5 gallons/acre. The highest dilution is based on 25 gallons of water per acre. A typical dilution rate is taken as 10 gallons/acre. Details regarding the calculation of field dilution rates are given in worksheet B01, and the calculations following this worksheet are summarized in worksheet B02.

It should be noted that the selection of application rates and dilution volumes in this risk assessment is intended to simply reflect typical or central estimates as well as plausible lower and upper ranges. In the assessment of specific program activities, the Forest Service will use

program specific application rates in the worksheets that are included with this report to assess any potential risks for a proposed application.

### 2.5. USE STATISTICS

The USDA Forest Service (USDA/FS 2002) tracks and reports use by geographical areas referred to as "*Regions*". As illustrated in Figure 2-2, the Forest Service classification divides the U.S. into nine regions designated from Region 1 (Northern) to Region 10 (Alaska). [Note: There is no *Region 7* in the Forest Service system.] As illustrated in Figure 2-2 and detailed further in Table 2-3, the heaviest used of glyphosate occurs in Region 5 (Pacific Southwest) in terms of the number of acres treated, the number of pounds used, and the application rate. Substantial glyphosate use also occurs in Regions 8 (Southern, 7.6%) and 9 (Eastern 9.3%) with moderate use in Region 6 (Pacific Northwest, 3.8%). Glyphosate use by the Forest Service in other regions is insubstantial – i.e., less than 2% of total.

Many formulations of glyphosate are used extensively in agriculture. A summary of the agricultural use of glyphosate is presented in Figure 2-3 (USGS 1998). These use statistics are for 1992, the most recent year for which data are available. As indicated in this figure, over 16,000,000 lbs of glyphosate are applied to crops annually, primarily to soybeans, corn, and citrus in the mid-west and in California. As noted in Table 2-3, the total annual use of glyphosate by the Forest Service for 2001 is about 45,000 lbs, which is 0.275 percent of the agricultural use. While the use of glyphosate by the Forest Service is not trivial, this use is less than that of agricultural uses by a factor of over 300. Thus, there is no basis for asserting that Forest Service programs will substantially contribute to general concentrations of glyphosate in Forest Service programs is discussed in detail in the human health risk assessment (Section 3) and the ecological risk assessment (Section 4).



**Figure 2-1**. Structure and dissociation constants (pK<sub>a</sub>) for the various forms of glyphosate.



**Figure 2-2**. Use of glyphosate by the USDA Forest Service in various regions of the United States.



Figure 2-3. Agricultural use of glyphosate in the United States for 1992 (USGS 1998).

CAS Number:	1071-83-6
Molecular weight:	169.07 (Budavari 1989)
Melting point (°C):	200 (USDA/ARS 1995)
Vapor pressure (mm Hg):	$< 7 \text{ x } 10^{-9} \text{ mm Hg} (25^{\circ} \text{ C}) (\text{Weber } 1991)$
Water solubility:	12 g/L (25°C) (USDA/ARS 1995)
÷	900 g/L, amine salt (Knisel et al. 1992)
	11.6 g/L (Schuette 1998)
Log K <sub>ow</sub> :	-3.39 (pH 1.77) (Chamberlain et al. 1996)
	-4.38 (pH 4.61) (Chamberlain et al. 1996)
	-4.85 (pH 6.86) (Chamberlain et al. 1996)
	-4.14 (pH 9.00) (Chamberlain et al. 1996)
	-3.5 (Schuette 1998)
Soil:water part. coef. K <sub>d</sub> :	61 g/m <sup>3</sup> (Schuette 1998)
	2100 (500 - 2600) (USDA/ARS 1995)
Soil adsorption K <sub>oc</sub> (L/kg):	554–34,000 (Piccolo et al; 1994)
	2,600–4,900 (Glass 1987)
	8 to >500,000 (Gerritse et al. 1996)
	54 (Knisel et al. 1992)
Foliar half-life (days):	$\simeq 1.6$ (Thompson et al. 1994)
	8–10 (Feng and Thompson 1990)
	10.6–26.6 (Newton et al. 1984)
	2.5 (Knisel et al. 1992)
	46 (Siltanen et al. 1981)
Soil half-life (days):	20–40 (Weber 1991)
	<60 (average) (WSSA 1989)
	45–60 (Feng and Thompson 1990)
	29–40 (Newton et al. 1984)
	30-40 (Smith and Aubin 1993)
	47 (Knisel et al. 1992)
	22.1 aerobic, 96.4 anaerobic (Schuette 1998)
	2.8-30 (dissipation, Hatfield 1996)
	37(2-174) for field dissipation (USDA/ARS 1995)
Water half-life (days):	50–70 (U.S. EPA/ODW 1992)
	14 (minimum rate) (Reinert and Rodgers 1987)
	42–70 (Reinert and Rodgers 1987)
	> 35 (Schuette 1998)
	3.5–11.2 days [surface water; some glyphosate in the water column
	was transferred to sediment and not degraded] (Goldsborough and
	Brown 1993)
Foliar wash-off fraction	0.6 (Knisel et al. 1992)

# Table 2-1. Physical, chemical, and biochemical properties of glyphosate
Use Classification	Acres	Pounds	Propor- tion of Use by Pounds	lbs/acre
Agricultural Weed Control	4,172.60	628.50	0.014	3.64
Aquatic Weed Control	4.50	4.70	0.000	1.04
Conifer and Hardwood Release	954.10	1,501.20	0.034	1.57
Conifer Release	7,855.00	26,032.50	0.582	3.31
Facilities Maintenance	49.80	48.64	0.001	0.98
Noxious Weed Control	6,368.60	6,768.54	0.151	1.06
Nursery Weed Control	247.30	718.15	0.016	2.90
Recreation Improvement	78.50	39.68	0.001	0.51
Right-of-Way	577.70	895.48	0.020	1.55
Seed Orchard Protection	16.00	10.00	0.000	0.63
Site Preparation	2,189.00	7,319.76	0.164	3.34
Wildlife Habitat Improvement	508.00	753.50	0.017	1.48
Grand Total	19,021.10	44,720.60	1	2.35

Table2-2: Use of glyphosate by USDA Forest Service in 2001 by Type of Use (USDA/FS 2002)

Forest	Acres	Pounds	lbs/acre	Proportion of Total Acres	Proportion of Total Lbs.
Northern (R1)	133.00	264.00	1.99	0.007	0.006
Rocky Mountain (R2)	264.00	182.00	0.69	0.014	0.004
Southwestern (R3)	3.00	3.00	1.00	0.000	0.000
Intermountain (R4)	405.00	261.00	0.64	0.021	0.006
Pacific Southwest (R5)	8,395.00	34,740.00	4.14	0.441	0.777
Pacific Northwest (R6)	1,003.00	1,706.00	1.70	0.053	0.038
Southern (R8)	3,888.00	3,419.00	0.88	0.204	0.076
Eastern (R9)	4,930.00	4,146.00	0.84	0.259	0.093
Total	19,021.00	44,721.00	2.35		

Table 2-3: Use of glyphosate by USDA Forest Service in 2001 by Region (USDA/FS 2002)

#### 3. HUMAN HEALTH RISK ASSESSMENT

## **3.1. HAZARD IDENTIFICATION**

**3.1.1. Overview.** The herbicidal activity of glyphosate is due primarily to the inhibition of the shikimate pathway which is involved in the synthesis of aromatic amino acids in plants and microorganisms. This metabolic pathway does not occur in humans or other animals and thus this mechanism of action is not directly relevant to the human health risk assessment. Two specific biochemical mechanisms of action have been identified or proposed for glyphosate: uncoupling of oxidative phosphorylation and inhibition of hepatic mixed function oxidases. Both glyphosate and the polyethoxylated tallow amine (POEA) surfactant used in Roundup will damage mucosal tissue, although the mechanism of this damage is likely to differ for these two agents. Many of the effects of acute oral exposure to high doses of glyphosate or Roundup are consistent with corrosive effects on the mucosa.

The available experimental studies indicate that glyphosate is not completely absorbed after oral administration and is poorly absorbed after dermal applications. Two dermal absorption studies have been published on glyphosate and both of these studies indicate that glyphosate is very poorly absorbed across the skin.

Like all chemicals, glyphosate as well as commercial formulations of glyphosate may be toxic at sufficiently high exposure levels. In rats and mice, acute oral  $LD_{50}$  values of glyphosate range from approximately 2,000 to 6,000 mg/kg. Formulations of glyphosate with a POEA surfactant have been used in many suicides and attempted suicides. Gastrointestinal effects (vomiting, abdominal pain, diarrhea), irritation, congestion, or other forms of damage to the respiratory tract, pulmonary edema, decreased urinary output sometimes accompanied by acute renal tubular necrosis, hypotension, metabolic acidosis, and electrolyte imbalances, probably secondary to the gastrointestinal and renal effects, are seen in human cases of glyphosate/surfactant exposure.

One of the more consistent signs of subchronic or chronic exposure to glyphosate is loss of body weight. This effect has been noted in mice, rats, dogs, and rabbits. This observation is consistent with experimental data indicating that glyphosate may be an uncoupler of oxidative phosphorylation. Other signs of toxicity seem general and non-specific. A few studies report changes in liver weight, blood chemistry that would suggest mild liver toxicity, or liver pathology. Changes in pituitary weight have also been observed. Signs of kidney toxicity, which might be expected based on the acute toxicity of glyphosate, have not been reported consistently and are not severe. Various hematological changes have been observed that may be secondary to mild dehydration.

Glyphosate has been specifically tested for neurotoxicity in rats after both acute and subchronic exposures and has been tested for delayed neurotoxicity in hens. In both the animal data as well as the clinical literature involving suicide attempts, there is no clear pattern suggestive of a specific neurotoxic action for glyphosate or its commercial formulations. The weight of evidence

suggests that any neurologic symptoms associated with glyphosate exposures are secondary to other toxic effects. No studies are reported that indicate morphologic abnormalities in lymphoid tissues which could be suggestive of an effect on the immune system. As discussed in the ecological risk assessment, one study has asserted that glyphosate causes immune suppression in a species of fish. This study, however, is deficient in several respects and does not provide a basis for impacting the hazard identification for effects on the immune system.

Only three specific tests on the potential effects of glyphosate on the endocrine system have been conducted and all of these tests reported no effects. All of these assays are *in vitro* – i.e., not conducted in whole animals. Thus, such studies are used qualitatively in the hazard identification to assess whether there is a plausible biologic mechanism for asserting that endocrine disruption is plausible. Because they are *in vitro* assays, measures of *dose* and quantitative use of the information in dose/response assessment is not appropriate. For glyphosate, these studies to not indicate a basis for suggesting that glyphosate is an endocrine disruptor. Nonetheless, glyphosate has not undergone an extensive evaluation for its potential to interact or interfere with the estrogen, androgen, or thyroid hormone systems. Thus, the assessment of the potential endocrine effects of glyphosate cannot be overly interpreted.

Glyphosate has been subject to multi-generation reproduction studies which measure overall effects on reproductive capacity as well as teratology studies which assay for a compounds ability to cause birth defects. Signs of teratogenic activity have not been observed in standard assays in both rats and rabbits. In a multi-generation reproduction study in rats, effects on the kidney were observed in male offspring. This effect is consistent with the acute systemic toxicity of glyphosate, rather than a specific reproductive effect. Several other subchronic and chronic studies of glyphosate have been conducted with no mention of treatment-related effects on endocrine glands or reproductive organs. A single study has reported substantial decreases in libido, ejaculate volume, sperm concentrations, semen initial fructose and semen osmolality as well as increases in abnormal and dead sperm in rabbits after acute oral exposures to glyphosate. This study is inconsistent with other studies reported on glyphosate and is poorly documented – i.e., specific doses administered to the animals are not specified. In addition, the use of gelatin capsules, as in this study results, in a high spike in body burden that is not typical or particularly relevant to potential human exposures – other than attempted suicides. Numerous epidemiological studies have examined relationships between pesticide exposures or assumed pesticide exposures in agricultural workers and reproductive outcomes. Of those studies that have specifically addressed potential risks from glyphosate exposures, adverse reproductive effects have not been noted.

Based on standard animal bioassays for carcinogenic activity *in vivo*, there is no basis for asserting that glyphosate is likely to pose a substantial risk. The Re-registration Eligibility Decision document on glyphosate prepared by the U.S. EPA indicates that glyphosate is classified as Group E: Evidence of non-carcinogenicity for humans. This classification is also indicated in U.S. EPA's most recent publication of tolerances for glyphosate and is consistent with an assessment by the World Health Organization. This assessment has been challenged based on some studies

that indicate marginal carcinogenic activity. As with any compound that has been studied for a long period of time and tested in a large number of different systems, some equivocal evidence of carcinogenic potential is apparent and may remain a cause of concern, at least in terms of risk perception. While these concerns are understandable, there is no compelling basis for challenging the position taken by the U.S. EPA and no quantitative risk assessment for cancer is conducted as part of the current analysis.

Glyphosate formulations used by the Forest Service are classified as either non-irritating or only slightly irritating to the skin and eyes in standard assays required for product registration. Based on a total of 1513 calls to a poison control center reporting ocular effects associated with the use of Roundup, 21% were associated with no injury, 70% with transient minor injury, 2% with some temporary injury. The most frequently noted symptoms included blurred vision, a stinging or burning sensation, lacrimation. No cases of permanent damage were reported.

Various glyphosate formulations contain a POEA surfactant at a level of up to about 20%. Other formulations of glyphosate recommend the use of a surfactant to improve the efficacy of glyphosate. While surfactants are typically classified as "inert" ingredients in herbicides, these compounds are not toxicologically inert and some surfactants may be more toxic than the herbicides with which they are used. Although surfactants may play a substantial role in the interpretation of a large number of suicides and attempted suicides involving the ingestion of glyphosate formulations, primarily Roundup, the acute mammalian toxicity of different glyphosate formulations do not appear to differ substantially. This is in contrast to the available data on the toxicity of various formulations to aquatic species, as detailed in the ecological risk assessment.

**3.1.2. Mechanisms of Action.** While the mechanism of action of glyphosate in plants is well-characterized, the mechanism by which glyphosate exerts toxic effects in humans or experimental mammals is not clear. Two specific biochemical mechanisms of action have been identified or proposed: uncoupling of oxidative phosphorylation and inhibition of hepatic mixed function oxidases. In addition, both glyphosate and the POEA surfactant used in Roundup will damage mucosal tissue, although the mechanism of this damage is likely to differ for these two agents.

The herbicidal activity of glyphosate is due primarily to the inhibition of the shikimate pathway which is involved in the synthesis of aromatic amino acids in plants and microorganisms (Section 4.1). This metabolic pathway does not occur in humans or other animals and thus this mechanism of action is not directly relevant to the human health risk assessment. Nonetheless, shikimate pathway inhibitors have been considered as antimicrobial agents for the control of pathogens (Roberts et al. 1998; Roberts et al. 2002; Schonbrunn et al. 2001) and glyphosate has been shown to be effective in prolonging survival in mice infected with a pathogen, *Cryptococcus neoformans* (Nosanchuk et al. 2001).

Oxidative phosphorylation is a fundamental metabolic process in which metabolic energy derived from the oxidation of nutrients is transferred to and stored in high-energy phosphate bonds. The uncoupling of this process results in energy loss in the organism and lead to death. Symptoms of

uncouplers of oxidative phosphorylation include increased heart rate (tachycardia), increased respiratory rate, labored breathing, profuse sweating, fever, metabolic acidosis, and weight loss (ATSDR 2001).

Based on a series of experiments using rat liver mitochondria exposed to the isopropanolamine salt of glyphosate without any surfactant (summarized in detail by U.S. EPA 1992), glyphosate appears to be an uncoupler of oxidative phosphorylation (Bababunmi et al. 1979, Olorunsogo 1982, Olorunsogo and Bababunmi 1980, Olorunsogo et al. 1977, Olorunsogo et al. 1979a,b). This effect has been noted after intraperitoneal doses as low as 15 mg/kg (Olorunsogo et al. 1979a). Many of the observations on whole animals and isolated mitochondria are consistent with an uncoupling of oxidative phosphorylation including decreased body weight, decreased food conversion efficiency and increased body temperature (Section 3.1.3). It is less clear that uncoupling of oxidative phosphorylation is a significant factor in acute exposures. Of the 97 patients covered in the Tominack et al. (1991) report, only seven individuals had mild elevations in body temperature (>99.5°F). In addition, acute gavage doses of 50, 100, or 200 mg glyphosate a.e./kg in rats was associated with hypothermia (a decrease in body temperature) rather than hyperthermia (Horner 1996a).

The other specific mechanism of action that may account for some the effects of glyphosate involves the inhibition of hepatic mixed-function oxidases. This is a class of enzymes comprised of various isozymes of cytochrome P-450 that is involved in the metabolism of a wide variety of endogenous compounds as well as xenobiotics. Decreases in hepatic mixed function oxidase activity in rats has been noted after doses of glyphosate (as Roundup 360 g/L) of 500 mg/kg/day for four days followed by doses of 300 mg/kg/day for 6 days (Hietanen et al. 1983). This decrease in mixed function oxidase activity is only suggestive of cytochrome P-450 inhibition in that a general decrease in mixed function oxidase activity could also be caused by direct liver damage. Nonetheless, it seems reasonable to suggest that this effect may have been caused by P-450 inhibition by glyphosate because glyphosate has been shown to inhibit cytochrome P-450 in plants (Lamb et al. 1998).

Many of the effects of acute oral exposure to high doses of glyphosate or Roundup are consistent with corrosive effects on the mucosa. Glyphosate, the POEA surfactant in Roundup, as well as Roundup itself all cause corrosive effects on the gastric mucosa as well as other tissue (Chang et al. 1999; Hung et al. 1997). While somewhat speculative, it is likely that the mechanisms for this effect differ between glyphosate and the POEA surfactant. As indicated in Section 2, glyphosate is a zwitterion that will have a net negative charge and can be expected to act as an acid at physiological pH. Thus, the effects of glyphosate on mucosal tissue may be due to the acidic action of glyphosate, similar to the effects of high concentrations of hydrochloric acid in dog (Talbot et al. 1991). As detailed in Section 3.1.11, the POEA surfactant behaves essentially like a soap to dissolve cell membranes.

**3.1.3. Pharmacokinetics and Metabolism.** The available experimental studies indicate that glyphosate is not completely absorbed after oral administration and is poorly absorbed after

dermal applications. Much of the early literature on the pharmacokinetics and metabolism of glyphosate is reviewed by WHO (1994) and Williams et al. (2000). After oral exposure only about 30% of glyphosate is absorbed from the gastrointestinal tract. This has also been noted in more recent studies by Davies (Davies 1996a; Davies 1996b; Davies 1996c; Davies 1996d; Davies 1996e). Although most unabsorbed glyphosate remains in the gastrointestinal tract (Davies 1996e), absorbed glyphosate is widely distributed in the body. Although glyphosate does not substantially concentrate and persist in any tissue, concentrations in the bone tend to be higher than in other tissue (WHO 1994; Davies 1996e). The only known metabolite of glyphosate is AMPA. While this is a common environmental metabolite formed in the degradation of glyphosate (Section 1.3.14.1), only trace amounts of AMPA are formed in mammals (Macpherson 1996; U.S. EPA/ODW 1992; WHO 1994) and most of the administered dose of glyphosate (>97%) is excreted unchanged (Williams et al. 2000).

As discussed further in the dose-response assessment (Section 3.3), dose levels expressed in mg/kg/day cause comparable effects over broad periods of exposure, which is consistent with the rapid elimination of and lack of toxic metabolites from glyphosate (Brewster et al. 1991, Monsanto Co. 1993a,b; NTP 1992).

Most of the occupational exposure scenarios and many of the exposure scenarios for the general public involve the dermal route of exposure. For these exposure scenarios, dermal absorption is estimated and compared to an estimated acceptable level of oral exposure based on subchronic or chronic toxicity studies. Thus, it is necessary to assess the consequences of dermal exposure relative to oral exposure and the extent to which glyphosate is likely to be absorbed from the surface of the skin. Two types of dermal exposure scenarios are considered: immersion and accidental spills. As detailed in SERA (2001a), the calculation of absorbed dose for dermal exposure scenarios involving immersion or prolonged contact with chemical solutions use Fick's first law and require an estimate of the permeability coefficient, K<sub>p</sub>, expressed in cm/hour. For exposure scenarios like direct sprays or accidental spills, which involve deposition of the compound on the skin's surface, dermal absorption rates (proportion of the deposited dose per unit time) rather than dermal permeability rates are used in the exposure assessment.

Two dermal absorption studies (Wester et al. 1991; Wester et al. 1996) have been published on glyphosate and both of these studies indicate that glyphosate is very poorly absorbed across the skin. Wester et al. (1991) assayed the dermal absorption of <sup>14</sup>C-labeled glyphosate in a Roundup formulation in both an *in vitro* system using skin from human cadavers and in the *in vivo* study in monkeys. *In vitro* skin preparations were exposed to undiluted Roundup formulations for up to 8 hours and 1:20 and 1:32 dilutions of Roundup similarly treated for up to 16 hours (Wester et al. 1991, Table 1, p. 728). Based on the 16 hour exposures to the dilute solutions, first-order dermal absorption rates ranged from  $1.3 \times 10^{-4}$  to  $1.0 \times 10^{-3}$  hour<sup>-1</sup> with an average value of  $4.1 \times 10^{-4}$  hour<sup>-1</sup>. Based on the 8 hour exposures to the concentrated Roundup, first-order dermal absorption rates ranged from  $7.5 \times 10^{-5}$  to  $5.0 \times 10^{-4}$  hour<sup>-1</sup>. Thus, glyphosate in undiluted Roundup – i.e., containing the POEA surfactant – does not appear to be more rapidly absorbed than glyphosate in a more dilute solution of the surfactant. The *in vivo* studies in monkeys indicated that about 1.5% of the

glyphosate was absorbed in 12 hours, corresponding to a first-order dermal absorption rate of  $1.3 \times 10^{-3}$  hour<sup>-1</sup> [ka = ln(1-proportion absorbed)/duration].

These experimental measurements of dermal absorption are very consistent with the standard methods used to estimate first-order dermal absorption rates (SERA 2001a). The details of the method specified in SERA (2001a) for estimating the first-order dermal absorption coefficient based on the molecular weight and octanol-water partition coefficient are given in worksheet A07a. The application of this method to glyphosate is detailed in worksheet B03 and yields a central estimate of  $7.43 \times 10^{-4}$  hour<sup>-1</sup> with a range of  $1.40 \times 10^{-4}$  to  $3.95 \times 10^{-4}$  hour<sup>-1</sup>.

Given the similarities between the estimated values of the first-order dermal absorption rates in worksheet B03 and the experimental values calculated from the study by Wester et al. (1991), the use of either set in this risk assessment makes relatively little difference. Nonetheless, the experimental values for human skin preparations from Wester et al. (1991) are used in all exposure assessments requiring first-order dermal absorption rates as specified in worksheet B05.

Wester et al. (1996) have reported a permeability coefficient,  $K_p$ , of  $4.59\pm1.56\times10^{-4}$  cm/hour with a lag time of 10.48 hours based on *in vitro* human skin preparation. As detailed in U.S. EPA/ORD (1992) different types of models with or without lag times may be used to estimate dermal permeability coefficients. Using the method recommended by U.S. EPA/ORD (1992), the estimated dermal permeability coefficient for glyphosate (excluding a lag time) is  $1.53\times10^{-6}$ cm/hour with a 95% confidence interval of  $3.47\times10^{-7}$  to  $6.27\times10^{-6}$  cm/hour. The details of the U.S. EPA/ORD (1992) method for estimating  $K_p$  based on the molecular weight and octanolwater partition coefficient are given in worksheet A07b. The application of this method to glyphosate is detailed in worksheet B04. Because of the differences in the underlying models – i.e., inclusion or exclusion of a lag time – the estimates based on the method of U.S. EPA/ORD (1992) and the experimental report from Wester et al. (1996) are not directly comparable. For this risk assessment, the simpler model and estimates from U.S. EPA/ORD (1992) are used.

This is a more conservative approach than using the lag-time model because washing with soap and water will effectively remove about 90% of glyphosate applied to the skin (Wester et al. 1994). Thus, under standard spill scenarios used in this risk assessment (Section 3.3), the pesticide is washed off the skin in less than the lag time noted by Wester et al. (1996) and hence exposure would be essentially zero, even though the Kp with lag-time reported by Wester et al. (1996) is substantially higher than the values estimated from the method of U.S. EPA/ORD (1992).

**3.1.4.** Acute Toxicity. Like all chemicals, glyphosate as well as commercial formulations of glyphosate may be toxic at sufficiently high exposure levels. In rats and mice, acute oral  $LD_{50}$  values of glyphosate range from approximately 2,000 to 6,000 mg/kg (Williams et al. 2000) and intraperitoneal  $LD_{50}$  values are about 10 times lower, ranging from 134 to 234 mg/kg (Bababunmi et al. 1978). As detailed further in Section 4.3.1. (Dose-Response Assessment for Terrestrial Animals), there appears to be no systematic differences in toxicity among species when doses of

glyphosate are expressed in units of mg/kg body weight. In experimental mammals, signs of acute toxicity after oral or intraperitoneal dosing include increased respiratory rates, elevated rectal temperature, and in some instances asphyxia convulsion. The primary pathological lesion is lung hyperemia (Bababunmi et al. 1978; Olorunsogo et al. 1977; Olorunsogo and Bababunmi 1980).

Formulations of glyphosate with a POEA surfactant have been used in many suicides and attempted suicides (Chang et al. 1999; Garcia-Repetto et al. 1998; Hung et al. 1997; Lee et al 2000; Lin et al. 1999; Temple and Smith 1992; Tominack et al. 1991; Sawada et al. 1988; Yang et al. 1997). Details of these studies are presented in Appendix 4. Gastrointestinal effects (vomiting, abdominal pain, diarrhea), irritation, congestion, or other forms of damage to the respiratory tract, pulmonary edema, decreased urinary output sometimes accompanied by acute renal tubular necrosis, hypotension, metabolic acidosis, and electrolyte imbalances, probably secondary to the gastrointestinal and renal effects, are seen in human cases of glyphosate/surfactant exposure.

In a recent analysis of poisoning incidents associated with suicides or attempted suicides in Taiwan (Lee et al. 2000), fatalities were associated with doses of glyphosate/surfactant formulations in the range of  $330\pm42$  mL and that survivors of poisonings were associated with doses of  $122\pm12$  mL. This is very similar to previous estimates of fatal and non-fatal doses (Talbot et al. 1991; Tominack et al. 1991). Assuming a body weight of about 60 kg, the lethal dose is about 5500 mg Roundup/kg bw [330 mL × ca. 1000 mg/mL ÷ 60 kg], very similar to the acute LD<sub>50</sub> of Roundup in rats.

As detailed in Section 3.1.11, the POEA surfactant use in glyphosate formulations (e.g., various formulations of Roundup) is a factor, and probably the dominant factor, in some of the effects seen in humans in cases of the suicidal ingestion of glyphosate formulations.

Glyphosate is commonly used as a model compound in the development of various in vitro screening methods (e.g., Bertheussen et al. 1997; El-Demerdash et al. 2001; Figenschau et al. 1997). Except as noted specifically in the following subsections, most of these studies do not contribute substantially to the risk assessment. Based on a clinical investigation of a poisoning incident, Sorensen and Gregersen (1999) have suggested that glyphosate IPA formulations may present a lower acute risk than glyphosate trimesium formulations such as Touchdown. While this may worth noting in terms of comparative risks across glyphosate formulations, only the IPA formulations used in Forest Service programs (Appendix 1) are specifically considered in this risk assessment.

**3.1.5.** General Subchronic or Chronic Systemic Toxicity. Systemic toxicity encompasses virtually any effects that a chemical has after the chemical has been absorbed. Certain types of effects, however, are of particular concern and involve a specific subset of tests. Such special effects are considered below for the nervous system (Section 3.1.6) and immune system (Section 3.1.7), development or reproduction (Section 3.1.8), and carcinogenicity or mutagenicity (Section 3.1.9). This section encompasses the remaining signs of general and non-specific toxicity.

One of the more consistent signs of subchronic or chronic exposure to glyphosate is loss of body weight. This effect has been noted in mice (U.S. EPA 1986a, NTP 1992), rats (Horner 1996b; Milburn 1996; NTP 1992, Stout and Ruecker 1990), dogs (Brammer 1996), and rabbits (Yousef et al. 1995). This observation is consistent with the work of Olorunsogo and coworkers, summarized in section 3.1.2, indicating that glyphosate may be an uncoupler of oxidative phosphorylation (U.S. EPA/ODW 1992, NTP 1992). Loss of body weight, particularly in studies using dietary exposure, can be secondary to decreased food consumption. In the NTP bioassay using mice, however, weight loss was noted at the two higher dose levels but there were no significant differences in food consumption between any of the treated groups and the control group. Similarly, in rabbits, the weight loss was not associated with a decrease in food consumption (Yousef 1995). In the NTP study using rats (NTP 1992), a slight decrease in food consumption was observed in the high dose group (50,000 ppm in the diet), which amounted to 91% of control values for females and 88% of control values for males. This behavior may account for the weight decrease in females, 95% of controls, and possibly for the weight decrease in males, 82% of controls. In the study by Horner (1996b), a dietary concentration of 20,000 ppm for 13 weeks resulted in a 12% decrease in body weight that could not be attributed to decreased food consumption. At a dietary concentration of 30,000 ppm, this effect has also been noted in dogs (Brammer 1996).

Other signs of toxicity seem general and non-specific. A few studies report changes in liver weight, blood chemistry that would suggest mild liver toxicity, or liver pathology (U.S. EPA 1986, NTP 1992, Stout and Ruecker 1990). Changes in pituitary weight have also been observed (Monsanto Co. 1985). Signs of kidney toxicity, which might be expected based on the acute toxicity of glyphosate, have not been reported consistently and are not severe (Monsanto Co. 1987, NTP 1992, U.S. EPA 1986). As summarized by NTP (1992), various hematological changes have been observed but are not considered severe and are attributed to mild dehydration.

**3.1.6. Effects on Nervous System.** Glyphosate has been specifically tested for neurotoxicity in rats after both acute (Horner, 1996a) and subchronic exposures (Horner, 1996b) and has been tested for delayed neurotoxicity in hens (Johnson 1997). It should be noted that the hen assay, while not involving a mammal, is the assay of choice for the hazard identification on agents causing delayed neurotoxicity (U.S. EPA/PPTS, 1998). In all three assays, glyphosate was negative for signs of neurotoxicity.

In the acute study by Horner (1996a), 10 male and 10 female rats were given doses of 50, 100, or 200 mg glyphosate a.e./kg and observed for two weeks. Initially – i.e., 6 hours after dosing – the animals exhibited decreased activity, subdued behavior, and hypothermia. However, there were no effects on landing foot splay, sensory perception, muscle strength, or locomotor activity and no abnormal histologic changes in the central or peripheral nervous system tissue. In the subchronic study (Horner, 1996b), groups of 12 male and 12 female rats were exposed to glyphosate in the diet at concentrations of 2000, 8000, or 20000 ppm for 13 weeks. Although effects were noted on growth and food utilization, there were no neurologic effects based on locomotor activity, no changes in brain weight or dimensions, and no evidence of damage to

nerve tissue (peripheral or central). In hens (n=20) given a single dose (gavage) of glyphosate at 2000 mg/kg, a slight decrease in brain AChE activity was observed but there were no signs of delayed locomotor ataxia and no signs of neuropathology (Johnson 1997).

Williams et al. (2000) also describe a study in which neurological examinations were conducted on dogs that received a single oral dose of 59 or 366 mg/kg of Roundup and the information is attributed to an unpublished study by Monsanto which is cited as Naylor (1988). This study was not identified in a search of the U.S. EPA/CBI files and has not been reviewed in preparation of this risk assessment. According to Williams et al. (2000):

"A detailed examination consisting of 12 different measurements of spinal, postural, supporting, and consensual reflexes was performed before treatment, during the postadministration observation period, and again on the following day. Reflexes appeared normal, and there were no clinical signs indicative of neuromuscular abnormalities."

In subchronic studies in mice and rats (NTP, 1992), morphological examinations were conducted of brain (including basal ganglia, a site of injury in Parkinsonism); however, it is unclear from the report whether or not spinal cord and sciatic nerve were examined. In any event, the NTP (1992) study did not report abnormal findings in these tissues, nor did it report clinical signs of neurotoxicity. The NTP (1992) study observed histological changes in salivary glands in both rats and mice. These changes were less severe in animals that received glyphosate in combination with a dosage of propranolol, an antagonist of  $\beta$ -adrenergic neurotransmitters. Propranolol also completely prevented similar changes produced by isoproterenol, a  $\beta$ -adrenergic agonist. NTP (1992) concluded from these results that glyphosate may have produced the salivary gland changes by acting through an adrenergic mechanism. This conclusion has been challenged as being difficult to reconcile with the absence of  $\beta$ -adrenergic effects (e.g., on heart rate and blood pressure) when glyphosate was administered intravenously to dogs or rabbits (Williams et al., 2000). However, it is possible that, rather than acting by a direct adrenergic mechanism, glyphosate could have produced an adrenergic-mediated stimulation of the salivary glands through some indirect mechanism exerted during prolonged repeated dosing. In a one-year feeding study of rats, Milburn (1996) also reports increased incidence of mild focal basophilia of the acinar cells of parotid salivary gland in both sexes at 20000 ppm but not at 2000 ppm. No signs of neurotoxicity, however, were noted and the mechanism of this effect is unclear.

Schiffman et al. (1995) conducted a study of the effects of glyphosate on taste response in gerbils. This study appears to be the only reported investigation of the effects of glyphosate on sensory mechanisms. Glyphosate (1 or 10 mM) applied to the tongue of anesthetized gerbils decreased taste receptor response to table salt, sugars, and acids. These tests on glyphosate involved exposure periods of one minute and were conducted along with tests on 10 other pesticides, with one minute rinses between each agent. The mechanism of this effect on the taste response has not been investigated and the implications in terms of dietary preferences in the field cannot be assessed. The effect could have been produced by a general biochemical alteration in the

epithelial cells of the tongue, including the specialized cells that detect taste (glyphosate has been shown to produce injury to the oral cavity), by chemical injury to the tongue, or by a direct neurotoxic effect on the sensory nerve endings. Thus, effects reported in Schiffman et al. (1995) cannot be classified clearly as a glyphosate-induced neurologic effect.

As with the animal data, no clear pattern suggestive of neurotoxicity is apparent in an extensive and detailed literature on health outcomes of accidental and intentional (e.g., suicide attempts) gross over-exposures to glyphosate or its commercial formulations (Chang et al., 1999; Dickson et al., 1988; Hung et al., 1997; Lee et al., 2000; Menkes et al., 1991; Pushnoy et al., 1998; Talbot et al., 1991; Temple and Smith, 1992; Tominack et al., 1991; Sawada et al., 1988; Sorensen and Gregersen, 1999). In the hundreds of reported cases, neurological symptoms that are unrelated to respiratory tract distress and shock (confusion, drowsiness, collapse, coma) associated with severe acute glyphosate toxicity cannot be identified. In a review of 92 cases, only 11 individuals were reported as having an abnormal mental state prior to the onset of severe respiratory and/or cardiovascular complications; most of these cases received atropine or pralidoxime, neurotoxicants used as antidotes for certain organophosphate insecticides that inhibit acetylcholinesterase (in these cases, organophosphate intoxication and cholinesterase inhibition was suspected, although glyphosate is not a potent cholinesterase inhibitor) (Tominack et al., 1991). In a review of 93 cases, 12 were reported as having neurological symptoms (confusion, coma) two of which occurred after cardiovascular resuscitation. The cause of symptoms in 10 other cases were not distinguished from secondary respiratory tract and/or cardiovascular distress (Talbot et al., 1991). Thus, the weight of evidence suggests that any neurologic symptoms associated with glyphosate exposures were secondary to other toxic effects.

Garry et al. (2002) has conducted a self-reporting survey of individuals exposed to herbicides and other pesticides, including glyphosate. This study reports that 6 or 14 children of parents who had used phosphonamino herbicides had parent-reported attention deficit disorder. While Garry et al. (2002) indicated that the odds ratio for this is statistically significant (OR=3.6; CI 1.35 to 9.65), it should be appreciated that the use of lay diagnosed disease and self-reported exposure histories diminishes the utility of this study for hazard identification.

Barbosa et al. (2001) reported a case of Parkinsonism in an adult male who was exposed to glyphosate. This study is essentially anecdotal and does not provide a clear or even credible causal relationship between glyphosate and neurotoxic effects. Nonetheless, the report by Barbosa et al. (2001) must be examined aggressively.

Parkinsonism is a degenerative disease of the central nervous system that impairs movement. The subject of the Barbosa et al. (2001) report was a 54-year old male who experienced an extensive dermal exposure to the herbicide while spraying a garden. The acute and transient symptoms included eye irritation (*conjenctival hyperemia*) and skin rash which progressed to blisters. One month after the exposure, the individual developed hand tremors. He was subsequently diagnosed with Parkinsonism, based on the results of a neurological examination and brain imaging.

Parkinsonism is a chronic degenerative disorder that could have been present in the patient prior to the exposure.

While the case reported by Barbosa et al. (2001) may have involved gross over-exposure to glyphosate, this over-exposure, in itself, is not dismissive of a possible neurologic risk. As noted above, extreme and sometimes fatal over-exposures to glyphosate are not generally associated with neurologic effects. In addition, there is an at least tenuous biological basis for suggesting a potential association. Glyphosate is a structural analog of glycine, a physiological agent that serves as an inhibitory neurotransmitter in the CNS. Glycine, which is also a naturally occurring amino acid and is essential for normal growth and development, has been implicated as an excitotoxin when present at high concentrations in brain tissue (Johnson and Ascher, 1987; Newell et al., 1997). Excitotoxicity has been hypothesized as a possible mechanism of Parkinsonism induced by the neurotoxicants MPTA (1-methyl-4-phenyl–2-3-6-tetrahydropyridine) and N-methylamino-L-alanine (Kanthasamy et al., 1997; Karcz et al., 1999; Spencer et al., 1987).

At this point, there is no evidence to conclude that glyphosate can produce or exacerbate Parkinsonism; indeed, the Barbosa et al. (2001) observation stands in contrast to the abundant case literature that suggests glyphosate is not a neurotoxicant in humans. However, the risk assessment community should be alert to any follow-up studies of glyphosate interactions with the pathophysiological mechanisms that underlie excitotoxicity and Parkinsonism, such as the NMDA (N-methyl-D-aspartate) receptor/ion channel complex. Nonetheless, the possible connection between the onset of Parkinsonism and the exposure to glyphosate cannot be established from the single case reported by Barbosa et al. (2001), as the apparent concurrence of the two effects could be coincidental. A coincidental association is suggested by the fact that no other cases of glyphosate-related Parkinsonism have been reported.

**3.1.7. Effects on Immune System.** Glyphosate has been tested specifically for effects on the immune system in both humans and experimental mammals. In experimental mammals, the only reported *in vivo* study (Blakley 1997) assayed for the effects of glyphosate on immune response to antigens. In this study, mice were exposed for 26 days to Roundup in drinking water (0, 0.35, 0.70, or 1.05 %) and humoral (antibody) immune response was assessed using sheep red blood cell challenge. The response in exposed mice was not different than that of control (unexposed) mice. This is consistent with *in vitro* assays using human immunocompetent cells — natural killer cells and cytotoxic T cells — which indicated that exposure to glyphosate or Roundup at concentrations ranging from 0.01 to 10 µmoles had no effect on immune system function (Flaherty et al. 1991). Further, there is no evidence that glyphosate or glyphosate formulations produce sensitization in acute dermal sensitization tests performed in guinea pigs (Stebbins and Brooks 1999e; Stebbins and Brooks 1999j; U.S. EPA/OPP 1993a,b,c; Williams et al. 2000). As noted in the previous discussions of subchronic (Section 3.1.5) and neurologic effects (Section 3.1.6) and detailed further in Appendices 5 and 6, no studies are reported that indicate morphologic abnormalities in lymphoid tissues which could be suggestive of an effect on the immune system.

In humans, experimental, clinical, and field studies have evaluated the ability of glyphosate formulations to induce allergic responses. Maibach (1986) exposed volunteers to Roundup and found that direct dermal application did not produce allergic or photoallergic responses. Williams et al. (2000) describes a study in which dermal exposure to Roundup (approximately 9 or 4.1% glyphosate as the isopropylamine salt) did not produce allergy or sensitization (Shelanski et al., 1973). A study of five forest workers who participated in mixing and spraving operations did not observe changes in blood leukocyte counts or symptoms of allergy (e.g., skin rash, respiratory symptoms) (Jauhiainen et al., 1991). Although cases of skin rashes following dermal exposures to glyphosate formulations have been reported (Barbosa et al., 2001), these effects are thought to derive primarily from irritation rather than allergy, based on observations of Maibach (1986). Hindson and Diffey (1984a) reported that a formulation of glyphosate used in the United Kingdom, Tumbleweed, could cause photosensitization. The effect, however, was subsequently attributed to an adjuvant, benzisothiazolone (Hindson and Diffey 1984b). Benzisothiazolone is not used in the glyphosate formulations covered by this risk assessment. Based on the study by Maibach (1986) using volunteers, there is no evidence that glyphosate itself causes photoirritation or photosensitization.

As noted in Section 4.1.3, El-Gendy et al. (1998) has reported that glyphosate caused immune suppression in a species of fish. As detailed in Section 4.1.31, this study was deficient in several respects and does not provide a basis for impacting the hazard identification for effects on the immune system.

**3.1.8. Effects on Endocrine Function.** In terms of functional effects that have important public health implications, effects on endocrine function would be expressed as diminished or abnormal reproductive performance. This issue is addressed specifically in the following section (Section 3.1.9). This section is limited to direct and largely mechanistic assays that can be used to assess potential direct action on the endocrine system.

Only three specific tests on the potential effects of glyphosate on the endocrine system have been conducted and all of these tests reported no effects. Glyphosate was inactive as an estrogen receptor agonist (*estrogenic activity*) in MCF-7 human breast cancer cells (Lin and Garry, 2000) as well as in yeast and trout hepatocyte assays (Petit et al., 1997). In a third assay, glyphosate did not inhibit steroid synthesis in MA-10 mouse Leydig tumor cells by disrupting expression of the steroidogenic acute regulatory (StAR) protein (Walsh et al., 2000). This protein mediates the rate-limiting step in the mitochondrial synthesis of steroid hormones (the transfer of cholesterol to the inner mitochondrial membrane). In the Walsh et al. (2000) study, however, Roundup did inhibit steroid synthesis, probably due to the effects of the surfactant on membrane function. All of these assays are *in vitro* – i.e., not conducted in whole animals. Thus, such studies are used qualitatively in the hazard identification to assess whether there is a plausible biologic mechanism for asserting that endocrine disruption is plausible. Because they are *in vitro* assays, measures of *dose* and quantitative use of the information in dose/response assessment is not appropriate. For glyphosate, these studies to not indicate a basis for suggesting that glyphosate is an endocrine disruptor.

Notwithstanding the negative results on endocrine function, the current RfD for glyphosate is based on reproductive effects, as discussed further in Sections 3.1.8 and 3.3. In addition, glyphosate has not undergone an extensive evaluation for its potential to interact or interfere with the estrogen, androgen, or thyroid hormone systems (i.e., assessments on hormone availability, hormone receptor binding or postreceptor processing as recommended by EDSTAC (1998). Thus, the assessment of the potential endocrine effects of glyphosate cannot be overly interpreted.

**3.1.9. Reproductive and Teratogenic Effects.** Glyphosate has been subject to multi-generation reproduction studies which measure overall effects on reproductive capacity as well as teratology studies which assay for a compounds ability to cause birth defects (Appendix 6).

Signs of teratogenic activity have not been observed in standard assays in both rats (Moxon 1996a; Farmer et al. 2000b; Rodwell et al. 1980a) and rabbits (Moxon 1996b; Rodwell et al. 1980b). Summaries of these studies are given in Appendix 6. No teratogenic effects in soft-tissue were observed in any study at doses of up to 3500 mg/kg/day. The only abnormal developement was delayed bone development (ossification). This was seen in rats at 3500 mg/kg/day (Rodwell et al. 1980a; Farmer et al. 2000b) and rabbits at 300 mg/kg/day (Moxon 1996b). Severe signs of maternal toxicity, including mortality, were observed in rats at 3500 mg/kg/day (Farmer et al. 2000b). Less severe signs of maternal toxicity (diarrhea, reduced fecal output, reduced food intake and body weight) were observed in rabbits at doses of 175 mg/kg/day and higher (Moxon 1996b).

In a multi-generation reproduction study in rats (Schroeder and Hogan 1981), unilateral focal tubular dilation of the kidney was observed in male  $F_{3b}$  pups at 30 mg/kg/day but not at 10 mg/kg/day. As discussed in section 3.3, the U.S. EPA has classified 30 mg/kg/day as the LOAEL and has based the RfD for glyphosate on the 10 mg/kg/day NOAEL for this effect. This effect is consistent with the acute systemic toxicity of glyphosate, rather than a specific reproductive effect.

Daruich et al. (2001) assayed effects of glyphosate on enzymatic activity in pregnant rats with a commercial formulation of glyphosate (specified as Herbicygon) that is used in Argentina. Changes in several biochemical parameters were noted but these were accompanied by significant decreases in food and water consumption. Since this study did not use a food and water restricted control, the observed effects cannot be attributed directly to glyphosate.

A 2-year dietary study, in which rats were exposed to 0, 2,000, 8,000 or 20,000 ppm glyphosate in diet, examined morphology of the reproductive organs, mammary glands, and all major endocrine glands, including the testis, ovary, pituitary, and thyroid (Stout and Ruecker, 1990). No treatment-related effects on reproductive organs or endocrine glands were observed at or below the maximally tolerated dose (20,000 ppm in diet) which resulted in decreased weight gain and histopathologic changes in liver, stomach, and eye lens. U.S. EPA (2001) summarized a study in which dogs were exposed to 0, 20, 100, and 500 mg/kg/day "glyphosate in gelatin

*capsules*" for 1 year (Reyna 1985). The summary notes that a decrease in absolute and relative pituitary weight was observed at the 100 and 500 mg/kg/day dose levels.

Subchronic studies, in which mice and rats were exposed to 3,125, 6,250, 12,500, 25,000, or 50,000 ppm glyphosate in the diet, examined morphology of all reproductive organs; mammary glands; and major endocrine glands, including adrenal, ovary, pancreas, parathyroid, pituitary, thymus and thyroid; the study also evaluated sperm counts and morphology and estrous cycle length (NTP, 1992). No treatment-related effects were observed on the morphology of reproductive organs or endocrine glands at or below the maximally tolerated dose (50,000 ppm in diet) which resulted in decreased weight gain in both rats and mice. A statistically significant decrease (20%) in sperm count was observed in male rats exposed to 25,000 or 50,000 ppm. NTP (1992) concluded that there was no evidence of adverse effects on the reproductive system of rats or mice, and summarized the findings as follows:

"Measures of sperm density, or the number of sperm/g caudal epididymal tissue, were reduced somewhat in male rats in the 2 highest dose groups (25,000, 50,000 ppm); other spermatozoal measurements were not different from controls in rats or mice. There was a slight lengthening of the estrous cycle in high dose female rats (50,000 ppm), but the biologic significance of these findings, if any, is not known."

Several other subchronic and chronic studies of glyphosate are noted in Williams et al. (2000), with no mention of treatment-related effects on endocrine glands or reproductive organs; however, the specific tissues that were evaluated are not reported.

Yousef et al. (1995) has reported substantial decreases in libido, ejaculate volume, sperm concentrations, semen initial fructose and semen osmolality as well as increases in abnormal and dead sperm in rabbits after acute exposures to glyphosate. The authors report that all of the effects were statistically significant at p<0.05. A serious limitation of this study is that the authors report the doses as proportions of 0.01 and 0.1 of the LD<sub>50</sub> but do not specify the actual doses. Using a reported rabbit LD<sub>50</sub> of 3,800 mg/kg (SERA 1996), the doses would correspond to 38 and 380 mg/kg. As discussed by Hastings (1995), the Yousef et al. (1995) study also does not specify the formulation that was used.

The toxicological significance of the observed effects described by Yousef et al. (1995) is clear. As noted above, however, a 3-generation study in rats found no treatment-related effects of glyphosate on mating, fertility, or reproductive parameters at doses of 3, 10, or 30 mg/kg body weight, although changes in kidney morphology were noted at the 30 mg/kg/day dose level (Schroeder and Hogan 1981). In addition and as also summarized above, very high dietary concentrations of glyphosate have not be associated with impaired reproductive performance or signs of damage in testicular tissue.

The basis for the inconsistency between the Yousef et al. (1995) study and all other studies that have assessed the reproductive effects of glyphosate cannot be identified unequivocally. As discussed by Williams et al. (2000), the Yousef et al. (1995) study can be criticized for a number of reporting and experimental design limitations or deficiencies. In addition, it should be noted that the rabbits in the Yousef et al. (1995) study were dosed by gelatin capsules whereas the Schroeder and Hogan (1981) multigeneration study involved dietary exposures. The use of gelatin capsules is a reasonable mode of administration but, like gavage exposures, it results in a high spike in body burden that is not typical or particularly relevant to potential human exposures – other than attempted suicides. On the other hand, dietary exposures, as used in the Schroeder and Hogan (1981) study, result in more gradual and steady exposures over the course of the day that are more comparable and relevant to potential human exposures.

In a subsequent study, Yousef et al. (1996) have demonstrated that glyphosate may reduce sperm motility in the range of 116  $\mu$ M to about 300  $\mu$ M in protein free media and 500  $\mu$ M to about 740  $\mu$ M in a media with protein. The mechanism of the this effect is not clear but it may be related to the ability of glyphosate to inhibit oxidative phosphorylation. While this *in vitro* study cannot be applied directly to the risk assessment, it is worth noting that the lowest reported effect concentration, 116  $\mu$ M, corresponds to a concentration of about 19.6 mg/L [116  $\mu$ Moles/L × 169.07  $\mu$ g/ $\mu$ Mole = 19,612  $\mu$ g/L], which is in turn about a factor of 10 above the NOAEL used in the dose-response assessment.

Numerous epidemiological studies have examined relationships between pesticide exposures or assumed pesticide exposures in agricultural workers and reproductive outcomes. Very few studies, however, have attempted to characterize exposures, either qualitatively or quantitatively, to specific pesticides (Arbuckle and Sever, 1998). Of those studies that have specifically addressed potential risks from glyphosate exposures, adverse reproductive effects have not been associated with glyphosate exposure.

The Ontario Farm Health Study collected information on pregnancy outcomes and pesticide use among Ontario farm couples. Three retrospective cohort studies of this group (Arbuckle et al. 2001; Curtis et al. 1999; Savitz et al. 1997) have examined relationships between exposures to glyphosate formulations (defined as self-reported participation in mixing and/or spraying operations) and reproductive outcomes. One study analyzed self-reported spontaneous miscarriages of 3,984 pregnancies among 1,898 couples who self-reported exposures to glyphosate formulations within a period beginning two months before pregnancy and ending the month of conception (Savitz et al., 1997). Risk of miscarriage was unrelated to self-reported exposure to glyphosate formulations. A second study of spontaneous abortions among 2,110 women and 3,936 pregnancies disaggregated the herbicide exposures into pre- and post-conception and spontaneous abortions into early- (< 12 wk) and late-term (12-19 wk) abortions (Arbuckle et al., 2001). Spontaneous abortions were not associated with post-conception exposure; however, the odds ratio for abortions and post-conception exposure was 1.4 (1.0-2.1), and for late-term abortions was 1.7 (1.0-2.9). The latter odds ratios were not adjusted for maternal age which is a risk factor for spontaneous abortion. When

maternal age was considered in a regression tree analysis, spontaneous abortions were found to be unrelated to glyphosate formulation use. Curtis et al. (1999) examined fecundity among 1,048 farm couples who self-reported exposures to glyphosate formulations within a period beginning 2 months prior to trying conception (to account for time of spermatogenesis) and ending at pregnancy. Fecundity was unrelated to glyphosate exposure.

Larsen et al. (1998a) examined relationships between use of pesticides and semen quality among farmers in Denmark. Participants in the study included 161 farmers who self-reported crop spraying with a variety of pesticides, that included Roundup (7% prevalence of use) and 87 farmers who did not use pesticides. Semen samples were collected at the start of the spraying season and 12-18 weeks after the first spraying. Evaluations included sperm count, morphology, chromatin structure and motility; and serum concentrations of reproductive hormones (testosterone, LH, FSH). Semen quality and reproductive hormone levels were unrelated to pesticide use. In a related study, fecundity was compared among farmers who did or did not participate in pesticide spraying operations (Larsen et al., 1998b). Fecundity was determined from the number of self-reported menstrual cycles or months between discontinuation of birth control and pregnancy. Participants included 450 traditional farmers who reported that they sprayed pesticides, 72 traditional farmers who did not participate in spraying operations, and 94 organic farmers who reported not using pesticides on their crops. Fecundity was unrelated to pesticide use or participation in pesticide spraying operations.

**3.1.10.** Carcinogenicity and Mutagenicity. Information regarding the mutagenicity and carcinogenicity of glyphosate has been reviewed in detail by U.S. EPA (U.S. EPA/ODW 1992; U.S. EPA/OPP 1993a,b,c), the World Health Organization (WHO 1994) as well as in the open literature (Cox 2002; Williams et al. 2000).

Based on standard animal bioassays for carcinogenic activity in vivo (Appendix 5), there is no basis for asserting that glyphosate is likely to pose a substantial risk. The Re-registration Eligibility Decision document on glyphosate (U.S. EPA/OPP 1993a) indicates that glyphosate is classified as Group E: Evidence of non-carcinogenicity for humans. This classification is also indicated in U.S. EPA's most recent publication of tolerances for glyphosate (U.S. EPA/OPP 2002). This is also consistent with the assessment by WHO (1994) and review by (Williams et al. 2000). Cox (2002) has challenged the interpretation of the cancer data but does not provide any reanalyses of the data. Tumors have been observed in some of the chronic toxicity studies (Appendix 3). As discussed in U.S. EPA/ODW (1992), the studies conducted before 1990 were judged by U.S. EPA as insufficient for evaluating the potential carcinogenicity of glyphosate because the observed responses were equivocal or the dose levels were inappropriate (i.e., the highest dose used was not the maximum tolerated dose). U.S. EPA requested the study by Stout and Ruecker (1990) and judged it to be adequate. Although the study indicated increases in some tumor types (pancreatic islet cell adenomas in low dose male rats, hepatocellular adenomas in male rats, and C-cell adenomas of the thyroid males and females), the effects were not dose related. Gold et al. (1997) reports cancer potency estimates of  $5.9 \times 10^{-5}$  to  $4.8 \times 10^{-4}$ (mg/kg/day)<sup>-1</sup> for glyphosate. The potency parameters provided by Gold et al. (1997), however,

are based on experimental data in which there were no statistically significant increases in tumor rates at any dose level.

Roundup has been shown to cause an increase in chromosomal aberrations in a plant (*Allium sp.*) associated with cell abnormalities in spindle fiber (Rank et al. 1993), DNA adduct formation in mice (Reluso et al. 1998) and single strand breaks in mice (Bolognesi et al. 1997). None of the *in vivo* studies using mammalian species or mammalian cell lines have reported mutagenic activity (i.e., NTP 1992, Rank et al. 1993). Two studies (Vyse and Vigfusson 1979, Vigfusson and Vyse 1980) report a significant increase in sister chromatid exchanges in human lymphocytes *in vitro*. The authors of these studies conclude from their results that glyphosate is, at most, slightly mutagenic. In addition, some positive assays in the fruit fly have been reported (Kale et al. 1995; Kaya et al., 2000) as well as positive results in lymphocyte cultures (Lioi et al. 1998a; Lioi et al. 1998b). Nonetheless, most of the other screening studies for mutagenicity are negative (Appendix 7). Based on the weight of evidence of all available studies, U.S. EPA (U.S. EPA/ODW 1992; U.S. EPA/OPP 1993a; U.S. EPA/OPP 2002) concluded that glyphosate is not mutagenic.

The human data on the potential carcinogenic activity of glyphosate is sparse. Hardell and Erikson (1999a) reported an increased cancer risk of non-Hodgkin lymphoma (NHL) in individuals in Sweden who have a history of exposure to glyphosate. The increased risk was not statistically significant. Acquavella et al. (1999) have criticized the methodology used by Hardell and Erikson (1999a). As part of the response to this criticism, Hardell and Erikson (1999b) reported that an additional analysis of their data pooled with data from another study demonstrated a statistically significant increase in NHL associated with exposures to glyphosate. Details of the pooled analysis are not provided by Hardell and Erikson (1999b).

These results are of concern to the Forest Service and the Forest Service requested that the U.S. EPA review these studies (Rubin 2000). The U.S. EPA (Tompkins 2000) replied that:

The Office of Pesticides Programs Health Effects Division has reviewed the journal article entitled "A Case-Control Study of Non-Hodgkin Lymphoma and Exposure to Pesticides" and concluded that the study does not change EPA's risk assessment for the currently registered uses of glyphosate.

This issue is also addressed in the most recent U.S. EPA/OPP (2002) assessment:

This type of epidemiologic evaluation does not establish a definitive link to cancer. Furthermore, this information has limitations because it is based solely on unverified recollection of exposure to glyphosate-based herbicides.

Given the marginal mutagenic activity of glyphosate and the failure of several chronic feeding studies to demonstrate a dose-response relationship for carcinogenicity and the limitations in the available epidemiology study, the Group E classification given by the U.S. EPA/OPP (1993a,

2002) appears to be reasonable. As with any compound that has been studied for a long period of time and tested in a large number of different systems, some equivocal evidence of carcinogenic potential is apparent and may remain a cause of concern, at least in terms of risk perception (e.g., Cox 2002). While these concerns are understandable, there is no compelling basis for challenging the position taken by the U.S. EPA and no quantitative risk assessment for cancer is conducted as part of the current analysis.

**3.1.11. Irritation and Sensitization.** Glyphosate formulations used by the Forest Service are classified as either non-irritating or only slightly irritating to the skin and eyes in standard assays required for product registration. Based on several eye and skin irritation studies submitted to the U.S. EPA as part of the registration process, the U.S. EPA/OPP (1993c) classifies glyphosate as mildly irritating to the eyes (Category III) and slightly irritating to the skin (Category IV). Literature on the irritant effects of unformulated glyphosate is summarized by Williams et al. (2000). The free acid of glyphosate is severely irritating to the eyes but the IPA salt of glyphosate, the form that is in all formulations used by the Forest Service, is nonirritating to the skin and eyes. As discussed in Section 3.1.14, POEA and other surfactants used in glyphosate formulations may be severely irritating to the eyes, skin, and other mucosal surfaces such as the gastrointestinal tract and lungs.

Based on a total of 1513 calls to a poison control center reporting ocular effects associated with the use of Roundup, 21% were associated with no injury, 70% with transient minor injury, 2% with some temporary injury. One case was classified as a major effect which took more than 2 weeks to resolve. This case, however, involved an individual exposed to a dilute solution of Roundup while wearing extended wear contact lenses. In addition, symptoms were apparent in both the exposed and unexposed eye. Thus, it is unclear if the ocular signs observed in this individual were attributable to the Roundup exposure. For all patients, the most frequently noted symptoms included blurred vision, a stinging or burning sensation, lacrimation. No cases of permanent damage were reported. (Acquavella et al. 1999).

**3.1.12.** Systemic Toxic Effects from Dermal Exposures. As discussed in section 3.1.3, glyphosate is poorly absorbed from the skin and thus systemic toxic effects from dermal exposure are likely to be less than those from oral exposure. In terms of acute exposures, however, there is relatively little apparent difference in the oral and dermal toxicity of glyphosate because glyphosate is relatively non-toxic by either route. For example, the acute oral toxicity of glyphosate expressed at the LD<sub>50</sub> in rats is listed by U.S. EPA/OPP (1993c) as >4320 mg/kg. Based on this LD<sub>50</sub>, glyphosate is classified as Category III for oral toxicity. Similarly, the acute dermal toxicity of glyphosate expressed at the LD<sub>50</sub> in rabbits is listed by U.S. EPA/OPP (1993c) as >2000 mg/kg and is also classified as Category III. In both of these LD<sub>50</sub> values, the "greater than" designation (>) indicates that less than 50% of the animals died at the maximum dose tested, in this example 4320 mg/kg for oral exposure and 2000 mg/kg for dermal exposure. The difference in these doses is an artifact of the highest doses used in the toxicity studies and does not suggest that glyphosate is more toxic by the dermal route than by the oral route of exposure.

As indicated in Appendix 3, all glyphosate formulations have dermal  $LD_{50}$  values in rats or rabbits that are >2000 mg/kg and most are >5000 mg/kg.

A more meaningful assessment of the dermal toxicity of glyphosate can be made from repeated dose 21-day studies. The U.S. EPA RED for glyphosate (U.S. EPA/OPP 1993c) cites a 1982 study (MRID 00098460) in which glyphosate was applied to the intact or abraded skin of rabbits at doses of 10, 1000 or 5000 mg/kg/day, five days per week, for three weeks. The only treatment related effects included slight irritation of the abraded skin (a local and not a systemic effect), decreased food consumption, and decreased serum lactic dehydrogenase activity at 5000 mg/kg/day. In a more recent but similarly designed study in rats (Pinto 1996), dermal doses of 250, 500, 1000 mg/kg/day caused no effects on body weight, food consumption, hematology, clinical chemistry, or organ weights and there were no signs of dermal irritation or pathologic changes in any tissue.

**3.1.13.** Inhalation Exposures [including Brown-and-Burn Operations]. Some volatile pesticides may present practical risks in normal applications but inhalation is not an important route of exposure for most low volatile herbicides (Dowling and Seiber 2002). Because of the low volatility rate for technical grade glyphosate (Tria 1994) and the available inhalation toxicity studies on a number of glyphosate formulations (Appendix 3), the U.S. EPA waived the requirement of an acute inhalation study for technical grade glyphosate (U.S. EPA 1993b, p. 10). The acute inhalation LC<sub>50</sub> value of the isopropylamine salt glyphosate is >6.37 mg/L – i.e., no mortality in any of five rats of each sex exposed to this concentration for four hours (Mcguirk 1999a). As indicated in Appendix 3, the short-term (typically 4 hours) inhalation LC<sub>50</sub> values for various glyphosate formulations range from >1.3 mg/L to >7.3 mg/L. The lowest LC<sub>50</sub> value for Credit, Honcho, Mirage, Roundup Original, and Roundup ProDry. This appears to reference the acute inhalation study by Dudek and Cortner (1998) in which 20% mortality (1/5 rats of each sex) was observed after 4-hour inhalation exposures to MON 77063 at a concentration of 2.6 mg/L.

A case of "*Roundup Pneumonitis*" has recently been reported by Pushnoy et al. (1998). This involved an individual with shortness of breath, respiratory irritation, and dizziness. Exposure to Roundup had involved disassembling sprayer equipment that had been used in the application of Roundup. As discussed by Goldstein et al. (1999), the plausibility of association between Roundup exposure and the development of these symptoms is tenuous given that this individual may have been exposed to diesel fuel aerosols, chlorinated solvents, smoking, or welding fumes. Jamison et al. (1986) has suggested a potential effect associated with glyphosate after inhalation exposures to flax dust. In this study, human volunteers were exposed to two different types of flax dust: one derived from glyphosate treated flax and the other derived from flax not treated with glyphosate. The glyphosate treated flax consistently caused a greater depression in respiratory function than the dust from flax not treated with glyphosate. As noted by the authors, the glyphosate residue on the flax six weeks prior to testing and it is likely that there was very little glyphosate residue on the flax. The authors also note that particles size distribution of the two dusts used in the study was not significantly different. Based on particle size distribution data

presented in this publication (Jamison et al. 1986, Table 1, p. 810), however, the glyphosate treated flax dust contained about 25% more particles in the 0-1 $\mu$  range. Particles in this range typically penetrate to the alveolar sacs (Razman and Klassen 1996). Thus, even though the distributions in the particle sizes for the two forms of flax may not be statistically significantly different, the higher concentration of respirable particles in the glyphosate treated flax may be contributed to the apparent difference in biological activity.

Although inhalation of glyphosate is not a typical route of exposure, it may occur during brownand-burn operations. Brown-and-burn operations are conducted 30 to 180 days after treatment with the herbicide. As discussed by Bush et al. (1987), the combustion of several herbicides does not result in exposure to toxic air concentrations of herbicides. These investigators, however, did not look specifically at glyphosate and did not take toxic combustion products into consideration.

The thermal degradation of glyphosate has been studied by Flora and Simon (1981). During combustion at temperatures ranging from 200°C to 240°C, glyphosate forms a polycondensate. This range of temperatures is typical of slow combustion but is far less than the 800–1,000°C temperatures of an actively burning wood stove or fireplace (Bush et al. 1987). No information is available regarding the toxicological properties of the combustion product identified by Flora and Simon (1981) or other combustion products of glyphosate.

**3.1.14. Role of Surfactant.** As summarized in Appendix 1, various glyphosate formulations contain a polyethoxylated tallow amine (POEA) surfactant at a level of up to about 20% (200 g/L) and Roundup Pro contains a phosphate ester neutralized polyethoxylated tallow amine surfactant at a level of 14.5% (145 g/L). Tallow contains a variety of fatty acids including oleic (37–43%), palmitic (24–32%), stearic (20–25%), myristic (3–6%), and linoleic (2–3%) acids as well as small amounts of cholesterol, arachidonic, elaidic, and vaccenic acids (Budavari 1989). The other formulations of glyphosate recommend the use of a surfactant to improve the efficacy of glyphosate. There is an extensive amount of literature on glyphosate indicating that the addition of surfactants can greatly enhance phytotoxicity of herbicides (De Ruiter and Meinen 1996, De Ruiter and Meinen 1998; Denis and Delrot 1997; Laerke 1995; Miller et al. 1998; O'Sullivan et al. 1981; Riechers et al. 1995; Sundaram 1990; Sundaram et al. 1996).

While surfactants are typically classified as "inert" ingredients in herbicides, these compounds are not toxicologically inert. It is beyond the scope of this risk assessment to specifically review all of the information available on surfactants. Much of the available information on the toxicity of surfactants used with glyphosate have been summarized in SERA (1997) and more recent studies (Chang et al. 1999; Garry et al. 1999; Lin and Garry 2000; Reluso et al. 1998) reenforce the conclusion reached in SERA (1997) that some surfactants may be more toxic than the herbicides with which they are used. Although surfactants may play a substantial role in the interpretation of a large number of suicides and attempted suicides involving the ingestion of glyphosate formulations, primarily Roundup, the acute mammalian toxicity of different glyphosate formulations do not appear to differ substantially (Appendix 3a and 3b). This is in contrast to the

available data on the toxicity of various formulations to aquatic species, as detailed in Section 4.1.3 and summarized in Appendix 3c.

While a number of surfactants may be used in conjunction with glyphosate, the most important to this risk assessment is the POEA or phosphate ester neutralized POEA. The POEA surfactant was originally used with glyphosate in Roundup formulations. The phosphate ester neutralized POEA is the surfactant currently used in Roundup Pro (Appendix 1). By far the most relevant study on the toxicity of the POEA surfactant used in Roundup involves a series of teratology studies in rats using glyphosate (98.7% purity), the POEA surfactant used in many glyphosate formulations, and a phosphate ester neutralized POEA (Farmer et al. 2000b). In this study, groups of pregnant female rats were dosed on days 6 through 19 of gestation with glyphosate at 300, 1000, or 3500 mg/kg/day, POEA at 15, 100, and 300 mg/kg/day, or the neutralized POEA at 15, 50, or 150 mg/kg/day. For glyphosate, severe maternal poisoning was observed at 3500 mg/kg/day and this was associated with reduced fetal body weights and sternal ossification, as well as fetal death. The NOAEL for both maternal and fetal toxicity was 1000 mg glyphosate/kg/day. The surfactants also caused mortality in dams at the highest doses tested: 300 mg/kg/day for POEA and 150 mg/kg/day for neutralized POEA. In addition, a dose of 100 mg POEA/kg/day caused mild clinical signs of toxicity and decreased food consumption in dams. No fetotoxic effects were reported at any dose level. Thus, in repeated dosing, the NOAEL for glyphosate of 1000 mg/kg/day was substantially higher than the NOAEL for either POEA (15 mg/kg/day) or neutralized POEA (50 mg/kg/day).

The potential role of the surfactant in the toxicity of Roundup was first emphasized by the Sawada et al. (1988) in their analysis of poisoning cases in humans. They indicate that the acute  $LD_{50}$  of POEA is *"less than one-third that of roundup and its active ingredient"* and reference this statement to a chapter by Atkinson (1985) in *The Herbicide Glyphosate* (Grossbard and Atkinson 1985). The Sawada reference has been quoted in turn by Martinez and Brown (1991) as indicating that "... *POEA by itself has a LD\_{50} of 1-2 g/kg"*. Atkinson (1985) does cite an  $LD_{50}$  of 4.3 g/kg for glyphosate [a rounding of the rat oral  $LD_{50}$  of 4,320 mg/kg reported in U.S. EPA (1986a) and earlier U.S. EPA reports] and indicates that this is about the same as the acute oral  $LD_{50}$  for isopropylamine salt in rats, 4.9 g/kg. Atkinson (1985), however, does not give an acute oral  $LD_{50}$  for POEA or any other surfactant. Although there is evidence that POEA is more toxic than glyphosate to aquatic species (Section 4), the acute oral toxicity of Roundup (glyphosate and surfactant,  $LD_{50}$  in rats of 5400 mg/kg) is almost the same as that of glyphosate ( $LD_{50}$  in rats of 5,600 mg/kg).

Based on these  $LD_{50}$  values, the  $LD_{50}$  of the surfactant can be estimated under the assumption of dose addition (Finney 1971). This assumption requires that the components in the mixture have the same mode of action. This assumption is not certain, but it is consistent with the observation by Talbot et al. (1991) that both glyphosate and POEA may exert some of their acute toxicity via irritation of biological membranes. The assumption of dose addition is also not interactive—that is, it assumes that the components in the mixture do not influence the toxicity of one another.

This assumption is conservative, compared with other non-interactive models of joint action (Mumtaz et al. 1994).

For some uniform measure of toxicity ( $\zeta$ ) (e.g., LD<sub>50</sub>), the toxicity of any mixture ( $\zeta_M$ ) is predicted, under the assumption of dose addition, by:

$$\zeta_M = \frac{\zeta_1}{(\pi_1 + \rho \pi_2)} \tag{3-1}$$

where  $\zeta_1$  is the effective exposure (e.g.,  $LD_{50}$  or  $LD_{95}$  values) for one compound,  $\pi_1$  and  $\pi_2$  are the proportions of each compound in the mixture, and  $\rho$  is the potency defined as  $\zeta_1 \div \zeta_2$ . Furthermore, given the toxicity of a defined mixture ( $\zeta_M$ ) and one of the components ( $\zeta_1$ ), the potency of the second component can be calculated as:

$$\rho = \frac{(\zeta_1/\zeta_M) - \pi_1}{\pi_2}$$
(3-2)

Here, the term *defined mixture* indicates that  $\pi_1$  and  $\pi_2$  are known. From this relationship, the effective exposure (i.e., toxic potency) of the second component ( $\zeta_2$ ) can be estimated as:

$$\zeta_2 = \frac{\zeta_1}{\rho} \tag{3-3}$$

Using the nominal LD<sub>50</sub> for Roundup of 5,400 mg/kg, a  $\pi_1$  of 0.356 for glyphosate (356 g/L), and  $\pi_2$  of 0.15 for POEA (150 g/L), the estimated LD<sub>50</sub> for POEA would be almost exactly 1,200 mg/kg, consistent with dose additivity. This approach, as detailed below, would be a misapplication of the above equations.

To estimate the toxicity of POEA from the Roundup (glyphosate+POEA) LD<sub>50</sub>, this LD<sub>50</sub> must be converted from units of glyphosate to total mixture mass (glyphosate+POEA). In other words, an LD<sub>50</sub> of 5,400 mg glyphosate/kg bw is equivalent to a combined mass (glyphosate and POEA) of about 7,560 mg [1.4·5,400 mg], since the ratio of POEA to glyphosate is approximately 0.4 [150 g/L ÷ 356 g/L]. Similarly, the correct  $\pi_1$  for glyphosate is about 0.7 [356 ÷ (150+356)] and the correct  $\pi_2$  for POEA is about 0.3 [150 ÷ (150+356)]. Using this approach, the potency of POEA relative to glyphosate is about 0.14 and the estimated oral LD<sub>50</sub> in rats is for POEA is about 40,000 mg/kg [5,600 mg/kg ÷ 0.14]. This estimate is consistent with the published results of Martinez, summarized in the following paragraph, in which no mortality was noted in rats after oral doses of up to 14,286 mg/kg POEA.

Martinez and coworkers (Martinez and Brown 1991; Martinez et al. 1990) conducted a series of experiments specifically designed to assess the role of the surfactant in the acute toxicity of Roundup. In these studies, compounds were administered to groups of five rats either by gavage [direct instillation into the stomach] or direct installation into the trachea. Oral exposures to

Roundup at doses of 1, 3, and 5 mL/animal caused 0%, 40%, and 100% mortality, respectively, over a 24-hour observation period. Taking an average body weight of 350 g/rat reported by Martinez and Brown (1991), the mid-dose level corresponds to approximately 3,050 mg/kg [3 mL  $\cdot$  356 mg a.e./mL  $\div$  0.350 kg], only somewhat less than and consistent with the reported LD<sub>50</sub> for Roundup of 5,400 mg/kg (Monsanto Co. 1982a,b). POEA, administered by gavage, caused no deaths at doses of 1, 3, and 5 mL/animal. Since ethoxylated surfactants generally have a density of about 1g/mL (Kosswig 1994, p. 789), the doses of POEA correspond to approximately 2,857, 8,571, and 14,286 mg/kg. The low acute oral toxicity of POEA is consistent with the similarity between the acute oral toxicity of glyphosate and Roundup, discussed above.

In the earlier study by Martinez et al. (1990), an oral dose with Roundup RTU or Roundup concentrate caused delayed (6 hours) pulmonary edema, consistent with clinical observations in humans, as summarized above. The authors concluded that "... delayed pulmonary edema combined with blood stained weeping from the nose, diarrhea, distended GI tract, and ascites is in excellent agreement with ... The clinical picture of ... hypovolemic shock", as described by Sawada et al. (1988). In the individuals involved in the Taiwan studies of glyphosate poisoning, however, hematocrit, blood urea nitrogen, and central venous pressure determinations were not consistent with hypovolemia.

Intratracheal instillations in rats resulted in much more toxic effects at much lower dose levels. Roundup at doses of 0.1, 0.2, and 0.4 mg/animal caused 80% mortality at the low dose and 100% mortality at the two higher doses as well as an increase in lung weights. POEA, at the same dose levels, caused 20%, 70%, and 100% mortality as well as increases in lung weights, although the increases were less than those observed with Roundup (Martinez and Brown 1991, Table 1, p. 44). Pathological examinations indicated that both Roundup, and to a lesser extent POEA, cause hemorrhaging and congestion of the lungs after intratracheal instillations. Martinez and Brown (1991) conclude that POEA potentiates the pulmonary toxicity of glyphosate. Since, however, these investigators did not test glyphosate alone, the basis for their conclusion is not clear.

Adam et al. (1997) have studied the effects of glyphosate, POEA, mixtures of glyphosate and POAE, as well as a commercial formulation of Roundup (41% glyphosate IPA and 18% POEA) in rats after gavage (oral) and intratracheal installations (i.e., directly to the lungs). Respiratory effects and pulmonary damage were more severe in the rats dosed with any of the POEA containing treatments than with glyphosate alone. Similarly, the gastrointestinal effects of the POEA containing treatments were uniformly more severe than seen in rats treated with glyphosate alone. Tai et al. (1990) reported that injections of Roundup in rats led to cardiac depression caused solely by POEA and partially antagonized by glyphosate.

Based on drinking water studies of both glyphosate and Roundup (glyphosate with POEA), the surfactant does not affect the rapid elimination rate of glyphosate in mammals (NTP 1992).

# **3.1.15.** Impurities and Metabolites.

**3.1.15.1.** *Aminomethylphosphonate (AMPA)* -- The primary metabolite of glyphosate in mammals and other organisms is aminomethylphosphonate (AMPA):

NH<sub>2</sub>-CH<sub>2</sub>-P-OH

which is formed together with glyoxylate (HCO-COOH).

In mammals, only very small amounts of AMPA, less than 1% of the absorbed dose, are formed (U.S. EPA/ODW 1992, Brewster et al. 1991). In addition, AMPA is an environmental metabolite of glyphosate. This is to say that AMPA is formed in environmental media such as soil and water as a breakdown product of glyphosate. For example, Mao (1996) found that AMPA reached approximately 21.2% of applied dose of glyphosate by day 181 under anaerobic conditions in pond water. In addition, glyphosate is readily metabolized by soil bacteria with AMPA as a major metabolite (Dick and Quinn 1995a; Dick and Quinn 1995b). It should be noted that AMPA is also formed in the degradation of amino(trimethylenephosphonic) acid, a compound used as a scale inhibitor and additive in washing agents (Schweinsberg et al. 1999)

These two differing sources of exposure – i.e., as an endogenous metabolite in mammals and as an environmental metabolite – must be handled differently in this risk assessment. The approach of examining the potential importance of the metabolism of a chemical agent by a mammal is common in the risk assessment of xenobiotics, which generally involve the formation of one or more mammalian metabolites, some of which may be more toxic than the parent compound. Usually, the parent compound is selected as the agent of concern because the toxicology studies and monitoring studies provide information about the agent. Thus, the dose metameter for the risk assessment is most clearly expressed in terms of the parent compound. In cases where a toxic metabolite is known to be handled differently by humans, this simple approach may be modified. There is no indication that such a modification is necessary for glyphosate. Thus, in terms of assessing direct exposures to technical grade glyphosate, the inherent exposures to AMPA as a metabolite are encompassed by the existing toxicity data on glyphosate.

This approach does not, however, encompass concern for exposures to AMPA as an environmental metabolite. As noted above, about 20% of applied dose of glyphosate may be found in water as AMPA after about six months. The toxicity and environmental fate of AMPA has been reviewed recently by WHO (1997), Cox (2002), and Williams et al. (2000). In addition, the U.S. EPA/OPP (2002) has reviewed this information and assessed the potential consequences of exposures to AMPA as an environmental degradate. Based on this review, the U.S. EPA/OPP (2002) concluded:

The nature of the residue in plants and animals is adequately understood and consists of the parent, glyphosate. The Agency has decided that only glyphosate parent is to be regulated in plant and animal commodities and that the major metabolite, AMPA (aminomethylphosphonic acid) is not of toxicological concern regardless of its levels in food. – U.S. EPA/OPP (2002, p. 17725)

While Cox (2002) has cited concerns for AMPA based on a limited subset of the literature on this compound, no formal dose-response and exposure assessment is presented that would argue against the position of U.S. EPA/OPP (2002). Further, the position taken in U.S. EPA/OPP (2002) is supported by the conclusions of the more extensive reviews by both WHO (1997) and Williams et al. (2002). Although data are sufficient to conduct a separate risk assessment on AMPA (e.g., Holson 1991a,b; Kier and Stegeman 1993; Stout 1991; Tompkins 1991), this would not be a judicious expenditure of limited resources on the part of the Forest Service. The position taken by U.S. EPA/OPP (2002) appears to be reasonable and is well-supported. Consequently, in this risk assessment, AMPA is not quantitatively considered in the dose-response and exposure assessments.

*3.1.15.2. N-nitrosoglyphosate (NNG) --* Glyphosate also contains N-nitrosoglyphosate (NNG) as an impurity:



### NNG

Nitroso compounds are characterized by the N=O group, a double bond between a nitrogen and oxygen. Nitrosamines are nitroso compounds in which the nitroso group is attached to a nitrogen atom, N-N=O. NNG contains the nitrosoamine group. Certain groups of nitrosoamines have served as model compounds in some of the classical studies on chemical carcinogenicity. While there is a general concern for the carcinogenic potential of nitroso compounds, the contribution of specific nitroso compounds to carcinogenic risk is difficult to quantify (Mirvish 1995).

The EPA re-registration document (RED) for glyphosate states:

Technical grade glyphosate contains N-nitrosoglyphosate (NNG) as a contaminant. Carcinogenicity testing of nitroso contaminants is normally required only in those cases in which the level of nitroso compounds exceeds 1.0 ppm. Analyses showed that greater than 92% of the individual technical glyphosate samples contained less than 1.0 ppm NNG. The Agency concluded that the NNG content of glyphosate was not toxicologically significant.

As part of the conduct of this risk assessment, data available to U.S. EPA for RED as well as the more recent data on the levels of *N-nitrosoglyphosate* and related compounds has been reviewed (Bernard 2002; Hirsch and Augustin 1987). This information is classified as trade secret under FIFRA and cannot be detailed in this risk assessment. Nonetheless, no information has been encountered in the CBI files or in the open literature that contradicts the above assessment in the RED. In addition, none of the recent reviews on the toxicity of glyphosate cite contamination with *N-nitrosoglyphosate* (*NNG*) as a concern (Cox 2002; WHO 1994; Williams 2000). Consequently, as with AMPA, a detailed dose-response and exposure assessment for NNG does not appear to be justified with in this risk assessment or as a separate assessment.

**3.1.15.3. 1,4-Dioxane** – 1,4-Dioxane, is a contaminant in POEA. U.S. EPA (U.S. EPA/IRIS 1992) considers dioxane to be a carcinogen, Class B2: Probable human carcinogen and has derived a cancer potency factor (referred to by U.S. EPA as a slope factor) of 0.011  $(mg/kg/day)^{-1}$ . This assessment has been reviewed by and is in concordance with the analysis by the Agency for Toxic Substances and Disease Registry (DeRosa et al. 1996). Dioxane is present in Roundup at a level of approximately 0.03% (Monsanto Co. 1990) or 300 mg/L (300 ppm). This is about a factor of 0.00084 less than the level of glyphosate in Roundup [300 mg dioxane/L in Roundup  $\div$  356,000 mg glyphosate/L in Roundup].

In a previous Forest Service risk assessment on glyphosate (USDA 1989a,b,c), it was demonstrated that the upper limit of risk associated with contamination was extremely low -e.g.,  $<1\cdot10^{-7}$  (Borrecco and Neisess 1991). The cancer potency factor used in this risk assessment was 0.0076 (mg/kg/day)<sup>-1</sup>, almost the same as the value currently recommended by U.S. EPA (i.e., both round to 0.01). Borrecco and Neisess (1991) derived toxicity based criteria for 1,4-dioxane and use the information to calculate margins of safety for exposure to 1,4-dioxane. According to the available toxicity data, dioxane does not present unique toxic effects; therefore, its toxicity, except for cancer) is likely to be encompassed by the available toxicity data on Roundup.

## **3.2. EXPOSURE ASSESSMENT**

3.2.1. Overview. Exposure assessments are developed for both workers and members of the general public. Two types of work exposure assessments are considered: general and accidental/incidental. The term *general* exposure assessment is used to designate those exposures that involve estimates of absorbed dose based on the handling of a specified amount of a chemical during specific types of applications. The accidental/incidental exposure scenarios involve specific types of events that could occur during any type of application. For general exposures in workers, exposure rates are expressed in units of mg of absorbed dose per kilogram of body weight per pound of chemical handled. For glyphosate, there are several worker exposure studies involving backpack applications that can be used to assess the quality general estimates used in many Forest Service risk assessments. These studies indicate that these general methods may be extremely conservative. Nonetheless, for this risk assessment, the standard worker exposure rates are used, recognizing that the upper range of exposures may overestimate risk. This conservative approach has little impact on the interpretation of risk because none of the worker exposures exceed a hazard quotient of unity. Central estimates of worker exposures span a very narrow range from 0.026 mg/kg/day to about 0.045 mg/kg/day. The upper range of exposures for the different application methods are about a factor of 10 higher, spanning a range from about 0.1 mg/kg/day to 0.3 mg/kg/day.

Under normal circumstances, members of the general public should not be exposed to substantial levels of glyphosate as a result of Forest Service activities. Nonetheless, any number of exposure scenarios can be constructed for the general public, depending on various assumptions regarding application rates, dispersion, canopy interception, and human activity. Several highly conservative scenarios are developed for this risk assessment. The two types of exposure scenarios developed for the general public include acute exposure and longer-term or chronic exposure. All of the acute exposure scenarios are primarily accidental. They assume that an individual is exposed to the compound either during or shortly after its application. Specific scenarios are developed for direct spray, dermal contact with contaminated vegetation, as well as the consumption of contaminated fruit, water, and fish. Most of these scenarios should be regarded as extreme, some to the point of limited plausibility. The longer-term or chronic exposure scenarios parallel the acute exposure scenarios for the consumption of contaminated fruit, water, and fish but are based on estimated levels of exposure for longer periods after application. Most acute accidental exposure scenarios for members of the general public are less than or similar to the general exposure scenarios in workers. The major exception is the scenario for an accidental spill of 200 gallons of a field solution into a small pond. This leads to modeled estimates of exposure in the range of 0.3 to about 4 mg/kg/day. This is an extraordinarily extreme and conservative scenario that is used in all Forest Service risk assessments. Most longer term estimates of exposure for members of the general public are much lower than exposure estimates for workers. The one exception involves the longer term consumption of contaminated fruit, which leads to timeweighted average estimated doses of 0.003 to 0.08 mg/kg/day.

**3.2.2. Workers.** A summary of the exposure assessments for workers is presented in Table 3-1. Two types of exposure assessments are considered: general and accidental/incidental. The term *general* exposure assessment is used to designate those exposures that involve estimates of absorbed dose based on the handling of a specified amount of a chemical during specific types of applications. The accidental/incidental exposure scenarios involve specific types of events that could occur during any type of application. Details regarding all of these exposure assessments are presented in the worksheets that accompany this risk assessment, as indicated in Table 3-1. These exposure assessments as well as other similar assessments for the general public (see Section 3.2.3) are based on the typical application rate of 2 lbs a.e./acre (see Section 2). The consequences of using different application rates in the range considered by the Forest Service are discussed further in the risk characterization (see Section 3.4).

**3.2.2.1. General Exposures** – As described in SERA (2001a), worker exposure rates are expressed in units of mg of absorbed dose per kilogram of body weight per pound of chemical handled. Based on analyses of several different pesticides using a variety of application methods, default exposure rates are estimated for three different types of applications: directed foliar (backpack), boom spray (hydraulic ground spray), and aerial. The specific rates generally used for each of these application methods is summarized in Table 3-2. As described in SERA (2001a), the ranges of estimated occupational exposure rates vary substantially among individuals and groups, (i.e., by a factor of 50 for backpack applicators and a factor of 100 for mechanical ground sprayers).

As detailed in Section 2, the most common method of application for glyphosate in Forest Service programs is backpack-applied directed foliar sprays. As indicated in Table 3-2, the default rates derived in SERA (2001a) for this method of application is 0.003 mg/kg bw per lb applied with a range of 0.0003 to 0.01 mg/kg bw per lb applied. For glyphosate, there are several worker exposure studies involving backpack applications that can be used to assess the quality of these values (Edmiston et al. 1995; Jauhianen et al. 1991; Lavy et al. 1992; Machado-Neto et al. 2000; Middendorf 1993; and Schneider et al. 1999). Three of these studies (Edmiston et al. 1995; Machado-Neto et al. 2000; Schneider et al. 1999) provide only deposition data and cannot be used directly to assess the use of the standard exposure rates summarized in Table 3-2. The other three studies (Jauhianen et al. 1991;Lavy et al. 1992; Middendorf 1993) involved backpack applications with both biomonitoring – i.e., urinary analysis – as well as deposition data as measures of exposure and are thus most relevant to the assessment of the general exposure values summarized in Table 3-2.

In the study by Jauhiainen et al. (1991), biological monitoring was conducted on five workers applying Roundup. Each worker handled an average of 9.8 L of an 8% solution of Roundup (360 g a.i./L or 270 g a.e/L). Thus, the amount of glyphosate acid handled each day was approximately 0.211 kg [9.8 L  $\times$  0.08  $\times$  0.270 kg/L] (Jauhiainen et al. 1991, p. 62, column one, top of page) or about 0.5 lbs. Urine samples [not total daily urine] were collected at the end of each work day for 1 week during the application period, and one sample was taken 3 weeks after the applications. The urine samples were assayed for glyphosate using gas

chromatography/electron capture with a limit of detection of 0.1 ng/ $\mu$ L or 0.1 mg/mL. No glyphosate was detected in any of the urine samples using this method.

One urine sample was assayed for glyphosate by gas chromatography/mass spectroscopy (GC/MS) and glyphosate was detected at a level of 0.085 ng/µL, equivalent to 0.085 µg/mL. Assuming that this urine sample was representative and using the default body weight of 70 kg and urinary output of 2000 mL/day (Worksheet A-02), the absorbed dose would be 0.17 mg or 170 µg [0.085 µg/mL × 2,000 mL] or 0.0024 mg/kg bw [0.17 mg  $\div$  70 kg]. The corresponding exposure rate would be 0.0048 mg/kg bw per lb a.e. applied [0.0024 mg/kg bw  $\div$  0.5 lb a.e.]. This value is very similar to the central estimate of 0.003 mg/kg bw per lb applied that is generally used for directed foliar applications (Table 3-2).

As with the study by Jauhianen et al. (1991), the Lavy et al. (1992) study involved applications of Roundup. Nursery workers applied Roundup to small weeds in a nursery bed by placing a 290 mL (2.5x3.5 cm) cylindrical metal shield surrounding the spray nozzle over the weed—to protect adjacent conifer seedlings—and then spraying the weeds with Roundup. Biological monitoring consisted of 5-day complete urine collections. In a total of 355 urine samples, no glyphosate was detected (limit of detection = 0.01 µg/mL). Assuming that the concentration of glyphosate in the urine was just below the limit of detection and assuming a urinary output of 2,000 mL (Worksheet A-02), the total absorbed dose would be 20 µg or 0.02 mg. The most exposed individual in this study weighed 63.5 kg and handled, on average, 0.54 kg [1.18 lbs] of glyphosate per day. Thus, the maximum absorbed dose of 0.02 mg corresponds to 0.0003 mg/kg bw [0.02 mg  $\div$  63.5 kg] and 0.00025 mg/kg bw per lb applied [0.0003 mg/kg  $\div$  1.18 lbs]. This is modestly below the lower range of the value of 0.0003 mg/kg bw per lb applied is generally used for directed foliar applications (Table 3-2). Based on passive monitoring, estimated exposure rates were about  $1.3 \times 10^{-3} (2.6 \times 10^{-4} \text{ to } 1.27 \times 10^{-2}) \text{ mg/kg bw per lb applied}. This central estimate and range is virtually identical to the values for directed foliar applications given in Table 3-2.$ 

The study by Middendorf (1993) also involved backpack (directed foliar) applications of Roundup, albeit in a more dilute mixture (2.3%). Middendorf (1993) provides data (urinary excretion, lbs applied, body weight, and deposition) on 15 workers at three different application sites. The average exposure rate for all workers was approximately 0.00032 mg/kg bw per lb applied with a range of 0.00013 to 0.001 mg/kg bw per lb applied. The central estimate from the Middendorf (1993) study is virtually identical to the lower range of 0.0003 mg/kg bw per lb typically used for directed foliar applications and the upper range noted in the Middendorf (1993) study is somewhat below the central estimate of 0.003 mg/kg bw given in Table 3-2.

The three worker studies (Jauhianen et al. 1991; Lavy et al. 1992; Middendorf 1993) that provide biomonitoring data sufficient to estimate absorbed doses in workers support the use of the exposure rates summarized in Table 3-2. If anything, the upper range of exposure – i.e., 0.01 mg/kg bw per lb applied – is likely to overestimate exposure. None of the estimates based on biomonitoring approach this rate. Nonetheless, for this risk assessment, the standard worker exposure rates are used, recognizing that the upper range of exposures may be extremely

conservative. As discussed further in Section 3.4 (Risk Characterization), this conservative approach has little impact on the interpretation of risk because none of the worker exposures exceed a hazard quotient of unity.

An estimate of the number of acres treated per hour is needed to apply these worker exposure rates. These values are taken from previous USDA risk assessments (USDA 1989a,b,c). The number of hours worked per day is expressed as a range, the lower end of which is based on an 8-hour work day with 1 hour at each end of the work day spent in activities that do not involve herbicide exposure. The upper end of the range, 8 hours per day, is based on an extended (10-hour) work day, allowing for 1 hour at each end of the work day to be spent in activities that do not involve herbicide exposure.

It is recognized that the use of 6 hours as the lower range of time spent per day applying herbicides is not a true lower limit. It is conceivable and perhaps common for workers to spend much less time in the actual application of a herbicide if they are engaged in other activities. Thus, using 6 hours can be regarded as conservative. In the absence of any published or otherwise documented work practice statistics to support the use of a lower limit, this conservative approach is used.

The range of acres treated per hour and hours worked per day is used to calculate a range for the number of acres treated per day. For this calculation as well as others in this section involving the multiplication of ranges, the lower end of the resulting range is the product of the lower end of one range and the lower end of the other range. Similarly, the upper end of the resulting range is the product of the upper end of one range and the upper end of the other range. This approach is taken to encompass as broadly as possible the range of potential exposures.

The central estimate of the acres treated per day is taken as the arithmetic average of the range. Because of the relatively narrow limits of the ranges for backpack and boom spray workers, the use of the arithmetic mean rather than some other measure of central tendency, like the geometric mean, has no marked effect on the risk assessment.

As detailed in worksheets C01a (directed foliar), C01b (broadcast foliar), and C01c (aerial), the central estimate of the amount handled per day is calculated as the product of the central estimates of the acres treated per day and the application rate. The ranges for the amounts handled per day are calculated as the product of the range of acres treated per day and the application rate. Similarly, the central estimate of the daily absorbed dose is calculated as the product of the central estimate of the exposure rate and the central estimate of the amount handled per day. The ranges of the daily absorbed dose are calculated as the range of exposure rates and the ranges for the amounts handled per day. The lower and upper limits are similarly calculated using the lower and upper ranges of the amount handled, acres treated per day, and worker exposure rate.

**3.2.2.2.** Accidental Exposures – Typical occupational exposures may involve multiple routes of exposure (i.e., oral, dermal, and inhalation); nonetheless, dermal exposure is generally the predominant route for herbicide applicators (Ecobichon 1998; van Hemmen 1992). Typical multi-route exposures are encompassed by the methods used in Section 3.2.2.1 on general exposures. Accidental exposures, on the other hand, are most likely to involve splashing a solution of herbicides into the eyes or to involve various dermal exposure scenarios.

Some glyphosate formulations can cause irritant effects in the skin and eyes (see Section 3.1.11). The available literature does not include quantitative methods for characterizing exposure or responses associated with splashing a solution of a chemical into the eyes; furthermore, there appear to be no reasonable approaches to modeling this type of exposure scenario quantitatively. Consequently, accidental exposure scenarios of this type are considered qualitatively in the risk characterization (Section 3.4).

There are various methods for estimating absorbed doses associated with accidental dermal exposure (U.S. EPA/ORD 1992, SERA 2001a). Two general types of exposure are modeled: those involving direct contact with a solution of the herbicide and those associated with accidental spills of the herbicide onto the surface of the skin. Any number of specific exposure scenarios could be developed for direct contact or accidental spills by varying the amount or concentration of the chemical on or in contact with the surface of the skin and by varying the surface area of the skin that is contaminated.

For this risk assessment, two exposure scenarios are developed for each of the two types of dermal exposure, and the estimated absorbed dose for each scenario is expressed in units of mg chemical/kg body weight. As specified in Table 3-2, the details of these exposure estimates are presented in the worksheets appended to this risk assessment.

Exposure scenarios involving direct contact with solutions of the chemical are characterized by immersion of the hands for 1 minute or wearing contaminated gloves for 1 hour. Generally, it is not reasonable to assume or postulate that the hands or any other part of a worker will be immersed in a solution of a herbicide for any period of time. On the other hand, contamination of gloves or other clothing is quite plausible. For these exposure scenarios, the key element is the assumption that wearing gloves grossly contaminated with a chemical solution is equivalent to immersing the hands in a solution. In either case, the concentration of the chemical in solution that is in contact with the surface of the skin and the resulting dermal absorption rate are essentially constant.

For both scenarios (the hand immersion and wearing the contaminated glove), the assumption of zero-order absorption kinetics is appropriate. Following the general recommendations of U.S. EPA/ORD (1992), Fick's first law is used to estimate dermal exposure. As discussed in Section 3.1.3, the experimental dermal permeability coefficient (Kp) for glyphosate is not used in this risk assessment because of the differences in the underlying models used for the exposure assessment

(no lag period) and the model used to estimate the experimental Kp that included a lag period. As further discussed in Section 3.1.3, this is the more conservative approach.

Exposure scenarios involving chemical spills on to the skin are characterized by a spill on to the lower legs as well as a spill on to the hands. In these scenarios, it is assumed that a solution of the chemical is spilled on to a given surface area of skin and that a certain amount of the chemical adheres to the skin. The absorbed dose is then calculated as the product of the amount of the chemical on the surface of the skin (i.e., the amount of liquid per unit surface area multiplied by the surface area of the skin over which the spill occurs and the concentration of the chemical in the liquid) the first-order absorption rate, and the duration of exposure. As summarized in Section 3.1.3, the first-order dermal absorption rates are taken from the study by Wester et al. (1991): an average value of  $4.1 \times 10^{-4}$  hour<sup>-1</sup> with a range of  $1.3 \times 10^{-4}$  to  $1.0 \times 10^{-3}$  hour<sup>-1</sup>. These values are included in Worksheet B05 rather than the values calculated in Worksheet B03, which are based on molecular weight and the K<sub>o/w</sub> for glyphosate.

For both scenarios, it is assumed that the contaminated skin is effectively cleaned after 1 hour. As with the exposure assessments based on Fick's first law, this product (mg of absorbed dose) is divided by body weight (kg) to yield an estimated dose in units of mg chemical/kg body weight. The specific equation used in these exposure assessments is taken from SERA (2000).

## 3.2.3. General Public.

**3.2.3.1.** *General Considerations* – Under normal circumstances, members of the general public should not be exposed to substantial levels of glyphosate as a result of Forest Service activities. Nonetheless, any number of exposure scenarios can be constructed for the general public, depending on various assumptions regarding application rates, dispersion, canopy interception, and human activity. Several highly conservative scenarios are developed for this risk assessment.

The two types of exposure scenarios developed for the general public include acute exposure and longer-term or chronic exposure. All of the acute exposure scenarios are primarily accidental. They assume that an individual is exposed to the compound either during or shortly after its application. Specific scenarios are developed for direct spray, dermal contact with contaminated vegetation, as well as the consumption of contaminated fruit, water, and fish. Most of these scenarios should be regarded as extreme, some to the point of limited plausibility. The longer-term or chronic exposure scenarios parallel the acute exposure scenarios for the consumption of contaminated levels of exposure for longer periods after application.

The exposure scenarios developed for the general public are summarized in Table 3-3, and the details regarding the assumptions and calculations involved in these exposure assessments are provided in worksheets D01-D09. The remainder of this section focuses on a qualitative description of the data supporting each of the assessments.

**3.2.3.2.** *Direct Spray* -- Direct sprays involving ground applications are modeled in a manner similar to accidental spills for workers (see Section 3.2.2.2.). In other words, it is assumed that the individual is sprayed with a solution containing the compound and that an amount of the compound remains on the skin and is absorbed by first-order kinetics. As with the similar worker exposure scenarios, the first-order absorption dermal absorption rates are taken from the study by Wester et al. (1991).

For direct spray scenarios, it is assumed that during a ground application, a naked child is sprayed directly with glyphosate. The scenario also assumes that the child is completely covered (that is, 100% of the surface area of the body is exposed), which makes this an extremely conservative exposure scenario that is likely to represent the upper limits of plausible exposure. An additional set of scenarios are included involving a young woman who is accidentally sprayed over the feet and legs. For each of these scenarios, some assumptions are made regarding the surface area of the skin and body weight. These assumptions are detailed and referenced in Worksheet A03.

**3.2.3.3.** Dermal Exposure from Contaminated Vegetation – In this exposure scenario, it is assumed that the herbicide is sprayed at a given application rate and that an individual comes in contact with sprayed vegetation or other contaminated surfaces at some period after the spray operation. For these exposure scenarios, some estimates of dislodgeable residue and the rate of transfer from the contaminated vegetation to the surface of the skin must be available. No such data are directly available for glyphosate, and the estimation methods of Durkin et al. (1995) are used as defined in worksheet D03. Other estimates used in this exposure scenario involve estimates of body weight, skin surface area, and first-order dermal absorption rates. The estimates of body weight and surface area are detailed in Worksheet A03. As with the direct spray scenarios, the first-order absorption dermal absorption rates are taken from the study by Wester et al. (1991).

**3.2.3.4.** *Contaminated Water* – Water can be contaminated from runoff, as a result of leaching from contaminated soil, from a direct spill, or from unintentional contamination from aerial applications. For this risk assessment, the two types of estimates made for the concentration of glyphosate in ambient water are acute/accidental exposure from an accidental spill and longer-term exposure to glyphosate in ambient water that could be associated with the application of this compound to a 10 acre block that is adjacent to and drains into a small stream or pond.

**3.2.3.4.1.** ACUTE EXPOSURE – Two exposure scenarios are presented for the acute consumption of contaminated water: an accidental spill into a small pond (0.25 acres in surface area and 1 meter deep) and the contamination of a small stream by runoff.

The accidental spill scenario assumes that a young child consumes contaminated water shortly after an accidental spill into a small pond. The specifics of this scenarios are given in Worksheet D05. Because this scenario is based on the assumption that exposure occurs shortly after the spill, no dissipation or degradation of glyphosate is considered. This is an extremely conservative scenario dominated by arbitrary variability. The actual concentrations in the water would depend

heavily on the amount of compound spilled, the size of the water body into which it is spilled, the time at which water consumption occurs relative to the time of the spill, and the amount of contaminated water that is consumed. Based on the spill scenario used in this risk assessment, the concentration of glyphosate in a small pond is estimated to range from 1 mg/L to 126 mg/L with a central estimate of about 18 mg/L (Worksheet D05).

The other acute exposure scenario for the consumption of contaminated water involves runoff into a small stream. There are several relevant monitoring studies that are useful for estimating exposure to glyphosate in streams. After an aerial application of Roundup at a rate of 2 kg a.i./ha [about 1.8 lb a.i./acre] over a 10 km<sup>2</sup> area in Vancouver Island, British Columbia, maximum concentrations in streams that were intentionally oversprayed reached about 0.16 mg/L and rapidly dissipated to less than 0.04 mg/L after 10 minutes. After a storm event, peak concentrations in stream water were less than 0.15 mg/L, rapidly dissipating to less than 0.02 mg/L before the end of the storm event (Feng et al. 1990, Kreutzweiger et al. 1989). At the same application rate, another Canadian study noted maximum stream concentrations of 0.109-0.144 mg/L, occurring 7-28 hours after aerial application. Similar results were noted in a study conducted in Oregon (Newton et al. 1984). Maximum water levels in streams reached 0.27 mg/L. This concentration was associated with repeated helicopter applications -i.e., direct spray) across a small stream at an application rate of 3.3 kg/ha (equivalent to 2.9 lbs/acre). In a more recent series of studies conducted in Oregon, Michigan, and Georgia, peak concentrations in streams shortly after application of glyphosate at 4.1 kg/ha (about 3.6 lbs/acre) ranged from less than 0.1 mg/L to about 1 mg/L (Newton et al. 1994, Figure 4, p. 1799). The upper range of 1 mg/L corresponds to 0.28 mg/L per lb applied. As reviewed by Neary and Michael (1996), some applications have resulted in much lower concentrations in streams, in the range of 0.003 to 0.007 mg/L per lb applied (Neary and Michael 1996, Table 11, p. 253).

While monitoring data provide practical and documented instances of water contamination, monitoring studies may not encompass a broad range of conditions which may occur during program applications – e.g., extremely heavy rainfall. Consequently, for this component of the exposure assessment, the monitored levels in ambient water are compared to modeled estimates based on GLEAMS (Groundwater Loading Effects of Agricultural Management Systems). GLEAMS is a root zone model that can be used to examine the fate of chemicals in various types of soils under different meteorological and hydrogeological conditions (Knisel et al. 1992). As with many environmental fate and transport models, the input and output files for GLEAMS can be complex. The general application of the GLEAMS model to estimating concentrations in ambient water are given in Attachment 2.

For the current risk assessment, the application site was assumed to consist of a 10 acre square area that drained directly into a small pond or stream. The pond dimensions (1000  $\text{m}^3$  or about 0.25 acres with an average depth of 1 meter) are the same as those used in the acute spill scenario.
The chemical specific values used in the GLEAMS modeling are summarized in Table 3-4. As discussed in Section 3.1.15, glyphosate degrades in the environment to metabolites whose toxicity appears to be of the same order or less than the toxicity of glyphosate. For the risk assessment, only glyphosate is modeled and the half-time in water is set to 1000 days – i.e., a gross overestimate that is intended to functionally ignore microbial degradation. This is a conservative approach in that the concentration of glyphosate (as opposed to less toxic degradation products) is maximized.

The GLEAMS modeling yielded estimates glyphosate runoff and percolation that were used to estimate concentrations in the stream adjacent to a treated plot, as detailed in Section 5.5 of Attachment 2. The results of the GLEAMS modeling for the small stream are summarized in Table 3-5 and the corresponding values for the small pond are summarized in Table 3-6. These estimates are expressed as the water contamination rates (WCR) - i.e., the concentration of the compound in water in units of mg/L normalized for an application rate of 1 lb a.e./acre.

Overall, the monitoring data are in relatively good agreement with the estimates from GLEAMS. The upper range of the estimates based on monitoring data – i.e., 0.28 mg/L per lb applied from Newton et al. 1994 – is very close to peak rates of about 0.2 to 0.4 mg/L per lb applied from the GLEAMS stream modeling (Table 3-6). The lower range of values from the monitoring data of 0.003 to 0.007 mg/L per lb applied (Neary and Michael 1996) is reasonably close to maximum values of 0.001 to 0.007 mg/L per lb applied from GLEAMS with rainfall rates of 15 inches per year or about 0.4 inches per storm event (Table 3-6).

Given the close correspondence between the monitoring data and modeling estimates of peak concentrations in stream water, the selection of monitoring data or modeling estimates makes very little difference to the exposure assessment. For this risk assessment, the range of WCR will be taken as 0.001 to 0.4 mg/L per lb applied per acre. The lower range is somewhat arbitrarily set: in very arid environments, no contamination is likely. The upper range of 0.4 mg/L per lb applied is based on the upper range of the modeled stream concentrations from GLEAMS based on sandy soil. The typical WCR is taken as 0.02 mg/L per lb applied per acre. This is the geometric mean of the range and the approximate value of maximum concentrations in stream water modeled for clay and loam soils at an annual rainfall rate of 50 inches per year.

**3.2.3.4.2.** *LONGER-TERM EXPOSURE* -- The scenario for chronic exposure to glyphosate from contaminated water is detailed in worksheet D07. This scenario assumes that an adult (70 kg male) consumes contaminated ambient water (2 liters/day) for a lifetime.

As with the above stream scenario, the estimated concentrations in pond water are based on modeled estimates from GLEAMS which are supported by monitoring data. The specific methods used to calculate the concentration of glyphosate in a small pond based on the GLEAMS output are detailed in Section 5.4 of Attachment 2.

The results of the GLEAMS modeling for the pond is summarized in Table 3-5 and the specific estimates of concentrations of glyphosate in ambient water that are used in this risk assessment are summarized in Worksheet B06. As with the corresponding values for a small stream, these estimates are expressed as the water contamination rates (WCR) in units of mg/L per lb applied per acre.

The typical WCR is taken as  $1 \mu g/L$  or 0.001 mg/L. This is about the average concentration that could be expected over a wide range of rainfall rates in clay or loam soil. The upper limit is taken as 0.008 mg/L, approximately the longer-term average concentration from sandy soil at rainfall rates of 25 inches per year. The lower limit of the WCR is taken as 0.0001 mg/L, about the average concentration from clay soil at an annual rainfall rate of 15 inches per year or from loam at an annual rainfall rate of 10 inches per year.

Monitoring data on glyphosate in pond water are reasonably consistent with these estimates. Over a period of 70 days after aerial application of 2.1 kg/ha (about 1.8 lbs/acre), Goldsborough and Brown (1993) reported concentrations of about 0.001–0.002 mg/L in the water of ponds that were less than 1 hectare (2.47 acres) in surface area and about 0.9 to 1.5 meters deep. Similarly, by 30 days after aerial applications of 3.7 lb/acre adjacent to small ponds (<1 m deep and 50 m<sup>2</sup> or 0.012 acres in surface area), monitored concentrations were in the range of 0.001 to 0.002 mg/L (Newton et al. 1994). Even in ponds that were directly sprayed with glyphosate at a rate of 0.89 kg a.i./ha (0.8 lbs a.i./acre), initial concentrations of between about 0.02 to 0.15 mg/L dissipated to about 0.001 mg/L by day 12 after application (Goldsborough and Beck 1989, Figure 1, p. 540).

**3.2.3.5.** Oral Exposure from Contaminated Fish -- Many chemicals may be concentrated or partitioned from water into the tissues of animals or plants in the water. This process is referred to as bioconcentration. Generally, bioconcentration is measured as the ratio of the concentration in the organism to the concentration in the water. For example, if the concentration in the organism is 5 mg/kg and the concentration in the water is 1 mg/L, the bioconcentration factor (BCF) is 5 L/kg [5 mg/kg ÷ 1 mg/L]. As with most absorption processes, bioconcentration depends initially on the duration of exposure but eventually reaches steady state. Details regarding the relationship of bioconcentration factor to standard pharmacokinetic principles are provided in Calabrese and Baldwin (1993).

Glyphosate has a relatively low potential for bioconcentration. In a bioconcentration study using <sup>14</sup>C-glyphosate, bioconcentration in carp exposed to levels in water of 5–50 µg/L ranged from about 10 after 1 day of exposure to about 40 after 14 days of exposure (Wang et al. 1994a). These estimates of bioconcentration, however, are based on total radioactivity rather than the identification of glyphosate residues. Consequently, the apparent bioconcentration appears to reflect the binding of glyphosate metabolites, including mineralized carbon, to fish tissue. Based on the study by Forbis (1989), the U.S. EPA/OPP (1993c, p. 36) used maximum bioconcentration factors of 0.38 for edible tissues and 0.52 for whole fish. Calabrese and Baldwin (1993) have reviewed a number of different methods for estimating BCF values in fish based on chemical and

physical properties. Using a log  $K_{o/w}$  of -4.85 at pH 6.86 (from Chamberlain et al. 1996 as summarized in Table 2-2), the estimated BCF values in fish would be well below unity, consistent with the study by Forbis (1989) and the BCF values used by U.S. EPA/OPP (1993c). For the current risk assessment, the values reported by Forbis (1989) and used by EPA/OPP (1993c) will be used to estimate dietary exposure to fish. These values are included in Worksheet B02 and used in all exposure assessments involving the consumption of contaminated fish. In the exposure assessment for humans, the assumption is made that the individual consumes only the edible portion of the fish. In the ecological risk assessment, the assumption is made that the predator completely consumes the fish and the whole body BCF is used.

For both the acute and longer-term exposure scenarios involving the consumption of contaminated fish, the water concentrations of glyphosate used are identical to the concentrations used in the contaminated water scenarios (see Section 3.2.3.4). The acute exposure scenario is based on the assumption that an adult angler consumes fish taken from contaminated water shortly after an accidental spill of 200 gallons of a field solution into a pond that has an average depth of 1 m and a surface area of 1000 m<sup>2</sup> or about one-quarter acre. No dissipation or degradation is considered. Because of the available and well-documented information and substantial differences in the amount of caught fish consumed by the general public and native American subsistence populations (U.S. EPA 1996), separate exposure estimates are made for these two groups, as illustrated in worksheet D08. The chronic exposure scenario is constructed in a similar way, as detailed in worksheet D09, except that estimates of glyphosate concentrations in ambient water are based on GLEAMS modeling as discussed in Section 3.2.3.4.

**3.2.3.6.** Oral Exposure from Contaminated Vegetation -- Under normal circumstances and in most types of applications, it is extremely unlikely that humans will consume vegetation contaminated with glyphosate. Any number of accidental scenarios could be developed involving either spraying of crops, gardens, or edible wild vegetation. Again, in most instances and particularly for longer-term scenarios, treated vegetation would probably show signs of damage from exposure to glyphosate (Section 4.3.2.4), thereby reducing the likelihood of consumption that would lead to significant levels of human exposure.

Notwithstanding that assertion, it is conceivable that individuals could consume contaminated vegetation that is accidentally sprayed. One of the more plausible scenarios involves the consumption of contaminated berries after the accidental spray of an area in which wild berries grow. The most relevant publication for assessing exposure from such a scenario is that of Siltanen et al. (1981). These investigators monitored levels of glyphosate on cowberries and bilberries after backpack sprays of Roundup at an application rate of 0.25 and 0.75 kg a.i./ha [0.22 and 0.67 lb a.i./acre]. At 6 days after treatment with 0.67 lb/acre, residues on cowberries were 1.6 mg/kg. At 7 days after treatment, residues on bilberries were 2.1 mg/kg. The residue data plotted over a 70-day post-application observation period are illustrated in Figure 3-2.

These data fit a first order model (p=0.004) with a dissipation rate of 0.015 day<sup>-1</sup>, which corresponds to a half-time of about 46 days. This model is indicated by the thick solid line in

Figure 3-1. Although the data fit a simple one-compartment first order model, visual inspection of the data suggests that a two-compartment first order model could also be applied.

The central estimate of residues immediately after application are approximately 1.6 ppm (mg/kg) with a 95% upper limit of 4 ppm. This corresponds to a residue rate of about 2.4 ppm per lb per acre [1.6 ppm  $\div$  0.67 lb a.i./acre] with an upper limit of 5.9 ppm per lb per acre [4 ppm  $\div$  0.67 lb a.i./acre]. As summarized in Worksheet A04, this is close to the estimate 1.5 ppm per lb per acre with an upper range of 7 ppm per lb per acre from the empirical relationships between application rate and concentration on fruit developed by Hoerger and Kenaga (1972) and somewhat lower than the estimate 7 ppm per lb per acre with an upper range of 15 ppm per lb per acre developed by Fletcher et al. (1994) which is in turn based on a re-analysis of data from Hoerger and Kenaga (1972). Because the study by Siltanen et al. (1981) uses glyphosate on two different types of fruit and because the results are reasonably consistent between the two types, there is no reason to use either the general relationships developed by Hoerger and Kenaga (1972) or Fletcher et al. (1994) for scenarios involving contaminated fruit. For other scenarios involving other types of vegetation, the estimates from Fletcher et al. (1994) are used since they are somewhat more conservative than the earlier estimates by Hoerger and Kenaga (1972).

The two accidental exposure scenarios developed for this exposure assessment include one scenario for acute exposure, as defined in Worksheet D03 and one scenario for longer-term exposure, as defined in Worksheet D04. In both scenarios, the concentration of glyphosate on contaminated vegetation is estimated using the empirical relationships between application rate and concentration on vegetation developed by Fletcher et al. (1994) which is in turn based on a re-analysis of data from Hoerger and Kenaga (1972). These relationships are defined in worksheet A04. For the acute exposure scenario, the estimated residue level is taken as the product of the application rate and the residue rate (Worksheet D03).

For the longer-term exposure scenario (D04), a duration of 90 days is used and the dissipation on the vegetation is estimated using the halftime of 46 days from Siltanen et al. (1981). As summarized in Table 2-2, this is the most conservative value - i.e., the longest halftime which leads to the highest time-weighted average residues. Although the duration of exposure of 90 days is somewhat arbitrarily chosen, this duration is intended to represent the consumption of contaminated fruit that might be available over one season. Longer durations could be used for certain kinds of vegetation but would lower the estimated dose (i.e., would result in a less conservative exposure assessment).

For the longer-term exposure scenarios, the time-weighted average concentration on fruit is calculated from the equation for first-order dissipation. Assuming a first-order decrease in concentrations in contaminated vegetation, the concentration in the vegetation at time t after spray,  $C_t$ , can be calculated based on the initial concentration,  $C_{\theta}$ , as:

$$\mathbf{C}_{\mathrm{t}} = \mathbf{C}_{\mathrm{0}} \times \mathrm{e}^{\mathrm{-kt}}$$

where k is the first-order decay coefficient  $[k=ln(2)\div t_{50}]$ . Time-weighted average concentration ( $C_{TWA}$ ) over time *t* can be calculated as the integral of  $C_t$  (De Sapio 1976, p. p. 97 ff) divided by the duration (*t*):

$$\mathbf{C}_{\mathrm{TWA}} = C_{\theta} (\mathbf{1} - \mathbf{e}^{-\mathbf{k} t}) \div (\mathbf{k} t).$$

For the acute exposure scenario, it is assumed that a woman consumes 1 lb (0.4536 kg) of contaminated fruit. Based on statistics summarized in U.S. EPA/ORD (1996) and presented in worksheet D04, this consumption rate is approximately the mid-range between the mean and upper 95% confidence interval for the total vegetable intake for a 64 kg woman. The range of exposures presented in Table 3-3 is based on the range of concentrations on fruit and the typical application rate for glyphosate. The longer-term exposure scenario is constructed in a similar way, except that the estimated exposures include the range of fruit consumption (Worksheet A03) as well as the range of concentrations on fruit.

A separate scenario involving the consumption of contaminated vegetation by drift rather than direct spray is not developed in this risk assessment. As detailed further in Section 3.4, this elaboration for glyphosate is not necessary because the direct spray scenario leads to estimates of risk that are below a level of concern. Thus, considering spray drift and a buffer zone quantitatively would have no impact on the characterization of risk.

### 3.3. DOSE-RESPONSE ASSESSMENT

**3.3.1. Overview.** Generally, the dose-response assessments used in Forest Service risk assessments adopt RfDs proposed by the U.S. EPA as indices of 'acceptable' exposure. An RfD is basically defined as a level of exposure that will not result in any adverse effects in any individual. The U.S. EPA RfDs are used because they generally provide a level of analysis, review, and resources that far exceed those that are or can be conducted in the support of most Forest Service risk assessments. In addition, it is desirable for different agencies and organizations within the federal government to use concordant risk assessment values.

The most recent RfD on glyphosate is that proposed by the U.S. EPA Office of Pesticide Programs. This RfD of 2 mg/kg/day was proposed originally in the RED for glyphosate and was also used in the recent glyphosate pesticide tolerances. This RfD is based on teratogenicity study in rabbits (Rodwell et al. 1980b) in which no effects observed in offspring at any dose levels and maternal toxicity was observed at 350 mg/kg/day with a NOAEL of 175 mg/kg/day. Using an uncertainty factor of 100 - 10 for sensitive individuals and 10 for species-to-species extrapolation – U.S. EPA/OPP derived the RfD of 2 mg/kg/day, rounding the value of 1.75 mg/kg/day to one significant digit.

For the current risk assessment, the RfD 2 mg/kg/day derived by U.S. EPA/OPP is used as the basis for characterizing risk from longer-term exposures in this risk assessment. For short-term exposures, the value of 2 mg/kg/day recommended by U.S. EPA's Office of Drinking Water is used. Since this is identical to the chronic RfD, this approach is equivalent to applying the same RfD to be short-term and long-term exposures. Given the lack of a significant dose-duration relationship for glyphosate, this approach seems appropriate.

**3.3.2. Existing Guidelines.** Generally, the dose-response assessments used in Forest Service risk assessments adopt RfDs proposed by the U.S. EPA as indices of 'acceptable' exposure. An RfD is basically defined as a level of exposure that will not result in any adverse effects in any individual. The U.S. EPA RfDs are used because they generally provide a level of analysis, review, and resources that far exceed those that are or can be conducted in the support of most Forest Service risk assessments. In addition, it is desirable for different agencies and organizations within the federal government to use concordant risk assessment values. The current risk assessment is somewhat complicated by the fact that the U.S. EPA has proposed two chronic RfDs for glyphosate and World Health Organization has proposed another comparable value, referred to as an ADI (acceptable daily intake).

The most recent RfD on glyphosate is that proposed by the U.S. EPA Office of Pesticide Programs. This RfD of 2 mg/kg/day was proposed originally in the RED for glyphosate (U.S. EPA/OPP 1993c) and was also used in the recent glyphosate pesticide tolerances (U.S. EPA/OPP 2002). This RfD is based on teratogenicity study in rabbits (Rodwell et al. 1980b) in which doses of 75, 175, or 350 mg/kg/day were administered by gavage on days 6-27 of gestation. As detailed in Appendix 6, no effects observed in offspring at any dose levels. Maternal toxicity – i.e., nasal discharge, diarrhea, altered physical appearance and death among dams – was observed

at 350 mg/kg/day. Using an uncertainty factor of 100 - 10 for sensitive individuals and 10 for species-to-species extrapolation – U.S. EPA/OPP (1993c) derived the RfD of 2 mg/kg/day, rounding the value of 1.75 mg/kg/day to one significant digit.

The U.S. EPA has also derived an RfD on glyphosate of 0.1 mg/kg/day (U.S. EPA/IRIS 1990). This RfD was originally derived in 1990 by the U.S. EPA Integrated Risk Information System (IRIS) workgroup and is the current (Nov. 2002) RfD posted on IRIS. This RfD is based on a dietary 3-generation reproduction study (Schroeder and Hogan 1981), which is also detailed in Appendix 6. In this study, rats were exposed to glyphosate in the diet with resulting dose rates of 0, 3, 10 and 30 mg/kg/day. No signs of maternal toxicity were observed. The only effect in offspring was an increase in the incidence of unilateral renal tubular dilation in male pups from the F3b mating. Thus, the NOAEL was identified as 10 mg/kg/day and an uncertainty factor of 100 was applied to derive an RfD of 0.1 mg/kg/day.

Unlike the two RfD values proposed by the U.S. EPA, the ADI proposed by WHO (1994) is not based on a reproductive toxicity study. Instead, WHO (1994) selected a life-time feeding study in rats (Lankas and Hogan 1981). The Lankas and Hogan (1981) study is detailed in Appendix 5 and involved dietary concentrations of 0, 30, 100, or 300 ppm for 26 months which corresponded to approximate daily doses (expressed as mg/kg body weight) of 0, 3.1, 10.3, or 31.5 mg/kg/day for males and 0, 3.4, 11.3, or 34.0 mg/kg/day for females. No effects were seen at any dose levels and thus WHO (1994) used a NOAEL of 31.5 mg/kg/day and uncertainty factor of 100. Rounding to one significant digit, the recommended ADI was set at 0.3 mg/kg/day.

The U.S. EPA/OPP will sometimes derive acute RfD values that can be used to assess risks associated with very short-term exposures -i.e., accidental spills. No acute RfD has been proposed, however, for glyphosate.

The Office of Drinking Water (U.S. EPA/ODW 1992) has proposed a 10-day health advisory for glyphosate of 17.5 mg/L and a longer-term health advisory of 1 mg/L. The longer-term health advisory is based on the U.S. EPA RfD of 0.1 mg/kg/day reference dose, as summarized above. The 10-day health advisory is based on the NOAEL of 175 mg/kg/day from Rodwell et al. 1980b, which is also summarized above. An uncertainty factor of 100 was applied to this NOAEL and the 10-day exposure limit was set at 1.75 mg/kg/day. This value was multiplied by 10 kg, the default weight for a child used by U.S. EPA/ODW (1992) and divided by 1 L, the default amount of water consumed by a child. Rounding the value of 1.75 mg/kg/day to one significant digit, this is equivalent to a short-term RfD of 2 mg/kg/day, identical to the chronic RfD currently recommended by U.S. EPA/OPP (1993c; 2002).

**3.3.3. Dose-Response and Dose-Severity Relationships.** There is a striking concordance between the available human and animal data on the acute toxicity of glyphosate. The dose-mortality data in humans is consistent with estimates of oral  $LD_{50}$  values in experimental mammals. Several different dose-response models can be use to quantitatively compare the lethality data on humans with those available on experimental mammals. In general, different

dose-response models yield similar results in the region of observed responses but may differ substantially in the low dose region. To estimate the  $LD_{50}$  in humans, variants of the multistage model were used, one non-threshold and one with a threshold. Both models yielded virtually identical estimates of the  $LD_{50}$ , approximately 3000 mg/kg, very similar to the range of 2000 to 6000 mg/kg reported in experimental mammals (Section 3.1.4).

For systemic toxic effects, it is generally assumed that population thresholds exist. In other words, below a certain dose, no individual in the population will respond. This assumption is fundamental to risk assessment methods for systemic toxic effects. For cancer, population thresholds are not generally assumed and non-threshold models are considered appropriate. The threshold version of the multi-stage model used in this analysis yielded an estimate of the threshold at about 445 mg/kg.

The dose-severity relationships for experimental mammals and humans are also similar, as illustrated in Figure 3-2. In this figure, the animal data are taken from Appendices 4 and 5. In cases where dietary exposure levels were not converted to units of dose in mg/kg/day, such conversions were made using the methods presented in U.S. EPA (1986b, Reference Values for Risk Assessment). The animal data are categorized using four standard severity levels: NOEL (no observed effect level), NOAEL (no observed adverse effect level), AEL (adverse effect level), and FEL (frank effect level), as discussed in SERA (2001a). Three different groups of end-points are presented: general systemic toxic effects (T), reproductive or developmental effects (R), and acute LD<sub>50</sub> values (A). The estimated human oral LD<sub>50</sub>, as estimated above, is plotted as a FEL with the LD<sub>50</sub> values from experimental mammals.

These studies span exposure periods ranging from 1 day to more than 2 years. The exposure axis is not presented in this figure. As discussed below, the duration of exposure is not an important variable in the toxicity of glyphosate, probably because of the rapid rate of glyphosate excretion. The data from study on which the RfD is based as well as the RfD itself are plotted and labeled with arrows.

Figure 3-2 also includes human data from the study by Tominack et al. (1991). These investigators report mean dose levels associated with four levels of severity. Patients in the least severe category were asymptomatic. The average amount of Roundup consumed by these patients was 31 mL. Assuming an average body weight of 60 kg for the individuals from Taiwan and using the concentration of 356 g of glyphosate a.e./L, this corresponds to an average dose of 184 mg/kg. This is plotted as a box just below the NOEL line in Figure 3-2 and labeled as "*No apparent effects*". Patients with transient signs or symptoms localized to the oral mucosa or gastrointestinal tract had, on average, consumed 72 mL ( $\approx$ 427 mg/kg). This point is labeled as "*Mild poisoning*" in Figure 3-2. Patient with "*Moderate poisoning*" had consumed on average 176 mL ( $\approx$ 1,044 mg/kg). These patients evidenced gastrointestinal tract irritation lasting less than 24 hours, transient decreases in blood pressure or decreased urinary output, transient hepatic or renal damage, acid-base disturbances, or pulmonary dysfunction which did not require intubation. "*Severe poisoning*", which included fatal cases, occurred in patients who had on average consumed 216 mL ( $\simeq$ 1,282 mg/kg). The publication by Tominack et al. (1991) also reports the variability of the doses associated with each of these severity levels.

For experimental mammals, the dose-severity relationships can be assessed using categorical regression analyses (Durkin et al. 1992; Hertzberg 1989; McCullagh 1980). This approach correlates categorical responses—such as NOELs, NOAELs, AELs, and FELs—with factors that may influence the response such as dose and duration of exposure. The method results in estimates of the probability of a group of animals subjected to a given exposure being classified into a given category. For the statistical analyses, data on NOELs and NOAELs were combined. This was done for two related reasons. First, the primary concern for this risk assessment is the delineation between regions of adverse and non-adverse effects. Thus, the distinction between a NOEL and NOAEL is not critical. Second, many reported NOELs could be artifacts of the level of detail at which the animals are examined. For example, simply because there are no adverse effects based on gross examination of organs does not mean that effects might not be seen if all organs were examined microscopically. Consequently, analyses were conducted using both four categories (NOELs, NOAELs, AELs, and FELs) as well as two categories (NOELs and NOAELs combined).

Initially, the categorical regression was conducted on both dose and duration of exposure. The effect of duration was not statistically significant (p=0.7267). This seems reasonable given the data on the influence of duration of exposure on toxicity. For example, all of the LD<sub>50</sub> values shown in Figure 3-2 involved single doses. Many of the AELs, some of which are doses at or above reported LD<sub>50</sub> values, involved exposure periods of up 2 years. This apparently anomalous result can be explained by two factors. First, all of the LD<sub>50</sub> studies involved intubations: the animal was given the total dose by stomach tube at one time. Most of the subchronic and chronic studies involved dietary exposures, in which the daily dose was spread out over the course of the day depending on the animals eating habits. Thus, the animals who were intubated in LD<sub>50</sub> studies received essentially more severe exposures for a given dose. Secondly, for chemicals that are eliminated rapidly and do not cause cumulative damage, there is often very little relationship between the duration of exposure and the severity of response for a fixed dose level.

Because of the lack of significance of duration, the analysis was re-run using only dose as the independent variable. The results of this analysis indicate that the probability of an adverse effect at the RfD of 0.1 mg/kg/day is 0.0005. At doses of 1 mg/kg/day the probability of observing an adverse effect is 0.003. At a dose that is 100 fold above the RfD (i.e., 10 mg/kg/day) the probability of an adverse effect is 0.12. This analysis suggest that the current RfD is highly protective and that the proposed alternative RfD of 2 mg/kg/day is also protective. At this higher level, the probability of an adverse effect is 0.006.

All of the above estimates are based on the two category analysis - the segregation of any adverse effect from non-adverse effects. They indicate the probability of a group of animals exposed at the specified dose level evidencing responses sufficiently, albeit perhaps minimally, severe to classify the dose level as *adverse* based on the responses observed in the group of animals.

The four category analysis can be used to estimate the probability of observing effects that would be classified as *frank signs of toxicity*. These effects are sufficiently severe that they can be observed in the whole organism without the use of invasive methods. The probability of a frank toxic effect at the RfD of 0.1 mg/kg/day is 0.00005. At the proposed alternative RfD of 2 mg/kg/day, the probability increases to only 0.0006.

The consistency between the categorical analysis using data on groups of experimental animals and dose-response analyses of the human lethality data using the multi-stage model is relatively good. At the estimated threshold for lethality, 445 mg/kg, the probability of observing a frank toxic effect is about 0.04. At this dose, the non-threshold version of the multi-stage model estimates the probability of mortality at about 0.02. At the estimated human LD<sub>50</sub> of about 3000 mg/kg, the categorical regression using two categories (NOELs and NOAELs combined as well as AELs and FELs combined) indicates the probability of observing an adverse or frank effect at 0.7. The four category model, however, substantially underestimates the probability of observing a FEL, 0.13. Visual inspection of Figure 3-2 suggests that this is attributable to the relatively small number of FELs in experimental mammals and the overlap of FELs with AELs. As discussed above, this overlap may be related to the rapid elimination and lack of cumulative damage in longer-term studies.

A somewhat more detailed analysis could be conducted on data collected by Tominack et al. (1991) that would provide information on the probabilities of individuals rather than groups being classified as adverse responders to given doses of glyphosate. The necessary data for such an analysis [amount consumed, body weight or sex/age, and severity classification] is not presented in the Tominack publication.

The significance of the categorical regression on animals and the available human data relates to the use of the uncertainty factor. As summarized in the previous section, the current RfD as well as the proposed U.S. EPA Office of Pesticides alternative use an uncertainty factor of 10 for species to species extrapolation (i.e., extrapolating from experimental animals to humans). This is a common default procedure. For glyphosate, however, the available data suggest that humans are no more sensitive to glyphosate than experimental mammals. This in turn suggests that the current and proposed RfD may be overly protective by a factor of 10 or greater. In other words, the RfDs suggest that no adverse effects are anticipated at doses of 0.1-2 mg/kg/day. The human data suggest that no frank adverse effects are likely at doses substantially above 10 mg/kg/day.

**3.3.4. RfD Values Used in Risk Assessment.** As with any chemical that has been extensively studied, the data base on glyphosate is large, complex, and open to differing interpretations. Based on the categorical regression analysis summarized in the previous section, all of the RfD values currently recommended would appear to be protective for both short- and long-term exposures. Notwithstanding this, however, the availability of three different chronic values must be reconciled explicitly.

The ADI of 0.3 mg/kg/day that is proposed by WHO (1994) is discussed by U.S. EPA/OPP (1993c, p. 19). The Schroeder and Hogan (1981) study used by WHO (1994) failed to identify an effect level. Thus, the NOAEL of 31.5 mg/kg/day identified in the Schroeder and Hogan (1981) study is essentially "free-standing". In other words, the true threshold for toxicity could be much higher. In such cases, it is generally appropriate to utilize a different study that identifies both a NOAEL and a LOAEL and base the RfD on the NOAEL. This is the approach taken in U.S. EPA/OPP (1993c), using a NOAEL of 175 mg/kg/day and a LOAEL of 350 mg/kg/day from the gavage teratogenicity study in rabbits (Rodwell et al. 1980b)

The Schroeder and Hogan (1981) study is cited but not discussed in U.S. EPA/OPP (1993c). In addition, U.S. EPA/OPP (1993c) does not discuss the RfD derived by U.S. EPA/IRIS (1990). Thus, the LOAEL of 30 mg/kg/day based on increase in renal tubular dilation in male F3b pups (Schroeder and Hogan 1981) is not specifically addressed by or compared to the NOAEL of 175 mg/kg/day used by U.S. EPA/OPP (1993c).

Nonetheless, the U.S. EPA/OPP (1993a) Science Chapter, which is a support document prepared for the RED, does explicitly discuss the renal tubular dilation reported by Schroeder and Hogan (1981) at 30 mg/kg/day and notes that this effect was not seen in a later two-generation reproductive study conducted by Reyna (1990) that involved doses of up to 1500 mg/kg/day. Thus, U.S. EPA/OPP (1993a) classified the effects reported by Schroeder and Hogan (1981) as "spurious". Given that the Reyna (1990) study involved three dose levels above 30 mg/kg/day – i.e., 100, 500, and 1500 mg/kg/day – and given that similar lesions have not been noted in other studies, this decision by U.S. EPA/OPP (1993a) appears to be reasonable.

An additional factor to consider in assessing the protectiveness of the U.S. EPA/OPP (1993c) RfD of 2 mg/kg/day and the corresponding NOAEL of 175 mg/kg/day is the report by Yousef et al. (1995) of increases in abnormal and dead sperm in rabbits after acute exposures to glyphosate. As discussed in some detail in Section 3.1.9, the Yousef et al. (1995) study is not well documented and does not specify the absolute doses used but only cites the doses relative to the  $LD_{50}$  for glyphosate, which is also not specified in the Yousef et al. (1995) study. Assuming the doses of 38 and 380 mg/kg estimated for this study (see Section 3.1.9) are reasonable approximations, the presumptive LOAEL of 38 mg/kg is not consistent with the better documented reproductive NOAEL of 30 mg/kg/day from Schroeder and Hogan (1981). [Note that the unilateral renal tubular dilation in male F3b pups reported at 30 mg/kg/day by Schroeder and Hogan (1981) is a LOAEL for systemic toxicity but, since no effects were observed on reproduction, this dose is a NOAEL for reproductive toxicity.] Thus, while the report by Yousef et al. (1995) must be considered, it is not supported by other well-conducted and welldocumented studies and does not impact the selection of the RfD for this current risk assessment.

Thus, for the current risk assessment, the RfD 2 mg/kg/day derived by U.S. EPA/OPP (1993c) and used by U.S. EPA/OPP (2002) will be used as the basis for characterizing risk from longer-term exposures in this risk assessment (Section 3.4). For short-term exposures, the value of 2 mg/kg/day recommended by U.S. EPA/ODW (1992) will be used. Since this is identical to the

chronic RfD, this approach is equivalent to applying the same RfD to be short-term and long-term exposures. Given the lack of a significant dose-duration relationship for glyphosate (Section 3.3.3.), this approach seems appropriate.

### **3.4. RISK CHARACTERIZATION**

**3.4.1. Overview.** The risk characterization for both workers and members of the general public are reasonably consistent in unambiguous. For both groups, there is very little indication of any potential risk at the typical application rate of 2 lbs a.e./acre. Even at the upper range of plausible exposures in workers, most hazard quotients are below the level of concern.

For workers, the highest hazard quotient – i.e., 0.2, the upper range for workers involved in broadcast ground spray – is below the level of concern by a factor of about 5. The highest hazard quotient for any accidental exposure scenario for workers - i.e., 0.006 for the upper range of the hazard quotient for spill over the lower legs for one hour - is lower than the level of concern by a factor of over 150. Confidence in these assessments is reasonably high because of the availability of dermal absorption data in human as well as worker exposure studies. The Forest Service may apply glyphosate at a maximum rate of 7 lbs a.e./acre, a factor of 3.5 higher than the typical application rate of 2 lbs a.e./acre. This has essentially no impact of the risk characterization for workers. The highest hazard quotient for the typical application rate is 0.2. For an application rate of 7 lbs a.e./acre, the corresponding hazard quotient would be higher by a factor of 3.5 or 0.7, which is still below the level of concern.

From a practical perspective, the most likely accidental exposure for workers that might require medical attention involves accidental contamination of the eyes. Glyphosate and glyphosate formulations are skin and eye irritants. Quantitative risk assessments for irritation are not normally derived, and, for glyphosate specifically, there is no indication that such a derivation is warranted. Glyphosate with the POEA surfactant is about as irritating as standard dish washing detergents, all purpose cleaners, and baby shampoos. As with the handling of any chemical, including a variety of common household products, reasonable care should be taken to avoid contact of skin and eyes.

The only area of remarkable uncertainty involving worker exposures concerns the potential health effects during brown-and-burn operations. The combustion of wood and wood by-products may produce a number of toxic compounds. This is a concern with brown-and-burn operations but does not pertain to the use of glyphosate or any other herbicide. The potential effects of combustion products is common to all risk assessments of materials that might be subject to burning. With the exception of some plastics, the combustion products of which are known to pose a risk to fire fighters, the combustion products of most chemicals have not been examined in detail. The necessity of addressing this data gap must be weighed against the need to address other data gaps on glyphosate and other chemicals. The combustion products of burning wood and vegetation are respiratory irritants as well as carcinogens, and exposure to these combustion products should be avoided. There is no basis for believing that the presence of low or even high levels of glyphosate residues will have a significant impact on this hazard.

For members of the general public, none of the longer-term exposure scenarios exceed or even approach a level of concern. Although there are several uncertainties in the longer-term exposure assessments for the general public, the upper limits for hazard indices are below a level of concern by factors of about 25 (longer term consumption of contaminated fruit) to over two million (2,500,000 for longer-term consumption of fish by the general population). The risk characterization is thus relatively unambiguous: based on the available information and under the foreseeable conditions of application and exposure, there is no route of exposure or exposure scenario suggesting that the general public will be at risk from longer-term exposure to glyphosate. As with the hazard characterization for workers, an application rate of 7.5 lbs a.e./acre makes no difference in the assessment of potential risks. At this application rate, the highest hazard quotient would be about  $0.14 [0.04 \times 3.5]$ , which is still below a level of concern by a factor of about 7.

One acute exposure scenario does exceed the level of concern at the upper range at the typical application rate of 2 lbs a.e./acre. The exposure scenario for the consumption of contaminated water after an accidental spill into a small pond results in an excursion above the RfD at the upper limit of exposure – i.e, a hazard quotient of 2. This exposure scenario is extreme to the point of limited plausibility. This sort of scenario is routinely used in Forest Service risk assessments as an index of the measures that should be taken to limit exposure in the event of a relatively large spill into a relatively small body of water. For glyphosate, as well as for most other chemicals, this exposure assessment indicates that such an event would require measures to ensure that members of the general public do not consume contaminated water.

At the highest application rate that might be used in Forest Service programs, the accidental spill scenario is the only other scenario that results in a hazard quotient above unity. At this application rate, the associated dose is about 14 mg/kg, which is still below the dose of 184 mg/kg associated with no apparent overt effects in humans by a factor of over 10.

**3.4.2. Workers.** A quantitative summary of the risk characterization for workers is presented in Table 3-7 for the typical application rate of 2 lbs a.e./acre. The quantitative risk characterization is expressed as the hazard quotient. For both general and accidental exposures, the hazard index is calculated as the estimated doses from Table 3-1 divided by the RfD of 2 mg/kg/day. As discussed in Section 3.3.2, there is no substantial dose-duration-effect relationship for glyphosate and the acute and chronic RfDs are identical.

Given the very low hazard quotients for accidental exposure, the risk characterization is reasonably unambiguous. None of the accidental exposure scenarios approach a level of concern. While the accidental exposure scenarios are not the most severe one might imagine (e.g., complete immersion of the worker or contamination of the entire body surface for a prolonged period of time) they are representative of reasonable accidental exposures. Given that the highest hazard quotient for any accidental exposure scenario - i.e., 0.0057 for the upper range of the hazard quotient for spill over the lower legs for one hour - is a factor of 175 lower than the level of concern, substantially more severe and much less plausible scenarios would be required to suggest a potential for systemic toxic effects from accidental exposures. Confidence in this assessment is reasonably high because of the availability of dermal absorption data in human. As

discussed in Section 3.2.2.2, the first-order dermal absorption rates used in this exposure scenario are taken from the study by Wester et al. (1991).

The hazard quotients for general or longer term exposures in workers are also unambiguous. Even at the upper range of plausible exposures all hazard quotients are below the level of concern. The highest hazard quotient – i.e., 0.2, the upper range for workers involved in broadcast ground spray – is below the level of concern by a factor of 5.

As noted in Section 2, the Forest Service may apply glyphosate at a maximum rate of 7 lbs a.e./acre, a factor of 3.5 higher than the typical application rate of 2 lbs a.e./acre. This has essentially no impact of the risk characterization for workers. As noted above, the highest hazard quotient for the typical application rate is 0.2. For an application rate of 7 lbs a.e./acre, the corresponding hazard quotient would be higher by a factor of 3.5 or 0.7, which is still below the level of concern.

As summarized in section 3.1.11, glyphosate and glyphosate formulations are skin and eye irritants. Quantitative risk assessments for irritation are not normally derived, and, for glyphosate specifically, there is no indication that such a derivation is warranted. As discussed by Maibach (1986), glyphosate with the POEA surfactant is about as irritating as standard dish washing detergents, all purpose cleaners, and baby shampoos. As with the handling of any chemical, including a variety of common household products, reasonable care should be taken to avoid contact of skin and eyes.

The only area of remarkable uncertainty concerns brown-and-burn operations. Glyphosate, like Roundup, does not appear to be very toxic by inhalation (Section 3.1.13). Although residues of glyphosate in air during brown-and-burn operations have not been measured, they are likely to be very low given that brown-and-burn operations take place about 30 to180 days after treatment with the herbicide and the foliar half-life is from 1.6 to 46 days. Consequently, there is no evidence to suggest that toxic levels of glyphosate are likely to be encountered.

The combustion of wood and wood by-products may produce a number of toxic compounds. This is a concern with brown-and-burn operations but does not pertain to the use of glyphosate or any other herbicide. Nevertheless, as discussed in section 3.1.13, glyphosate forms a polycondensate on combustion at temperatures ranging from 200 to 240°C. It is likely that other combustion products are formed under different combustion conditions. No information is available regarding the inhalation toxicity of the polycondensate or other possible combustion products.

The potential effects of combustion products is common to all risk assessments of materials that might be subject to burning. With the exception of some plastics, the combustion products of which are known to pose a risk to fire fighters, the combustion products of most chemicals have not been examined in detail. The necessity of addressing this data gap must be weighed against the need to address other data gaps on glyphosate and other chemicals. The combustion products

of burning wood and vegetation are respiratory irritants as well as carcinogens, and exposure to these combustion products should be avoided. There is no basis for believing that the presence of low or even high levels of glyphosate residues will have a significant impact on this hazard.

**3.4.3. General Public.** The quantitative hazard characterization for the general public is summarized in Table 3-8 for the typical application rate of 2 lbs a.e./acre. Like the quantitative risk characterization for workers, the quantitative risk characterization for the general public is expressed as the hazard quotient using the RfD of 2 mg/kg/day for both acute and longer-term exposures.

None of the longer-term exposure scenarios exceed or even approach a level of concern. Although there are several uncertainties in the longer-term exposure assessments for the general public, as discussed in Section 3.2.3, the upper limits for hazard indices are below a level of concern by factors of about 25 (longer term consumption of contaminated fruit) to over two million (about 2,325,000 for longer-term consumption of fish by the general population). The risk characterization is thus relatively unambiguous: based on the available information and under the foreseeable conditions of application and exposure, there is no route of exposure or exposure scenario suggesting that the general public will be at risk from longer-term exposure to glyphosate. As with the hazard characterization for workers, an application rate of 7.5 lbs a.e./acre makes no difference in the assessment of potential risks. At this application rate, the highest hazard quotient would be  $0.14 [0.04 \times 3.5]$ , which is still below a level of concern by a factor of about 7.

One acute exposure scenario does exceed the level of concern at the upper range at the typical application rate of 2 lbs a.e./acre. The exposure scenario for the consumption of contaminated water after an accidental spill into a small pond results in an excursion above the RfD at the upper limit of exposure – i.e., a hazard quotient of 2. As detailed in Section 3.2.3.4.1, this exposure scenario is extreme to the point of limited plausibility. This sort of scenario is routinely used in Forest Service risk assessments as an index of the measures that should be taken to limit exposure in the event of a relatively large spill into a relatively small body of water. For glyphosate, as well as for most other chemicals, this exposure assessment indicates that such an event would require measures to ensure that members of the general public do not consume contaminated water. As detailed in Table 3-3, the upper range of exposure, it is far below doses that would likely result in overt signs of toxicity. As detailed in Section 3.3.3, a dose of 184 mg/kg as Roundup – i.e., glyphosate plus surfactant – was not associated with any overt signs of toxicity in humans – and mild signs of toxicity were apparent at doses of 427 mg/kg, over 100 times higher than the upper range of 4.1 mg/kg in the accidental spill scenario.

At the highest application rate that might be used in Forest Service programs, the accidental spill scenario remains the only scenario that results in a hazard quotient above unity. At this application rate, the associated dose is about 14 mg/kg, which is still below the dose of 184 mg/kg associated with no apparent overt effects in humans.

**3.4.4.** Sensitive Subgroups. No reports were encountered in the literature leading to the identification of sensitive subgroups. There is no indication that glyphosate causes sensitization or allergic responses, which does not eliminate the possibility that some individuals might be sensitive to glyphosate as well as many other chemicals.

**3.4.5.** Connected Actions. There is very little information available on the interaction of glyphosate with other compounds. As summarized in Section 3.1.2, there are some data suggesting that glyphosate may inhibit hepatic mixed-function oxidases. This is a very important system of enzymes in the metabolism of many xenobiotics. While the inhibition of hepatic mixed-function oxidases is a plausible mechanism of interaction, this does not lead to any definite conclusions regarding the potential influence of glyphosate on the toxicity of other chemicals. In any event, this mechanism of action would probably be relevant only at very high doses, substantially above exposures that can be anticipated in Forest Service programs.

**3.4.6.** Cumulative Effects. As noted above, this risk assessment specifically considers the effect of repeated exposures and no adverse effects are anticipated. As discussed in the dose-response and dose-severity relationships (see section 3.3.3), the daily dose rather than the duration of exposure determines the toxicological response. Consequently, repeated exposure to levels below the toxic threshold should not be associated with cumulative effects.

It is possible and even likely that some individuals will be exposed to multiple sources of glyphosate as a result of Forest Service programs. For example, an individual consuming contaminated fish might also consume contaminated water and/or vegetation. For glyphosate, these multiple sources of exposure as inconsequential. As detailed in Table 3-8, the only substantial exposure scenario for acute exposures is the consumption of contaminated water after a spill into a small pond. All other plausible combinations of exposures would not exceed a level of concern. Similarly, for longer term exposures, the over-riding route of exposure is the consumption of contaminated vegetation. Adding all other sources of exposure would have no substantial impact on the hazard quotient.

Lastly, individuals may be exposed to glyphosate from other sources not related to Forest Service activities. For example, glyphosate has a number of approved uses on crops and some exposure to glyphosate in crop residues is likely. The U.S. EPA/OPP (2002) has estimated that residues of glyphosate on crops could account for about 1% of the RfD for the U.S. population overall (U.S. EPA/OPP 2002, p. 17726). This amounts to a hazard quotient of 0.01, substantially higher than any of the central estimates of the hazard quotients for longer term exposure scenarios associated with Forest Service programs (Table 3-8). Thus, except for the upper limit of exposure in the consumption of contaminated vegetation, the use of glyphosate by the Forest Service is not likely to substantially increase an individuals potential exposure to glyphosate from other sources. For the consumption of contaminated vegetation, the upper limit of exposure combined with the U.S. EPA estimate of other dietary exposures would not exceed the RfD – i.e., 0.14 + 0.01 = 0.15.

a i	Do	Exposure		
Scenario	Central	Lower	Upper	Assessment Worksheet
	General Exposure	s (dose in mg/kg/day)		
Directed ground spray (Backpack)	2.63e-02	9.00e-04	1.60e-01	C01a
Broadcast ground spray (Boom spray)	4.48e-02	1.32e-03	3.02e-01	C01b
Aerial applications	2.94e-02	4.80e-04	1.60e-01	C01c
	Accidental/Incidental Exp	oosures (dose in mg/kg/e	vent)	
Immersion of Hands, 1 minute	7.20e-06	7.10e-07	6.05e-05	C02a
Contaminated Gloves, 1 hour	4.32e-04	4.26e-05	3.63e-03	C02b
Spill on hands, 1 hour	9.44e-04	1.20e-04	4.61e-03	C03a
Spill on lower legs, 1 hour	2.33e-03	2.95e-04	1.14e-02	C03b

# Table 3-1: Summary of Worker Exposure Scenarios

Table 3-2: Occupational Exposure Rates used in risk assessments <sup>a</sup>						
Western Course	Rate (mg/kg bw per lb applied)					
worker Group	Central	Lower	Upper			
Directed foliar	0.003	0.0003	0.01			
Broadcast foliar	0.0002	0.00001	0.0009			
Aerial	0.00003	0.000001	0.0001			
<sup>a</sup> Taken from SERA (2	2001a).					

	Target		Dose (mg/kg/da	y)	Worksheet
Scenario		Central	Lower	Upper	
Acute/Accidental Exposures					
Direct spray, entire body	Child	3.57e-02	4.53e-03	1.74e-01	D01a
Direct spray, lower legs	Woman	3.58e-03	4.55e-04	1.75e-02	D01b
Dermal, contaminated vegetation	Woman	2.19e-03	6.98e-04	5.31e-03	D02
Contaminated fruit	Woman	8.06e-03	8.06e-03	1.47e-01	D03
Contaminated water, spill	Child	1.37e+00	3.33e-01	4.10e+00	D05
Contaminated water, stream	Child	3.01e-03	9.17e-05	9.02e-02	D06
Consumption of fish, general public	Man	1.56e-02	6.23e-03	3.12e-02	D08a
Consumption of fish, subsistence populations	Man	7.59e-02	3.04e-02	1.52e-01	D08b
Chronic/Longer Term Expos	ures				
Contaminated fruit	Woman	4.41e-03	4.41e-03	8.04e-02	D04
Consumption of water	Man	5.71e-05	4.00e-06	5.49e-04	D07
Consumption of fish, general public	Man	1.09e-07	1.09e-08	8.69e-07	D09a
Consumption of fish, subsistence populations	Man	8.79e-07	8.79e-08	7.04e-06	D09b

## Table 3-3: Summary of Exposure Scenarios for the General Public

Parameter	Clay	Loam	Sand	Comment/ Reference
Halftimes (days)				
Aquatic Sediment	203	203	203	Dix 1998
Foliar	10	10	10	Note 1
Soil	30	30	30	Note 2
Water	1000	1000	1000	Note 3
Ko/c	2000	2000	2000	Note 4
K <sub>d</sub>	2600	2100	500	Note 5
Water Solubility, mg/L	12000	12000	12000	Note 6
Foliar wash-off fraction	0.5	0.5	0.5	Leung 1994

**Table 3-4:** Glyphosate specific pesticide parameters used in GLEAMS modeling and estimation of concentrations in ambient water

Note 1 Central value from Feng and Thompson 1990 and Newton et al. 1984.

Note 2 Typical value from Table 2-2

Note 3 Gyphosate is stable in water at neutral pH. Ignore microbial degradation for modeling.

Note 4 Highly variable. Used geometric mean from Gerritse et al. 1996

Note 5 Use ranges from USDA/ARS 1995. Central value for loam, lower value for sand, and upper value for clay.

Note 6 Value for acid given by Tomlin (1994) and USDA/ARS (1995)

	Concentrations in Ambient Water (µg/L per lb/acre)						
Annual Rainfall	Cla	Clay		Loam		Sand	
	Average	Maximum	Average	Maximum	Average	Maximum	
5	0.00000	0.00000	0.00000	0.00000	0.00000	0.00000	
10	0.00012	0.00030	0.09440	0.17565	1.10442	2.05992	
15	0.13807	0.25089	0.34458	0.63596	3.18058	5.97802	
20	0.29624	0.54061	0.73480	1.35134	5.60349	10.69358	
25	0.47944	0.74475	1.14161	2.02281	8.10309	15.85858	
50	0.75344	1.65665	1.39087	3.09125	4.34293	15.06299	
100	0.97995	3.20707	1.53840	5.39600	3.30534	20.35153	
150	1.03871	4.27997	1.49796	6.84364	2.73199	22.74196	
200	1.03666	5.06852	1.41487	7.92040	2.33566	25.83345	
250	1.01076	5.78976	1.32712	8.79284	2.04561	27.84026	

**Table 3-5:** Estimated concentrations of glyphosate in a small (0.25 acre, 1 meter deep) pond adjacent to a 10 acre plot based on GLEAMS modeling with different soil types and annual rainfall rates and using a normalized application rate of 1 lb/acre.

**Table 3-6:** Estimated concentrations of glyphosate in a small stream (4,420  $m^3$ /day) adjacent to a 10 acre plot based on GLEAMS modeling with different soil types and annual rainfall rates and using a normalized application rate of 1 lb/acre.

Annual Rainfall	Clay		Lo	am	Sand	
	Average	Maximum	Average	Maximum	Average	Maximum
5	0.00000	0.00000	0.00000	0.00000	0.00000	0.00000
10	0.00001	0.00492	0.00703	0.62660	0.02064	2.13822
15	0.01267	1.12481	0.02556	2.33913	0.05947	6.64604
20	0.02713	2.44600	0.05435	5.10484	0.10512	12.47003
25	0.05311	4.88123	0.08585	8.26234	0.15311	19.10679
50	0.18321	18.06811	0.26291	28.05858	0.37979	56.63863
100	0.45951	52.78010	0.59880	77.47465	0.73684	140.44210
150	0.69588	92.44662	0.86220	132.26409	0.98828	227.07054
200	0.88933	135.15046	1.06546	190.09709	1.17292	314.68937
250	1.04689	180.02965	1.22476	250.14160	1.31401	402.83056

Concentrations in Ambient Water (µg/L per lb/acre)

Acute RfD	2	mg/kg/day	Sect. 3.3.3.	
Chronic RfD	2	mg/kg/day	Sect. 3.3.3.	
	Hazard	Exposure		
Scenario	Central	Lower	Upper	Assessment Worksheet
	General Exposu	res [using Chronic RfI	<b>)</b> ]	
Directed ground spray (Backpack)	1e-02	5e-04	8e-02	C01a
Broadcast ground spray (Boom spray)	2e-02	7e-04	2e-01	C01b
Aerial applications	1e-02	2e-04	8e-02	C01c
	Accidental/Incidental	Exposures [using Acu	te RfD]	
	Accidental/Incidental Hazard	Exposures [using Acu Quotient Based on Ac	te RfD] rute RfD	Exposure
Scenario	Accidental/Incidental Hazard Central	Exposures [using Acu Quotient Based on Ac Lower	te RfD] pute RfD Upper	Exposure Assessment Worksheet
Scenario Immersion of Hands, 1 minute	Accidental/Incidental Hazard Central 4e-06	Exposures [using Acu Quotient Based on Ac Lower 4e-07	te RfD] pute RfD Upper 3e-05	Exposure Assessment Worksheet C02a
Scenario Immersion of Hands, 1 minute Contaminated Gloves, 1 hour	Accidental/Incidental Hazard Central 4e-06 2e-04	Exposures [using Acu Quotient Based on Ac Lower 4e-07 2e-05	te RfD] pute RfD Upper 3e-05 2e-03	Exposure Assessment Worksheet C02a C02b
Scenario Immersion of Hands, 1 minute Contaminated Gloves, 1 hour Spill on hands, 1 hour	Accidental/Incidental Hazard Central 4e-06 2e-04 5e-04	Exposures [using Acu Quotient Based on Ac Lower 4e-07 2e-05 6e-05	te RfD] Pute RfD Upper 3e-05 2e-03 2e-03	Exposure Assessment Worksheet C02a C02b C03a

## **Table 3-7:** Summary of risk characterization (HQ's<sup>1</sup>) for workers.

<sup>1</sup> Hazard quotient is the level of exposure divided by the RfD then rounded to one significant decimal place or digit. See Table 3-1 for summary of exposure assessment.

Chronic RfD		2	mg/kg/day	Sect. 3.3.3.	
Acute RfD		2	mg/kg/day	Sect. 3.3.3.	
	Target		Hazard Quotient		Worksheet
Scenario		Central	Lower	Upper	
Acute/Accidental Exposures					
Direct spray, entire body	Child	2e-02	2e-03	9e-02	D01a
Direct spray, lower legs	Woman	2e-03	2e-04	9e-03	D01b
Dermal, contaminated vegetation	Woman	1e-03	3e-04	3e-03	D02
Contaminated fruit	Woman	4e-03	4e-03	7e-02	D03
Contaminated water, spill	Child	7e-01	2e-01	2	D05
Contaminated water, stream	Child	2e-03	5e-05	5e-02	D06
Consumption of fish, general public	Man	8e-03	3e-03	2e-02	D08a
Consumption of fish, subsistence populations	Man	4e-02	2e-02	8e-02	D08b
Chronic/Longer Term Expo	sures				
Contaminated fruit	Woman	2e-03	2e-03	4e-02	D04
Consumption of water	Man	3e-05	2e-06	3e-04	D07
Consumption of fish, general public	Man	5e-08	5e-09	4e-07	D09a
Consumption of fish, subsistence populations	Man	4e-07	4e-08	4e-06	D09b

<sup>1</sup> Hazard quotient is the level of exposure divided by the RfD. See Table 3-3 for a summary of the exposure assessments.



**Figure 3-1:** Residues of glyphosate on cowberries and bilberries after the application of glyphosate at 0.67 lb/acre (data from Siltanen et al. 1981).



Figure 3-2: Dose-Severity Relationships for glyphosate [see text for details]

### 4. ECOLOGICAL RISK ASSESSMENT

#### 4.1. HAZARD IDENTIFICATION

**4.1.1. Overview.** There are several standard toxicity studies in experimental mammals that were conducted as part of the registration process and there is a large body of published information on the toxicity of glyphosate to mammals. Just as these studies are used in the human health risk assessment to identify the potential toxic hazards associated with exposures to glyphosate, they can also be used to identify potential toxic effects in wildlife mammalian species. Loss of body weight is the most commonly seen effect of glyphosate in mammals. Inhibition of oxidative phosphorylation has been implicated as a possible mechanism by which glyphosate causes weight loss in experimental mammals; however, there is not adequate information about terrestrial wildlife from which to make a further assessment about the importance of this mechanism. As in the human health risk assessment, the potential significance of non-specific toxic effects can be assessed from the available toxicity studies in mammals. Because toxicity data in mammals are available in few species of experimental mammals, the use of these data to assess the potential hazards to large number of diverse mammalian wildlife species is an uncertain process. Nonetheless, there do not appear to be any systematic differences among mammalian species, including humans, when comparable toxicity values are expressed in units of mg/kg/day. While the available data are limited, this apparent consistency among species diminishes concern with the use of data based on a limited subset of species to characterize risk for terrestrial mammals in general.

In assessing potential effects in birds, the most relevant data for this risk assessment are the standard dietary and bird reproduction studies required for registration as well as the acute oral  $LD_{50}$  studies. The available toxicity studies do not suggest any specific or unique toxicity in birds compared to mammals. As in mammals, there is suggestive evidence that glyphosate may inhibit oxidative phosphorylation and consequently reduce food conversion efficiency. Also consistent with the data in experimental mammals is the apparent lack of teratogenic activity in birds.

The honey bee is the standard test organism for assessing the potential effects of pesticides on terrestrial invertebrates and there is a standard set of studies available on this species. In addition, studies are available on a relatively wide range of other terrestrial invertebrates including earthworms, isopods, snails, spiders, butterflies, and other terrestrial arthropods.

Glyphosate is readily metabolized by soil bacteria and many species of soil microorganisms can use glyphosate as sole carbon source. Nonetheless, microorganisms, like higher plants, do have the shikimate pathway for the production of aromatic amino acids. At the molecular level, glyphosate occupies the binding site of phosphoenol pyruvate, the second substrate of 5enolpyruvylshikimate 3-phosphate synthase, mimicking an intermediate state of the ternary enzyme-substrate complex. This inhibits the shikimic acid pathway in plants, effectively blocking the synthesis of certain phenolic compounds and the synthesis of aromatic amino acids. This, in turn, leads to a variety of toxic effects in plants, including the inhibition of photosynthesis in plants as well as inhibition of respiration and nucleic acid synthesis in plants and microorganisms. Since glyphosate inhibits this pathway, toxicity to microorganisms may be expected and glyphosate has been considered as an antimicrobial agent for human pathogens. Nonetheless, there is very little information suggesting that glyphosate will be harmful to soil microorganisms under field conditions and a substantial body of information indicating that glyphosate is likely to enhance or have no effect on soil microorganisms. Most field studies involving microbial activity in soil after glyphosate exposures note an increase in soil microorganisms or microbial activity and the application of glyphosate may cause transient increases in soil fungi that may be detrimental to some plants. While the mechanism of this apparent enhancement is unclear, it is plausible that glyphosate was used as a carbon source by the fungi and/or treatment with glyphosate resulted in increased nutrients for fungi in the soil. There is no indication that the transient enhancement in populations of soil fungi or bacteria will result in any substantial or lasting damage to soil ecology.

In higher plants, inhibition of the shikimic acid pathway leads to an inhibition or cessation of growth, cellular disruption, and, at sufficiently high levels of exposure, plant death. The time course for these effects can be relatively slow, depending on the plant species, growth rate, climate, and application rate. Gross signs of toxicity include wilting and yellowing of the vegetation, followed by browning, breakdown of plant tissue, and, ultimately, root decomposition. Standard toxicity studies are available on seedling emergence and vegetative vigor in a number of different plant species. The drift studies are also highly relevant to the assessment of risk in that unintended drift is one of the more plausible exposure scenarios for nontarget terrestrial plant species. The lowest reported effect level in drift studies is 1/33 of an application rate of 1.121 kg/ha that was associated with transient damage in soybeans. This treatment corresponds to 0.034 kg/ha [1.121 kg/ha ÷ 33] or about 0.03 lb/acre. At much higher concentrations – in the range of 0.7 lbs/acre – there is a plausible basis for concern that exposure to substantial glyphosate drift may have long term impacts on bryophyte and lichen communities.

In addition to the laboratory bioassays or field observations on single species, there are a number field studies that have assessed the effects of glyphosate on groups of terrestrial organisms, both animals and plants. These studies indicate that effects on terrestrial animals are likely to be secondary to effects on vegetation when glyphosate is applied at application rates comparable to or greater than those contemplated by the Forest Service. In some cases, the effects noted in filed studies appeared to be beneficial to some species under study. In most cases, the effects noted were changes in population density that reflected changes in food availability or suitable habitat.

In aquatic species, the acute lethal potency of glyphosate and glyphosate formulations has been relatively well-defined. These values are typically expressed as time-specific  $LC_x$  values where x is the estimate of the proportion of fish that die – e.g., 96 hour  $LC_{50}$ . A large number of acute  $LC_{50}$  values have been determined in various species of fish. As in the human health risk assessment, the formulation of glyphosate with surfactants, especially the POEA surfactant commonly used in glyphosate formulations, has a pronounced effect on the acute lethal potency of glyphosate.

The U.S. EPA typically uses  $LC_{50}$  values or fractions of  $LC_{50}$  values as the basis for characterizing risk, as in the U.S. EPA RED on glyphosate. A common concern with this approach is that more subtle non-lethal effects that may impact of the stability of fish populations in the field may not be properly assessed. The available information on the sub-lethal effects associated with glyphosate is summarized in this risk assessment and NOEC (no observable effect concentration) values form the basis of the risk characterization.

Lastly, field studies are available on the effects of glyphosate applications on fish populations. As with the risk characterization for terrestrial species, these studies have limitations in terms of their quantitative use in a risk assessment but are nonetheless highly relevant to the risk assessment and may be used to further assess the quality of the risk characterization based on laboratory bioassay.

The toxicity of glyphosate to aquatic plants has been evaluated by U.S. EPA based on studies submitted for the registration of glyphosate. In addition, several studies are available from the open literature as well as more recent studies submitted to U.S. EPA. These studies are available for both algae and aquatic macrophytes. As would be expected from a herbicide, glyphosate is much more toxic to aquatic plants than animals.

### 4.1.2. Toxicity to Terrestrial Animals.

**4.1.2.1.** *Mammals* – As summarized in the human health risk assessment (Section 3.1), there are several standard toxicity studies in experimental mammals that were conducted as part of the registration process and there is a large body of published information on the toxicity of glyphosate to mammals. Just as these studies are used in the human health risk assessment to identify the potential toxic hazards associated with exposures to glyphosate, they can also be used to identify potential toxic effects in wildlife mammalian species.

Loss of body weight is the most commonly seen effect of glyphosate in mammals. Inhibition of oxidative phosphorylation has been implicated as a possible mechanism by which glyphosate causes weight loss in experimental mammals (see section 3.1.2); however, there is not adequate information about terrestrial wildlife from which to make a further assessment about the importance of this mechanism. As in the human health risk assessment, the potential significance of non-specific toxic effects can be assessed from the available toxicity studies (Appendices 3 through 6).

The WHO (1994) criteria document summarizes a study in heifers that is cited on several material safety data sheets (MSDS's in C&P Press, 2002) for glyphosate formulations. According to WHO (1994), this study involved dosing of Brahman-cross heifers with Roundup at 400, 500, 630 or 790 mg/kg body weight per day by nasogastric intubation. At 790 mg/kg, some animals died with labored breathing and pneumonia from the aspiration of rumen contents. This is consistent with lung damage seen in experimental mammals (Section 3.1.4). Additional signs of toxicity at 500, 630 and 790 mg/kg body weight included diarrhoea and decreased food intake. No adverse effects were observed at 400 mg Roundup/kg bw (equivalent to 215 mg a.i./kg bw or about 160 mg a.e./kg bw). Although this report is attributed to Monsanto Inc, this study was not

found in a search of the U.S. EPA studies submitted for registration. The MSDS for Glyphosate Original (C&P Press, 2002) is consistent with this summary but the MSDS's for for Aqua Neat, Debit TMF, Eagre, Forester's Non-Selective Herbicide, and Glyphosate contain a different summary:

This product was administered to Brahman-cross heifers by gavage at daily doses of 0, 540, 830, 1290, or 2000 mg/kg for 7 consecutive days. Clinical signs of toxicity, including loss of appetite, diarrhea, and death (1290 and 2000 mg/kg) were observed at 830 mg/kg or above. The no-effect level was considered to be 540 mg/kg/day.

The reason for differences between these summaries is unclear.

A reported incident of fatal poisoning of a horse by drift from glyphosate was investigated by the Texas Department of Agriculture (1992), which made the determination that glyphosate was not the cause of death and that the horse died of natural causes.

Because toxicity data in mammals are available in few species of experimental mammals, the use of these data to assess the potential hazards to large number of diverse mammalian wildlife species is an uncertain process. One approach to this process involves identifying patterns of toxicity in mammals of various sizes (i.e., allometric relationships as discussed in SERA 2001a, Section 3.2.). As detailed in Section 4.3.2 (Dose Response Assessment for Terrestrial Species), there do not appear to be any systematic differences among mammalian species, including humans, when comparable toxicity values are expressed in units of mg/kg/day. While the available data are limited, this apparent consistency among species diminishes concern with the use of data based on a limited subset of species to characterize risk for terrestrial mammals in general.

**4.1.2.2.** *Birds* – The most relevant data for this risk assessment are the standard dietary and bird reproduction studies required for registration as well as the acute oral  $LD_{50}$  studies. The toxicity of glyphosate on acute gavage administration to birds is >2000 mg/kg (Fink et al. 1978), comparable to that seen in experimental mammals. No effects on reproduction have been seen in bobwhite quail (Fink 1975) and mallard ducks (Fink and Beavers 1978) at concentration of up to 1000 ppm.

While the available toxicity studies do not suggest any specific or unique toxicity in birds, the study in zebra finches by Evan and Batty (1986) noted pronounced weight loss in animals after dietary exposures to 5000 ppm glyphosate for 3 to 7 days accompanied by a 20-30% decrease in food consumption but a 30-60% decrease in body weight. This is consistent with observations in experimental mammals suggesting that glyphosate may inhibit oxidative phosphorylation and consequently reduce food conversion efficiency.

Also consistent with the data in experimental mammals is the apparent lack of teratogenic activity in birds. There is no indication that glyphosate or Roundup causes birth defects in birds (Batt et

al. 1980, Hoffman and Albers 1984). The study by Hoffman and Albers (1984) is somewhat difficult to interpret because of the way in which doses are expressed—lb/acre at 100 gallons/acre. In this study, eggs were immersed in various concentrations of several pesticides, including glyphosate, for approximately 30 seconds and observed throughout development. The reported  $LC_{50}$  for glyphosate from Roundup is 178 lbs/acre at 100 gallons/acre. This probably corresponds to a concentration of 80.1 kg  $\div$  378.5 L

 $(178 \text{ lbs} \cdot 0.45 \text{ kg/lb}) \div (100 \text{ gallons} \cdot 3.785 \text{ L/gallon})$ 

or approximately 200 g/L, which corresponds to a solution of about 20% (w/v). This is consistent with the NOEL reported by Batt et al. (1980), which involved a less severe exposure-immersion in a 5% solution for 5 seconds. It should also be noted that the apparent application rate of 178 lbs/acre is substantially higher than the maximum annual labeled application rate of 10.6 lbs/acre.

**4.1.2.3.** *Invertebrates* – The honey bee is the standard test organism for assessing the potential effects of pesticides on terrestrial invertebrates and there is a standard set of studies available on this species (Palmer and Beavers 1997c; Palmer and Krueger, 2001a; Palmer and Krueger, 2001b). In addition, studies are available on a relatively wide range of other terrestrial invertebrates including earthworms, isopods, snails, spiders, butterflies, and other terrestrial arthropods.

In standard oral and contact bioassay summarized by U.S. EPA/OPP (1993c), the LD<sub>50</sub> values for bees was over 100 µg/bee. Three more recent studies have been submitted to the U.S. EPA that are consistent with these earlier reports. In an acute contact toxicity assay, no effects were seen at 100 µg/bee (Palmer and Beavers 1997c). Similar results have been reported recently for a newer formulation, MON 77360, in which the NOEC based on mortality in contact toxicity was also 100 µg (Palmer and Krueger 2001a). This dose is classified as a NOEC because mortality (3/60 animals) was not significantly different from mortality in the matched solvent control (0/60, p=0.12 using the Fisher exact test). Combining the matched solvent control (0/60) with the negative control (0/60) for a combined control response of 0/120, the mortality of 3/60 animals is statistically significant (p=0.0358 using the Fisher exact test) although low (3/60 = 5%). No mortality (0/60) was observed at the next lower dose (50 µg/bee) or at any of the other lower doses down to 6.25 µg/bee.

In an acute dietary study (Palmer and Krueger 2001b), the 48 h oral  $LD_{50}$  is reported as >100 µg/bee based on a 11.7% mortality (7/60) at the highest dose tested. The NOEC is reported as 50 µg/bee based on 5% mortality (3/60). Again, this response rate is not significant with respect to solvent matched controls (0/60) but is significant is solvent and negative controls are combined (0/120, p=0.0358 using the Fisher exact test). It should be notes that high mortality (26/60) was observed at 12.5 µg/bee but this was attributed to an unidentified failure in the test apparatus that resulted in apparently substantial direct contact with the test solution. While this sort of unexpected low dose response is noteworthy, the low mortality at higher doses (i.e., 1/60 at 25

 $\mu$ g/bee and 3/60 at 50  $\mu$ g/bee) supports the assessment of Palmer and Krueger (2001b) that the high mortality at 12.5  $\mu$ g/bee was an aberration.

Glyphosate has been tested as an insecticide for spider mites, *Tetranychus urticae*, a pest species on apple trees (Ahn et al. 1997) as well as for toxicity to *Typhlodromus pyri*, an important predator of spider mites (Weppleman 1998b). Direct foliar spray of glyphosate IPA at 0.593 to 4.74 mg ai per leaf (kidney bean plants) had no effect on the spider mite based on mortality in eggs, larva, nymphs or adults (Ahn et al. 1997) an was essentially ineffective as an insecticide.

Applications equivalent to 10 L/ha RoundUp ULTRA (glyphosate isopropylame salt at 360 g/L or an application rate of 3.6 kg a.i./ha) applied to glass slides caused 100% mortality in spider mites after 24 hours of contact and was classified as "harmful" (Weppleman 1998a). In a similar assay using *Aphidius rhopalosiphi* (a beneficial wasp that is a parasite of the cereal aphid), the same contact exposure also resulted in 100% mortality after 24 hours. The relevance of the studies by Weppleman (1998a and 1998b) to the assessment of potential effects under normal use is unclear. As noted in Weppleman (1998a),

the 5% v/v test solution of Roundup ULTRA produced a wet sticky layer on the treated glass plates that resulted in alterations of the moving behavior of the wasps to the point of sticking.

In other words, it appears the application of the glyphosate formulation to the glass slides cause the test organism to stick to the slides and this may have contributed to the observed mortality.

Haughton et al. (1999; 2001a; 2001b) have conducted a series of laboratory and field studies on the effects of glyphosate on the spider, *Lepthyphantes tenuis*. Direct spray laboratory bioassays at rates equivalent to 180, 360, 720, 1080, 1440, and 2160 g/ha resulted in low rates of mortality that were not dose related (Haughton et al. 2001a). In the field, application rates of 360, 720, and 1440 g ae/ha resulted in decreased spider populations that were attributed to secondary effects from changes in the vegetation (Haughton et al. 2001b). No substantial effects were observed in spider populations at application rates of 90 or 180 g a.e./ha (Haughton et al. 1999).

Data on other arthropods are less detailed but also indicate a low potential for a direct toxic effect from glyphosate. In a laboratory study in which isopods were exposed to leaf litter at levels equivalent to application rates of 2.1 kg/ha, the effect on litter degradation depended on the tree species. Direct toxic effects—evidenced by increased mortality—could not be ruled out but were not statistically significant (Eijsackers 1992). Samsoe-Petersen (1995) report no measurable effect on rove beetle (mortality and egg production) after spray of substrate with 1% Roundup (3.6 g/L) at 6  $\mu$ L/cm<sup>2</sup>. Bramble et al. (1997) conducted a series of studies on effects of using herbicides (including glyphosate) in rights-of-way maintenance compared to using mechanical maintenance and noted no significant or substantial differences in butterfly populations.

Three studies are available relating to the potential effects of glyphosate on earthworms. In a laboratory study, effects on earthworm cultures treated at levels equivalent to application rates of 0.7 to 2.8 g glyphosate/ha included decreased growth rates and early mortality (Springett and Gray 1992). The direct relevance of this study is limited, however, because the exposure conditions (spraying twice weekly on culture dishes) do not closely approximate field conditions. Dalby et al. (1995) report no effects on earthworms in applications designed to mimic agricultural use. This study, however, does not report exposures either as g/ha or ppm soil and thus cannot be used directly in this risk assessment. The soil  $LC_{50}$  for glyphosate to *Aporrectodea caliginosa*, a worm common in Libya, has been reported to be 246 to 177 mg glyphosate/kg soil dry weight over exposure periods of 8 to 37 days (Mohamed et al. 1995).

One study is available on the toxicity of glyphosate to a terrestrial snail, *Helix aspersa*, Brown garden snail, in which diets containing 4994 ppm glyphosate resulted in no mortality over a 14 day exposure period. Assuming a 30% food consumption factor for this species (APHIS 1993), this corresponds to a dose of about 1,500 mg/kg (4994 ppm  $\times$  0.3 mg/kg bw ppm = 1498.2 mg/kg bw).

**4.1.2.4. Soil Microorganisms**. As noted in Section 3.1.15.1, glyphosate is readily metabolized by soil bacteria with AMPA as a major metabolite. In addition, many species of soil microorganisms can use glyphosate as sole carbon source (Dick and Quinn 1995a; Dick and Quinn 1995b; Dotson et al. 1996; Wardle and Parkinson 1992a). Microorganisms, like higher plants, do have the shikimate pathway for the production of aromatic amino acids. Since glyphosate inhibits this pathway, toxicity to microorganisms may be expected (Cox 2002; Issa 1999). As noted in Section 3.1.2, glyphosate has been considered as an antimicrobial agent for human pathogens. Nonetheless, there is very little information suggesting that glyphosate will be harmful to soil microorganisms under field conditions and a substantial body of information indicating that glyphosate is likely to enhance or have no effect on soil microorganisms (Busse et al. 2001; Wardle and Parkinson 1990a,b; Wardle and Parkinson 1991).

As reviewed by Cox (2002), a number of studies have demonstrated effects on soil microorganisms under laboratory conditions that are consistent with the ability of glyphosate to inhibit the shikimate pathway. For example, Issa (1999) noted that the growth of soil algae and cyanobacteria may be inhibited by glyphosate in artificial culture media at concentrations of 5 and 20 mM – i.e., about 845 and 3,380 mg/L. Roundup was a more potent inhibitor than glyphosate IPA, which was in turn a more potent inhibitor than the free acid of glyphosate. The decreased growth was associated with shikimate accumulation and was antagonized by excess aromatic amino acids. Thus, this inhibition is consistent with the inhibition of the shikimate pathway.

Wan et al. (1998) noted an inhibition of extraradical mycelial growth in *Glomus intraradices* after 14 days exposure in a preparation with carrot roots in a culture medium containing 0.5 ppm glyphosate. This effect, however, was attributed to an effect of glyphosate on the carrot roots rather than a direct toxic effect on the fungi. Direct toxic effects on soil fungi in culture medial have been demonstrated by Chakravarty and Sidhu (1987) at concentrations of 10 ppm or greater.

At application rates of 0.54 kg/ha, a transient decrease in populations of soil fungi and bacteria was noted after 2 months but no effect was apparent after 6 months. Similarly, at an application rate of 3.23 kg/ha, no effect was seen on soil fungi and bacteria after 10-14 months (Chakravarty and Chatarpaul 1990). A transient decrease in soil microbial activity was also noted by Wardle and Parkinson (1992b) after the application of glyphosate at 5 kg/ha. Sannino and Gianfreda (2001) report that glyphosate inhibited soil phosphatase activity at 20 mM. This inhibition, however, was attributed to competitive inhibition of p-nitrophenylphosphate, the substrate used in the phosphatase assay, by glyphosate. Thus, the inhibition of phosphatase activity was an artifact of the assay method rather than an indication of glyphosate toxicity.

Soil concentrations of 100 ppm of glyphosate or AMPA had no significant effect on soil denitrification (Pell et al. 1998). Bromilow et al. (1996) noted no effects on soil fertility in repeated applications over 14 years – 1980 to 1993 – of glyphosate at 1.4 kg/ha based on assays for microbial biomass and crop productivity.

Several field studies involving microbial activity in soil after glyphosate exposures note an increase rather than decrease in soil microorganisms or microbial activity (Haney et al. 2002; Hart and Brookes 1996; Laatikainen and Heinonen-Tanski 2002; Nicholson and Hirsch 1998). As discussed by Kremer (2002), the application of glyphosate may cause transient increases in soil fungi that may be detrimental to some plants. For example, Descalzo et al. (1996a,b) has shown that inoculation of soil with various pathogenic soil fungi may result in an apparent enhancement of glyphosate toxicity.

**4.1.2.5.** Toxicity to Terrestrial Plants. Studies on the mechanism of action of glyphosate are numerous (Anthelme and Marigo 1998; Green et al. 1992; Hernandez et al. 1999; Hernandez et al. 2000; Hetherington et al. 1998; Jain et al. 2002; Pline et al. 2002; Uotila et al. 1995; Singh and Shaner 1998; Schonbrunn et al. 2001). At the molecular level, glyphosate occupies the binding site of phosphoenol pyruvate, the second substrate of 5-enolpyruvylshikimate 3-phosphate synthase, mimicking an intermediate state of the ternary enzyme-substrate complex. This inhibits the shikimic acid pathway in plants, effectively blocking the synthesis of certain phenolic compounds and the synthesis of aromatic amino acids. This, in turn, leads to a variety of toxic effects in plants, including the inhibition of photosynthesis, respiration, and nucleic acid synthesis.

At the level of the whole plant, inhibition of the shikimic acid pathway leads to an inhibition or cessation of growth, cellular disruption, and, at sufficiently high levels of exposure, plant death. The time course for these effects can be relatively slow, depending on the plant species, growth rate, climate, and application rate. Gross signs of toxicity, which may not be apparent for 2–4 days in annuals or for more than 7 days in perennials, include wilting and yellowing of the vegetation, followed by browning, breakdown of plant tissue, and, ultimately, root decomposition.

There are a large number of efficacy studies on glyphosate (e.g., Bariuan et al. 1999). For the most part, this risk assessment is not concerned with efficacy studies and these studies are not

covered. Nonetheless, some efficacy studies focus on understanding the pharmacology of glyphosate in plants and such studies are germane to assessing potential effects in nontarget vegetation. Glyphosate is absorbed primarily through the foliage, and the absorption is rapid. Approximately 33% of the applied glyphosate is absorbed within a few hours after application. Glyphosate absorption by plants may be enhanced by high humidity (Schonherr 2002). Because glyphosate is strongly adsorbed to soil, relatively little if any absorption occurs through the roots (Smith and Oehme 1992). The production of <sup>14</sup>C from plant associated material does not appear to be correlated with soil microbial biomass (Von Wiren-Lehr et al. 1997). In actively growing plants, translocation involves cell to cell transport through the cuticle followed by long distance transport via vascular tissue. In dormant plants, transport is much slower and may be negligible. Glyphosate is not extensively metabolized or detoxified in plants. In plants that share a common seedpiece or propaglue node, such as sugar cane, translocation from plant to plant can result in injury to plants that are not treated directly (Dal Piccolo et al. 1980). At least in sugar beets, the difference between tolerant and susceptible strains is in the rate of excretion of glyphosate (Geiger et al. 1999). The retention of glyphosate on foliage is affected by the use of adjuvants with a wash off of about 50% with adjuvants and 64% without adjuvants (Leung 1994). As with many herbicides, glyphosate may produce a hormetic response in some species, causing a stimulation of growth at low concentrations and an inhibition only at higher concentrations (Schabenberger et al. 1999). This type of response has also been noted in other studies (e.g., yields of smartweed and soybeans in Figure 2, p. 1195 of Fletcher et al. 1996).

As noted in the U.S. EPA/OPP (1993c) RED for glyphosate, data submitted up to 1993 did not include bioassays for vegetative vigor (i.e., direct application to foliage after plants have emerged) or sufficient studies on the effects of drift on non-target species. The only data on toxicity to terrestrial plants summarized in the RED involved seedling emergence assays using a 50% solution of glyphosate IPA. In this assay (Bohn 1987), glyphosate applications of up to 10 lb a.i./acre resulted in <25 % effect on the monocots and dicots tested.

Since the publication of the RED, additional studies have been submitted to the U.S. EPA or published in the open literature on seedling emergence (Willard 1996; Everett et al. 1996a; Suma et al. 1995) and vegetative vigor (Chetram and Lucash 1994; Everett et al. 1996b). In addition, several studies have been conducted on the effects of drift on nontarget plant species (Al-Khatib and Peterson 1999; Bhatti et al. 1997; De Jong and de Haes 2001; Fletcher et al. 1996; Gilreath et al. 2001; Marrs and Frost 1997; Newmaster et al. 1999).

The studies on seedling emergence submitted to the U.S. EPA involve a water dispersible granule formulation (Willard 1996) and a wettable powder formulation (Everett et al. 1996a). Additional details on the formulations cannot be disclosed in this risk assessment. Both studies were consistent with earlier reports indicating no adverse effects in monocots and dicots at application rates of about 4.5 lb a.e./acre. In an open literature publication by Shuma e tal. (1995), a Roundup formulation (356 g a.i./L) was associated with complete inhibition of seed viability when applied to oats 15 days after anthesis (flowering or seed formation) at an application rate of 1.76 kg/ha (about 1.6 a.i. lb/acre) and applications as low as 0.44 kg/ha partially inhibited seed
germination. This study is not inconsistent with the studies by Everett et al. (1996a) or Willard (1996). These latter studies involved standard test protocols in which seeds were exposed directly rather than harvested from plants after anthesis.

The two vegetative vigor studies (Chetram and Lucash 1994; Everett et al. 1996b) both follow similar experimental designs. Various monocots and dicots are grown from seeds to emergence prior to any herbicide application. After the leaves emerge, the plants are sprayed at various rates – expressed in the same units as application rate – to assess the consequences of direct spray and drift. In the study by Everett et al. (1996b) glyphosate was applied as a wettable powder formulation. The most sensitive species was the dicot, oilseed rape (*Brassica napus*), with an EC<sub>25</sub> of 0.083 kg a.e./ha (0.07 lb a.e./acre) and EC<sub>5</sub> of 0.042 kg a.e./ha (0.037 lb a.e./acre). The least sensitive species was the monocot, Purple Nutsedge (*Cyperus rotundus*), with an EC<sub>25</sub> of 0.891 kg a.e./ha (0.79 lb a.e./acre) and EC<sub>5</sub> of 0.58 kg a.e./ha (0.52 lb a.e./acre). The EC<sub>5</sub> values are essentially NOECs. In the study by Chetram and Lucash (1992), the most sensitive species based on reduced growth were tomato and radish, with a NOEC of 0.035 lb a.e./acre and the least sensitive species were ryegrass, corn, and onions with a NOEC of 0.56 lb a.e./acre.

The drift studies are also highly relevant to the assessment of risk in that unintended drift is one of the more plausible exposure scenarios for nontarget terrestrial plant species (Section 4.2). The lowest reported effect level in drift studies is the report by Al-Khatib and Peterson (1999) that 1/33 of an application rate of 1.121 kg/ha was associated with transient damage in soybeans, based on an assessment of visual injury, over a 30 day period after application but no net decrease in soybean production by the end of the season. This treatment corresponds to 0.034 kg/ha [1.121 kg/ha  $\div$  33] or about 0.03 lb/acre. A study by the same authors found that grapes were much less sensitive, evidencing damage at exposures equivalent to one-third of the application rate. A grass (*Poa annua*) and a dicot (*Brassica napus*) both exhibited substantial damage at deposition rates of over 1000 µg/m<sup>2</sup> or about 1.8 lbs/acre. Fletcher et al. (1996) found that simulated drift in the range of 0.4% to 0.8% of an application rate of 0.43 kg/ha had no marked effect on canola, smartweed, soybean or sunflower plants.

The study by Newmaster et al. (1999) suggests that some bryophytes and fungi may be sensitive to long term effects after glyphosate exposure. The  $EC_{50}$  for a decrease in relative abundance two years after application is about 0.8 kg/ha or 0.7 lbs/acre (Newmaster et al. 1999, Figure 3, p. 1105). In addition, changes in relative abundance were apparent at six weeks after application (Newmaster et al. 1999, Figure 7, p. 1108). The statistical analyses presented by Newmaster et al. (1999) involves the use of a non-threshold polynomial model. While this may be a reasonable method for quantifying effects among the two herbicides studied (glyphosate and triclopyr), this may be less appropriate for risk assessment, as discussed further in Section 4.3 (dose-response assessment). Nonetheless, this study does appear to present a plausible basis for concern that exposure to substantial glyphosate drift may have long term impacts on bryophyte and lichen communities.

**4.1.2.6.** *Field Studies* – In addition to the laboratory bioassays or field observations on single species, there are a number field studies that have assessed the effects of glyphosate on groups of terrestrial organisms, both animal and plant (Appendix 8).

These studies indicate that at application rates comparable to or greater than those contemplated by the Forest Service effects on terrestrial animals are likely to be secondary to effects on vegetation. This has been demonstrated for moose (Santillo 1994), small mammals (Anthony and Morrison 1985; D'Anieri et al. 1987; Ritchie et al. 1987; Santillo et al. 1989a; Sullivan 1990; Sullivan et al. 1997; Sullivan et al. 1998a), rabbits (Hjeljord et al. 1988), birds (Cayford 1988; Easton and Martin 1998; Freemark and Boutin 1995; Linz et al. 1994 and 1996 ; Linz and Blixt 1997; MacKinnon and Freedman 1993; Solberg and Higgins 1993), carabid beetles (Brust 1990; Duchesne et al. 1999), and various other invertebrates (Byers and Bierlein 1984; Freemark and Boutin 1995; Moldenke 1992; Santillo et al. 1984; Yokoyama and Pritchard 1984).

In some cases, the effects noted in these studies appeared to be beneficial to the species under study [e.g., increased use by water fowl associated with an increase in open water after treatment with Rodeo (Solberg and Higgins 1993)]. In most cases, the effects noted were changes in population density that reflected changes in food availability or suitable habitat.

One report does indicate that use of glyphosate in conifer release will cause an initial decrease in leaves on deciduous trees that serve as a food source for white-tailed deer. This effect, however, is transient and followed by an increase in forb abundance (Vreeland et al. 1998). While glyphosate will certainly impact some plant species, no effects on plant diversity were noted in a study by Miller et al. (1999). Glyphosate residues or perhaps residues of adjuvants used with glyphosate have been shown to affect grazing preference in cattle (Jones and Forbes 1984) but not sheep (Kisseberth et al. 1986).

A concern with the direct use of field studies in a risk assessment is that field studies, like many epidemiology studies, may be difficult to interpret because of the nature of the "control group" and because some studies may not be sufficiently sensitive to detect subtle adverse effects. These concerns are reasonable. Nonetheless, it is worth noting that Sullivan et al. (1998b) was able to detect a decline in vole populations with an concomitant increase in chipmunk and deer mice populations due to a shift in the nature of the available food. While this is not to suggest that all of the field studies summarized in Appendix 8 are or would be highly sensitive to subtle changes and effects, it does support the qualitative use of these studies – along with the more quantitative use of laboratory toxicity studies – in the risk characterization (Section 4.4).

### 4.1.3. Aquatic Organisms

### 4.1.3.1. Toxicity to Fish

**4.1.3.1.1. Overview** – As with terrestrial species, the acute lethal potency of glyphosate and glyphosate formulations has been relatively well-defined. These values are typically expressed as time-specific  $LC_x$  values where *x* is the estimate of the proportion of fish that die – e.g., 96 hour  $LC_{50}$ . A large number of acute  $LC_{50}$  values have been determined in various species of fish.

These are summarized in Appendix 9 and discussed further in Section 4.1.3.1.2. As in the human health risk assessment (Section 3.1.14), the formulation of glyphosate with surfactants, especially the POEA surfactant commonly used in glyphosate formulations, has a pronounced effect on the acute lethal potency of glyphosate. These data are detailed in Section 4.1.3.1.3.

The U.S. EPA typically uses  $LC_{50}$  values or fractions of  $LC_{50}$  values as the basis for characterizing risk, as in the U.S. EPA/OPP (1993c) RED on glyphosate. A common concern with this approach is that more subtle non-lethal effects that may impact of the stability of fish populations in the field may not be properly assessed. The available information on the sub-lethal effects associated with glyphosate is summarized in Section 4.1.3.1.4 and, as discussed further in Section 4.3. (dose-response assessment), NOEC (no observable effect concentration) values form the basis of the risk characterization in this risk assessment.

Lastly, field studies are available on the effects of glyphosate applications on fish populations. As with the risk characterization for terrestrial species (Section 4.1.2.6), these studies have limitations in terms of their quantitative use in a risk assessment but are nonetheless highly relevant to the risk assessment and may be used to further assess the quality of the risk characterization based on laboratory bioassay. Thus, these studies are summarized in Section 4.1.3.1.5.

**4.1.3.1.2.** Standard Toxicity Studies – Glyphosate and glyphosate formulations have been tested in a large number of fish species and this information is summarized in Appendix 9. As summarized by U.S. EPA/OPP (1993c, p. 40), the 96-hour  $LC_{50}$  values in freshwater fish range from 86 (70-106) mg/L in rainbow trout for a 83% pure sample of technical grade glyphosate to 140 mg/L (120-170) mg/L in rainbow trout for a 97.6% pure sample of technical grade glyphosate (U.S. EPA/OPP 1993c, p. 40). Based on these bioassays, U.S. EPA/OPP (1993c) classified technical grade glyphosate as non-toxic to practically non-toxic in freshwater fish. Some bioassays of technical grade glyphosate resulted in much lower  $LC_{50}$  values – i.e., 10 mg/L for trout in soft-water (Wan et al. 1989) and about 3 mg/L for unfed flagfish (Holdway and Dixon 1988).

Some formulations of glyphosate can be much more toxic to fish than technical grade glyphosate. As also summarized by U.S. EPA/OPP (1993c) and detailed in Appendix 9, the 96-hour  $LC_{50}$  values for formulated glyphosate in freshwater fish range from 1.3 (1.1-16) mg/L in rainbow trout for a 41% glyphosate formulation to >1000 mg/L mg/L in rainbow trout and bluegill sunfish for a 62.4% glyphosate formulation (U.S. EPA/OPP 1993c, pp. 42-43). Based on these  $LC_{50}$  values, U.S. EPA/OPP (1993c) classified glyphosate formulations as moderately toxic to practically nontoxic in freshwater fish. The less toxic surfactants appear are identified only as "AA surfactant", "W" surfactant, and "x-77" surfactant. The "x-77" surfactant appears to be X-77<sup>®</sup>, a non-ionic alkylphenol ethoxylate-based surfactant supplied by Loveland Industries. As detailed in Appendix 9, this surfactant modestly increases the toxicity of Rodeo - e.g., decreases the  $LC_{50}$  value by about 30% (Mitchell and Chapman 1985a). The identity of the other surfactants in these formulations is not specified in U.S. EPA/OPP (1993c). Abdelghani et al. (1997) have noted that

Syndets surfactant, a surfactant used with glyphosate and other herbicides, is much more toxic to fish (as well as crawfish) than Roundup (Appendices 11 and 12).

Because of the high toxicity of some glyphosate formulations and some surfactants, the U.S. EPA/OPP (1993c) required testing of one surfactant used in glyphosate, MON0818. This surfactant was classified as highly toxic to slightly toxic, with  $LC_{50}$  values in the range of 0.65 (0.54-2.7) mg/L to 2.0 (1.5-2.7) mg/L in rainbow trout (U.S. EPA/OPP 1993c, p. 43). While the U.S. EPA/OPP (1993c) does not specifically identify MON0818, the study cited for the toxicity of this compound references Folmar et al. (1979). As discussed further in Section 4.1.3.1.3, Folmar et al. (1979) tested the POEA surfactant used in Roundup.

Based on the low acute toxicity of formulated glyphosate to marine organisms (as summarized in Section 4.1.3.3), the U.S. EPA/OPP (1993c, p. 49) waived the requirement for testing of marine or estuarine fish species.

In terms of the dose-response assessment for longer term exposures to glyphosate, the most relevant study remains the life cycle toxicity studies done in fathead minnow. As summarized in the U.S. EPA/OPP (1993c), no effect on mortality or reproduction was observed at a concentration of 25.7 mg/L using 87.3% pure technical grade glyphosate. No other chronic toxicity studies have been encountered in fish either in the published literature or the more recent studies submitted to U.S. EPA.

4.1.3.1.3. Effect of Surfactants – Most of the surfactant studies summarized in U.S. EPA (1993b) were conducted in the 1970's or 1980's and the formulations were identified in these studies by codes assigned by the registrant. Thus, while it is possible to specifically identify some formulation bioassays with some current formulations, the correspondence of other formulation bioassays to currently available formulations is unclear. Even with the more recent studies, the name of the formulation is not always clearly specified. For example, Roundup ULTRA (NOS) has a 96-hour LC<sub>50</sub> value of 7.9 (6.2 - 10) mg/L with an NOEC of 6.2 mg/L in rainbow trout (Drottar and Swigert 1998b) and a 96-hour LC<sub>50</sub> value of 7.7 (6.5 - 11) mg/L with an NOEC of 3.9 mg/L in bluegill sunfish (Drottar and Swigert 1998c). These studies, however, do not specifically identify the formulation as Roundup ULTRAMAX but do indicate that the formulation was a liquid. Because only two Roundup ULTRA formulations have been identified, Roundup ULTRAMAX as a liquid formulation and Roundup ULTRADRY as a granular formulation, it would appear that the studies by Drottar and Swigert involved ULTRAMAX. However, the composition of the Roundup ULTRA used in the Drottar and Swigert studies is different from the composition of Roundup ULTRAMAX given on the product label (Appendix 1). This difficulty is clearly associating formulation data with specific studies makes formulation specific assessments difficult.

In an attempt to clarify this issue, Appendix 3c summarizes the available ecological information from all of the MSDS's for the formulations that are labeled for forestry applications. It is apparent that these formulations fall into relatively clear groups. The most toxic formulations

appear to be Credit Systemic, Credit, Glyfos, Glyphosate, Glyphosate Original, Prosecutor Plus Tracker, Razor SPI, Razor, Roundup Original, Roundup Pro Concentrate, and Roundup UltraMax. It may be presumed that these formulations contain the most toxic surfactants. Other formulations such as Aqua Neat, Aquamaster, Debit TMF, Eagre, Foresters' Non-Selective Herbicide, Glyphosate VMF, and Roundup Custom are much less acutely toxic. Some of these, however, require or recommend the use of a surfactant and this would likely increase the toxicity of the formulation.

For this risk assessment, the uncertainties involving the presence or absence of a surfactant and the possibly differing effects of using various surfactants cannot be resolved with certainty. As detailed in the dose-response assessment, the toxicity of glyphosate is characterized based on the use of a surfactant, either in the formulation or added as an adjuvant in a tank mixture.

As noted in the previous subsection, the POEA surfactant used in some glyphosate formulations is substantially more toxic to aquatic species than glyphosate and substantially more toxic than other surfactants that may be used with glyphosate. Two aquatic toxicity studies (Folmar et al. 1979, Wan et al. 1989) have been conducted on glyphosate, the POEA surfactant, and a Roundup formulation which permit a quantitative assessment of the relative toxicities of glyphosate and POEA as well as an assessment of potential for toxicologic interactions (i.e., synergism or antagonism) in combined exposures to these agents. Both of these studies indicate that POEA is substantially more toxic than glyphosate and that POEA surfactant is the primary toxic agent of concern.

The study by Folmar et al. (1979) is summarized in Table 4-1. As indicated in the first column of this table, these investigators conducted bioassays on four species of fish and one invertebrate (midge larvae). The following three columns give the  $LC_{50}$  values for glyphosate, POEA, and Roundup, respectively. For fish, the 96-hour  $LC_{50}$  values are given in the table. Folmar et al. (1979) report  $LC_{50}$  values for 24 and 48 hours but these values are not substantially different from those at 96 hours. The fifth column calculates the relative potency ( $\rho$ ) of POEA with respect to glyphosate as the  $LC_{50}$  of glyphosate divided by the corresponding  $LC_{50}$  for POEA. In other words, for rainbow trout at pH 6.5, the  $LC_{50}$  for POEA is 7.4 mg/L and the corresponding  $LC_{50}$  for glyphosate is 140 mg/L. Thus, the relative potency of POEA with respect to glyphosate is about 19 [140 ÷ 7.4 = 18.92].

In mixtures, the concept of relative potency provides an explicit tool for identifying the most significant toxic agent(s) in a mixture as well as for assessing potential interactions among agents in a mixture (Durkin 1981, Mumtaz et al. 1994). For example, for a mixture of two agents with the same potency present in a mixture in proportions of  $\pi_1$  and  $\pi_2$ , the fractional contribution of each agent to the toxicity of the mixture is simply the proportion ( $\pi_1$  or  $\pi_2$ ) of the agent in the mixture. When the potencies differ, both agents contribute equally to the toxicity of the mixture when  $\pi_1$  is equal to  $\rho \pi_2$ . As above,  $\rho$  is defined here as the LC<sub>50</sub> of component 1 divided by the LC<sub>50</sub> of component 2.

In Roundup, glyphosate is present at 356 g/L and POEA is present at 150 g/L. The proportion of glyphosate in Roundup ( $\pi_{G}$ ), ignoring the only other constituent which is water, is about 0.7 [356÷(356+150)]. Similarly, the proportion of POEA ( $\pi_{s}$  for proportion of surfactant) in the mixture is about 0.3 [150÷ (356+150)]. Both constituents would contribute equally to the mixture if the relative potency of POEA was about 2.3 [0.7÷0.3]. The relative potency of POEA with respect to glyphosate is much greater than 2.3, at least for fish species (Table 4-1). Thus, POEA is the more significant toxic agent in the mixture.

The magnitude of the difference can be expressed in various ways, the simplest of which is the ratio of the concentrations or equivalently the ratios of the proportions adjusted for the difference in potency:

For example, if the relative potency is 70, as it is in Table 4-1 for rainbow trout at pH 7.2, POEA may be said to contribute 30 [70  $\cdot$  0.3  $\div$  0.7] times more than glyphosate to the toxicity of the mixture.

This method of describing relative toxic contribution is based on the assumption that the components in the mixture do not affect one another (i.e., there are no toxicological interactions). For terrestrial plants, such interactions have been clearly documented. One method for assessing whether or not similar interactions are plausible in aquatic species is to compare the observed  $LC_{50}$  values for Roundup to the  $LC_{50}$  values that would be predicted by one model of non-interactive joint action, simple similar action (Finney 1971, Durkin 1981). Using this assumption, the expected  $LC_{50}$  can be calculated as:

$$LC_{50_{Roundup}} = \frac{LC_{50_{Ghphosate}}}{(\pi_G + \rho \pi_S)}$$
(4-2)

where  $\pi$  and  $\rho$  are as defined above.

The predicted  $LC_{50}$  values for Roundup based on this assumption are presented in the second to the last column of Table 4-1, and the ratio of the predicted to observed  $LC_{50}$  values are given in the last column. Ratios >1 suggest some form of greater than additive toxicity, and, conversely, ratios <1 indicate less than additive toxicity. Note also that the observed  $LC_{50}$  values for Roundup are presented as the total concentration of glyphosate and POEA. In other words, the  $LC_{50}$  values for Roundup reported in Folmar et al. (1979) are multiplied by 1.42 ((352+150)÷352) and give the  $LC_{50}$  values in units of weight of both glyphosate and POEA. These units are required for the above equation 4-2.

As indicated in Table 4-1, there is a tendency for the toxicity of glyphosate to decrease (i.e., the  $LC_{50}$  values increase) as the pH increases, although the changes are not substantial. The effect of pH on POEA is also not substantial but the effect seems to be the opposite of the effect that pH has on glyphosate. In all of the bioassays, the surfactant is more toxic than glyphosate. Because

of the effect of pH on toxicity, the relative potency of POEA increases as pH increases. At all pH levels, the ratio of predicted to observed  $LC_{50}$  values for Roundup does not deviate remarkably or systematically from unity, suggesting that no substantial interactions take place between these two compounds.

A similar analysis of the results presented by Wan et al. (1989) are summarized in Table 4-2. In general, this study agrees well with the earlier study by Folmar et al. (1979). In all cases, the surfactant is substantially more toxic than glyphosate. The effect of pH is more consistent and more substantial: the toxicity of glyphosate decreases and the toxicity of the surfactant increases with increasing pH. This is discussed further by Wan et al. (1992) but no additional experimental data are provided. Consequently, the relative potency of the surfactant to glyphosate also increases with increasing pH. The LC<sub>50</sub> values reported in Wan et al. (1989) for Roundup are expressed as "*mg product/L*." In calculating the expected LC<sub>50</sub> values for Roundup in Table 4-2, it is assumed that these LC<sub>50</sub> values include the concentrations of both glyphosate and the surfactant. As indicated in the last column of this table, the ratio of the predicted to observed LC<sub>50</sub> values for Roundup are consistently <1, indicating a less than additive interaction. Similarly, Wan et al. (1991) found no significant interaction between glyphosate and an indicator dye, referred to as basacid blue.

The significance of this information on the toxicity of the POEA surfactant to the current risk assessment is that much of the toxicity and all of the available monitoring data used in the risk assessment for aquatic species is on glyphosate rather than the surfactant. Because POEA is the toxic agent of primary concern in some Roundup formulations, the monitoring data used in the exposure assessment and toxicity data used in the dose response assessment must be adjusted, as discussed in Section 4.3, to consider the differences in potency between these two agents.

The only reported potential adverse effect in fish associated with field applications of a glyphosate/surfactant mixture is given by Trumbo (2002). This report involved the application of Rodeo (1.5%) and the surfactant R-11 (0.5%) to three sites for the control of purple loosestrife. Water samples were collected from water near the application. At one site, glyphosate was monitored at 0.85 mg/L and the surfactant was monitored at 0.4 mg NPE/L and 0.0125 mg/L. When fathead minnows were exposed to this water in the laboratory, 30% mortality was noted after 96 hours and this mortality was significantly (p<0.05) greater than control mortality. As discussed by Trumbo (2002), it is unlikely that the mortality was associated directly with glyphosate but the 96-hour LC<sub>50</sub> for R-11 is about 4 mg/L or one-tenth of the monitored concentration of NPE and it is likely that the mortality was attributable to the surfactant.

**4.1.3.1.4.** Sub-lethal Effects – The sub-lethal effects of glyphosate are less well characterized than its acute lethal potency. As noted in Appendix 9, non-lethal NOEC concentrations are available in many of the studies that report  $LC_{50}$  values. These NOEC values may be regarded as information on "sub-lethal" exposures in that no lethality was observed. In terms of this risk assessment, however, the term sublethal is not intended to apply to endpoints that may be precursor effects leading to mortality such as various forms of necrosis or other degenerative

changes in organs associated with the lethality. In addition, the term *sublethal* is not intended to apply to levels of exposure in which no mortality was observed. Such effects will be referred as *nonlethal* endpoints. Rather, *sublethal* will be used to designate endpoints that may lead to harmful but nonlethal effects that may impact the ability of wildlife species to maintain normal populations. In other words, the term *sub-lethal* is intended to designate effects that may impact reproduction, behavior, or the ability to respond to other stressors.

Six studies have reported and focused on acute effects other than mortality in fish (El-Gendy et al. 1998; Grisolia 2002; Janz et al. 1991; Morgan et al. 1991; Neskovic et al. 1996b; Szarek et al. 2000). The studies by Szarek et al. (2000) involve observations of nonlethal endpoints that do not substantially impact the hazard identification for fish. Janz et al. (1991) noted that short-term exposures at 5% to 85% of the 96 hour  $LC_{50}$  values of several glyphosate formulations do not induce indicators of physiological stress assayed as changes in biochemical parameters in blood. Morgan et al. (1991) noted that trout do not exhibit avoidance responses to glyphosate formulations at concentrations less than the 96-hour LC<sub>50</sub>. Behavioral changes - i.e., changes in coughing and ventilation rates, changes in swimming, loss of equilibrium, and changes in coloration – were observed at 25% of the  $LC_{50}$  values over exposure periods of up to 96 hours. The study by Grisolia (2002) is of minimal relevance because it involved intra-abdominal injections in a erythrocyte micronucleus assay in *Tilapia rendalli*. The results of the assay were positive and the authors interpreted this as indicating that fish DNA might be more sensitive as an indicator of potential mutagenic effects. While this interpretation may have some merit, it does not add substantially to large data base available in a variety of mammalian systems, as discussed in Section 3.1.10. The study by Szarek et al. (2000) involved very brief exposures of carp to Roundup concentrations that are far greater than the  $LC_{50}$  values – i.e., 1 hour exposures to 205 mg a.e./L and one-half hour exposures to and 410 mg a.e./L. All fish died during these exposures. Changes were observed in the mitochondria of carp hepatocytes. The observed effects may be due to the uncoupling oxidative phosphorylation (Section 3.1.2). Conversely, given that all fish died during exposure, these effects may represent normal post-mortem pathology. In either event, this is not suggestive of a sub-lethal effect that is relevant to the assessment of population level effects.

The studies by Neskovic et al. (1996b) and El-Gendy et al. (1998) report effects that could be viewed as true sub-lethal toxicity. The study by Neskovic et al. (1996b) noted histologic changes in the gills, kidneys, and liver of carp, *Cyprinus carpio*. In this study, carp were exposed to *technical grade* glyphosate but the purity was only 62%, much lower than that used in current commercial formulations. Nonetheless, the 96-hour LC<sub>50</sub> value for the technical grade glyphosate in carp is reported as 620 (607-638) mg/L, which is higher than values for more highly purified forms of glyphosate (Section 4.1.3.1.2) in trout and bluegill sunfish. The sub-lethal studies were conducted over 14-days of exposure to concentrations of 2.5, 5, 10 mg a.e./L. At 10 mg/L abnormal histopathologic changes were noted in the gills and liver. At 5 mg/L abnormal histopathologic changes were noted only in the gills. These changes were accompanied by increased alkaline phosphatase activity. While these effects cannot be directly associated with

potential longer term effects on fish populations, the histologic changes in the gills and liver would be classified as adverse.

In terms of impacting the risk assessment, however, it should be noted that a full life-cycle toxicity study has been conducted in fathead minnow, a standard chronic toxicity that was required by and accepted by the U.S. EPA (1993a). In this study, the NOEC was 25.7 mg/L (U.S. EPA, 1993a, p. 41). It is important to note that the NOEC from this full life-cycle toxicity study not only indicates a lack of mortality but also indicates that the fish were able to reproduce normally. Thus, it is conceivable, based on the work of Neskovic et al. (1996b), that the some at least transient histopathologic effects could occur at the NOEC was 25.7 mg/L. Nonetheless, in terms of the risk assessment, the life cycle NOEC of 25.7 mg/L remains the most appropriate basis for risk characterization.

El-Gendy et al. (1998) published a study on potential effects of glyphosate on immune function in fish. This is the only study that has reported any effect on immune function in any species. In this study, Bolti fish (*Tilapia nilotica*) were exposed for up to 4 weeks to glyphosate. However, neither the formulation of glyphosate nor the specific concentration used in the study are reported. Instead, exposure level was described as "1/1000 of the field recommended concentration" and the formulation is given only as "glyphosate 48% SC".

This study examined a number of important immunologic endpoints including:

- Proliferative response of splenocytes (LT) to the T-cell mitogens phytohemagglutinin (PHA) and concanavalin A (Con A) and to the B-cell mitogen lipopolysaccharide (LPS) using This is an *in vitro* assay for cell-mediated immunity.

- The Plaque Forming Cell (PFC) assay following *in vitro* immunization with sheep red blood cells (SRBC). This is a key assay to determine effects on humoral (antibody in circulation) immunity.

- The quantification of serum anti-SRBC levels. This endpoint is also an assay for humoral immunity.

- The electrophoretic evaluation of serum protein fractions. This is a general parameter for detection of overt/non-specific toxicity.

However, there are several aspects of this study that pose difficulties in interpreting the data.

Firstly, it is stated that the LT assay was performed on blood samples taken at 1 hr, 24 hr, 2 and 4 weeks from the time of treatment. It is assumed that for each of these treatment dates a new set of cultures would be set up. Therefore one would expect to have stimulation index (SI) values for the control for each of the mitogens tested at each time point. This is not the case since SI values for all three mitogens are presented only once. Furthermore, it is not clear for which time point the stated SI values are (see Table 1 in El-Gendy et al. 1998). Secondly, the authors report data

for the anti-SRBC titres (Table 3 in El-Gendy et al. 1998) at 1 hr, 24 hr, 2 and 3 weeks. No data are presented for optimizing the number of SRBC injected. The schedule of immunization (one injection vs multiple injections) with SRBC is not stated by the authors. It is rather odd that statistically significant depressed anti-SRBC titres are noted within one hr following treatment. Further, no data are presented on the preimmunization level of anti-SRBC in the control and treated. Also only one control value is presented and the time point to which this value applies is not specified. Finally, no control values are presented for each of the time points to which the treated groups should be compared.

The PFC assay is carried out in vitro using several treatment levels in  $\mu$ M quantities. Data from this assay is questionable for the following reasons: It is not clear whether the assay was performed in groups of fish separate from those which were immunized for anti-SRBC *in vivo*; there is evidence from Table 2 in El-Gendy et al. 1998 that the concentrations used in this assay are cytotoxic to spleen cells. Thus, the issue of direct toxicity of the chemicals in question on cells of the immune system is a very important issue. Ideally there should be very little toxicity when one deals with immunologic assays. In addition, the data on protein levels and serum fractions are inconclusive. Lastly, and most importantly, the authors do not mention any infections of the fish and have not challenged the fish with any infectious agent to test for a potential ecological effects, the failure to test for susceptibility to infections greatly reduces the utility of this study. Thus, it can not be concluded from the data presented in this study that the effects reported on the immune system represent a direct toxic effect on the immune parameters examined. Given the reported cytotoxicity, it is plausible that the reported immune effects are the result of general cytotoxicity rather than due to specific effects on immune function.

In addition to the above noted deficiencies, the study by El-Gendy et al. (1998) is inconsistent with a full life-cycle toxicity study conducted in fathead minnow (U.S. EPA, 1993a, p. 41), as discussed above. While El-Gendy et al. (1998) do not report the concentration tested in their study, the study required by U.S. EPA defines clearly a NOEC for an exposure over a life span. If glyphosate had caused any substantial impairment of immune function in this assay, signs of the immune impairment – i.e., increased infections – should have been apparent. Thus, in terms of the ecological risk assessment, the study by El-Gendy et al. (1998) has no substantial impact.

**4.1.3.1.5.** *Field Studies* – Several field studies are available indicating that the application of glyphosate to control aquatic weeds is beneficial to fish populations. Caffrey (1996) evaluated the efficacy of glyphosate in the control of emergent weeds along the river Boyne in Ireland. Glyphosate was applied as a "5L/ha" formulated product that is not otherwise specified. In other words, the information in this publication is not sufficient to calculate exposures either as lb/acre or concentration of glyphosate in water. While no rigorous studies of fish populations were conducted, anecdotal accounts from local anglers indicated that brown trout and salmon populations were enhanced and that the fish were observed to spawn in newly cleared areas. Similarly, Olaleye and Akinyemiju (1996) report a beneficial effect on fish populations in Nigeria when Roundup (360 g/L) was used for aquatic weed control and Kruger et al. (1996) report no

adverse effects when Roundup (360 g/L) was used for aquatic weed control in commercial carp production facilities. In an abstract, D'Silva et al. (1997) report that glyphosate was the least toxic herbicide, compared to 2,4-D, diquat, fluridone, endothall, in terms of sub-lethal effects in largemouth bass. This publication, however, provides little detail and a full publication has not been encountered in the literature.

4.1.3.2. Amphibians – The observation of hind limb deformities in free-living amphibians has substantially increased concern for the effects of xenobiotics on populations of amphibians (e.g., Quellet et al. 1997). Glyphosate IPA, Roundup, and the POEA surfactant used in Roundup have been specifically tested for malformations in the frog embryo teratogenesis assay (Perkins et al. 2000). In this assay, frog (Xenopus laevis) embryos are exposed to the test solution in petri dishes for 96-hours. As in the bioassay in fish, the least toxic agent was glyphosate IPA with an LC<sub>50</sub> of 5407 mg a.e./L and an LC<sub>5</sub> of 3779 mg a.e./L, indicating that glyphosate IPA is less toxic to frog embryos than to fish. Also as with fish, the most toxic agent was the POEA surfactant with  $LC_{50}$  of 2.7 mg/L and an  $LC_5$  of 2.2 mg/L. The Roundup formulation has an intermediate toxicity with an LC<sub>50</sub> of 9.4 mg a.e./L and an LC<sub>5</sub> of 6.4 mg a.e./L. The LC<sub>50</sub> values are comparable to the those of fish. No reported hind limb abnormalities were noted. The only abnormalities specified in the publication include uncoiling of the gut, edema, blistering, abnormal pigmentation, and axial twisting in control embryos. No statistically significant increase in abnormalities were seen in any groups exposed to glyphosate IPA, Roundup, and the POEA surfactant at levels that were not lethal. The precise number and nature of abnormalities in the groups exposed to lethal concentrations of glyphosate IPA, Roundup, and the POEA surfactant are, however, not specified.

Smith (2001) assayed another formulation of glyphosate, Kleeraway Grass and Weed Killer RTU (Monsanto), that contains glyphosate IPA at 0.75% as well as an ethoxylated tallowamine surfactant. Bioassays were conducted on tadpoles (1 week post-hatching) of the western chorus frog, *Pseudacris triseriata*, and the plains leopard frog (*Rana blairi*). The concentrations used in the bioassays are specified in the publication as 0.0001, 0.001, 0.01, 0.1 dilutions of the formulated product. A 0.75% formulation contains 7.5 g/L. Thus, the concentrations used in this study correspond to 0.75 mg IPA/L, 7.5 mg IPA/L, 75 mg IPA/L, and 750 mg IPA/L or 0.56 mg a.e./L, 5.6 mg a.e./L, 560 mg a.e./L. The test protocol involved a 24-hour exposure period followed by a two week observation period to detect sub-lethal toxicity. In *Pseudacris triseriata*, 100% mortality was observed at all concentrations above 0.56 mg a.e./L and 55% mortality was observed at this concentration. During the post-exposure observation period, 4/9 animals died in first 2 days. In an initial experiment with *Rana blairi*, all tadpoles died at all concentrations. In a repeat experiment using older tadpoles (not otherwise specified), all animals survived at 0.56 mg a.e./L. In both species, normal growth and development was observed over the two week observation period in all survivors.

This very high sensitivity of tadpoles reported by Smith (2001) is not consistent with the study by Bidwell and Gorrie (1995), who assayed the toxicity of glyphosate and "Roundup 360" in four species of frogs from western Australia. For juvenile frogs, 48-hour  $LC_{50}$  values were 51.8 mg

a.e./L for Roundup 360 and 83.6 mg/L for technical grade glyphosate. For tadpoles, 48-hour  $LC_{50}$  values were 11.6 mg a.e./L for Roundup 360 and 121 mg/L for technical grade glyphosate. Thus, while the tadpoles were somewhat more sensitive than juveniles, the reported  $LC_{50}$  values are in the range of those seen in fish.

An additional formulation of glyphosate, Roundup Biactive is available in Australia (Monsanto Australia Limited, 2000). This formulation is less toxic than Roundup to aquatic species based on bioassays in various species of frogs (Mann and Bidwell 1999).

Cole et al. (1997) report no effect on populations of six species of amphibians (based on capture rates) among clearcut sites with and without glyphosate applications. Species included rough-skin newt, ensatina, Pacific giant salamander, Dunn's salamander, western redback salamander, and relegged frog. Removal of red alder from the habitat, reduced amphibian populations regardless of the method used to remove the alder.

**4.1.3.3.** Aquatic Invertebrates – As with the toxicity data on fish, the U.S. EPA (1993c) summarizes a standard set of bioassays of toxicity to aquatic invertebrates. For aquatic invertebrates, however, the  $LC_{50}$  values are typically given based on a 48 hour rather than 96 hour exposure period. The 48-hour  $LC_{50}$  of technical grade glyphosate (83%) to *Daphnia magna* is listed as 780 mg/L, substantially higher than the 96-hour  $LC_{50}$  values in freshwater fish. For the common midge, *Chironomus plumosus*, the 48-hour  $LC_{50}$  of 96.7% technical grade glyphosate is listed as 55 (31-97) mg/L (U.S. EPA/OPP 1993c, p. 44). As with fish, the toxicity of some formulated glyphosate products is much greater, with  $LC_{50}$  values as low as 3 mg/L (U.S. EPA/OPP 1993c, p. 46). These and other more recent studies submitted to U.S. EPA (e.g., Long et al. 1996a; Long et al. 1996b; Drottar and Swigert 1998a) on the toxicity of glyphosate and glyphosate formulations are summarized in Appendix 3 (comparison of formulations) and Appendix 10 (toxicity to aquatic invertebrates).

In addition to registrant submitted studies, Alberdi et al. (1996) has published a comparative study on the toxicity of a glyphosate formulation to two species of daphnids, *Daphnia magna* and *Daphnia spinulata*. The formulation of glyphosate tested in this study contained glyphosate IPA 48% and a different surfactant - characterized as a oxide-coco-amide-propyl dimethyl-amine (15%). The 48 hour EC<sub>50</sub>s at IPA are reported as 66.18 mg/L in *D. spinulata* and 61.72 (58.8-64.2) mg/L in *D. magna*. These are comparable to the LC<sub>50</sub> of 72 mg/L for a glyphosate formulation containing the "W" surfactant (U.S. EPA/OPP 1993c).

One chronic life cycle toxicity study with *Daphnia magna* is reported in U.S. EPA/OPP (1993c) with an MATC of between 50 and 96 mg/L. This is comparable to the NOEC of 25.7 mg/L in the life-cycle study in fish (Section 4.1.3.1.2). As with the fish study, this is the most relevant bioassay in aquatic invertebrates for assessing the longer term effects of glyphosate in ambient water.

The effects of glyphosate have also been determined in an aquatic snail, *Pseudosuccinea columella*, intermediate host of the sheep liver fluke. Tate et al. (1997) assayed glyphosate acid for sub-lethal effects on egg production at concentrations of 0.1, 1, and 10 mg a.e./L for 3-generations. No marked effects were noted on the first or second generations. In the third generation, snail embryos exposed to 1 mg/L developed much faster than those exposed at 0.1 or 10 mg/L and faster than control snails. Hatching, however, was inhibited at 10 mg/L and inhibited slightly at 0.1 mg/L but egg-laying capacity increased at both of these concentrations. In a follow up study, Tate et al. (2000) noted effects on concentrations of amino acids in snails (specifically alanine, glycine, glutamic acid and threonine) at the same concentrations. Effects on concentrations of some proteins have also been noted by Christian et al. (1993) for this species of snail. The mechanism for the effect of glyphosate on amino acid and protein metabolism is not known. In terms of reproductive effects that might be significant, the Tate et al. (1997) study suggests that some changes could be observed at concentrations as low as 0.1 mg/L but that the mixed effects of glyphosate on egg-laying capacity and hatching could be off-setting in terms of total reproductive capacity.

Various field studies have not noted any remarkable effects on aquatic invertebrates. At application rates of 1 L Rodeo/ha for the control of purple loosestrife, Gardner and Grue (1996) noted no adverse effects on aquatic invertebrates. At application rates of 0.94 or 1.48 kg a.i./ha as glyphosate IPA (Rodeo), Hagg (1986) found no indication of lethality in two water hyacinth weevils, *Neochetin eichhorniae* and *N. bruchi*. In a forest pond mesocosm, Hildebrand et al. (1980) found no differences in invertebrate survival over an 8 day period after sprays of 2.2 kg/ha, 22 kg/ha and 220 kg/ha. Lastly, no indication of short or long term (119 days) effects were noted after the application of a Rodeo and X-77 mixture in control of smooth cordgrass in a marine estuary. In this study, Rodeo was applied at a rate of 4.7 L/ha and X-77 was applied at a rate of 1 L/ha (Simenstad et al. 1996).

**4.1.3.4. Aquatic Plants** – The toxicity of glyphosate to aquatic plants has been evaluated by U.S. EPA/OPP (1993c) based on studies submitted for the registration of glyphosate. In addition, several studies are available from the open literature as well as more recent studies submitted to U.S. EPA. These studies are detailed in Appendix 11 for both algae and aquatic macrophytes. As would be expected from a herbicide, glyphosate is much more toxic to aquatic plants than animals.  $EC_{50}$  values for technical grade glyphosate in algae reported by U.S. EPA/OPP (1993c, p. 51) are as low as 0.85 mg/L for *Skeletonema costatum* (a marine species). The lower value of 0.85 mg/L for *Skeletonema costatum* appears to be the most sensitive bioassay reported. In a more recent study on this species, Smyth et al. (1996c) report an  $EC_{50}$  of 12 (7.6-19) mg/L a 5-day with an NOEC of 1.8 mg/L and a LOEC of 3.2 mg/L.

Some species appear to be much more tolerant than *Skeletonema costatum*. Smyth et al. (1996d) report EC<sub>50</sub> of 17 (13-24) mg/L for a freshwater diatom *Navicula pelliculosa*, with growth enhancement observed at 1.8 mg/L and no inhibition observed at concentrations below 32 mg/L – i.e., the EC<sub>50</sub> was estimated by extrapolation. The EC<sub>50</sub> value reported for this species in EPA/OPP (1993c, p. 51) is to 39.9 mg/L. At stimulation of chlorophyll-a synthesis has also been

reported by Wong (2000) in *Scenedesmus quadricauda* at low (0.02 mg/L) concentrations. The most tolerant species of algae appears to be *Chlorella fusca*, with an EC<sub>50</sub> of 377 mg/L for growth inhibition (Faust et al. 1994).

Saenz et al. (1997) have reported that a commercial formulation of glyphosate, referred to as "Ron-do", is about equitoxic in *Scenedesmus acutus* and *Scenedesmus quadricauda* with glyphosate  $EC_{50}$  values of about 7.2 to 10.2 mg/L and  $EC_{50}$  values for Ron-do of about 99 mg/L when expressed as glyphosate concentrations. No information on this formulation, however, has been encountered and it does not appear to be distributed in the United States.

Freshwater aquatic macrophytes appear to be as sensitive to glyphosate as algae. The U.S. EPA/OPP (1996c) report a 7-day EC<sub>50</sub> of 21.5 mg/L in duckweed for technical grade glyphosate, very similar to the EC<sub>50</sub> reported for the freshwater diatom, *Navicula pelliculosa*. More recently, Smyth et al. (1996a) reported an EC<sub>50</sub> in duckweed for frond number 12 (11-14) mg/L, with a NOEC of 3 mg/L and a LOEC of 6 mg/L.

Austin et al. (1991) studied the effects of glyphosate on periphyton in artificial streams and noted an increase in periphyton concentrations at concentrations of 0.0019 - 0.2874 mg/L. The authors suggest that algae are using glyphosate as a phosphorous source and that glyphosate could contribute to eutrophication of coastal oligotrophic (nutrient poor and oxygen rich) waterways and this could effect salmonid populations. No specific data supporting this supposition, however, is reported. In a pond study, Perschbacher et al. (1997) report no adverse effect on plankton after an application of glyphosate at a rate of 0.43 kg/ha or about 0.4 lbs/acre.

### 4.2. EXPOSURE ASSESSMENT

**4.2.1. Overview.** Terrestrial animals might be exposed to any applied herbicide from direct spray, the ingestion of contaminated media (vegetation, prey species, or water), grooming activities, or indirect contact with contaminated vegetation. The highest exposures for terrestrial vertebrates will occur after the consumption of contaminated vegetation or contaminated insects. In acute exposure scenarios, doses as high as 225 mg/kg are estimated. Other routes of exposure, like the consumption of contaminated water or direct spray, lead to lower levels of exposure. In chronic exposure scenarios, the estimated daily doses at the upper limits of exposure are in the range of about 50 to 80 mg/kg/day and are associated with highly conservative assumptions regarding the consumption of contaminated vegetation.

The primary hazards to non-target terrestrial plants are associated with unintended direct deposition or spray drift. Unintended direct spray will result in an exposure level equivalent to the application rate. At least some plants that are sprayed directly with glyphosate at or near the recommended range of application rates will be damaged. Based on the AgDRIFT model, no more than 0.0058 of the application rate would be expected to drift 100 m offsite after low boom ground applications. In order to encompass a wide range of field conditions, GLEAMS simulations were conducted for clay, loam, and sand at annual rainfall rates from 5 to 250 inches. Under arid conditions (i.e., annual rainfall of about 10 inches or less), there is no or very little runoff. Under these conditions, degradation, not dispersion, accounts for the decrease of glyphosate results in runoff losses that range from about negligible up to about 45% of the application rate, depending primarily on the amount of rainfall rather than differences in soil type.

The potential for effects on aquatic species are based on estimated concentrations of glyphosate in water that are identical to those used in the human health risk assessment without further elaboration. For an accidental spill, the central estimate for the concentration of glyphosate in a small pond is estimated at about 18.2 mg/L with a range from 1.8 to 127 mg/L. For longer term exposure scenarios, the expected concentrations of glyphosate in ambient water range from 0.0001 to 0.008 mg/L with a central value of 0.001 mg/L.

**4.2.2. Terrestrial Animals.** Terrestrial animals might be exposed to any applied herbicide from direct spray, the ingestion of contaminated media (vegetation, prey species, or water), grooming activities, or indirect contact with contaminated vegetation.

In this exposure assessment, estimates of oral exposure are expressed in the same units as the available toxicity data (i.e., oral  $LD_{50}$  and similar values). As in the human health risk assessment, these units are usually expressed as mg of agent per kg of body weight and abbreviated as mg/kg body weight. For dermal exposure, the units of measure usually are expressed in mg of agent per cm<sup>2</sup> of surface area of the organism and abbreviated as mg/cm<sup>2</sup>. In estimating dose, however, a distinction is made between the exposure dose and the absorbed dose. The *exposure dose* is the amount of material on the organism (i.e., the product of the residue level in mg/cm<sup>2</sup> and the amount of surface area exposed), which can be expressed either as mg/organism or mg/kg body

weight. The *absorbed dose* is the proportion of the exposure dose that is actually taken in or absorbed by the animal.

For the exposure assessments discussed below, general allometric relationships are used to model exposure. In the biological sciences, allometry is the study of the relationship of body size or mass to various anatomical, physiological, or pharmacological parameters (e.g., Boxenbaum and D'Souza 1990). Allometric relationships take the general form:

#### $y = aW^x$

where W is the weight of the animal, y is the variable to be estimated, and the model parameters are **a** and **x**. For most allometric relationships used in this exposure assessment, **x** ranges from approximately 0.65 to 0.75. These relationships dictate that, for a fixed level of exposure (e.g., levels of a chemical in food or water), small animals will receive a higher dose, in terms of mg/kg body weight, than large animals.

Estimates of exposure are given for both a small and a large mammal as well as a small and a large bird. For many compounds, allometric relationships for interspecies sensitivity to toxicants indicate that for exposure levels expressed as mg toxicant per kg body weight (mg/kg body weight), large animals, compared with small animals, are more sensitive.

As discussed in Section 3.1.4 and 3.1.5, the limited data on glyphosate do suggest that larger mammals, specifically the dog, appear to be more sensitive to glyphosate than smaller mammals (i.e., rats and mice) but the data are not adequate to support the development of quantitative allometric relationships for toxicity. There are no data to assess species sensitivity in small and large birds.

The exposure assessments for terrestrial animals are summarized in Table 4-3. As with the human health exposure assessment, the computational details for each exposure assessment presented in this section are provided in the attached worksheets (worksheets F01 through F14).

**4.2.2.1.** *Direct Spray* – In the broadcast application of any herbicide, wildlife species may be sprayed directly. This scenario is similar to the accidental exposure scenarios for the general public discussed in section 3.2.3.2. In a scenario involving exposure to direct spray, the extent of dermal contact depends on the application rate, the surface area of the organism, and the rate of absorption.

For this risk assessment, three groups of direct spray exposure assessments are conducted. The first, which is defined in worksheet F01, involves a 20 g mammal that is sprayed directly over one half of the body surface as the chemical is being applied. The range of application rates as well as the typical application rate is used to define the amount deposited on the organism. The absorbed dose over the first day (i.e., a 24-hour period) is estimated using the assumption of first-order dermal absorption. In the absence of any data regarding dermal absorption in a small mammal,

the estimated absorption rate for humans is used (see section 3.1.7). An empirical relationship between body weight and surface area (Boxenbaum and D'Souza 1990) is used to estimate the surface area of the animal. The estimates of absorbed doses in this scenario may bracket plausible levels of exposure for small mammals based on uncertainties in the dermal absorption rate of glyphosate.

Other, perhaps more substantial, uncertainties affect the estimates for absorbed dose. For example, the estimate based on first-order dermal absorption does not consider fugitive losses from the surface of the animal and may overestimate the absorbed dose. Conversely, some animals, particularly birds and mammals, groom frequently, and grooming may contribute to the total absorbed dose by direct ingestion of the compound residing on fur or feathers. Furthermore, other vertebrates, particularly amphibians, may have skin that is far more permeable than the skin of most mammals (Moore 1964).

Quantitative methods for considering the effects of grooming or increased dermal permeability are not available. As a conservative upper limit, the second exposure scenario, detailed in worksheet F02a, is developed in which complete absorption over day 1 of exposure is assumed.

Because of the relationship of body size to surface area, very small organisms, like bees and other terrestrial insects, might be exposed to much greater amounts of glyphosate per unit body weight, compared with small mammals. Consequently, a third exposure assessment is developed using a body weight of 0.093 g for the honey bee (USDA/APHIS 1993). Because there is no information regarding the dermal absorption rate of glyphosate by bees or other invertebrates, this exposure scenario, detailed in worksheet F02b, also assumes complete absorption over the first day of exposure.

Direct spray scenarios are not given for large mammals. As noted above, allometric relationships dictate that large mammals will be exposed to lesser amounts of a compound in any direct spray scenario than smaller mammals. As detailed further in Section 4.4, the direct spray scenarios for the small mammal are substantially below a level of concern. Consequently, elaborating direct spray scenarios for a large mammal would have no impact on the characterization of risk.

**4.2.2.2.** Indirect Contact – As in the human health risk assessment (see section 3.2.3.3), the only approach for estimating the potential significance of indirect dermal contact is to assume a relationship between the application rate and dislodgeable foliar residue. The study by Harris and Solomon (1992) is used to estimate that the dislodgeable residue will be approximately 10 times less than the nominal application rate.

Unlike the human health risk assessment in which transfer rates for humans are available, there are no transfer rates available for wildlife species. As discussed in Durkin et al. (1995), the transfer rates for humans are based on brief (e.g., 0.5- to 1-hour) exposures that measure the transfer from contaminated soil to uncontaminated skin. Species of wildlife are likely to spend longer periods of time, compared to humans, in contact with contaminated vegetation.

It is reasonable to assume that for prolonged exposures a steady-state may be reached between levels on the skin, rates of absorption, and levels on contaminated vegetation, although there are no data regarding the kinetics of such a process. The bioconcentration data on glyphosate (section 3.2.3.5) as well as its high water solubility and low octanol/water partition coefficient suggest that glyphosate is not likely to partition from the surface of contaminated vegetation to the surface of skin, feathers, or fur. Thus, a plausible but conservative partition coefficient is unity (i.e., the concentration of the chemical on the surface of the animal will be equal to the dislodgeable residue on the vegetation).

Under these assumptions, the absorbed dose resulting from contact with contaminated vegetation will be one-tenth that associated with comparable direct spray scenarios. As discussed in the risk characterization for ecological effects (section 4.4), the direct spray scenarios result in exposure levels far below those of toxicological concern. Consequently, details of the indirect exposure scenarios for contaminated vegetation are not further elaborated in this document.

**4.2.2.3.** *Ingestion of Contaminated Vegetation or Prey* – Since glyphosate will be applied to vegetation, the consumption of contaminated vegetation is an obvious concern and separate exposure scenarios are developed for acute and chronic exposure scenarios for a small mammal (Worksheets F04a and F04b) and large mammal (Worksheets F10, F11a, and F11b) as well as large birds (Worksheets F12, F13a, and F13b).

A small mammal is used because allometric relationships indicate that small mammals will ingest greater amounts of food per unit body weight, compared with large mammals. The amount of food consumed per day by a small mammal (i.e., an animal weighing approximately 20 g) is equal to about 15% of the mammal's total body weight (U.S. EPA/ORD 1989). When applied generally, this value may overestimate or underestimate exposure in some circumstances. For example, a 20 g herbivore has a caloric requirement of about 13.5 kcal/day. If the diet of the herbivore consists largely of seeds (4.92 kcal/g), the animal would have to consume a daily amount of food equivalent to approximately 14% of its body weight [(13.5 kcal/day  $\div$  4.92 kcal/g), the animal would have to consume a daily amount of food equivalent to consume a daily amount of food equivalent to approximately 2.46 kcal/g) $\div$ 20g = 0.274] (U.S. EPA/ORD 1993, pp.3-5 to 3-6). For this exposure assessment, the amount of food consumed per day by a small mammal weighing 20 g is estimated at about 3.6 g/day from the general allometric relationship for food consumption in rodents (U.S. EPA/ORD 1993, p. 3-6).

A large herbivorous mammal is included because empirical relationships of concentrations of pesticides in vegetation, discussed below, indicate that grasses may have substantially higher pesticide residues than other types of vegetation such as forage crops or fruits (Worksheet A04). Grasses are an important part of the diet for some large herbivores, but most small mammals do not consume grasses as a substantial proportion of their diet. Thus, even though using residues from grass to model exposure for a small mammal is the most conservative approach, it is not generally applicable to the assessment of potential adverse effects. Hence, in the exposure

scenarios for large mammals, the consumption of contaminated range grass is modeled for a 70 kg herbivore, such as a deer. Caloric requirements for herbivores and the caloric content of vegetation are used to estimate food consumption based on data from U.S. EPA/ORD (1993). Details of these exposure scenarios are given in worksheets F10 for acute exposures as well as Worksheets F11a and F11b for longer-term exposures.

For the acute exposures, the assumption is made that the vegetation is sprayed directly – i.e., the animal grazes on site – and that the animal consumes 100% of the animals diet is contaminated. While appropriately conservative for acute exposures, neither of these assumptions are plausible for longer-term exposures. Thus, for the longer-term exposure scenarios for the large mammal, two sub-scenarios are given. The first is an on-site scenario that assumes that a 70 kg herbivore consumes short grass for a 90 day period after application of the chemical. The contaminated vegetation accounts for 10 to 100% of the diet assuming that the animal would spend 10 to 100% of the grazing time at the application site. Because the animal is assumed to be feeding at the application site, drift is set to unity - i.e., direct spray. This scenario is detailed in Worksheet 12a. The second sub-scenario is similar except the assumption is made that the animal is grazing at distances of 25 to 100 feet from the application site (lowing risk) but that the animal consumes 100% of the diet from the contaminated area (increasing risk). For this scenario, detailed in Worksheet F12b, AgDRIFT is used to estimate deposition on the off-site vegetation. Drift estimates from AgDrift are summarized in Worksheet A06 and this model is discussed further in Section 4.2.3.2.

The consumption of contaminated vegetation is also modeled for a large bird. For these exposure scenarios, the consumption of range grass by a 4 kg herbivorous bird, like a Canada Goose, is modeled for both acute (Worksheet F12) and chronic exposures (Worksheets F13a and F13b). As with the large mammal, the two chronic exposure scenarios involve sub-scenarios for on-site as well as off-site exposure.

For this component of the exposure assessment, the estimated amounts of pesticide residue in vegetation are based on the relationship between application rate and residue rates on different types of vegetation. As summarized in Worksheet A04, these residue rates are based on estimated residue rates from Fletcher et al. (1994).

Similarly, the consumption of contaminated insects is modeled for a small (10g) bird. No monitoring data have been encountered on the concentrations of glyphosate in insects after applications of glyphosate. The empirical relationships recommended by Fletcher et al. (1994) are used as surrogates as detailed in Worksheet F14. To be conservative, the residue rates from small insects are used – i.e., 45 to 135 ppm per lb/ac – rather than the residue rates from large insects – i.e., 7 to 15 ppm per lb/ac. As detailed in Worksheet F14, this approach yields dose estimates of about 20 to 800 mg/kg bw for a small bird. This is higher than any monitored residues in terrestrial animals by factors or about 10 to 400 and is likely to be grossly conservative (Newton et al. 1984, p. 1148, Table II).

As indicated in Section 3.2.3.6, the empirical relationships recommended by Fletcher et al. (1994) for fruit may somewhat overestimate concentrations on fruit based on the study by Siltanen et al. (1981). As indicated in Worksheet A04, Fletcher et al. (1994) estimate residue rates on leaves at 45 to 135 ppm per lb applied and on short grasses as 85 to 240 ppm per lb applied. Newton et al. (1994) found that initial residues on "herbaceous vegetation" were typically in the range of 360 to 1273 ppm after the application of glyphosate at a rate of 4.12 kg/ha or about 3.7 lb/acre (Newton et al. 1994, , p. 1798, Table 3). This corresponds to residue rates of about 97 to 344 ppm per lb applied per acre [360ppm to 1273 ppm  $\div$  3.7 lb/acre]. This suggests that the estimates of glyphosate on grasses could be somewhat underestimated using the residue rates from Fletcher et al. (1994). The potential impact of this underestimate is discussed further in the risk characterization.

In addition to the consumption of contaminated vegetation and insects, glyphosate may reach ambient water and bioconcentrate in fish. Thus, a separate exposure scenario is developed for the consumption of contaminated fish by a predatory bird in both acute (worksheet F08) and chronic (worksheet F09) exposures. Because predatory birds usually consume more food per unit body weight than do predatory mammals (U.S. EPA 1993, pp. 3-4 to 3-6), separate exposure scenarios for the consumption of contaminated fish by predatory mammals are not developed.

**4.2.3. Terrestrial Plants.** In general, the primary hazard to non-target terrestrial plants associated with the application of most herbicides is unintended direct deposition or spray drift. In addition, herbicides may be transported off-site by percolation or runoff or by wind erosion of soil.

**4.2.3.1.** *Direct Spray* – Unintended direct spray will result in an exposure level equivalent to the application rate. For many types of herbicide applications - e.g., rights-of-way management, it is plausible that some non-target plants immediately adjacent to the application site could be sprayed directly. This type of scenario is modeled in the human health risk assessment for the consumption of contaminated vegetation.

**4.2.3.2.** *Off-Site Drift* – Because off-site drift is more or less a physical process that depends on droplet size and meteorological conditions rather than the specific properties of the herbicide, estimates of off-site drift can be modeled using AgDRIFT (Teske et al. 2001). AGDRIFT is a model developed as a joint effort by the EPA Office of Research and Development and the Spray Drift Task Force, a coalition of pesticide registrants. AGDRIFT is based on the algorithms in FSCBG (Teske and Curbishley. 1990), a drift model previously used by USDA.

For aerial applications, AGDRIFT permits very detailed modeling of drift based on the chemical and physical properties of the applied product, the configuration of the aircraft, as well as wind speed and temperature. For ground applications, AGDRIFT provides estimates of drift based solely on distance downwind as well as the types of ground application: low boom spray, high boom spray, and orchard airblast. Representative estimates based on AGDRIFT (Version 1.16) are given in Worksheet A06). For the current risk assessment, the AGDRIFT estimates are used

for consistency with comparable exposure assessments conducted by the U.S. EPA. In addition, AGDRIFT represents a detailed evaluation of a very large number of field studies and is likely to provide more reliable estimates of drift. Further details of AGDRIFT are available at <a href="http://www.agdrift.com/">http://www.agdrift.com/</a>.

Estimates of drift for ground applications is given in Worksheet A06. In ground broadcast applications, glyphosate will typically be applied by low boom ground spray and thus these estimates are used in the current risk assessment. Drift associated with backpack (directed foliar applications) are likely to be much less. This is discussed further in the risk characterization.

Drift distance can be estimated using Stoke's law, which describes the viscous drag on a moving sphere. According to Stoke's law:

$$v = \frac{D^2 \cdot g}{18n}$$
  
or  
$$v = 2.87 \cdot 10^5 \cdot D^2$$

where v is the velocity of fall (cm sec<sup>-1</sup>), D is the diameter of the sphere (cm), g is the force of gravity (980 cm sec<sup>-2</sup>), and n is the viscosity of air (1.9  $\cdot$  10<sup>-4</sup> g sec<sup>-1</sup> cm<sup>-1</sup> at 20°C) (Goldstein et al. 1974).

In typical backpack ground sprays, droplet sizes are greater than 100  $\mu$ , and the distance from the spray nozzle to the ground is 3 feet or less. In mechanical sprays, raindrop nozzles might be used. These nozzles generate droplets that are usually greater than 400  $\mu$ , and the maximum distance above the ground is about 6 feet. In both cases, the sprays are directed downward.

Thus, the amount of time required for a 100  $\mu$  droplet to fall 3 feet (91.4 cm) is approximately 3.2 seconds,

$$91.4 \div (2.87 \cdot 10^5 (0.01)^2).$$

The comparable time for a 400  $\mu$  droplet to fall 6 feet (182.8 cm) is approximately 0.4 seconds,

$$182.8 \div (2.87 \cdot 10^{5}(0.04)^{2}).$$

For most applications, the wind velocity will be no more than 5 miles/hour, which is equivalent to approximately 7.5 feet/second (1 mile/hour = 1.467 feet/second). Assuming a wind direction perpendicular to the line of application, 100  $\mu$  particles falling from 3 feet above the surface could drift as far as 23 feet (3 seconds  $\cdot$  7.5 feet/second). A raindrop or 400  $\mu$  particle applied at 6 feet above the surface could drift about 3 feet (0.4 seconds  $\cdot$  7.5 feet/second).

For backpack applications, wind speeds of up to 15 miles/hour are allowed in Forest Service programs. At this wind speed, a 100  $\mu$  droplet can drift as far as 68 feet (3 seconds  $\cdot$  15  $\cdot$  1.5 feet/second). Smaller droplets will of course drift further, and the proportion of these particles in the spray as well as the wind speed will affect the proportion of the applied herbicide that drifts off-site.

**4.2.3.3.** *Runoff* – Glyphosate or any other herbicide may be transported to off-site soil by runoff or percolation. Both runoff and percolation are considered in estimating contamination of ambient water. For assessing off-site soil contamination, however, only runoff is considered. This is similar to the approach used by U.S. EPA (1995) in their exposure assessment for terrestrial plants. The approach is reasonable because off-site runoff will contaminate the off-site soil surface and could impact non-target plants. Percolation, on the other hand, represents the amount of the herbicide that is transported below the root zone and thus may impact water quality but should not affect off-site vegetation.

Based on the results of the GLEAMS modeling (Section 3.2.3.4.2), the proportion of the applied glyphosate lost by runoff was estimated for clay, loam, and sand at rainfall rates ranging from 5 inches to 250 inches per year. These results are summarized in Worksheet G04 and indicate that runoff will be negligible in relatively arid environments but may reach up to about 30% to 45% of the applied amount at high rainfall rates.

**4.2.3.4.** *Wind Erosion* – Wind erosion is a major transport mechanism for soil (e.g., Winegardner 1996) and is associated with the environmental transport of herbicides (Buser 1990). Although numerous models were developed for wind erosion (e.g., Strek and Spaan 1997, Strek and Stein 1997), the quantitative aspects of soil erosion by wind are extremely complex and site specific. Field studies conducted on agricultural sites found that annual wind erosion may account for soil losses ranging from 2 to 6.5 metric tons/ha (Allen and Fryrear 1977). The upper range reported by Allen and Fryrear (1977) is nearly the same as the rate of 2.2 tons/acre (5.4 tons/ha) recently reported by the USDA (1998). The temporal sequence of soil loss (i.e., the amount lost after a specific storm event involving high winds) depends heavily on soil characteristics as well as meteorological and topographical conditions.

This risk assessment uses average soil losses ranging from 1 to 10 tons/ha·year, with a typical value of 5 tons/ha·year. The value of 5 tons/ha·year is equivalent to 500 g/m<sup>2</sup> [1 ton=1000 kg and 1 ha = 10,000 m<sup>2</sup>] or 0.05 g/cm<sup>2</sup> [1m<sup>2</sup>=10,000 cm<sup>2</sup>]. Thus, using a soil bulk density of 1.5 g/cm<sup>3</sup> (Knisel et al. 1992, p. 56), the depth of soil removed from the surface per year would be 0.033 cm[(0.05 g/cm<sup>2</sup>)÷ (1.5 g/cm<sup>3</sup>)]. The average amount per day would be about 0.00007 cm/day [0.033 cm per year ÷ 365 days/year]. The upper range of the typical daily loss would thus be about 0.00009 cm/day.

The amount of glyphosate that might be transported by wind erosion depends on several factors, including the application, the depth of incorporation into the soil, the persistence in the soil, the wind speed, and the topographical and surface conditions of the soil. Under desirable conditions,

like relatively deep (10 cm) soil incorporation, low wind speed, and surface conditions that inhibit wind erosion, it is likely that wind transport of glyphosate would be neither substantial or nor significant.

Any number of undesirable exposure scenarios could be constructed. As a reasonable 'worst case' scenario, it is assumed that glyphosate is applied to arid soil, that it is incorporated into the top 1 cm of soil, that minimal rainfall occurs for a 2-month period, that the degradation and dispersion of glyphosate in the soil is negligible over the 2-month period, and that local conditions favor a high rate of soil loss (i.e., smooth, sandy surface with high wind speeds) that is a factor at the upper limit of the typical rate (i.e., 0.00009 cm/day). Under those conditions, 0.0054 [0.00009 cm/day × 60 days  $\div$  1 cm] of the applied glyphosate would be lost due to wind erosion. This is virtually identical to the estimates of off-site contamination from low-boom applications at a distance of 100 feet from the application site and is greater than drift that would be expected 500 feet offsite (0.0016 for low-boom applications from Worksheet A06) by a factor about 3 [0.0054  $\div$  0.0016 = 3.375]. Thus, in areas where wind erosion of soil may occur, wind erosion could be a more important mode of offsite movement than drift during application.

The deposition of the glyphosate contaminated soil also will vary substantially with local conditions. Under desirable conditions, the soil might be dispersed over a very large area and be of no toxicological consequence. In some cases, however, local topographical conditions might favor the deposition and concentration of contaminated dust from a large treated area into a relatively small off-site area. An objective approach for modeling these types of events was not available in the literature. For this risk assessment, neither concentration nor dispersion is considered quantitatively.

**4.2.4. Aquatic Organisms.** The potential for effects on aquatic species are based on estimated concentrations of glyphosate in water that are identical to those used in the human health risk assessment (Section 3.2.3.4). Thus, for an accidental spill, the central estimate for the concentration of glyphosate in a small pond is estimated at about 18.2 mg/L with a range from 1.8 to 127 mg/L (Worksheet D05). For longer term exposure scenarios, the expected concentrations of glyphosate in ambient water range from 0.0001 to 0.008 mg/L with a central value of 0.001 mg/L (Worksheet B06).

# 4.3. DOSE-RESPONSE ASSESSMENT

**4.3.1. Overview.** For mammals, the toxicity data used to characterize risk are identical to those used in the human health risk assessment – i.e., a NOAEL of 175 mg/kg with an associated LOAEL of 350 mg/kg. The 175 mg/kg NOAEL and 350 mg/kg LOAEL values are used for both the acute and chronic risk assessments. This approach is taken because of the lack of a substantial dose-duration or dose-severity relationship for glyphosate. For birds, a dose of 100 mg/kg is used as a NOAEL for characterizing chronic risks. It should be noted that this dose is very close to the NOAEL of 175 mg/kg used for mammals and is consistent with the apparent lack of variability in the toxicity of glyphosate among species. As in the assessment for mammals, this NOAEL is based on a repeated dose study for reproductive effects. The acute NOAEL is taken as 562

mg/kg from a five-day dietary studies in bobwhite quail and mallard ducks. Toxicity to terrestrial invertebrates is characterized using a standard set of studies in honey bees. The NOEC used in this risk assessment is taken as  $50 \mu g$ /bee.

The assessment of potential effects in plants is based on standard toxicity studies required for pesticide registration involving pre-emergence and post-emergence exposures. In seedling emergence assays, very high concentrations – i.e., 10 lb a.i./acre or about 7.5 lbs a.e./acre – will modestly inhibit seed germination in both monocots and dicots. The NOEC for seed germination is 4.5 lb a.e./acre in both monocots and dicots. This value is used to assess the consequences of off-site movement of glyphosate in runoff. Glyphosate appears to be more toxic in vegetative vigor assays – i.e., direct application to the foliage of growing plants. The lowest reported NOEC for growth in standard bioassays required for registration is 0.035 lb a.e./acre. The highest reported NOEC for growth is 0.56 lb a.e./acre. This range of values for sensitive and relatively insensitive species is used to assess the consequences of off-site drift of glyphosate.

The dose-response assessment for fish is substantially complicated by information indicating that some fish species such as salmonids are more sensitive to glyphosate than other species of fish and by information indicating that some surfactants are very toxic to fish and may substantially increase to the toxicity of glyphosate to fish. These factors are further complicated by gaps in the available data. Given the apparently high sensitivity of some salmonids to glyphosate, it would be desirable to have a life cycle toxicity study or at least an egg-and-fry study available on salmonids. In addition, given the apparently high toxicity of surfactant formulations compared to technical grade glyphosate, a life cycle toxicity study on at least one formulation containing a toxic surfactant would be desirable. Such studies, however, are not available. Consequently, an approximation method commonly used is mixtures risk assessment (the relative potency method) is employed to estimate a chronic NOEC of 2.57 mg/L for technical grade glyphosate in sensitive species of fish based on an observed NOEC value of 25.7 mg/L in tolerant species of fish. Similarly, NOEC values for glyphosate formulations containing toxic surfactants are estimated at 0.36 mg/L for sensitive species and 0.64 mg/L for tolerant species. A similar approach is used estimate the potential for acute effects based on 96-hour  $LC_{50}$  values.  $LC_{50}$  values rather than data on sublethal effects are used to characterize risks from acute exposures because most of the data on sublethal effects are based on very short-term exposures to concentrations in the range of 96-hour  $LC_{50}$  values. Most of the available toxicity data suggest that amphibians are no more sensitive to glyphosate than fish. Consequently, a separate dose-response assessment for amphibians is not conducted in this risk assessment.

The issues in the dose-response assessment for aquatic invertebrates are very similar to those encountered in the dose-response assessment for fish. There is sufficient data to assert that some glyphosate formulations that contain toxic surfactants may be much more toxic to aquatic invertebrates than technical grade glyphosate. There is only one chronic study on technical grade glyphosate and no chronic studies on glyphosate formulations. Similar to the approach used in the dose-response assessment for fish, a chronic NOEC of 50 mg/L for technical grade glyphosate is used to estimate a chronic NOEC of 0.7 mg/L for glyphosate formulations containing toxic

surfactants. The potential for acute effects in aquatic invertebrates are based on  $LC_{50}$  values of 780 mg/L for technical grade glyphosate and 11mg/L for glyphosate formulations containing toxic surfactants.

Glyphosate appears to be about equally toxic to both algae and aquatic macrophytes. In terms of growth inhibition, the NOEC of 3 mg/L in duckweed is used to characterize risk due to inhibition. At lower concentrations – i.e., in the range of 0.002 mg/L to 0.3 mg/L or higher – stimulation of algal growth may be a more common response and has been noted in several studies.

### 4.3.2. Toxicity to Terrestrial Organisms.

**4.3.2.1.** *Mammals*– As summarized in the dose-response assessment for the acute and chronic human health risk assessment (Section 3.3.3.), the NOAEL in experimental mammals is taken as 175 mg/kg with an associated LOAEL of 350 mg/kg.

The application of these NOAEL and LOAEL values to small rodents is clearly appropriate, since the NOAEL and LOAEL come from a study in rabbits (Rodwell et al. 1980b). Ecological risk assessments, however, are intended to encompass a wide range of mammalian species, from very small animals such as mice and voles to large mammals such as deer. For many chemicals, systematic differences in species sensitivity are apparent and generally indicate that small animals are less sensitive (i.e., have higher toxicity values) than large animals. This is not the case for glyphosate. Toxicity values for rats are very similar to the toxicity estimates in humans. For example, as discussed in Section 3.3.3, the threshold for toxicity based on an analysis of lethality data is about 445 mg/kg, virtually identical to the LOAEL of 350 mg/kg in rabbits from the study by Rodwell et al. (1980b). The estimated  $LD_{50}$  in humans is approximately 3,000 mg/kg which is consistent with oral  $LD_{50}$  values of glyphosate in mammals which range from approximately 2,000 to 6,000 mg/kg (Williams et al. 2000). Thus, for the ecological risk assessment, NOAEL of 175 mg/kg is used directly for both small and large mammals.

The 175 mg/kg NOAEL and 350 mg/kg LOAEL values are used for both the acute and chronic risk assessments. This approach is taken because of the lack of a substantial dose-duration or dose-severity relationship for glyphosate. It may be argued that this approach is somewhat conservative in that the 175 mg/kg/day NOAEL is based on a teratology study in rabbits involving a 21-day exposure period – i.e., days 6-27 of gestation – in which the compound was administered by gavage. Most acute exposure scenarios considered in this risk assessment will involve peak exposures that will occur over a much shorter period – i.e., a 1-day maximum concentration in water or on vegetation. Shorter-term toxicity values that might be used, however, would have very little impact on the risk characterization. As summarized above, the threshold for acute lethality is about 445 mg/kg, virtually identical to the LOAEL of 350 mg/kg in rabbits from the study by Rodwell et al. (1980b). Because of concern for nonlethal adverse effects, the estimated NOAEL for lethality would not be an appropriate basis for risk characterization.

**4.3.2.2.** *Birds* – As noted in Section 4.1.2.2, glyphosate has been classified by the U.S. EPA/OPP 1993c as no more than slightly toxic to birds. As an index of potential toxicity from acute exposure, the U.S. EPA/OPP 1993c uses the gavage study by Fink et al. (1978) in which the  $LD_{50}$  was >2000 mg/kg in bobwhite quail. The more recent studies by Palmer and Beavers (1997a,b) indicate five day dietary  $LC_{50}$  values of greater than 5620 ppm in both bobwhite quail and mallard ducks. These dietary values are actually an NOEC in that no mortality or signs of toxicity were observed in any test animals.

For longer-term effects, U.S. EPA/OPP 1993c uses the dietary NOAELs of 1000 ppm in bobwhite quail (Fink 1975) and mallard ducks (Fink and Beavers 1978). Both of these studies were assays for reproductive toxicity, a relevant and sensitive endpoint for the ecological risk assessment. In this risk assessment, the acute dietary studies by Palmer and Beavers (1997a,b) will be used to assess the effects from acute exposures. For longer term exposures, the reproductive NOAEL of 1000 ppm (Fink 1975; Fink and Beavers 1978) will be used in the risk characterization.

The dietary concentrations will be converted to doses expressed as mg/kg body weight. This approach is taken because the direct use of dietary concentrations from laboratory studies may be under-protective. Laboratory diets generally involve the use of dry food. Dry laboratory chow usually has a higher caloric content than food consumed in the wild, if only because most food consumed in the wild has a high water content. In addition, most reported concentrations of a pesticide in environmental samples are given on a wet (natural) weight rather than a dry (dedicated) weight basis. Consequently, animals tend to eat greater amounts of food in the wild than they do under laboratory conditions (U.S. EPA/ORD 1993) and thus ingested doses expressed as mg/kg bw/day will be higher in free living animals than in laboratory animals for a fixed concentration in food.

Because of these relationships, Forest Service risk assessments use doses expressed as mg/kg body weight for both the exposure and dose-response assessments. As detailed in the worksheets, information on caloric requirements and caloric values of different foods is used to estimate the amount of a particular food that an animal will use.

Based on average measured food consumption and body weight from other laboratory toxicity studies on mallard ducks and pheasant, the daily food consumption rates of the birds are approximately 10% to 20% of the body weight. Taking a conservative value of 10% (i.e., a value that leads to the lowest estimate of dose), the 1000 ppm benchmark dietary concentration cited by U.S. EPA corresponds to a daily dose of 100 mg/kg bw.

For the current risk assessment, the dose of 100 mg/kg bw will be used as a NOAEL for characterizing chronic risks to for birds. It should be noted that this dose is very close to the NOAEL of 175 mg/kg used for mammals and is consistent with the apparent lack of variability in the toxicity of glyphosate among species. As in the assessment for mammals, this NOAEL is based on a repeated dose study for reproductive effects but will be applied to both acute and

longer-term exposures. The acute NOAEL will be taken as 562 mg/kg bw from the acute dietary studies by Palmer and Beavers (1997a,b).

**4.3.2.3.** *Terrestrial Invertebrates* – As discussed in Section 4.1.2.3, a standard set of studies are available on the toxicity of glyphosate to honey bees (Palmer and Beavers 1997c; Palmer and Krueger, 2001a; Palmer and Krueger, 2001b). Palmer and Krueger (2001a) report an NOEC of 100 µg/bee and the U.S. EPA/OPP (1993c) classifies glyphosate as practically non-toxic based on an LD<sub>50</sub> of >100 µg/bee. The NOEC used in this risk assessment, however, will be taken as 50 µg/bee. As detailed in Section 4.1.2.3, the dose of 100 µg/bee from the study by Palmer and Krueger (2001a) was associated with 5% mortality (3/60) and this response was statistically significant when untreated and solvent controls are pooled.

Taking the NOEC of 50 µg/bee and using a body weight of 0.093 g for the honey bee (USDA/APHIS 1993), the 50 µg/bee dose corresponds to about 540 mg/kg bw [0.050 mg/0.000093 kg = 537.6 mg/kg]. This value will be used in the risk characterization for assessing effects of direct contact on terrestrial invertebrates.

Given the large number of species of terrestrial invertebrates, the use of data from a single species for the risk characterization obviously leads to uncertainty in the risk assessment. As noted in Section 4.1.2.3, several additional studies are available on other terrestrial invertebrates. These studies, however, cannot be used quantitatively in the dose-response assessment either because of the way in which exposures were conducted or characterized. Nonetheless, they provide information that can be used in the risk characterization and these studies are discussed further in Section 3.4.

**4.3.2.4.** Soil Microorganisms. As noted in Section 4.1.2.4, soil microorganisms posses the shikimate pathway and a number of studies have demonstrated inhibition of microbial growth in laboratory culture. This is consistent with transient decreases in the populations of soil fungi and bacteria in the field after the application of glyphosate at application rates that are substantially less than those used in Forest Service programs – i.e., 0.54 kg/ha or about 0.5 lbs/acre from Chakravarty and Chatarpaul (1990). As also discussed in Section 4.1.2.4, several field studies note an increase rather than decrease in soil microorganisms or microbial activity, including populations of fungal plant pathogens, in soil after glyphosate exposures. While the mechanism of this apparent enhancement is unclear, it is plausible that glyphosate treatment resulted in an increase in the population of microorganisms in soil because glyphosate was used as a carbon source and/or treatment with glyphosate resulted in increased nutrients for microorganisms in the soil secondary to damage to plants.

**4.3.2.5.** *Terrestrial Plants (Macrophytes)* – As discussed in Section 4.1.2.5., standard toxicity studies required for pesticide registration are available on pre-emergence and post-emergence exposures. In seedling emergence assays, very high concentrations – i.e., 10 lb a.i./acre or about 7.5 lbs a.e./acre – will modestly inhibit seed germination in both monocots and dicots (Bohn 1987). The NOEC for seed germination is 4.5 lb a.e./acre in both monocots and dicots (Everett

et al. 1996a; Willard 1996). This value is used in Worksheet G04 to assess the consequences of off-site movement of glyphosate in runoff.

Glyphosate appears to be more toxic in vegetative vigor assays – i.e., direct application to the foliage of growing plants. The lowest reported NOEC for growth in standard bioassays required for registration is 0.035 lb a.e./acre, reported for tomato and radish in the study by Chetram and Lucash (1992). The highest reported NOEC for growth is 0.56 lb a.e./acre for ryegrass, corn, and onions, also from the study by Chetram and Lucash (1992). This range of values for sensitive and relatively insensitive species is very similar to the range of LC<sub>5</sub> values reported in the vegetative vigor assay of Everett et al. (1996b). The report by Al-Khatib and Peterson (1999) that transient visual injury occurs in soybeans at an application rate of about 0.03 lb a.e./acre is not consistent with the study by Chetram and Lucash (1992), in which no visual damage to soybeans was evident at 0.07 lb a.e./acre. Nonetheless, variations among studies is not uncommon and the study by Al-Khatib and Peterson (1999) appears to have been well-conducted.

In terms of the risk assessment, growth is a more relevant functional endpoint and will be used in preference to transient visual damage for the quantitative characterization of risk. Thus, as indicated in Worksheet G05, the NOEC values for sensitive and tolerant species are taken from the study by Chetram and Lucash (1992).

As also noted in Section 4.1.2.5, exposures substantially above the NOEC of 0.07 lb/acre - i.e., in the range of 0.7 lbs/acre – may have long term impacts on bryophyte and lichen communities (Newmaster et al. 1999). This is not a highly sensitive endpoint compared to the much lower NOEC values used above for the quantitative dose-response assessment.

# 4.3.3. Aquatic Organisms.

**4.3.3.1.** Fish – As detailed in Section 4.1.3.1.2, U.S. EPA/OPP (1993c) classified technical grade glyphosate as non-toxic to practically non-toxic in freshwater fish and  $LC_{50}$  values for glyphosate are in the range of 70 to 170 mg/L. In addition, the U.S. EPA/OPP (1993c) used the NOEC of 25.7 mg/L from life cycle toxicity study on technical grade glyphosate using fathead minnow and concluded that: "technical glyphosate should not cause acute or chronic adverse effects to aquatic environments. Therefore, minimal risk is expected to aquatic organisms from the technical glyphosate".

The selection of the toxicity values by U.S. EPA/OPP (1993c), however, does not explicitly address the higher toxicity of some glyphosate formulations containing surfactants nor does it address the higher sensitivity of some species of fish to technical grade glyphosate. As discussed in Section 4.1.3.1.3, some formulations of glyphosate contain surfactants which are highly toxic to fish and the 96-hour  $LC_{50}$  values for these formulations can be in the range of 1 mg/L, substantially below the NOEC for glyphosate in the fathead minnow life cycle toxicity study. In terms of sensitive species, trout and other salmonids have much lower  $LC_{50}$  than those cited by

U.S. EPA/OPP (1993c), with the lowest  $LC_{50}$  value for salmonids of 10 mg glyphosate/L, for trout in soft-water (Table 4-2).

The data necessary to address these issues directly are not available. Given the apparently high sensitivity of some salmonids to glyphosate, it would be desirable to have a life cycle toxicity study or at least an egg-and-fry study available on salmonids. In addition, given the apparently high toxicity of surfactant formulations compared to technical grade glyphosate, a life cycle toxicity study on at least one formulation containing a toxic surfactant would be desirable.

In the absence of these types of studies, the relative potency method is the only remaining approach to assessing the potential consequences of longer-term exposures of more toxic formulations to more sensitive species. As discussed in U.S. EPA/ORD (2000), the relative potency method involves the assumption that the ratio of toxicity values are equal among differing bioassays. As applied to the current problem, the relative potency method involves the assumption that the ratio of toxic for glyphosate and glyphosate formulations can be used to assess the chronic NOEC for a glyphosate formulation.

Specifically, the LC<sub>50</sub> values of a glyphosate formulation containing a surfactant to the fathead minnow is 2.3 (1.9-2.8) mg/L (U.S. EPA/OPP 1993c, p. 42). The LC<sub>50</sub> value of technical grade glyphosate (96.7%) to the fathead minnow is 97 (79-120) mg/L (U.S. EPA/OPP 1993c, p. 40). Using the central estimates, the formulated product is more toxic than technical grade glyphosate by a factor of about 40 [97 mg/L  $\div$  2.3 mg/L = 42.2]. Taking the life cycle NOEC of 25.7 mg/L for technical grade glyphosate in the fathead minnow and using the relative potency method, the life cycle NOEC for the formulation is estimated at 0.64 mg/L [25.7 mg/L  $\div$  40].

The issue of species sensitivity may be addressed in a similar manner. As noted above, the lowest  $LC_{50}$  value for technical grade glyphosate in salmonids of 10 mg/L – rainbow trout in soft-water (Wan et al. 1989 as summarized in Table 4-2 of this risk assessment). Thus, salmonids may be more sensitive than fathead minnows to technical grade glyphosate by a factor of about 10 [97 mg/L  $\div$  10 mg/L]. Taking the life cycle NOEC of 25.7 mg/L for technical grade glyphosate in the fathead minnow and using the relative potency method for species sensitivity, the life cycle NOEC for technical grade glyphosate in rainbow trout is estimated at about 2.57 mg/L [25.7 mg/L  $\div$  10].

A similar approach may be taken to estimate the life cycle NOEC for trout. As discussed above, the life cycle NOEC for the surfactant formulation in fathead minnow is estimated at 0.64 mg/L. The lowest  $LC_{50}$  value in the fathead minnow is 2.3 mg/L. In salmonids, the lowest reported  $LC_{50}$  value for a surfactant formulation is 1.3 mg/L (U.S. EPA/OPP 1993c, rainbow trout, p. 42). This is lower than any of the  $LC_{50}$  values for surfactant formulations of glyphosate published in the open literature. Thus, for surfactant containing formulations, the sensitivity of salmonids relative to minnows is a factor of about 1.8 [2.3 mg/L  $\div$  1.3 mg/L= 1.77]. Thus, the NOEC for the surfactant formulation in salmonids is estimated at 0.36 mg/L [0.64 mg/L  $\div$  1.8 = 0.355].

Thus, the following chronic NOEC values are used in Worksheet G03 to characterize risk to fish:

typical fish, technical grade glyphosate:	25.7	mg/L [observed]
sensitive fish, technical grade glyphosate:	2.57	mg/L [estimated]
typical fish, glyphosate with surfactant:	0.64	mg/L [estimated]
sensitive fish, glyphosate with surfactant:	0.36	mg/L [estimated]

For acute toxicity, the data on sub-lethal effects, summarized in Section 4.1.3.1.4, could be considered but reported concentrations are generally above the  $LC_{50}$  values. Most of the studies summarized in Section 4.1.3.1.4 involve very short-term exposures to relatively high concentrations and are focused on understanding the mechanism of action of glyphosate. These studies, however, are not appropriate for the acute dose-response assessment.

As an alternative, the  $LC_{50}$  values will be used to characterize the risk of observing mortality in fish associated with peak concentrations of glyphosate. While the use of  $LD_{50}$  and  $LC_{50}$  values is generally avoided, the use of 96-hour  $LC_{50}$  values to assess the risks of peak exposure is inherently conservative in that most peak exposures will occur for a period of time much shorter than 96-hours. Thus, the following 96-hour  $LC_{50}$  values are used for characterizing the risks of short-term exposures in Worksheet G03:

typical fish, technical grade glyphosate:	97	mg/L
sensitive fish, technical grade glyphosate:	10	mg/L
typical fish, glyphosate with surfactant:	2.3	mg/L
sensitive fish, glyphosate with surfactant:	1.3	mg/L

The application of the relative potency method to the available toxicity data in fish is attended by substantial uncertainties. By definition, the relative potency method is used only when significant information is lacking and no better methods are available. The assumption that relative potencies are constant across species and different endpoints – i.e.,  $LC_{50}$  values and chronic NOECs – cannot be supported by data because no such data are available. If the data were available to support this assumption, the data would be used in preference to the relative potency method. In addition, the application of the relative potency method to estimating the chronic toxicity of glyphosate-surfactant mixtures assumes that the surfactant will coexist with glyphosate and exert an influence on the toxicity of glyphosate as environmental dissipation and degradation occur. This is not likely to be the case and the use of the relative potency method for chronic toxicity of glyphosate-surfactant mixtures may be grossly conservative. Nonetheless, in the absence of data on the chronic toxicity of surfactant formulations to potentially sensitive fish species, this approach will be used in the current risk assessment.

**4.3.3.2.** *Amphibians* – No separate dose-response assessment is conducted in this risk assessment for amphibians. As detailed in Section 4.1.3.2, glyphosate and glyphosate formulations have been tested in a number of different bioassays with amphibians and there is no indication that glyphosate or glyphosate formulations induce deformities in amphibians. Most of the available

toxicity data suggest that amphibians are no more sensitive to glyphosate than fish. The report by Smith (2001) does suggest a greater sensitivity of tadpoles to a specific formulation. However, this study is not supported by the other available amphibian studies on glyphosate and glyphosate formulations. In addition, this study involves a formulation that is not used in Forest Service programs.

**4.3.3.3.** Aquatic Invertebrates – The issues in the dose-response assessment for aquatic invertebrates are very similar to those encountered in the dose-response assessment for fish (Section 4.3.3.1). There is sufficient data to assert that some glyphosate formulations that contain toxic surfactants may be much more toxic to aquatic invertebrates than technical grade glyphosate. There is only one chronic study on technical grade glyphosate and no chronic studies on glyphosate formulations.

One quantitative difference, however, involves the apparent magnitude of the differences in toxicity among technical grade glyphosate and various glyphosate formulations. The 48-hour  $LC_{50}$  value for technical grade glyphosate to *Daphnia magna* is listed in U.S. EPA (1993c, p. 45) as 780 mg/L. The 48-hour  $LC_{50}$  values for *Daphnia magna* of various glyphosate formulations, however, range from 3 mg/L (surfactant not specified) to >1000 mg/L (glyphosate with X-77 surfactant). The one chronic life cycle toxicity study with *Daphnia magna* using technical grade glyphosate reports an MATC of between 50 and 96 mg/L, similar to the NOEC of 25.7 mg/L in the life-cycle study in fish.

The application of the relative potency method to the invertebrate data on surfactant formulations of glyphosate can be done as with fish, as detailed in Section 4.3.3.1. However, the high variability in toxicity of the surfactant formulations to daphnids must be acknowledged.

As summarized in Appendix 3c, the formulations of glyphosate that are registered for forestry uses have a wide but narrower range of toxicity than the range reported in U.S. EPA (1993c). Based on the data presented in Appendix 3c, the acute toxicity of glyphosate formulations to *Daphnia magna* ranges from 11 mg/L (e.g., Roundup Pro, Roundup Pro Concentrate, Roundup UltraMax) to 930 mg/L (e.g., Aqua Neat, Aquatmaster).

The 48-hour  $LC_{50}$  value for technical grade glyphosate to *Daphnia magna* will be taken as 780 mg/L. Recognizing that some surfactant formulations may be at least somewhat less toxic than technical grade glyphosate, it will be assumed that formulations summarized in Appendix 3c that have a low toxicity – i.e., 930 mg/L – are either technical grade glyphosate or glyphosate with a surfactant of low toxicity. It will be assumed that the formulations with a substantially lower  $LC_{50}$  contain a toxic surfactant. The extremely toxic product reported in U.S. EPA/OPP (1993c) that has an  $LC_{50}$  value of 3 mg/L will not impact the dose-response assessment because none of the products that might be used in Forest Service programs (Appendix 3c) have an  $LC_{50}$  value that is this low.

Thus, the relative potency of the most toxic formulations that might be used in Forest Service programs is estimated as 71 [780 mg/L  $\div$  11 mg/L = 70.91] and the MACT for the chronic toxicity of these formulations is estimated as 0.7 mg/L [50 mg/L  $\div$  71]. Consequently, the values used to characterize risk to aquatic invertebrates in Worksheet G03 are:

Acute $LC_{50}$ , non-toxic formulation:	780	mg/L
Acute $LC_{50}$ , toxic formulation:	11	mg/L
Chronic NOEC, non-toxic formulation:	50	mg/L
Chronic NOEC, toxic formulation:	0.7	mg/L

It should be noted that the toxicities on other intermediate formulations could be calculated in the same manner as above. The potential need for such calculations is discussed further in Section 3.4 (Risk Characterization).

**4.3.3.4.** Aquatic Plants – As discussed in Section 3.1.3.4 and detailed in Appendix 11, glyphosate appears to be about equally toxic to both algae and macrophytes. A greater complication in the characterization of ecological effects may involve the enhancement of algal populations at low concentrations of glyphosate. It is unclear whether this is a hormetic effect or simply a stimulation of algal growth due to the utilization of glyphosate as a nutrient source by algae.

In terms of growth inhibition, the NOEC of 3 mg/L in duckweed (Smyth et al. 1996a) will be used to characterize risk due to inhibition. At lower concentrations – i.e., in the range of 0.002 mg/L to 0.3 mg/L or higher – stimulation of algal growth may be a more common response and has been noted in several studies (Austin et al. 1991; Smyth et al. 1996d; Wong 2000).

#### 4.4. RISK CHARACTERIZATION

**4.4.1. Overview.** The current risk assessment for glyphosate generally supports the conclusions reached by U.S. EPA: *Based on the current data, it has been determined that effects to birds, mammals, fish and invertebrates are minimal.* At the typical application rate of 2 lbs a.e./acre, none of the hazard quotients for acute or chronic scenarios reach a level of concern even at the upper ranges of exposure for terrestrial organisms. For the application rate of 7 lbs a.e./acre, central estimates of the hazard quotients somewhat exceed the level of concern for the direct spray of a honey bee. That the upper range of the hazard quotients, the level of concern is exceeded modestly in acute scenarios for a large mammal consuming contaminated vegetation and a small bird consuming insects. In the chronic exposure scenarios, the hazard quotient for a large bird concern by a factor of about 3. As with all longer term exposure scenarios involving the consumption of contaminated vegetation – i.e., vegetation directly sprayed at the highest application rate – would reduce and perhaps eliminate the possibility of any animal actually consuming this vegetation over a prolonged period.

For relatively tolerant nontarget species of plants, there is no indication that glyphosate is likely to result in damage at distances as close as 25 feet from the application site. For sensitive species at the upper range of application rates, there is a modest excursion about the NOEC at offsite distances of 100 feet or less. It should be noted, however, that all of these drift estimates are based on low-boom ground sprays. Many applications of glyphosate are conducted by directed foliar applications using backpacks. In such cases, little if any damage due to drift would be anticipated. Nontarget terrestrial plants are not likely to be affected by runoff of glyphosate under any conditions.

The primary hazards to fish appear to be from acute exposures to the more toxic formulations. At the typical application rate of 2 lbs a.e./acre, the hazard quotients for the more toxic formulations at the upper ranges of plausible exposure indicate that the  $LC_{50}$  values for these species will be not reached or exceeded under worst-case conditions. At an application rate of 7 lbs a.e./acre, the acute exposures are estimated to slightly exceed the  $LC_{50}$  value for typical species and exceed the  $LC_{50}$  value for sensitive species by a factor of about 2. In these worst-case scenarios, the exposure estimates are based on a severe rainfall (about 7 inches over a 24 hour period) in an area where runoff is favored – a slope toward a stream immediately adjacent to the application site. This is a standard worst-case scenario used in Forest Service risk assessments to guide the Forest Service in the use of herbicides. This risk characterization strongly suggests that the use of the more toxic formulations near surface water is not prudent.

The use of less toxic formulations result in acute hazard quotients that do not approach a level of concern for any species. Nonetheless, the hazard quotient of 0.08 for sensitive species at an application rate of 2 lbs/acre is based on an  $LC_{50}$  value rather than a NOEC. Thus, the use of glyphosate near bodies of water where sensitive species of fish may be found (i.e., salmonids) should be conducted with substantial care to avoid contamination of surface water. Concern for

potential effects on salmonids is augmented by the potential effects of low concentrations of glyphosate on algal populations.

The likelihood of direct acute toxic effects on aquatic invertebrates or longer term direct effects on any fish species seems extremely remote based on central estimates of the hazard quotient and unlikely base on upper ranges of the hazard quotient. The hazard quotient of 0.044 for longer term effects of the more toxic formulations on sensitive fish is based on an estimated NOEC and thus is not, in itself, of substantial concern. Aquatic plants appear to be somewhat less sensitive to glyphosate than the most sensitive aquatic animals. There is no indication that adverse effects on aquatic plants are plausible.

# 4.4.2. Terrestrial Organisms

**4.4.2.1.** *Terrestrial Animals*– The quantitative risk characterization for terrestrial animals is summarized in Table 4-4. These hazard quotients are calculated by dividing the exposure assessments summarized in Table 4-1 by the toxicity values specified at the bottom of Table 4-4. As indicated at the bottom of Table 4-4, all of the hazard quotients are based on NOEC values. In addition, all hazard quotients are based on an application rate of 2 lbs a.e./acre, the typical application rate use in Forest Service programs. At this application rate, the level of concern is unity – i.e. at a hazard quotient of 1.0 or less, there is no plausible basis for asserting that adverse effects are likely to occur. As indicated in Section 2.4, the upper range of the application rate that is proposed in Forest Service programs is 7 lbs a.e./acre. At this application rate, the level of concern for the hazard quotients in Table 4-4 is about 0.3 [2 lbs a.e./acre  $\div$  7 lbs a.e./acre = 0.2857]. Levels of concern for other application rates may be similarly calculated.

Based on the typical application rate of 2 lbs a.e./acre, none of the hazard quotients for acute or chronic scenarios reach a level of concern even at the upper ranges of exposure. This is consistent with the risk characterization given by U.S. EPA/OPP (1993c, p. 53): *Based on the current data, it has been determined that effects to birds, mammals, fish and invertebrates are minimal.* 

For the application rate of 7 lbs a.e./acre, central estimates of the hazard quotients somewhat exceed the level of concern -i.e., the hazard quotient exceeds 0.3 - for the direct spray of a honey bee. That the upper range of the hazard quotients, the level of concern is exceeded modestly in acute scenarios for a large mammal consuming contaminated vegetation and a small bird consuming insects. In the chronic exposure scenarios, the hazard quotient for a large bird consuming contaminated vegetation on site exceeds the level of concern by a factor of about 2.8  $[0.83 \div 0.3]$ .

The direct spray of a honey bee at an application rate of 7 lbs a.e./acre corresponds to a dose of 1120 mg/kg bw. It is unclear if this would be associated with detectable toxic effects. Based on the study by Palmer and Krueger (2001a), a dose of 100  $\mu$ g/bee, corresponding to about 1080 mg/kg bw, was associated with 5% mortality (3/60). It should also be noted that this risk characterization applies only to bees that are directly sprayed at the maximum application rate (7

lbs/acre) and does not consider the effects of foliar interception. Thus, under actual field conditions, substantial mortality in bees would not be expected.

For a large mammal, dose associated with the acute consumption of contaminated vegetation at an application rate of 7 lbs a.e./acre is about 340 mg/kg (97.1 mg/kg from Table 4-3 × 7 lbs/acre  $\div$  2 lbs/acre). This dose is below the LD<sub>50</sub> value for mammals (2,000 to 6,000 mg/kg) by a factor of about 6 to 18. Thus, it is not likely that substantial mortality would be observed. For mammals, however, the NOEL of 175 mg/kg is associated with a LOAEL of 350 mg/kg, nearly identical with the upper range of exposure – i.e., 340 mg/kg. As detailed in Section 3.3.2 (the dose-response assessment for the human health risk assessment), the dose of 350 mg/kg/day results in some mortality in pregnant rabbits (Rodwell et al. 1980b). Thus, while *substantial* mortality would not be anticipated – i.e., the exposure is well below the LD<sub>50</sub> – mortality in some animals would be plausible.

For a small bird consuming contaminated insects after the application of 7 lbs a.e./acre, the estimated dose is about 787 mg/kg [225 mg/kg from Table 4-3  $\times$  7 lbs/acre  $\div$  2 lbs/acre]. This dose is well below the LD<sub>50</sub> but somewhat higher than the acute NOAEL of 562 mg/kg. As discussed in Section 4.2.2.3, this exposure assessment may be extremely conservative. The residue rates from small insects are used rather than the residue rates from large insects and this makes a substantial difference. In addition, as detailed in Worksheet F14, this exposure scenario assumes that 100% of the insects consumed by the bird on the day of exposure (i.e., no dissipation is assumed) were directly sprayed.

In the longer term exposure scenario for a large bird consuming vegetation at the application site, the average daily dose is about 291 mg/kg/day [83.2 mg/kg  $\times$  7 lbs/acre  $\div$  2 lbs/acre]. This is a standard exposure scenario used in all Forest Service risk assessments. Nonetheless, as with all longer term exposure scenarios involving the consumption of contaminated vegetation, the plausibility of this exposure scenario is limited because damage to the treated vegetation – i.e., vegetation directly sprayed at the highest application rate – would reduce and perhaps eliminate the possibility of any animal actually consuming this vegetation over a prolonged period. On the other hand, as discussed in Section 4.2.2.3, residue rates estimated by Fletcher et al. (1994) and used in the exposure assessment are below some monitored values for glyphosate on herbaceous vegetation (Newton et al. 1994). Thus, while the longer term estimates of exposure may be overestimated by not considering damage to vegetation, this may be somewhat offset by an underestimate of initial residues.

Another factor that must be considered in the interpretation of these hazard quotients is the information from field studies. As discussed in Section 4.1.2.6, several field studies report effects on terrestrial animals that appear to be associated with changes in habitat and, in some cases, the effects appear to be beneficial.

**4.4.2.2.** *Terrestrial Plants*– The quantitative risk characterizations for terrestrial plants are summarized in Worksheet G04 for the offsite movement of glyphosate in runoff and Worksheet G05 for offsite movement of glyphosate by drift and wind erosion.

The runoff estimates are based on GLEAMS modeling using three different soils (clay, loam, and sand) at annual rainfall rates of 5 to 250 inches and using the typical application rate that the Forest Service is considering, 2 lb/acre. The toxicity index is based on the pre-emergence NOAEL of 4.5 lb/acre. As with the corresponding assessment of effects in terrestrial mammals, the level of concern is unity for an application rate of 2 lbs/acre and about 0.3 for an application rate of 7 lbs/acre. Based on this index of toxicity, plant species are not likely to be affected by runoff of glyphosate under any conditions.

Hazard quotients for offsite drift (Worksheet G05) are based on the NOEC value of 0.035 lb/acre for sensitive plant species as well as the NOAEL of 0.56 lb/acre for tolerant plant species, as detailed in Section 4.3.2.5. The estimates for offsite drift encompass plausible exposures attributable to wind erosion, as discussed in Section 4.2.2.4. For relatively tolerant species, there is no indication that glyphosate is likely to result in damage at distances as close as 25 feet from the application site. For sensitive species at the upper range of application rates, there is a modest excursion about the NOEC (hazard quotients of 0.3 to 1.1 with a level of concern of 0.3) at offsite distances of 100 feet or less. It should be noted, however, that all of these drift estimates are based on low-boom ground sprays. Many applications of glyphosate are conducted by directed foliar applications using backpacks. In such cases, little if any damage due to drift would be anticipated.

**4.4.3.** Aquatic Organisms. The quantitative risk characterization for aquatic species is summarized in Table 4-5. As in the corresponding tables for terrestrial species, the exposure component of the hazard quotients are based on the typical application rate of 2 lbs a.e./acre. Thus, the level of concern for this application rate is unity and the level of concern for the maximum application rate of 7 lbs a.e./acre is 0.3.

As discussed in previous sections of this risk assessment, some formulations of glyphosate are much more acutely toxic to fish and aquatic invertebrates than technical grade glyphosate or other formulations of glyphosate. This difference in acute toxicity among formulations appears to be due largely to the use of surfactants that are toxic to fish and invertebrates.

The primary hazards to fish appear to be from acute exposures to the more toxic formulations. As summarized in Table 4-5, the hazard quotients for the more toxic formulations at the upper ranges of exposure – i.e., 0.35 for typical species and 0.62 for sensitive species – indicate that the  $LC_{50}$  values for these species will be not reached or exceeded under worst-case conditions due to runoff after a high rainfall event into a small stream. At an application rate of 7 lbs a.e./acre, the acute exposures are estimated to slightly exceed the  $LC_{50}$  value for typical species and exceed the  $LC_{50}$  value for sensitive species by a factor of about 2. In these worst-case scenarios, the exposure estimates are based on a severe rainfall (about 7 inches over a 24 hour period) in an area
where runoff is favored – a slope toward a stream immediately adjacent to the application site. This, again, is a standard worst-case scenario used in Forest Service risk assessments to guide the Forest Service in the use of herbicides. Although this scenario is not likely to occur, the study by Trumbo (2002) found monitored concentrations of glyphosate at 0.85 mg/L, virtually identical to the modeled value of 0.8 mg/L at an application rate of 2 lbs a.e./acre. Thus, this risk characterization strongly suggests that the use of the more toxic formulations near surface water is not prudent.

The use of less toxic formulations result in hazard quotients that do not approach a level of concern for any species. Nonetheless, the hazard quotient of 0.08 for sensitive species at an application rate of 2 lbs/acre is based on an  $LC_{50}$  value rather than a NOEC. Thus, the use of glyphosate near bodies of water where sensitive species of fish may be found (i.e., salmonids) should be conducted with substantial care to avoid contamination of surface water.

Concern for potential effects on salmonids is augmented by the potential effects of low concentrations of glyphosate on algal populations. As discussed in Section 4.1.3.4, glyphosate may stimulate rather than inhibit algal growth at low concentrations. While this might not be classified as an adverse effect on algae, Austin et al. (1991) has suggested that this stimulation could contribute to eutrophication of waterways with salmonid populations. No field studies have been encountered in the literature suggesting that this effect has been observed. Nonetheless, the central estimate of the projected longer-term concentrations of glyphosate in water is 0.002 mg/L with an upper range of 0.056 mg/L at and application rate of 2 lbs a.e./acre. At an application rate of 7 lbs a.e./acre, the corresponding concentrations are about 0.007 mg/L to 0.2 mg/L. An increase in periphyton concentrations in artificial streams has been reported by Austin et al. (1991) at concentrations which encompass these ranges – i.e., 0.0019 - 0.2874 mg/L. This is supported by Wong (2000), who reported an increase in chlorophyll-a synthesis by *Scenedesmus quadricauda* at a concentration of 0.02 mg/L.

The likelihood of direct acute toxic effects on aquatic invertebrates or longer term direct effects on any fish species seems extremely remote based on central estimates of the hazard quotient and unlikely base on upper ranges of the hazard quotient. The hazard quotient of 0.044 for longer term effects of the more toxic formulations on sensitive fish is based on an estimated NOEC and thus is not, in itself, of substantial concern.

A very important limitation to this risk characterization, however, is that no chronic studies are available on the formulations that appear to have a very high acute toxicity. Thus, as detailed in Section 4.3.3.1 on fish and Section 4.3.3.3 on aquatic invertebrates, the relative potency method was used in an attempt to quantify potential chronic effects from exposure to for both sensitive species, primarily salmonids, as well as the chronic effects of formulations that contain the more toxic surfactants. While the relative potency method is the most reasonable approach that is apparent given the lack of more directly relevant data, it increases uncertainty in the risk characterization for longer term exposures.

Aquatic plants appear to be somewhat less sensitive to glyphosate than the most sensitive aquatic animals. There is no indication that adverse effects on aquatic plants are plausible. Unlike the case with aquatic animals, even short-term toxicity studies in aquatic plants use endpoints involving changes in population density. Thus, both the short-term and longer-term hazard quotients given in Table 4-3 can be legitimately used to characterize risk.

	Observed LC <sub>50</sub> values								
Species/Assay/Study	Glyphosate	POEA	Roundup <sub>b</sub>	ρ	Predicted LC <sub>50</sub>	Pred.÷ Obs.			
рН 6.5									
Rainbow trout	140	7.4	10.8	19	22	2.0			
Bluegills	140	1.3	6.0	108	3.1	0.5			
pH 7.2, 96 hr unless specified									
Midge larvae, 48 hr.	55	13	25	4.2	28	1.1			
Rainbow trout	140	2	11.8	70	6.5	0.6			
Fathead minnow	97	1.0	3.2	97	3.2	1.0			
Channel catfish	130	13	18	10	35	1.9			
Bluegills	140	3.0	7.1	47	9.5	1.3			
рН 9.5									
Rainbow trout	240	0.65	2.0	369	2.1	1.1			
Bluegills	220	1.0	2.6	220	3.3	1.3			

Table 4-1. Estimates of relative potency and toxicological interaction of glyphosate and POEA<sup>a</sup>

<sup>a</sup>Data from Folmar et al. (1979).

<sup>b</sup>Value reported by Folmar as mg a.i multiplied 1.42 to account for added mass of surfactant.

 $\rho = LC_{50}$  of glyphosate  $\div LC_{50}$  of POEA.

	Observed 96-hour LC <sub>50</sub> Values							
Species/Assay/Study	Glyphosate	POEA	Roundup <sup>a</sup>	ρ	Predicted LC <sub>50</sub>	Pred.÷ Obs.		
Soft Water pH 6.3								
Coho	27	4.6	32	5.9	10.9	0.34		
Chum	10	2.7	20	3.7	5.5	0.28		
Chinook	19	2.8	33	6.8	6.9	0.21		
Pink	14	4.5	33	3.1	8.5	0.26		
Rainbow	10	2	33	5	4.5	0.13		
		Soft Wa	ater pH 7.2					
Coho	36	3.2	27	11.3	8.8	0.33		
Chum	22	4.2	19	5.2	9.7	0.51		
Chinook	30	2.8	27	10.7	7.5	0.28		
Pink	23	2.8	31	8.2	7.2	0.23		
Rainbow	22	2.5	15	8.8	6.6	0.44		
Hard Water pH 8.2								
Coho	210	1.8	13	117	5.9	0.45		
Chum	202	1.4	11	144	4.6	0.41		
Chinook	220	1.7	17	129	5.6	0.32		
Pink	380	1.4	14	261	4.6	0.33		
Rainbow	220	1.7	14	129	5.6	0.40		

Table 4-2. Estimates of relative potency and toxicological interaction of glyphosate and POEA in five species of salmonids <sup>a</sup>

<sup>a</sup> Data from Wan et al. (1989)

<sup>b</sup> As reported by Wan et al. (1989) in units of mg product/L.

		Worksheet				
Scenario	Central	Central Lower Upper				
Acute/Accidental Exposures						
Direct spray						
small mammal, first-order absorption	4.75e-01	1.51e-01	1.15e+00	F01		
small animal, 100% absorption	4.85e+01	4.85e+01	4.85e+01	F02a		
bee, 100% absorption	3.21e+02	3.21e+02	3.21e+02	F02b		
Contaminated vegetation						
small mammal	8.57e-01	8.57e-01	2.11e+00	F03		
large mammal	3.44e+01	3.44e+01	9.71e+01	F10		
large bird	5.38e+01	5.38e+01	1.52e+02	F12		
Contaminated water						
small mammal, spill	2.66e+00	1.06e+00	5.32e+00	F05		
stream	5.86e-03	2.93e-04	1.17e-01	F06		
Contaminated insects						
small bird	7.50e+01	7.50e+01	2.25e+02	F14		
Contaminated fish						
predatory bird, spill	9.45e-01	1.89e-01	2.83e+00	F08		
Longer-term Exposures						
Contaminated vegetation						
small mammal, on site	4.69e-02	2.35e-02	2.31e-01	F04a		
off-site	4.74e-04	1.36e-04	4.31e-03	F04b		
large mammal, on site	5.65e+00	1.88e+00	5.32e+01	F11a		
off-site	1.90e-01	1.09e-01	9.94e-01	F11b		
large bird, on site	8.84e+00	2.95e+00	8.32e+01	F13a		
off-site	2.98e-01	1.71e-01	1.56e+00	F13b		
Contaminated water						
small mammal	2.93e-04	2.93e-05	2.34e-03	F07		
Contaminated fish						
predatory bird	1.04e-04	5.20e-06	1.25e-03	F09		

**Table 4-3:** Summary of Exposure Scenarios for Terrestrial Animals for an Application Rate of 2 lbs a.e./acre.

	Hazard Quotient <sup>2</sup>		
Scenario	Central	Lower	Upper
Acute/Accidental Exposures			
Direct spray			
small mammal, first-order absorption	3e-03	9e-04	7e-03
small animal, 100% absorption	3e-01	3e-01	3e-01
bee, 100% absorption	6e-01	6e-01	6e-01
Contaminated vegetation			
small mammal	5e-03	5e-03	1e-02
large mammal	2e-01	2e-01	6e-01
large bird	1e-01	1e-01	3e-01
Contaminated water			
small mammal, spill	2e-02	6e-03	3e-02
small mammal, stream	3e-05	2e-06	7e-04
Contaminated insects			
small bird	1e-01	1e-01	4e-01
Contaminated fish			
predatory bird, spill	2e-03	3e-04	5e-03
Longer-term Exposures			
Contaminated vegetation			
small mammal, on site	3e-04	1e-04	1e-03
off-site	3e-06	8e-07	2e-05
large mammal, on site	3e-02	1e-02	3e-01
off-site	1e-03	6e-04	6e-03
large bird, on site	9e-02	3e-02	8e-01
off-site	3e-03	2e-03	2e-02
Contaminated water			
small mammal	2e-06	2e-07	1e-05
Contaminated fish			
predatory bird	1e-06	5e-08	1e-05
Toxicity Indices <sup>3</sup>			
Acute toxicity value for	mammal - NOAEL	175	mg/kg
Chronic toxicity value for	mammal - NOAEL	175	mg/kg/day
Acute toxicity value	for bird - NOAEL	562	mg/kg
Chronic toxicity value f	for birds - NOAEL	100	mg/kg/day
Honey Bee Acu	te toxicity - NOEC	540	mg/kg

**Table 4-4:** Summary of quantitative risk characterization for terrestrial animals at an application rate of 2 lbs a.e./acre<sup>1</sup>

<sup>1</sup> See Table 4-1 for summary of exposure assessments.

<sup>2</sup>Estimated dose  $\div$  toxicity index

<sup>3</sup> See chemical specific worksheet notes for a discussion of the dose-response assessments.

Hazard Quotients	Central	Lower	Upper	Toxicity Value (mg/L)	End- point
Fish, Acute Exposures					
Typical species, less toxic formulation	4.1e-04	2.1e-05	8.2e-03	97	LC <sub>50</sub>
Sensitive species, less toxic formulation	4.0e-03	2.0e-04	8.0e-02	10	LC <sub>50</sub>
Typical species, more toxic formulation	1.7e-02	8.7e-04	3.5e-01	2.3	LC <sub>50</sub>
Sensitive species, more toxic formulation	3.1e-02	1.5e-03	6.2e-01	1.3	LC <sub>50</sub>
Fish, Chronic Exposures					
Typical species, less toxic formulation	7.8e-05	7.8e-06	6.2e-04	25.7	NOEC
Sensitive species, less toxic formulation	7.8e-04	7.8e-05	6.2e-03	2.57	NOEC
Typical species, more toxic formulation	3.1e-03	3.1e-04	2.5e-02	0.64	NOEC
Sensitive species, more toxic formulation	5.6e-03	5.6e-04	4.4e-02	0.36	NOEC
Aquatic Invertebrates, Acute Exposures					
Less toxic formulation	5.1e-05	2.6e-06	1.0e-03	780	LC <sub>50</sub>
Most toxic formulation	3.6e-03	1.8e-04	7.3e-02	11	LC <sub>50</sub>
Aquatic Invertebrates, Chronic Exposures					
Less toxic formulation	4.0e-05	4.0e-06	3.2e-04	50	NOEC
Most toxic formulation	2.9e-03	2.9e-04	2.3e-02	0.7	NOEC
Aquatic Plants					
Acute	1.3e-02	6.7e-04	2.7e-01	3	NOEC
Chronic	6.7e-04	6.7e-05	5.3e-03	3	NOEC
Exposures (mg/L)	Central	Lower	Upper	Works	heets
Acute	0.04	0.002	0.8	F06	Strea m
Longer-term	0.002	0.0002	0.016	F09	Pond

Table 4-5: Quantitative Risk Characterization for Aquatic Species at an Application Rate of 2 lbs a.e./acre.

See Sections 4.3.3.2, 4.3.3.3, and 4.3.3.4 for a discussion and derivation of the toxicity values for fish, invertebrates, and aquatic plants, respectively. See Worksheets F06 and F09 for the derivation of the concentrations in streams and ponds, respectively.

## **5. LIST OF WORKS CONSULTED**

APHIS (Animal and Plant Health Inspection Service, USDA). 1993. Nontarget Risk Assessment for the MEDFLY Cooperative Eradication Program. Dated February 1993. USDA/APHIS, Riverdale, MD.

ATSDR (Agency for Toxic Substances and Disease Registry. 2001. ATSDR ToxProfiles 2002 on CD-ROM, Available from ATSDR, Division of Toxicology, www.atsdr.gov.

Abdelghani AA; Tchounwou PB; Anderson AC; Sujono H; Heyer LR; Monkiedje A. 1997. Toxicity evaluation of single and chemical mixtures ofRoundup, Garlon-3A, 2-4,D, and Syndets surfactant to channel catfish (Ictaluruspunctatus), bluegill sunfish (Lepomis microchirus), and crawfish (Procambarus spp.). Environmental Toxicology and Water Quality. 12:237-243..

Acquavella JF; Weber JA; Cullen MR; Cruz OA; Martens MA; Holden LR; Riordan S; Thompson M; Farmer D. 1999a. Human ocular effects from self-reported exposures to Roundup herbicides. Hum Exp Toxicol 1999 Aug;18(8):479-86.

Acquavella J; Farmer D; Cullen MR. 1999b. A case-control study of non-hodgkin lymphoma and exposure to pesticides. Cancer. 86(4):729-31.

Adam A; Marzuki A; Abdul Rahman H; Abdul Aziz M. 1997. The oral and intratracheal toxicities of roundup and its components to rats. Vet Hum Toxicol. 39(3):147-51.

Ahn Y-J; Cho J-R; Kim Y-J; Yoo J-K; Lee J-O. 1997. Toxicity of the herbicide glufosinateammonium to tetranychus urticae (Acari: tetranychidae) under laboratory and field conditions. Pesticide Science. 51(4): 455-461.

Al-Khatib K; Peterson D. 1999. Soybean (Glycine max) response to simulated drift from selected sulfonylurea herbicides, dicamba, glyphosate, and glufosinate. Weed Technology. 13(2): 264-270.

Alberdi JL; Saenz ME; Di Marzio WD; Tortorelli MC. 1996. Comparative acute toxicity of two herbicides, paraquat and glyphosate, to Daphnia magna and D. spinulata. Bull Environ Contam Toxicol. 57(2): 229-35.

Albrechtsen HJ; Mills MS; Aamand J; Bjerg PL. 2001. Degradation of herbicides in shallow danish aquifers: an integrated laboratory and field study. Pest Manag Sci. 57(4):341-50.

Alferness PL; Wiebe LA. 2001. Determination by glyphosate and aminomethylphosphonic acid in crops by capillary gas chromatography with mass-selective detection: collaborative study. J AOAC Int. 84(3): 823-46.

Alibhai MF; Stallings WC. 2001. Closing down on glyphosate inhibition--with a new structure for drug discovery. Proc Natl Acad Sci USA. 98(6): 2944-6.

Allen RR; Fryrear DW. 1977. Limited tillage saves soil, water, and energy. ASAE Annual Meeting, NC State Univ., Raleigh, NC. June 26-29, 1977. 14 pp.

Anthelme F; Marigo G. 1998. Glyphosate uptake in catharanthus roseus cells: involvement of a plasma membrane redox system?. Pesticide Biochemistry and Physiology. 62(2): 73-86.

Anthony RG; Morrison ML. 1985. Influence of glyphosate herbicide on small mammal populations in western Oregon USA. Northwest Sci. 59(3): 159-168.

Anton FA; Cuadra LM; Gutierrez P; Laborda E; Laborda P. 1993. Degradational behavior of the pesticides glyphosate and diflubenzuron in water. Bull Environ Contam Toxicol. 51(6):881-8.

Anton FA; Laborda E; De Ariz M. 1994. Acute toxicity of the herbicide glyphosate to fish. Chemosphere. 28(4): 745-753.

Arbuckle TE; Sever LE. 1998. Pesticide exposures and fetal death: A review of the epidemiologic literature. Critical Reviews in Toxicology. 28(3): 229-270.

Arbuckle TE; Lin Z; Mery LS. 2001. An exploratory analysis of the effect of pesticide exposure on the risk of spontaneous abortion in an Ontario farm population. Environ Health Perspect. 109(8): 851-7.

Atkinson D. 1985. The toxicological properties of glyphosate - a summary. In: The Herbicide Glyphosate. London, England: Butterworth and Co. Ltd. pp. 127-133.

Austin AP; Harris GE; Lucey WP. 1991. Impact of an organophosphate herbicide (Glyphosate) on periphyton communities developed in experimental streams. Bull Environ Contam Toxicol. 47(1):29-35.

Bababunmi EA; Olorunsogo OO; Bassir O. 1978. Toxicology of glyphosate in rats and mice. Toxicol Appl Pharmacol 45(1): 319-320.

Bababunmi EA; Olorunsogo OO; Bassir O. 1979. The uncoupling effect of n(phosphonomethyl)glycine on isolated rat liver mitochondria. Biochem Pharmacol; 28 (6) 1979 925-928. (Cited in U.S. EPA/ODW 1992).

Baird DD; Upchurch RP; Homesley WB; Franz JE. 1971. Introduction of a new broadspectrum pobroadspectrum postemergence herbicide class with utility for herbaceous perennial weed control. Proc of the North Central Weed Control Conference. 26: 64-68. (Summarized in Franz, 1985).

Barbosa ER; Leiros Da Costa MD; Bacheschi LA; Scaff M; Leite CC. 2001. Parkinsonism after glycine-derivate exposure. Mov Disord. 16(3): 565-8.

Barclay J. 1987a. Product Chemistry To Support the Registration of Rodeo Herbicide. (53.8% Isopropylamine Salt of Glyphosate): Product Identity and Composition: Laboratory Project ID: MSL-6710: R.D. No. 805. Unpublished compilation prepared by Monsanto Agricultural Co. 36 p. MRID 40298901.

Barclay J. 1987b. Product Chemistry To Support the Registration of Rodeo Herbicide. (53.8% Isopropylamine Salt of Glyphosate): Analysis and Certification of Product Ingredients: Laboratory Project ID: MSL-6711: R.D. No. 805. Unpublished compilation prepared by Monsanto Agricultural Co. 58 p. MRID 40298902.

Barclay J. 1987c. Product Chemistry To Support the Registration of Rodeo Herbicide. (53.8% Isopropylamine Salt of Glyphosate): Physical and Chemical Characteristics: Laboratory Project ID: MSL-6712: R.D. No. 805. Unpublished study prepared by Monsanto Agricultural Co. 15 p. MRID 40298903.

Bariuan JV; Reddy KN; Wills GD. 1999. Glyphosate injury, rainfastness, absorption, and translocation in purple nutsedge (Cyperus rotundus). Weed Technology. 13 (1): 112-119.

Batt BD; Black JA; Cowan WF. 1980. The effects of glyphosate herbicide on chicken egg hatchability. Can J Zool. 58(10): 1940-2.

Bauer KH; Knepper TP; Maes A; Schatz V; Voihsel M. 1999. Analysis of polar organic micropollutants in water with ion chromatography-electrospray mass spectrometry. J Chromatogr A. 837(1-2): 117-28.

Bechtel C. 1993. Acute Inhalation Study of MON 65005 Herbicide. (in Rats): Lab Project Number: EHL 93053: ML-93-125: 1198. Unpublished study prepared by Monsanto Co. 49 p. MRID 43020903.

Bechtel C; Becktame S. 1997. Acute Inhalation Study (in Rats) of MON 58442: Lab Project Number: EHL 96196: ML-97-006: 1366. Unpublished study prepared by Monsanto Co. 52 p. (in Rats) of MON 58442: Lab Project Number: EHL 96196: ML-97-006: 1366. Unpublished study prepared by Monsanto Co. 52 p. MRID 44272304.

Beltran J; Gerritse RG; Hernandez F. 1998. Effect of flow rate on the adsorption and desorption of glyphosate, simazine and atrazine in columns of sandy soils. European J Soil Sci. 49 (1): 149-156.

Benova DK; Roupova I; Yagova A; Vuglenov A; Bineva M. 1989. Mutagenicity studies on six pesticides in mice. Environ Mol Mutagen. 14(Suppl 15): 19.

Bergvinson DJ; Borden JH. 1992. Enhanced woodpecker predation on the mountain pine beetle, dendroctonus ponderosae hopk., in glyphosate-treated lodgepole pines. Can Entomol. 124(1): 159-165.

Bernard S. 2002. Determination of N-Nitroso-Glyphosate and N-Nitroso-Iminodiacetic Acid in Five Batches of Glyphosate Technical: Lab Project Number: PR99/031. Unpublished study prepared by HELM. 42 p. MRID 45685903.

Bertheussen K; Yousef MI; Figenschau Y. 1997. A new sensitive cell culture test for the assessment of pesticide toxicity. J Environ Sci Health B. 32(2): 195-211.

Beyers DW. 1995. Acute toxicity of rodeo herbicide to Rio Grande silvery minnow as estimated by surrogate species: Plains minnow and fathead minnow. Arch Environl Contam Toxicol. 29(1): 24-26.

Bhatti MA; Al-Khatib K; Parker R. 1997. Wine grape (Vitis vinifera) response to fall exposure of simulated drift from selected herbicides. Weed Technology; 11(3): 532-536.

Bidwell J; Gorrie J. 1995. Acute Toxicity of a Herbicide (Roundup) to Selected Frog Species: Final Report. Unpublished study prepared by Curtin University of Technology. 13 p.. (Roundup) to Selected Frog Species: Final Report. Unpublished study prepared by Curtin University of Technology. 13 p. MRID 43839601.

Bidwell JR: Gorrie JR. 1995. Acute toxicity of a herbicide to selected frog species. Final Report. Department of Environmental Protection, Perth, Western Australia. Technical Series 79, June 1995. 14 p..

Blakley BR. 1997. Effect of roundup and tordon 202c herbicides on antibody production in mice. Vet Hum Toxicol. 39(4): 204-6.

Bleeke M. 2002. Magnitude of Glyphosate Residues in Roundup Ready Alfalfa Raw Agricultural Commodities Following Topical Application of Roundup Ultra Herbicide: Lab Project Number: MSL-16761: 00-63-R-7. Unpublished study prepared by Grayson Research, Monsanto Comany, a. MRID 45646001.

Blondell J. 1997. Epidemiology of pesticide poisoning in the United States with special reference to occupational cases. Occupational Medicine (Philadelphia). 12(2): 209-220.

Boateng JO; Haeussler S; Bedford L. 2000. Boreal plant community diversity 10 years after glyphosate treatment. Western Journal Applied Forestry. 15(1): 15-26.

Bohn J. 1987. An Evaluation of the Preemergence Herbicidal Ac- tivity of CP-70139: Lab Project ID: 056337. Unpublished study prepared by Monsanto Agricultural Co. 25 p. Cited in U.S. EPA/OPP 1993c. MRID 40159301.

Bolognesi C; Bonatti S; Degan P; Gallerani E; Peluso M; Rabboni R; Roggieri P; Abbondandolo A. 1997. Genotoxic activity of glyphosate and its technical formulation roundup. Journal of Agricultural and Food Chemistry. 45(5): 1957-1962.

Bonnette K. 1993a. Acute Oral Toxicity Study in Rats with MON 65005 Herbicide: Final Report: Lab Project Number: SB-93-126: 1198: 3044.441. Unpublished study prepared by Springborn Labs, Inc. 18 p.. MRID 43020902.

Bonnette K. 1993b. Primary Eye Irritation Study in Rabbits With MON 65005: Final Report: Lab Project Number: 3044.401: SB-92-510: 1198. Unpublished study prepared by Springborn Labs, Inc. 18 p. MRID 43020904.

Bonnette K. 1993c. Dermal Sensitization Study in Guinea Pigs with MON 65005 Herbicide-Modified Buehler Design: Final Report: Lab Project Number: 3044.444: SB-93-129: 1198. Unpublished study prepared by Springborn Labs, Inc. 22 p. MRID 43035903.

Boocock MR; Coggins JR. 1983. Kinetics of 5-enolpyruvylshikimate-3-phosphate synthase inhibition by glyphosate. Febs Lett. 154(1):127-33.

Borjesson E; Torstensson L. 2000. New methods for determination of glyphosate and (Aminomethyl)phosphonic acid in water and soil. J Chromatogr A. 886(1-2): 207-16.

Borrecco JE; Neisess J. 1991. Risk Assessment for the Impurities 2-butoxyethanol and 1,4-Dioxane Found in Garlon 4 and Roundup Herbicide Formulations. Forest Pest Management, Pacific Southwest Region, Report No. R91-2, Feb 25, 1991.

Bowmer KH; Boulton P MD; Short DL; Higgins ML. 1986. Glyphosate-sediment interactions and phytotoxicity in turbid water. Pestic Sci. 17(2): 79-88.

Boxenbaum J; D'Souza R. 1990. Interspecies pharmacokinetic scaling, biological design and neoteny. Adv Drug Res. 19: 139-195.

Boyd RS; Freeman JD; Miller JH; Edwards MB. 1995. Forest herbicide influences on floristic diversity seven years after broadcast pine release treatments in central georgia, USA. New Forests. 10(1): 17-37.

Brakel G. 1999. Product Chemistry--Glyphosate Concentrate: Lab Project Number: GHB/99.09.23/GLY 441: 99.09.23/GLY 441/USA: VAM 027-01. Unpublished study prepared by Cheminova Agro A/S. 125 p. {OPPTS 830.1550, 830.1600, 830.1650, 830.1670, 830.1750, 830.1800, 830.6303,. MRID 44971101.

Bramble WC; Yahner RH; Byrnes WR. 1997. Effect of herbicides on butterfly populations of an electric transmission right-of-way. Journal of Arboriculture. 23(5): 196-206.

Brammer A. 1996. Glyphosate Acid: 1 Year Dietary Toxicity Study in Dogs: Lab Project Number: CTL/P/5079: PD1006. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 498 p.. MRID 44320614.

Breeze V; Thomas G; Butler R. 1992. Use of a model and toxicity data to predict the risks to some wild plant species from drift of four herbicides. Ann Appl Biol. 121(3): 669-677.

Brewster DW; Warren J; Hopkins W. 1991. Metabolism of glyphosate in Sprague-Dawley rats: tissue distribution, identification, and quantitation of glyphosate-derived materials following a single oral dose. Fundam Appl Toxicol. 17(1):43-51.

Bromilow RH; Evans AA; Nicholls PH; Todd AD; Briggs GG. 1996. The effect on soil fertility of repeated applications of pesticides over 20 years. Pesticide Science. 48(1): 63-72.

Brooks K. 1999. Glyphosate IPA Salt (NAF-63): Acute Eye Irritation Study in New Zealand White Rabbits: Lab Project Number: 991182. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2400}. (NAF-63): Acute Eye Irritation Study in New Zealand White Rabbits: Lab Project Number: 991182. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2400}. MRID 44918604.

Buchanan R. 1998. Product Chemistry Data to Support the Registration of MON 77063: Lab Project Number: MSL-15380: MSL-15379: PC63-98-1. Unpublished study prepared by Monsanto Company. 68 p. {OPPTS 830.1550, 830.1600, 830.1620, 830.1670, 830.1700, 830.1750, 830.1800, 830. MRID 44615501.

Budavari S. (Ed). 1989. The Merck Index: An Encyclopedia of Chemicals, Drugs, and Biologicals, 11th ed. Merck & Co., Inc., Rahway, New Jersey.

Burgat V; Keck G; Guerre P; Bigorre V; Pineau X. 1998. Glyphosate toxicosis in domestic animals: a survey from the data of the centre national d'informations toxicologiques veterinaires (Cnitv). Vet Hum Toxicol. 40(6): 363-7.

Burgess D; Hicks S. 1991. Acute Toxicity of AMPA to Daphnia magna: Lab Project Number: 38988: AB/90/401: 1251. Unpublished study prepared by ABC Labs, Inc. 20 p.. MRID 43334715.

Burnmaster DE. 1998. Lognormal distribution for total water intake and tap water intake by pregnant and lactating women in the United States. Risk Analysis. 18(5): 215-219.

Buser HR. 1990. Atrazine and other s-triazine herbicides in lakes and in rain in Switzerland. Environ. Sci. Technol. 24(7): 1049-1058.

Bush PB; Neary DG, McMahon CK; Taylor JW. 1987. Suitability of hardwoods treated with phenoxy pyridine herbicides for use as firewood. Arch Environ Contam Toxicol. 16:333-341.

Busse MD; Ratcliff AW; Shestak CJ; Powers RF. 2001. Glyphosate toxicity and the effects of long-term vegetation control on soil microbial communities.. Soil Biology and Biochemistry. 33: 1777-1789..

Byers RA; Bierlein DL. 1984. Continuous Alfalfa: Invertebrate Pests During Establishment. J Econ Entomol. 77(6): 1500-1503.

C&P Press (Chemical and Pharmaceutical Press). 2002. Product Labels and Material Safety Data Sheets. Greenbook.net. http://www.greenbook.net/.

Caffrey JM. 1996. Glyphosate in fisheries management. Hydrobiologia. 340(1-3): 259-263.

CalEPA (California Environmental Protection Agency). 1997. Public health goal for glyphosate in drinking water.. Pesticide and Environmental Toxicology Section, Office of Environmental Health Hazard Assessment, CalEPA. December 1997. 19p..

Calabrese EJ; Baldwin LA. 1993. Performing Ecological Risk Assessments. Lewis Publishers, Boca Raton, LA, pp. 12-24.

Callander R. 1996. Glyphosate Acid: An Evaluation of Mutagenic Potential Using S. typhimurium and E. coli: Lab Project Number: CTL/P/4874: YV3611. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 32 p.. MRID 44320617.

Campbell JL; Smith MA; Eiteman MA; Williams PL; Boeniger MF. 2000. Comparison of solvents for removing pesticides from skin using an in vitro porcine model. AIHAJ. 61(1): 82-8.

Cantrell RL; Hyland JR. 1985. Application techniques. In: A Guide to Silvicultural Herbicide Use in the Southern United States. Auburn University School of Forestry, Alabama Agricultural Experiment Station. November 1985. 612 pp.

Carson DB; Heitkamp MA; Hallas LE. 1997. Biodegradation of nphosphonomethyliminodiacetic acid by microorganisms from industrial activated sludge. Can J Microbiol. 43(1): 97-101. Cessna AJ; Waddington J. 1995. Dissipation of glyphosate and its metabolite AMPA in established crested wheatgrass following spring application. Can J Plant Sci. 75(3): 759-762.

Chakravarty P; Chatarpaul L. 1990. Non-target effect of herbicides: II. the influence of glyphosate on ectomycorrhizal symbiosis of red pine (Pinus resinosa) under greenhouse and field conditions. Pestic Sci. 28(3): 243-248.

Chakravarty P; Sidhu SS. 1987. Effect of glyphosate hexazinone and triclopyr on in-vitro growth of five species of ectomycorrhizal fungi. Eur J for Pathol. 17(4-5): 204-210.

Chamberlain K; Evans AA; Bromilow RH. 1996. 1-octanol/water partition coefficient (Kow) and pka for ionisable pesticides measured by a ph-metric method. Pest Sci. 47(3): 265-271.

Chan KY; Leung SC. 1986. Effects of paraquat and glyphosate on growth, respiration, and enzyme activity of aquatic bacteria. Bull Environ Contam Toxicol. 36(1): 52-9.

Chang CY; Peng YC; Hung DZ; Hu WH; Yang DY; Lin TJ. 1999. Clinical impact of upper gastrointestinal tract injuries in glyphosate-surfactant oral intoxication. Hum Exp Toxicol. 18(8): 475-8.

Cherepenko E; Karpenko O. 1999. Uptake of the herbicidal glyphosate by Escherichia coli K-12. Biosci Rep. 19(1): 43-9.

Chetram R; Lucash K. 1994. Tier 2 Vegetative Vigor Nontarget Phytotoxicity Study Using Glyphosate: Lab Project Number: 93235: MSL-13320: R.D. 1219. Unpublished study prepared by Pan-Agricultural Labs, Inc. 240 p.. MRID 43088701.

Chin K; Wyder MA; Kaneshiro ES. 1999. Glyphosate reduces organism viability and inhibits growth in vitro of pneumocystis. J Eukaryot Microbiol. 46(5): 139s-141s.

Christian FA; Jackson RN; Tate TM. 1993. Effect of sublethal concentrations of glyphosate and dalapon on protein and aminotransferase activity in pseudosuccinea columella. Bull Environ Contam Toxicol. 51(5): 703-9.

Clapp M; Sargent D. 1997. Glyphosate Trimesium: Reassessment of the Reference Dose. (RFD), and Assessment of Safety to Children and Infants: Lab Project Number: CTL/P/5457. Unpublished study prepared by Central Toxicology Laboratory. 18 p. MRID 44246703.

Clapp M; Fletcher A; Rhodes M et al. 1998. Zeneca's Response to the Toxicology Section of the HED Risk Assessment for Sulfosate Dated July, 10 1998. Lab Project Number: MER980828. Unpublished study prepared by Zeneca Ag Products. 22 p. MRID 44673601.

Clay P. 1996. Glyphosate Acid: L5178Y TK. K(+/-) Mouse Lymphoma Gene Mutation Assay: Lab Project Number: CTL/P/4991: VV0123. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 29 p. MRID 44320618.

Clegg BS; Stephenson GR; Hall JC. 1999. Development of an enzyme-linked immunosorbent assay for the detection of glyphosate. J Agric Food Chem. 47(12): 5031-7.

Clements C; Ralph S; Petras M. 1997. Genotoxicity of select herbicides in Rana catesbeiana tadpoles using the alkaline single-cell gel DNA electrophoresis (Comet) assay. Environ Mol Mutagen. 29(3): 277-88.

Clydesdale FM. 1997. Food Additives: Toxicology, Regulation, and Properties. CRC Press, Boca Raton, Florida. CD-ROM Database.

Cole EC; Mccomb WC; Newton M; Chambers CL; Leeming JP. 1997. Response of amphibians to clearcutting, burning, and glyphosate application in the Oregon coast range. Journal of Wildlife Management. 61(3): 656-664.

Coret J; Chamel A. 1995. Effects and possible mode of action of some nonionic surfactants on the diffusion of 14C-glyphosate and 14C-chlorotoluron across isolated plant cuticles. Pesticide Science. 43(2): 163-166.

Cornish PS. 1992. Glyphosate residues in a sandy soil affect tomato transplants. Aust J Exp Agric. 32(3): 395-399.

Coupe RH; Thurman EM; Zimmerman LR. 1998. Relation of useage to the occurrence of cotton and rice herbicides in three streams of the mississippi delta. Envi Sci Technol. 32(23): 3673-3680.

Coupland D; Lutman P JW. 1982. The movement of glyphosate from treated to adjacent untreated plants. Ann Appl Biol. 101(2): 315-322.

Cox C. 1998. Toxic Secrets, "Inert" Ingredients in Pesticides, 1987-1997. Available at www.pesticide.org/factsheets.html.

Cox C. 2000. Triclopyr Herbicide Fact Sheet. J Pesticide Reform. 20(4): 12-19.

Cox C. 2002. Herbicide Factsheet: Glyphosate (Roundup). J Pesticide Reform. 18(3): 3-17. Available at www.pesticide.org/factsheets.html.

Crissman J; Brooks K. 1999a. Glyphosate IPA Salt (NAF-545): Acute Oral Toxicity Study in Fischer 344 Rats: Lab Project Number: 991032. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1100}. (NAF-545): Acute Oral Toxicity Study in Fischer 344 Rats: Lab Project Number: 991032. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1100}. MRID 44863801.

Crissman J; Brooks K. 1999b. Glyphosate IPA Salt (NAF-545): Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: 99103. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1200}. (NAF-545): Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: 99103. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1200}. MRID 44863802.

Crissman J; Brooks K. 1999c. Glyphosate IPA Salt (NAF-545): Acute Eye Irritation Study in New Zealand White Rabbits: Lab Project Number: 991035. Unpublished study prepared by The Dow Chemical Company. 16 p. {OPPTS 870.2400}. (NAF-545): Acute Eye Irritation Study in New Zealand White Rabbits: Lab Project Number: 991035. Unpublished study prepared by The Dow Chemical Company. 16 p. {OPPTS 870.2400}. MRID 44863804.

Crissman J; Brooks K. 1999d. Glyphosate IPA Salt (NAF-545): Acute Dermal Irritation Study in New Zealand White Rabbits: Lab Project Number: 991034. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2500}. (NAF-545): Acute Dermal Irritation Study in New Zealand White Rabbits: Lab Project Number: 991034. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2500}. MRID 44863805.

Curtis KM; Savitz DA; Weinberg CR; Arbuckle TE. 1999. The Effect of Pesticide Exposure on TIme to Pregnancy. Epidemiology. 10(2): 112-117.

D'Anieri P; Leslie DM; McCormack ML. 1987. Small Mammals in glyphosate-treated clearcuts in northern Maine. Canadian Field-Naturalist. 101: 547-550.

D'Silva ET; Winter JD; Patino R. 1997. The stress response and development of juvenile largemouth bass exposed to sublethal concentrations of aquatic herbicides. Annual Meeting of the Ecological Society of America Held Jointly with the Nature Conservancy on Changing Ecosystems: Natural and Human Influences, Albuquerque, New Mexico, Usa, August 10-14, 1997 Bulletin of the Ecological Society of America; 78 (4 Supp.

Dalby PR; Baker GH; Smith SE. 1995. Glyphosate 2 4-DB and dimethoate effects on earthworm survival and growth. Soil Biology and Biochemistry. 27(12): 1661-1662.

Daniel O; Meier M; Schlatter J; Frischknecht P. 1999. Selected phenolic compounds in cultivated plants: Ecologic functions, health implications, and modulation by pesticides. Environmental Health Perspectives. 107(Suppl 1): 109-114.

Daruich J; Zirulnik F; Gimenez MS. 2001. Effect of the herbicide glyphosate on enzymatic activity in pregnant rats and their fetuses. Environ Res. 85(3): 226-31.

Davies D. 1996a. Glyphosate Acid: Excretion and Tissue Retention of a Single Oral Dose. (10 mg/kg) in the Rat: Lab Project Number: CTL/P/4940: UR0506. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 38 p. MRID 44320620.

Davies D. 1996b. Glyphosate Acid: Excretion and Tissue Retention of a Single Intravenous Dose. (10 mg/kg) in the Rat: Lab Project Number: CTL/P/4941: UR0508. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 40 p. MRID 44320621.

Davies D. 1996c. Glyphosate Acid: Excretion and Tissue Retention of a Single Oral Dose. (1000 mg/kg) in the Rat: Lab Project Number: CTL/P/4942: UR0507. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 40 p. MRID 44320622.

Davies D. 1996d. Glyphosate Acid: Excretion and Tissue Retention of a Single Oral Dose. (10 mg/kg) in the Rat Following Repeat Dosing: Lab Project Number: CTL/P/4944: UR0517. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 40 p. MRID 44320623.

Davies D. 1996e. Glyphosate Acid: Whole Body Autoradiography in the Rat. (10 mg/kg): Lab Project Number: CTL/P/4943: UR0509. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 26 p. MRID 44320625.

Day GM; Hart BT; Mckelvie ID; Beckett R. 1997. Influence of natural organic matter on the sorption of biocides onto goethite, II. glyphosate. Environmental Technology. 18(8): 781-794.

De Jong FM; Udo De Haes HA. 2001. Development of a field bioassay for the side-effects of herbicides on vascular plants using Brassica napus and Poa annua. Environ Toxicol. 16(5): 397-407.

De Jonge H; De Jonge L. 1999. Influence of pH and solution composition on the sorption of glyphosate and prochloraz to a sandy loam soil. Chemosphere. 39(5): 753-763.

De Marco A; De Simone C; Raglione M; Testa A; Trinca S. 1992. Importance of the type of soil for the induction of micronuclei and the growth of primary roots of vicia faba treated with the herbicides atrazine, glyphosate and maleic hydrazide. Mutat Res. 279(1): 9-13.

DeRosa, CT; Wilbur, S; Holler, J; Richter, P; Stevens, Y-W. 1996. Health Evaluation of 1,4-Dioxane. Toxicology and Industrial Health. 12(1): 1-43.

De Ruiter H; Meinen E. 1998. Influence of water stress and surfactant on the efficacy, absorption, and translocation of glyphosate. Weed Science. 46 (3): 289-296.

De Ruiter H; Uffing A JM; Meinen E. 1996. Influence of surfactants and ammonium sulfate on glyphosate phytotoxicity to quackgrass (Elytrigia repens). Weed Technology. 10(4): 803-808.

De Sapio R. 1976. Calculus for the Life Sciences. W.H. Freeman and Company, San Francisco, California. 740 pp..

Delcenserie R; Yzet T; Duchmann J-C; Delamarre J; Dupas J-L; Capron J-P. 1997. Pseudoappendicular syndrome late after Roundup intoxication. Gastroenterologie Clinique Et Biologique. 21(5): 435-437.

Denis M-H; Delrot S. 1997. Effects of salts and surfactants on foliar uptake and long distance transport of glyphosate. Plant Physiology and Biochemistry (Paris). 35(4): 291-301.

Descalzo RC; Punja ZK; Levesque CA; Rahe JE. 1996a. Assessment of host specificity among different species of glyphosate synergistic pythium. Mycological Research. 100(12): 1445-1453.

Descalzo RC; Punja ZK; Levesque CA; Rahe JE. 1996b. Identification and role of phythium species as glyphosate synergists on bean (Phaseolus vulgaris) grown in different soils. Mycological Research. 100 (8): 971-978.

Diamond G; Durkin P. 1997. Effects of Surfactants on the Toxicity of Glyphosate, with Specific Reference to Rodeo.. Prepared under USDA FS Contract No. 53-3187-5-12. Final Report. February 6, 1997..

Dick RE; Quinn JP. 1995a. Glyphosate-degrading isolates from environmental samples: occurrence and pathways of degradation. Appl Microbiol Biotechnol. 43(3): 545-50.

Dick RE; Quinn JP. 1995b. Control of glyphosate uptake and metabolism in Pseudomonas sp. Fems Microbiology Letters. 134(2-3): 177-182.

Dickson SJ; Meinhold RH; Beer ID; Koelmeyer TD. 1988. Rapid determination of glyphosate in postmortem specimens using 31P NMR. J Anal Toxicol. 12(5): 284-6.

Dix M. 1998. Glyphosate Acid-Determination of Aquatic Metabolism under Anaerobic Conditions: Lab Project Number: 13582.0795.6101.755: 96-6-6536: 4.3.07. Unpublished study prepared by Springborn Laboratories, Inc. 87 p.. MRID 44621801.

Dos Santos D. 1996. Toxicity of herbicides for spirodela punctata (G. f. w. meyer) thompson and salvinia minima baker (Aquatic macrophytes). Cientifica (Jaboticabal). 24(2): 353-362.

Dotson SB; Smith CE; Ling CS; Barry GF; Kishore GM. 1996. Identification, characterization, and cloning of a phosphonate monoester hydrolase from burkholderia caryophilli pg2982. J Biol Chem. 271(42): 25754-61.

Drottar K; Swigert J. 1998a. MON 65005: A 48-Hour Static Acute Toxicity Test with the Cladoceran (Daphnia magna):. Final Report: Lab Project Number: 1413: WL-95-257: 139A-185. Unpublished study prepared by Wildlife International Ltd. 41 p. MRID 44538201.

Drottar K; Swigert J. 1998b. MON 65005: A 96-Hour Static Acute Toxicity Test with the Rainbow Trout (Oncorhynchus mykiss):. Final Report: Lab Project Number: WL-95-255: 139A-183: 1413. Unpublished study prepared by Wildlife International Ltd. 45 p. MRID 44538202.

Drottar K; Swigert J. 1998c. MON 65005: A 96-Hour Static Acute Toxicity Test with the Bluegill (Lepomis macrochirus):. Final Report: Lab Project Number: 139A-184: WL-95-256: 1413. Unpublished study prepared by Wildlife International Ltd. 43 p. MRID 44538203.

Duchesne LC; Lautenschlager RA; Bell FW. 1999. Effects of clearcutting and plant competition control methods on carabid (Coleoptera: Carabidae) assemblages in northwestern Ontario. Environ. Monit. Assess. 56(1): 87-96..

Dudek B. 1987. Acute Toxicity of Rodeo Herbicide Administered by Inhalation to Male and Female Sprague-Dawley Rats: Lab Pro- ject ID: EHL 86105. Unpublished study prepared by Monsanto Co. 38 p. MRID 40201101.

Dudek R; Cortner D. 1998. Acute Inhalation Study of MON 77063 (in Rats): Lab Project Number: EHL 98007: ML-98-006: MSL-15390. Unpublished study prepared by Monsanto Company. 55 p. (in Rats): Lab Project Number: EHL 98007: ML-98-006: MSL-15390. Unpublished study prepared by Monsanto Company. 55 p. MRID 44615504.

Dudek B; Liley M. 1999. Acute Inhalation Study MON 78063 in Rats: Lab Project Number: MSE-N 99006: ML-99008: MSL-16012. Unpublished study prepared by Monsanto Co. 39 p. {OPPTS 870.1300}. MRID 44872704.

Duffell S. 1998. Pathology Working Group Peer Review of Ventricular Dilatation of the Brain in Dogs: Lab Project Number: CTL/P/6038. Unpublished study prepared by Central Toxicology Laboratory. 47 p. MRID 44673602.

Durkin PR. 1981. An Approach to the Analysis of Toxicant Interactions in the Aquatic Environment. Aquatic Toxicology and Hazard Assessment, ASTM 4th Symposium on Aquatic Toxicology, pp. 388-401.

Durkin PR; Rubin L; Withey J; Meylan W. 1995. Methods of assessing dermal absorption with emphasis on uptake from contaminated vegetation. Toxicol Indust Health. 11(1): 63-79.

EDSTAC (Endocrine Disruptor Screening and Testing Advisory Committee). 1998. Endocrine Disruptor Screening and Testing Advisory Committee Final Report. Dated August, 1998. Available at http://www.epa.gov/oscpmont/oscpendo/history/finalrpt.htm.

Easton WE; Martin K. 1998. The effect of vegetation management on breeding bird communities in British Columbia. Ecological Applications. 8(4): 1092-1103.

Eberbach P. 1998. Applying non-steady-state compartmental analysis to investigate the simultaneous degradation of soluble and sorbed glyphosate (N-(phosphonomethyl)glycine) in four soils. Pesticide Science. 52(3): 229-240.

Eberbach PL. 1999. Influence of incubation temperature on the behavior of triethylamineextractable glyphosate (N-phosphonomethylglycine) in four soils. J Agric Food Chem. 47(6): 2459-67.

Eberbach PL; Douglas LA. 1983. Persistence of glyphosate in a sandy loam soil. Soil Biol Biochem. 15 (4): 485-488.

Ecobichon DJ. 1998. Occupational Hazards of Pesticide Exposure - Sampling, Monitoring, Measuring. Taylor & Francis, Philadelphia, PA. 251 pp.

Edmiston S; Spencer J; Orr K; Cowan C; Margetich S. 1995. Exposure of Herbicide Handlers in the CalTrans Vegetation Control Program: 1993-1994. California EPA Health & Safety Report HS-1700, dated April 27, 1995, amended Jun1, 1995.

El-Demerdash FM; Yousef MI; Elagamy EI. 2001. Influence of paraquat, glyphosate, and cadmium on the activity of some serum enzymes and protein electrophoretic behavior (In vitro). J Environ Sci Health B. 36(1): 29-42.

el-Gendy KS; Aly NM; El-Sebae AH. 1998. Effects of edifenphos and glyphosate on the immune response and protein biosynthesis of bolti fish (Tilapia nilotica). J Environ Sci Health B 33(2): 135-49.

Estok D; Freedman B; Boyle D. 1989. Effects of the herbicides 2,4-D, glyphosate, hexazinone, and triclopyr on the growth of three species of ectomycorrhizal fungi. Bull Environ Contam Toxicol. 42(6): 835-9.

Evans DD; Batty MJ. 1986. Effects of high dietary concentrations of glyphosate Roundup on a species of bird Poephila-guttata marsupial sminthopsis-macroura and rodent indigenous to australia. Environ Toxicol Chem. 5(4): 399-402.

Everett C; Fleming T; Cole J. 1996a. Glyphosate Acid: A Tier II Glasshouse Study to Assess the Effects on Seedling Emergence of Terrestrial Non-Target Plants: Lab Project Number: 95JH113: RJ2008B. Unpublished study prepared by Zeneca Limited. 32 p. MRID 44320635.

Everett C; Fleming T; Cole J. 1996b. Glyphosate Acid: A Tier II Glasshouse Study to Assess the Effects on Vegetative Vigour of Terrestrial Non-Target Plants: Lab Project Number: 95JH114: RJ2009B. Unpublished study prepared by Zeneca Limited. 55 p.. MRID 44320636.

FAO and WHO Working Groups. 1986. Glyphosate. FAO Plant Production and Protection Paper. 78(2): 63-76.

Farmer DR; Kaempfe TA; Heydens WF; Kelce WR. 2000a. Multi-generation reproduction studies with glyphosate in rats. Toxicologist. 54(1): 395.

Farmer DR; Kaempfe TA; Heydens WF; Kelce WR. 2000b. Developmental toxicity studies with glyphosate and selected surfactants in rats. Teratology. 61(6): 446.

Faust M; Altenburger R; Boedeker W; Grimme LH. 1994. Algal toxicity of binary combinations of pesticides. Bull Envi Contam Toxicol. 53(1): 134-141.

Feng JC; Thompson DG. 1990. Fate of glyphosate in a canadian forest watershed: 2. Persistence in foliage and soils. J Agric Food Chem. 38 (4): 1118-1125.

Feng JC; Thompson DG; Reynolds PE. 1990. Fate of glyphosate in a canadian forest watershed: 1. Aquatic residues and off-target deposit assessment. J Agric Food Chem; 38 (4) 1990 1110-1118.

Fickel J; Mahlburg W. 1999. Product Chemistry Requirements for Credit Herbicide: Lab Project Number: 99-1A. Unpublished study prepared by Nufarm, Inc. 18 p. {OPPTS 830.1550, 830.1600, 830.1620, 830.1650, 830.1670, 830.1750}. MRID 44872601.

Figenschau Y; Yousef MI; Sveinbjornsson B; Bertheussen K. 1997. A sensitive serum-free colorimetric assay for the detection of cytotoxic effects of pesticides. J Environ Sci Health B. 32(2): 177-94.

Fink R. 1975. Final Report: One Generation Reproduction Study-- Mallard Duck: Project No. 139-101.. (Unpublished study received Sep 26, 1975 under 6G1679; prepared by Truslow Farms, Inc., submitted by Monsanto Co., Washington, D.C.; CDL:096483-N). Summarized in U.S. EPA/OPP 1993c. MRID 00036328.

Fink R; Beavers J. 1978. Final Report: One-generation Reproduc- tion Study--Mallard Duck: Glyphosate Technical: Project No. 139-143.. (Unpublished study received Nov 13, 1978 under 524-308; prepared by Wildlife International Ltd., submitted by Monsanto Co., Washington, DC; CDL:235924-A). Summarized in U.S. EPA/OPP 1993c. MRID 00111953. Flaherty DK; Gross CJ; Mcgarity KL; Winzenburger PA; Wratten SJ. 1991. The effect of agricultural herbicides on the function of human immunocompetent cells: II. effects on natural killer cell and cytotoxic t cell function. In Vitro Toxicol; 4 (2) 1991 145-160.

Fletcher JS; Nellessen JE; Pfleeger TG. 1994. Literature review and evaluation of the EPA foodchain (Kenega) nomogram, an instrument for estimating pesticide residues on plants. Environ. Toxicol. Chem. 13(9):1383-1391.

Fletcher JS; Pfleeger TG; Ratsch HC; Hayes R. 1996. Potential impact of low levels of chlorsulfuron and other herbicides on growth and yield of nontarget plants. Environ Toxicol Chem. 15(7): 1189-1196.

Flora T; Simon Z. 1981. Thermal degradation of glyphosate herbicide. Magyai Kemiai Floyoirat. 87: 419-424. (Cited in Smith and Oehme 1992).

Folmar LC. 1976. Overt avoidance reaction of rainbow trout fry to nine herbicides. Bull Environ Contam Toxicol 15(5): 509-514.

Folmar LC. 1978. Avoidance chamber responses of mayfly nymphs exposed to eight herbicides. Bull Environ Contam Toxicol. 19(3): 312-318.

Folmar LC; Sanders HO; Julin AM. 1979. Toxicity of the herbicide glyphosate and several of its formulations to fish and aquatic invertebrates. Arch Environ Contam Toxicol. 8(3): 269-278.

Forlani G; Mangiagalli A; Nielsen E; Suardi CM. 1999. Degradation of the phosphonate herbicide glyphosate in soil: evidence for a possible involvement of unculturable microorganisms. Soil Biol Biochem. 31 (7): 991-997.

Fox V; Mackay J. 1996. Glyphosate Acid: Mouse Bone Marrow Micronucleus Test: Lab Project Number: CTL/P/4954: SM0796. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 30 p. MRID 44320619.

Franz JE. 1985. Discovery, development and chemistry of glyphosate.. In: The Herbicide Glyphosate, Grossbard, E; Atkinson, D; eds. London, England: Butterworths. p. 3-17. (Cited in SERA 1996).

Freemark K; Boutin C. 1995. Impacts of agricultural herbicide use on terrestrial wildlife in temperate landscapes: A review with special reference to North America. Agriculture Ecosystems Environ. 52(2-3): 67-91.

Friis M. 1995a. Product Chemistry--Glyphosate Technical--FIFRA Guidelines Series 61: Lab Project Number: MVF/00.03.95/GLY-440/USA. Unpublished study prepared by Cheminova Agro A/S. 26 p. MRID 43632701.

Friis M. 1995b. Product Chemistry--Glyphosate Technical--FIFRA Guidelines Series 62: Analysis and Certification of Product Ingredients: Lab Project Number: MVF/01.03.95/GLY-440/USA: TEM 005-02. Unpublished study prepared by Cheminova Agro A/S. 601 p.. MRID 43728001.

Friis M. 1995c. Product Chemistry--Glyphosate Technical--Guidelines Series 63: Physical and Chemical Characteristics: Lab Project Number: MVF/00.02.95/PMG-USA: HUK 6758-676/5: 95-3-5748. Unpublished study prepared by Cheminova Agro A/S. 114 p.. MRID 43728002.

Garcia AM; Benavides FG; Fletcher T; Orts E. 1998. Paternal exposure to pesticides and congenital malformations. Scand J Work Environ Health. 24(6): 473-480.

Garcia-Repetto R; Soria ML; Gimenez MP; Menendez M; Repetto M. 1998. Deaths from pesticide poisoning in spain from 1991 to 1996. Vet Human Toxicol. 40(3): 166-168.

Gard JK; Feng PC; Hutton WC. 1997. Nuclear magnetic resonance timecourse studies of glyphosate metabolism by microbial soil isolates. Xenobiotica. 27(7): 633-44.

Gardner; SC; Grue CE. 1996. Effects of Rodeo and Garlon 3A on nontarget wetland species in central Washington. Environ Toxicol Chem. 15(4): 441-451.

Gardner SC; Grue CE; Grassley JM; Lenz LA; Lindenauer JM; Seeley ME. 1997. Single species algal (Ankistrodesmus) toxicity tests with Rodeo and Garlon 3a. Bull Environ Contam Toxicol. 59(3): 492-9.

Garry VF; Burroughs B; Tarone R; Kesner JS. 1999. Herbicides and adjuvants: An evolving view. Toxicol Indust Health. 15(1-2): 159-167.

Garry VF; Harkins ME; Erickson LL; Long-Simpson LK; Holland SE; Burroughs BL. 2002. Birth defects, season of conception, and sex of children born to pesticide applicators living in the Red River Valley of Minnesota, USA. Environ Health Perspect. 110 Suppl 3:441-9.

Geiger CP; Calabrese EJ. 1985. The effects of five widely used pesticides on erythrocytes of the dorset sheep, an animal model with low erythrocyte glucose-6-phosphate dehydrogenase (G-6-pd) activity. Journal of Environmental Science and Health, Part A. 20(5): 521-527.

Geiger DR; Shieh W-J; Fuchs MA. 1999. Causes of self-limited translocation of glyphosate in beta vulgaris plants. Pesticide Biochem Physiol. 64(2): 124-133.

Gerritse RG; Beltran J; Hernandez F. 1996. Adsorption of atrazine, simazine, and glyphosate in soils of the gnangara mound, western australia. Australian Journal of Soil Research; 34 (4) 1996 599-607.

Gibb TJ; Buhler WG. 1998. Infectivity of steinernema carpocapsae (Rhabditida: steinernematidae) in sterilized and herbicide-treated soil. Journal of Entomological Science; 33 (2) 1998 152-157.

Gladen BC; Sandler DP; Zahm SH; Kamel F; Rowland AS; Alavanja M CR. 1998. Exposure opportunities of families of farmer pesticide applicators. American Journal of Industrial Medicine; 34 (6) 1998 581-587.

Glass RL. 1987. Adsorption of glyphosate by soils and clay minerals. J Agric Food Chem; 35 (4) 1987 497-500.

Gohre K; Casida JE; Ruzo LO. 1987. N oxidation and cleavage of the amino acid derived herbicide glyphosate and anilino acid of the insecticide fluvalinate. J Agric Food Chem; 35 (3) 1987 388-392.

Gold LS; Stern BR; Slone TH; Brown JP; Manley NB; Ames BN. 1997. Pesticide residues in food: investigation of disparities in cancer risk estimates. Cancer Letters; 117 (2) 1997 195-207.

Goldsborough LG; Beck AE. 1989. Rapid dissipation of glyphosate in small forest ponds. Arch Environ Contam Toxicol; 18 (4) 1989 537-544.

Goldsborough LG; Brown DJ. 1988. Effect of glyphosate roundup formulation on periphytic algal photosynthesis. Bull Environ Contam Toxicol; 41 (2) 1988 253-260.

Goldsborough LG; Brown DJ. 1993. Dissipation of glyphosate and aminomethylphosphonic acid in water and sediments of boreal forest ponds. Environ Toxicol Chem. 12(7): 1139-1147.

Goldstein RS; Schnellmann RG. 1996. Toxic Responses of the Kidney. In: Casarett and Doull's Toxicology: The Basic Science of Poisons. 5th ed., CD Klaassen, MO Amdur, J Doull, Eds. McGraw-Hill, New York, New York. pp. 417-442..

Goldstein A; Aronow L; Kaman SM. 1974. Principles of Drug Action: The Basis of Pharmacology. 2nd ed. John Wiley and Sons, New York, NY. 854 pp..

Goldstein DA; Johnson G; Farmer DR; Martens MA; Ford JE; Cullen MR. 1999. Pneumonitis and herbicide exposure. Chest 1999 Oct;116(4):1139-40.

Goure W. 1992. Addendum to MSL-7633: Irrigated Crop Study. Determination of Glyphosate Residues in Crops, Irrigation Water, Sediment and Soil Following Treatment of Irrigation Source with Rodeo Herbicide: Supplement to MRID 40541305: Unpublished study prepared by Monsa. MRID 42372505.

Gout E; Bligny R; Genix P; Tissut M; Douce R. 1992. Effect of glyphosate on plant cell metabolism. 31p and 13c nmr studies. Biochimie 1992 Sep-oct;74(9-10):875-82.

Granby K; Vahl M. 2001. Investigation of the herbicide glyphosate and the plant growth regulators chlormequat and mepiquat in cereals produced in denmark. Food Addit Contam. 18(10):898-905.

Grande M; Andersen S; Berge D. 1994. Effects of pesticides on fish: experimental and field studies. Norwegian Journal of Agricultural Sciences Suppl. 13: 195-209. MRID 45202909.

Green TH; Minogue PJ; Brewer CH; Glover GR; Gjerstad DH. 1992. Absorption and translocation of carbon-14 glyphosate in four woody plant species. Can J for Res; 22 (6) 1992 785-789.

Grey L; Nguyen B; Yang P. 2001. Liquid chromatography/electrospray ionization/isotopic dilution mass spectrometry analysis of n-(phosphonomethyl) glycine and mass spectrometry analysis of aminomethyl phosphonic acid in environmental water and vegetation matrixes. J APAC Int. 84(6):1770-80.

Grisolia CK. 2002. A comparison between mouse and fish micronucleus test using cyclophosphamide, mitomycin CoA and various pesticides. Mutat Res. 518(2):145-50.

Guo YL; Wang B-J; Lee C-C; Wang J-D. 1996. Prevalence of dermatoses and skin sensitisation associated with use of pesticides in fruit farmers of southern taiwan. Occupational and Environmental Medicine, Vol 53, No 6, Pages 427-431, 15 References, 1996.

Haag KH. 1986. Effects of herbicide application on mortality and dispersive behavior of the water hyacinth weevils neochetina-eichhorniae and neochetina-bruchi coleoptera curculionidae. Environ Entomol; 15 (6) 1986 1192-1198.

Haney RL; Senseman SA; Hons FM. 2002. Effect of roundup ultra on microbial activity and biomass from selected soils. J Environ Qual. 31(3):730-5.

Hardell L; Eriksson M. 1999a. A Case-Control Study of Non-Hodgkin Lymphoma and Exposure to Pesticides.. Cancer 85(6):1353-1360..

Hardell L; Eriksson M. 1999b. Author Reply.. Cancer. 86(4): 730-731..

Harris SA; Solomon KR. 1992. Human exposure to 2,4-D following controlled activities on recently sprayed turf. J Environ Sci Health. B27(1): 9-22.

Hart MR; Brookes PC. 1996. Soil microbial biomass and mineralisation of soil organic matter after 19 years of cumulative field applications of pesticides. Soil Biology & Biochemistry; 28 (12) 1996 1641-1649.

Hartman WA; Martin DB. 1984. Effect of suspended bentonite clay on the acute toxicity of glyphosate to daphnia-pulex and lemna-minor. Bull Environ Contam Toxicol; 33 (3) 1984 355-361.

Hastings C. 1995. Review of: "Yousef, et al. (1995) Toxic effects of carbofuran and glyphosate on semen characteristics in rabbits. J. Environ. Sci. Health B30. 0(4):513-534". Unpublished study prepared by Monsanto Co. 2 p. MRID 43839502.

Hatfield M. 1996. Field Soil Dissipation of Glyphosate Acid: Interim Report: Lab Project Number: AA950252: AA950252.CA: AA950252.NC. Unpublished study prepared by American Agricultural Services, Inc. and EN-CAS Analytical Labs. 575 p. MRID 44125719.

Haughton AJ; Bell JR; Boatman ND; Wilcox A. 1999. The effects of different rates of the herbicide glyphosate on spiders in arable field margins. Journal of Arachnology; 27 (1) 1999 249-254.

Haughton AJ; Bell JR; Wilcox A; Boatman ND. 2001a. The effect of the herbicide glyphosate on non-target spiders: Part I. direct effects on Lepthyphantes tenuis under laboratory conditions. Pest Manag Sci. 57(11):1033-6.

Haughton AJ; Bell JR; Boatman ND; Wilcox A. 2001b. The effect of the herbicide glyphosate on non-target spiders: Part II. indirect effects on Lepthyphantes tenuis in field margins. Pest Manag Sci. 57(11):1037-42.

He M; Yang ZY; Nie YF; Wang J; Xu P. 2001. A new type of class I bacterial 5enopyruvylshikimate-3-phosphate synthase mutants with enhanced tolerance to glyphosate. Biochim Biophys Acta. 1568(1):1-6.

Heinonen-Tanski H. 1989. The effect of temperature and liming on the degradation of glyphosate in two arctic forest soils. Soil Biol Biochem; 21 (2) 1989 313-318.

Hendricks CW; Rhodes AN. 1992. Effect of glyphosate and nitrapyrin on selected bacterial populations in continuous-flow culture. Bull Environ Contam Toxicol 1992 Sep;49(3):417-24.

Henry CJ; Higgins KF; Buhl KJ. 1994. Acute toxicity and hazard assessment of Rodeo, X-77 Spreader, and Chem-Trol to aquatic invertebrates. Environ Contam Toxicol. 27: 392-399.

Hernandez A; Garcia-Plazaola JI; Becerril JM. 1999. Glyphosate effects on phenolic metabolism of nodulated soybean (Glycine max l. merr.). J Agric Food Chem 1999 Jul;47(7):2920-5.

Hernandez F; Hidalgo C; Sancho JV. 2000. Determination of glyphosate residues in plants by precolumn derivatization and coupled-column liquid chromatography with fluorescence detection. J Aoac Int 2000 May-jun;83(3):728-34.

Herre B; Barclay J. 1994a. Product Chemistry Data to Support the Registration of Accord Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: R/D/1276: MSL/6717. Unpublished study prepared by Monsanto Agricultur. MRID 43404901.

Herre B; Barclay J. 1994b. Product Chemistry Data to Support the Registration of Rodeo Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: MSL/6713: R/D/1276. Unpublished study prepared by Monsanto Agricultura. MRID 43404904.

Herre B; Barclay J. 1994c. Product Chemistry Data to Support the Registration of Accord Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: R/D/1276: MSL/6717. Unpublished study prepared by Monsanto Agricultur. MRID 43404901.

Herre B; Barclay J. 1994d. Product Chemistry Data to Support the Registration of Accord Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: R/D/1276: MSL/6717. Unpublished study prepared by Monsanto Agricultur. MRID 43404901.

Herre B; Barclay J. 1994e. Product Chemistry Data to Support the Registration of Accord Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: R/D/1276: MSL/6717. Unpublished study prepared by Monsanto Agricultur. MRID 43404901.

Herre B; Barclay J. 1994f. Product Chemistry Data to Support the Registration of Accord Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: R/D/1276: MSL/6717. Unpublished study prepared by Monsanto Agricultur. MRID 43404901.

Herre B; Barclay J. 1994g. Product Chemistry Data to Support the Registration of Accord Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: R/D/1276: MSL/6717. Unpublished study prepared by Monsanto Agricultur. MRID 43404901.

Herre B; Barclay J. 1994h. Product Chemistry Data to Support the Registration of Rodeo Herbicide. Part 4 of 4. Twelve-Month Storage Stability Study in Typical Container at Room Temperature: Lab Project Number: MSL/6713: R/D/1276. Unpublished study prepared by Monsanto Agricultura. MRID 43404904.

Hetherington PR; Marshall G; Kirkwood RC; Warner JM. 1998. Absorption and efflux of glyphosate by cell suspensions. Journal of Experimental Botany; 49 (320) 1998 527-533.

Hietanen E; Linnainmaa K; Vainio H. 1983. Effects of phenoxyherbicides and glyphosate on the hepatic and intestinal biotransformation activities in the rat. Acta Pharmacol Toxicol (copenh) 1983 Aug;53(2):103-12.

Hildebrand LD; Sullivan DS; Sullivan TP. 1980. Effects of roundup herbicide on populations of daphnia magna in a forest pond. Bull Environ Contam Toxicol 1980 Sep;25(3):353-7.

Hildebrand LD; Sullivan DS; Sullivan TP. 1982. Experimental studies of rainbow trout populations exposed to field applications of roundup herbicide. Arch Environ Contam Toxicol 1982;11(1):93-8.

Hindson C; Diffey B. 1984. Phototoxicity of glyphosate in a weedkiller. Contact Dermatitis 1984 Jan;10(1):51-2.

Hirsch R; Augustin D. 1987. Nitrosamine Analyses of Roundup Herbicide, Rodeo Herbicide, MON 0139 and Polado Technical: Laboratory Project ID R. D. No. 835. Unpublished study prepared by Monsanto Agricultural Company. 212 p.. MRID 40405401.

Hoffman DJ; Albers PH. 1984. Evaluation of potential embryotoxicity and teratogenicity of 42 herbicides, insecticides and petroleum contaminants to mallard (Anas platyrhynchos) eggs. Arch Environ Contam Toxicol; 13 (1) 1984 15-28.

Hogendoorn EA; Ossendrijver FM; Dijkman E; Baumann RA. 1999. Rapid determination of glyphosate in cereal samples by means of pre-column derivatisation with 9-fluorenylmethyl chloroformate and coupled-column liquid chromatography with fluorescence detection. J Chromatogr A. 833(1):67-73.

Holck AR; Meek CL. 1987. Dose-mortality responses of crawfish and mosquitoes to selected pesticides. J Am Mosq Control Assoc 1987 Sep;3(3):407-11.

Holdway DA; Dixon DG. 1988. Acute toxicity of permethrin or glyphosate pulse exposure to larval white sucker (Catostomus commersoni) and juvenile flagfish (Jordanella floridae) as modified by age and ration level. Environ Toxicol Chem; 7 (1) 1988 63-68.

Holland M. 1990. Results of the Analyses of AMPA in a 48-Hour Acute Study with Daphnia magna: Lab Project Number: 90187/DAPHNIA: 90187: ML/90/403. Unpublished study prepared by Monsanto Environmental Health Lab. 21 p. MRID 43334716.

Holson J. 1991a. A Dose Range-Finding Developmental Toxicity Study of AMPA in Rats: Final Report: Lab Project Number: 50146: WI/90/247: 1251. Unpublished study prepared by WIL Research Laboratories, Inc. 157 p.. MRID 43334704.

Holson J. 1991b. A Developmental Toxicology Study of AMPA in Rats: Final Report: Lab Project Number: 50159: WI/90/266: 1251. Unpublished study prepared by WIL Research Laboratories, Inc. 326 p. MRID 43334705.

Hontis A. 1995a. Residues of Glyphosate/AMPA in Winter Oilseed Rape Following Applications of MON 52276, MON 44068, and Roundup Herbicide Two Weeks Before Harvest: U.K. Field Trials 1992: Lab Project Number: MLL 30321: R.D. 1312: 92-GLY-02. Unpublished study prepared by. MRID 43827801.

Hontis A. 1995b. Glyphosate and AMPA Residues in Wheat and Barley Following Application of MON 52276, MON 44068 and Roundup Herbicide One Week Before Crop Harvest: French Trials 1991: Lab Project Number: MLL 30281: R.D. 1312: 91-GL-01. Unpublished study prepared by Monsa. MRID 43845401.

Hoogheem TJ. 1989. The safety of roundup pesticide. Jama 1989 Nov 17;262(19):2679.

Horner S. 1993a. Glyphosate Trimesium: Acute Neurotoxicity Study in Rats: Lab Project Number: CTL/P/3813: AR5425. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 328 p.. MRID 43132301.

Horner S. 1993b. Glyphosate Trimesium: Subchronic Neurotoxicity Study in Rats: Lab Project Number: CTL/P/3831: PR0887. Unpublished study prepared by Zeneca Central Toxicology Lab. 358 p.. MRID 43151202.

Horner S. 1996a. Glyphosate Acid: Acute Neurotoxicity Study in Rats: Lab Project Number: CTL/P/4866: AR5968. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 623 p.. MRID 44320610.

Horner S. 1996b. Glyphosate Acid: Subchronic Neurotoxicity Study in Rats: Lab Project Number: CTL/P/4867: PR1009. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 729 p.. MRID 44320612.

Howd RA; Brown JP; Morry DW; Wang YY; Bankowska J; Budroe JD; Campbell M; Dibartolomeis MJ; Faust J; Jowa L; Lewis D; Parker T; Polakoff J; Rice DW; Salmon AG; Tomar RS; Fan AM. 2000. Development of California public health goals (PHGS) for chemicals in drinking water. J Appl Toxicol 2000 Sep-oct;20(5):365-80.

Hung DZ; Deng JF; Wu TC. 1997. Laryngeal survey in glyphosate intoxication: a pathophysiological investigation. Hum Exp Toxicol 1997 Oct;16(10):596-9.

Hung D-Z; Deng J-F; Wu T-C. 1997. Laryngeal survey in glyphosate intoxication: a pathophysiological investigation. Human & Experimental Toxicology; 16 (10) 1997 596-599.

ICRP (International Commission on Radiologic Protection). 1975. Report of the Task Group on Reference Man. Recommendations of the International Commission on Radiological Protection (ICRP) Publ. No. 23. Pergamon Press, New York, NY. .

Ibrahim HZ; Salem MH; Bertheussen K; Seehy MA; Helmi S; Yousef MI. 1995. Toxic effects of carbofuran and glyphosate on semen characteristics in rabbits. J Environ Sci Health Part b 1995;b30(4):513-34.

Issa AA. 1999. Interference of glyphosate with the shikimate pathway by cyanobacteria in chemostat culture. Microbios; 100 (395) 1999 47-55.

Ito N; Hagiwara A; Tamano S; Futacuchi M; Imaida K; Shirai T. 1996. Effects of pesticide mixtures a the acceptable daily intake levels on rat carcinogenesis. Food and Chemical Toxicology; 34 (11-12) 1996 (1997) 1091-1096.

Jackson JR. 1988. Toxicity of herbicide containing glyphosate. Lancet 1988 Feb 20;1(8582):414.

Jain M; Choudhary D; Kale RK; Bhalla-Sarin N. 2002. Salt- and glyphosate-induced increase in glyoxalase I activity in cell lines of groundnut (Arachis hypogaea). Physiol Plant. 114(4):499-505.

Jamison JP; Langlands JH; Lowry RC. 1986. Ventilatory impairment from pre-harvest retted flax. Br J Ind Med 1986 Dec;43(12):809-13.

Jauhiainen A; Rasanen K; Sarantila R; Nuutinen J; Kangas J. 1991. Occupational exposure of forest workers to glyphosate during brush saw spraying work. Am Ind Hyg Assoc J 1991 Feb;52(2):61-4.

Jensen PC. 1989. Exposure to roundup. South Med J 1989 Jul;82(7):934.

Johnson A. 1997. Acute Delayed Neurotoxicity Study in the Domestic Hen: Glyphosate Acid: Lab Project Number: ISN 361/960244. Unpublished study prepared by Huntingdon Life Sciences Ltd. 81 p.. MRID 44354001.

Johnson JW; Ascher P. 1987. Glycine potentiates the NMDA response in cultured mouse brain neurons. Nature. 325: 529-531.

Kale PG; Petty Bt Jr; Walker S; Ford JB; Dehkordi N; Tarasia S; Tasie BO; Kale R; Sohni YR. 1995. Mutagenicity testing of nine herbicides and pesticides currently used in agriculture. Environ Mol Mutagen 1995;25(2):148-53.

Kammerer M. 1995. Intoxication by glyphosate based herbicides. Recueil De Medecine Veterinaire De L'ecole D'alfort; 171 (2-3) 1995 149-152.

Kamrin MA. 1997. Pesticide profiles toxicity environmental impact and fate. Pesticide Profiles: Toxicity, Environmental Impact, and Fate Xix+676p Crc Press Publishers Inc: Boca Raton, Florida, Usa; London, England, UK ISBN 1-56670-190-2; 0 (0) 1997 Xix+676p.

Kaya B; Creus A; Yanikoglu A; Cabre O; Marcos R. 2000. Use of the drosophila wing spot test in the genotoxicity testing of different herbicides. Environ Mol Mutagen 2000;36(1):40-6.

Khan SU; Young JC. 1977. N-nitrosamine formation in soil from the herbicide glyphosate. J Agric Food Chem 1977 Nov-dec;25(6):1430-2.

Kier L; Stegeman S. 1993. Mouse Micronucleus Study of AMPA: Lab Project Number: 90170: ML/90/404: 1251. Unpublished study prepared by Monsanto Environmental Health Lab. 47 p.. MRID 43334708.

Kilbride KM; Paveglio FL. 2001. Long-term fate of glyphosate associated with repeated rodeo applications to control smooth cordgrass (Spartina alterniflora) in willapa bay, washington. Arch Environ Contam Toxicol 2001 Feb;40(2):179-83.

Kilbride KM; Grue CE; Simenstad CA; Fresh KL. 1996. Use of rodeo and X-77 spreader to control smooth cordgrass (Spartina alterniflora) in a southwestern washington estuary: 1. Environmental fate. Environmental Toxicology and Chemistry; 15 (6) 1996 961-968.

Kirk A. 2000. Rationale to Support Follow-On Reduced Risk of Monsanto Petition Proposing Uses of Roundup Ultra Herbicide Over-the-Top of Glyphosate-Tolerant Wheat: Lab Project Number: 1527. Unpublished study prepared by Monsanto Company. 142 p.. MRID 45290401.

Kisseberth WC; Buck WB; Mansfield ME; Manuel RK. 1986. Preferential grazing by cattle on glyphosate-treated fescue pastures. Am J Vet Res 1986 Mar;47(3):696-8.

Klimek M; Lejczak B; Kafarski P; Forlani G. 2001. Metabolism of the phosphonate herbicide glyphosate by a non-nitrate-utilizing strain of penicillium chrysogenum. Pest Manag Sci. 57(9):815-21.

Klinger J; Lang M; Sacher F; Brauch H-J; Maier D; Worch E. 1998. Formation of glyphosate and AMPA during ozonation of waters containing ethylenediaminetetra(methylenephosphonic acid). Ozone Science & Engineering; 20 (2) 1998 99-110.

Knight H; Cox C. 1998. Secrets: Toxic Inert Ingredients in Pesticides. Northwest Coalition for Alternatives to Pesticides, Available at www.pesticide.org/factsheets.html.

Knisel WG; Davis FM; Leonard RA. 1992. GLEAMS Version 2.0 User Manual. U.S. Department of Agriculture, Agricultural Research Service, Southeast Watershed Research Laboratory, Tifton, GA. 202pp.

Kremer RJ. 2002. Affected Environment and Environmental Consequences" in Chapter Three of "Eradication of Arundo in Big Tujunga". Memo to Santa Monica Mountains Coalition for Alternatives to Toxics. Copy courtesy of Dave Bakke, USDA/FA..

Kreutzweiser DP; Kingsbury PD; Feng JC. 1989. Drift response of stream invertebrates to aerial applications of glyphosate. Bull Environ Contam Toxicol 1989 Mar;42(3):331-8.

Krueger A; Okoniewska G; Pochitonow Z; Krol Z; Garnett RP. 1996. Glyphosate as a management tool in carp fisheries. Hydrobiologia; 340 (1-3) 1996 253-257.

Krzysko-Lupicka T; Orlik A. 1997. The use of glyphosate as the sole source of phosphorus Or carbon for the selection of soil-borne fungal strains capable to degrade this herbicide. Chemosphere; 34 (12) 1997 2601-2605.

Krzysko-Lupicka T; Strof W; Kubs K; Skorupa M; Wieczorek P; Lejczak B; Kafarski P. 1997. The ability of soil-borne fungi to degrade organophosphonate carbon-to-phosphorus bonds. Appl Microbiol Biotechnol 1997 Oct;48(4):549-52.

Kubena LF; Smalley HE; Farr FM. 1981. Influence of glyphosate (N-(phosphonomethyl)glycine) on performance and selected parameters in broilers. Poult Sci. 60(1): 132-6.

Kudzin ZH; Gralak DK; Drabowicz J; Luczak J. 2002. Novel approach for the simultaneous analysis of glyphosate and its metabolites. J Chromatogr A. 947(1):129-41.

Laatikainen T; Heinonen-Tanski H. 2002. Mycorrhizal growth in pure cultures in the presence of pesticides. Microbiol Res. 2002;157(2):127-37.

Laerke PE; Streibig JG. 1995. Foliar absorption of some glyphosate formulations and their efficacy on plants. Pesticide Science; 44 (2) 1995 107-116.

Lamb DC; Kelly DE; Hanley SZ; Mehmood Z; Kelly SL. 1998. Glyphosate is an inhibitor of plant cytochrome P450: Functional expression of Thlaspi arvensae cytochrome p45071b1/reductase fusion protein in escherichia coli. Biochem Biophys Res Commun. 244(1):110-4.

Lankas GR; Hogan GK. 1981. A Lifetime Feeding Study of Glyphosate (Roundup^(R) Technical) in Rats: Project No. 77- 2062.. (Unpublished study received Jan 20, 1982 under 524-308; prepared by Bio/dynamics, Inc., submitted by Monsanto Co., Washington, D.C.; CDL:246617-A; 246618; 246619; 246620; 246621). MRID 00093879.

Larsen SB; Giwercman A; Spano M; Bonde JP; Group AS. 1998a. A longitudinal study of semen quality in pesticide spraying danish farmers. Reproductive Toxicology; 12 (6) 1998 581-589.

Larsen SB; Joffe M; Bonde JP; Group AS. 1998b. Time to pregnancy and exposure to pesticides in danish farmers. Occupational and Environmental Medicine; 55 (4) 1998 278-283.

Lavy TL; Cowell JE; Steinmetz JR; Massey JH. 1992. Conifer seedling nursery worker exposure to glyphosate. Arch Environ Contam Toxicol; 22 (1) 1992 6-13.

Lee HL; Chen KW; Chi CH; Huang JJ; Tsai LM. 2000. Clinical presentations and prognostic factors of a glyphosate-surfactant herbicide intoxication: A review of 131 cases. Acad Emerg Med 2000 Aug;7(8):906-10.

Leung JW. 1994. A fluorometric method to determine rainfastness, volatilization and photostability of glyphosate from glass slides, after application of vision with two adjuvants. Journal of Environmental Science and Health Part b Pesticides Food Contaminants and Agricultural Wastes; 29 (2) 1994 341-363.

Levesque CA; Rahe JE. 1992. Herbicide interactions with fungal root pathogens with special reference to glyphosate. Cook, R J (ed) Annual Review of Phytopathology, Vol 30 X+692p Annual Reviews Inc: Palo Alto, California, USA Illus ISBN 0-8243-1330-5; 0 (0) 1992 579-602.

Levine TE. 1996. The regulation of inert ingredients in the United States. Chapter 1 in Pesticide Formulation and Adjuvant Technology. CL Foy and DW Pritchard (eds). CRC Press. Boca Raton, Florida. Pp. 1-11.

Li AP; Long TJ. 1988. An evaluation of the genotoxic potential of glyphosate. Fundamental and Applied Toxicology, Vol 10, No 3, Pages 537-546, 22 References, 19881988.

Lin N; Garry VF. 1998. An in vitro pilot study of potential developmental toxicity of herbicides and fungicides commonly used in the Red River Valley of Minnesota. 29th Annual Meeting of the Environmental Mutagen Society, Anaheim, California, Usa, March 21-26, 1998 Environmental and Molecular Mutagenesis; 31 (suppl 29) 1998 45.

Lin N; Garry VF. 2000. In vitro studies of cellular and molecular developmental toxicity of adjuvants, herbicides, and fungicides commonly used in Red River Valley, Minnesota. J Toxicol Environ Health a 2000 Jul 28;60(6):423-39.

Lin CM; Lai CP; Fang TC; Lin CL. 1999. Cardiogenic shock in a patient with glyphosatesurfactant poisoning. J Formos Med Assoc. 98(10):698-700.

Linden E; Bengtsson BE; Svanberg O; Sundstrom G. 1979. The acute toxicity of 78 chemicals and pesticide formulations against two brackish water organisms, the bleak (Alburnus alburnus) and the harpacticoid nitocra spinipes. Chemosphere 8(11-12): 843-851.

Linz GM; Blixt DC. 1997. Black terns benefit from cattail management in the northern great plains. Colonial Waterbirds; 20 (3) 1997 617-621.

Linz GM; Blixt DC; Bergman DL; Bleier WJ. 1996. Response of ducks to glyphosate-induced habitat alterations in wetlands. Wetlands. 16 (1):38-44.

Linz GM; Blixt DC; Bergman DL; Bleier WJ. 1996. Responses of red-winged blackbirds, yellow-headed blackbirds and marsh wrens to glyphosate-induced alterations in cattail density. Journal of Field Ornithology; 67 (1) 1996 167-176.

Linz GM; Bergman DL; Blixt DC; Mcmurl C. 1997. Response of american coots and soras to herbicide-induced vegetation changes in wetlands. Journal of Field Ornithology; 68 (3) 1997 450-457.

Lioi MB; Scarf'i MR; Santoro A; Barbieri R; Zeni O; Di Berardino D; Ursini MV. 1998a. Genotoxicity and oxidative stress induced by pesticide exposure in bovine lymphocyte cultures in vitro. Mutat Res 1998 Jul 17;403(1-2):13-20.

Lioi MB; Scarfi MR; Santoro A; Barbieri R; Zeni O; Salvemini F; Di Berardino D; Ursini MV. 1998b. Cytogenetic damage and induction of pro-oxidant state in human lymphocytes exposed in vitro to gliphosate, vinclozolin, atrazine, and dpx-e9636. Environ Mol Mutagen 1998;32(1):39-46.

Lockhart W; Billeck B; Baron C. 1989. Bioassays with a floating aquatic plant (Lemna minor) for effects of sprayed and dissolved glyphosate. Hydrobiologia 188/189:353-359. MRID 44125713.

Long K; Shillabeer N; Caunter J; et al. 1996a. Glyphosate Acid: Acute Toxicity to Mysid Shrimp (Mysidopsis bahia): Lab Project Number: BL5713/B: AB0503/H. Unpublished study prepared by Brixham Environmental Laboratory. 20 p. (Mysidopsis bahia): Lab Project Number: BL5713/B: AB0503/H. Unpublished study prepared by Brixham Environmental Laboratory. 20 p. MRID 44320633.

Long K; Shillabeer N; Caunter J; et al. 1996b. Glyphosate Acid: Acute Toxicity to Larvae of the Pacific Oyster. (Crassostrea gigas): Lab Project Number: BL5714/B: AB0503/G: Unpublished study prepared by Brixham Environmental Laboratory. 20 p. MRID 44320634.

Lowcock LA; Sharbel TF; Bonin J; Ouellet M; Rodrigue J; Desgranges J-L. 1997. Flow cytometric assay for in vivo genotoxic effects of pesticides in green frogs (Rana clamitans). Aquatic Toxicology (amsterdam); 38 (4) 1997 241-255.

Lund-Hoie K; Friestad HO. 1986. Photodegradation of the herbicide glyphosate in water. Bull Environ Contam Toxicol 1986 May;36(5):723-9.

Lund-Hoie K; Rognstad A. 1990. Effect of foliage-applied imazapyr and glyphosate on common forest weed species and norway spruce. Crop Prot; 9 (1) 1990 52-58.

Lystbaek K. 1994. Product Chemistry--GLYFOS: Lab Project Number: KLY/00.08.94/GLYFOS-USA: 6756-676/4. Unpublished study prepared by Cheminova Agro A/S. 265 p. MRID 43530001.

Ma J; Liang W. 2001. Acute Toxicity of 12 Herbicides to the Green Algae Chlorella pyrenoidosa and Scenedesmus obliquus. Bull Env Contam Toxicol.

MacKinnon DS; Freedman B. 1993. Effects of silvicultural use of the herbicide glyphosate on breeding birds of regenerating clearcuts in Nova Scotia, Canada. J. Appl. Ecol. 30(3): 395-406.

Machado-Neto JG; Bassini AJ; Aguiar LC. 2000. Safety of working conditions of glyphosate applicators on eucalyptus forests using knapsack and tractor powered sprayers. Bull Environ Contam Toxicol 2000 Mar;64(3):309-15.

Mack RB. 1993. The night the light went off in sestos. roundup (Glyphosphate) poisoning. N C Med J 1993 Jan;54(1):35-6.

Macpherson D. 1996. Glyphosate Acid: Biotransformation in the Rat: Lab Project Number: CTL/P/5058: UR0511. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 45 p. MRID 44320624.

Maibach HI. 1986. Irritation sensitization photoirritation and photosensitization assays with a glyphosate herbicide. Contact Dermatitis; 15 (3) 1986 152-156.

Mallat E; Barcelo D. 1998. Analysis and degradation study of glyphosate and of aminomethylphosphonic acid in natural waters by means of polymeric and ion-exchange solid-phase extraction columns followed by ion chromatography-post-column derivatization with fluorescence detection. J Chromatogr A. 823(1-2):129-36.

Mann RM; Bidwell JR. 1999. The toxicity of glyphosate and several glyphosate formulations to four species of southwestern australian frogs. Arch Environ Contam Toxicol 1999 Feb;36(2):193-9.
Mann RM; Bidwell JR. 1999. The toxicity of glyphosate and several glyphosate formulations to four species of southwestern australian frogs. Archives of Environmental Contamination and Toxicology; 36 (2) 1999 193-199.

Manugistics. 1995. Statgraphics Plus for Windows. Version 3. Available from Manugistics, Inc. Rockville, Maryland.

Mao J. 1996. Glyphosate Acid--Determination of Aquatic Metabolism Under Anaerobic Conditions: Interim Report: Lab Project Number: 13582.0795.6101.755: 96-6-6536. Unpublished study prepared by Springborn Labs., Inc. 76 p. MRID 44125718.

Marc J; Mulner-Lorillon O; Boulben S; Hureau D; Durand G; Belle R. 2002. Pesticide roundup provokes cell division dysfunction at the level of cdk1/cyclin b activation. Chem Res Toxicol. 15(3):326-31.

Marquardt S. 1998. Toxic Secrets, "Inert" Ingredients in Pesticides, 1987-1997. Available at www.pesticide.org/factsheets.html.

Marrs TC. 1993. Organophosphate poisoning. Pharmacol Ther 1993;58(1):51-66.

Marrs RH; Frost AJ. 1997. A microcosm approach to the detection of the effects of herbicide spray drift in plant communities. Journal of Environmental Management; 50 (4) 1997 369-388.

Marrs RH; Frost AJ; Plant RA; Lunnis P. 1993. Determination of buffer zones to protect seedlings of non-target plants from the effects of glyphosate spray drift. Agric Ecosyst Environ. 45 (3-4): 283-293.

Marselas G; Beavers J; Jaber M. 1990a. ICIA0224: A One-Generation Reproduction Study with the Bobwhite Quail (Colinus virginianus):. Lab Project Number: 123-150. Unpublished study prepared by Wildlife International Ltd. 120 p. MRID 44578301.

Marselas G; Beavers J; Jaber M. 1990b. ICIA0024: A One-Generation Reproduction Study with the Mallard (Anas platyrhynchos):. Lab Project Number: 123-151. Unpublished study prepared by Wildlife International Ltd. 120 p. MRID 44578302.

Martinez TT; Brown K. 1991. Oral and pulmonary toxicology of the surfactant USED in roundup herbicide. Proc West Pharmacol Soc 1991;34:43-6.

Martinez TT; Long WC; Hiller R. 1990. Comparison of the toxicology of the herbicide roundup by oral and pulmonary routes of exposure. Proc West Pharmacol Soc 1990;33:193-7.

Mason RW; Johnson BL. 1987. Ergonomic factors in chemical hazard control. In: Handbook of Human Factors. Salveny, G; ed. John Wiley and Sons, New York, NY. pp. 772-741.

Maule A; Wright SJL. 1984. Herbicide effects on the population growth of some green algae and cyanobacteria. J. Appl. Bacteriol. 57(2): 369-379.

McCann M. 1999. Magnitude of Glyphosate Residues in Corn Raw Agricultural Commodities Following Preharvest Application of Roundup Ultra Herbicide: Lab Project Number: MSL-14917: 1501: 97-63-R-1. Unpublished study prepared by Texas A&M University and American Agricultur. MRID 45066901.

McConnell JS; Hossner LR. 1985. pH-dependent adsorption isotherms of glyphosate. J Agric Food Chem; 33 (6) 1985 1075-1078.

McGregor D; Anderson D. 1999. DNA damage and repair in mammalian cells in vitro and in vivo as indicators of exposure to carcinogens. IARC Sci Publ 1999(146):309-54.

McGuirk R. 1999a. Glyphosate IPA Salt (NAF-552): Acute Aerosol Inhalation Toxicity Study in Fischer 344 Rats: Lab Project Number: 991076. Unpublished study prepared by The Dow Chemical Company. 63 p. {OPPTS 870.1300}. (NAF-552): Acute Aerosol Inhalation Toxicity Study in Fischer 344 Rats: Lab Project Number: 991076. Unpublished study prepared by The Dow Chemical Company. 63 p. {OPPTS 870.1300}. MRID 44863403.

McGuirk R. 1999b. Glyphosate IPA Salt (NAF-545): Acute Aerosol Inhalation Toxicity Study in Fischer 344 Rats: Lab Project Number: 991069. Unpublished study prepared by The Dow Chemical Company. 70 p. {OPPTS 870.1300}. (NAF-545): Acute Aerosol Inhalation Toxicity Study in Fischer 344 Rats: Lab Project Number: 991069. Unpublished study prepared by The Dow Chemical Company. 70 p. {OPPTS 870.1300}. MRID 44863803.

McGuirk R. 1999c. Glyphosate IPA Salt (NAF-546): Acute Aerosol Inhalation Toxicity Study in Fischer 344 Rats: Lab Project Number: 991075. Unpublished study prepared by The Dow Chemical Company. 107 p. {OPPTS 870.1300}. (NAF-546): Acute Aerosol Inhalation Toxicity Study in Fischer 344 Rats: Lab Project Number: 991075. Unpublished study prepared by The Dow Chemical Company. 107 p. {OPPTS 870.1300}. MRID 44918603.

McKee M; Hastings C. 1995. Acute Toxicity of a Herbicide to Selected Frog Species: Report Review. Unpublished study prepared by Monsanto Co. 13 p. MRID 43839602.

Menkes DB; Temple WA; Edwards IR. 1991. Intentional self-poisoning with glyphosatecontaining herbicides. Hum Exp Toxicol 1991 Mar;10(2):103-7.

Merritt CR. 1982. The influence of form of deposit on the phytotoxicity of 4-chloro-2methylphenoxyacetic acid, paraquat and glyphosate applied as individual drops. Ann Appl Biol; 101 (3) 1982 (recd 1983) 527-532. Michael JL; Neary DG. 1993. Herbicide dissipation studies in southern forest ecosystems. Environ Tox Chem. 12: 405-410.

Middendorf PJ. 1993. Forest worker exposures to glyphosate during directed foliar application of Roundup herbicide. Rep. Proj.#A-8196-000. Georgia Tech. Res. Instit., Tech. Environ. Sci and Technol. Lab. 71 p..

Mierkowski M. 1999. Product Chemistry Data to Support the Registration of MON 78063 Herbicide: Lab Project Number: RDNO 1479: MSL-15863: PC63-99-02-0. Unpublished study prepared by Monsanto Co. 63 p. {OPPTS 830.1550, 830.1620, 830.1700, 830.1750, 830.1800, 830.6302, 830.6. MRID 44872701.

Milburn G. 1996. Glyphosate Acid: One Year Dietary Toxicity Study in Rats: Lab Project Number: CTL/P/5143: PR1012. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 842 p.. MRID 44320613.

Miles CJ; Moye HA. 1988. Extraction of glyphosate herbicide from soil and clay minerals and determination of residues in soils. J Agric Food Chem; 36 (3) 1988 486-491.

Miller R. 2002a. Product Identity and Composition, Description of Materials Used to Produce the Product, Description of the Formulation Process and Discussion of the Formation of Impurities: Nufarm Credit Duo XL: Lab Project Number: CHB-RP-PVC-091-T-1. Unpublished study. MRID 45615101.

Miller R. 2002b. Preliminary Analysis, Certified Limits and Enforcement Analytical Methods: Nufarm Credit Duo XL: Lab Project Number: CHB-RP-PCV-092-T-1: CHB-OP-MTH-096-P-3. Unpublished study prepared by Phibro-Tech, Inc. 23 p. {OPPTS 830.1700, 830.1750, 830.1800}. MRID 45615102.

Miller R. 2002c. Product Properties: Color, Physical State, Odor, Storage Stability, Corrosion Characteristics, pH, Viscosity, Density: Nufarm Credit Duo XL: Lab Project Number: CHB-RP-PCV-093-T-1: CHB-OP-MTH-004-P-4: CHB-OP-MTH-001-P-7. Unpublished study prepared by Phi. MRID 45615103.

Miller JJ; Hill BD; Lindwall CW. 1995. Residue detections in soil and shallow groundwater after long-term herbicide applications in southern alberta. Canadian Journal of Soil Science; 75 (3) 1995 349-356.

Miller DK; Griffin JL; Richard E P JR. 1998. Johnsongrass (Sorghum halepense) control and rainfastness with glyphosate and adjuvants. Weed Technology; 12 (4) 1998 617-622.

Miller JH; Boyd RS; Edwards MB. 1999. Floristic diversity, stand structure, and composition 11 years after herbicide site preparation. Canadian Journal of Forest Research; 29 (7) 1999 1073-1083.

Mirvish SS.. 1995. Role of N-nitroso compounds (NOC) and N-nitrosation in etiology of gastric, esophageal, nasopharyngeal and bladder cancer and contribution to cancer of known exposures to NOC. Canter Lett. 93(1): 17-48..

Mitchell D Chapman P. 1985. Acute Toxicity of Rodeo/X-77 to Chinook Salmon: Project ID. EV-84-190. Unpublished study prepared by EVS Consultants. 22 p. MRID 40579303.

Mitchell D; Chapman P. 1985a. Acute Toxicity of Rodeo/X-77 to Rainbow Trout: Project ID. EV-84-187. Unpublished study prepared by EVS Consultants. 29 p. MRID 40579301.

Mitchell D; Chapman P. 1985b. Acute Toxicity of Rodeo/X-77 to Coho Salmon: Project ID. EV-84-188. Unpublished study prepared by EVS Consultants. 22 p. MRID 40579302.

Mitchell D; Chapman P. 1987a. Acute Toxicity of Rodeo/X-77 to Chinook Salmon: Project ID: EV-86-181. Unpublished study prepared by EVS Consultants. 21 p.. MRID 40579305.

Mitchell D; Chapman P. 1987b. Acute Toxicity of Rodeo/X-77 to Rainbow Trout: Project ID: EV-86-182. Unpublished study prepared by EVS Consultants. 21 p. MRID 40579306.

Mitchell D; Chapman P. 1987c. Seawater Challenge Testing of Chinook Salmon Smolts Following Rodeo/X-77 Herbicide Exposure: Pro- ject ID: EV-86-100. Unpublished study prepared by EVS Consultants.. MRID 40579307.

Mitchell DG; Chapman PM; Long TJ. 1987a. Acute toxicity of roundup and rodeo herbicides to rainbow trout, chinook, and coho salmon. Bull Environ Contam Toxicol. 39(6): 1028-35.

Mitchell, DG; Chapman, PM; Long, TJ.. 1987b. Seawater challenge testing of coho salmon smolts following exposure to Roundup herbicide. Environ. Toxicol. Chem. 6(11): 875-878.

Mohamed AI; Nair GA; Kassem HH; Nuruzzaman M. 1995. Impacts of pesticides on the survival and body mass of the earthworm aporrectodea caliginosa (Annelida: oligochaeta). Acta Zoologica Fennica; 0 (196) 1995 344-347.

Moldenke AR. 1992. Non-target impacts of management practices on the soil arthropod community of ponderosa pine plantations. Proc. Ann. For. Veg. Manage. Conf. pp. 78-103.

Monsanto Co. 1983. [Product Chemistry Data for Rodeo Herbicide and Glyphosate]. Unpublished compilation. 7 p.. MRID 00152394.

Monsanto Co. 1992. Submission of spray drift reports involving Roundup and Accord herbicides. Data received under FIFRA 6(a). )(2) status. Transmittal of 1 study. MRID 42317200.

Monsanto Co. 1993. Submission of Acute Toxicology Data in Support of Application for Registration of MON 65005 Herbicide. Transmittal of 3 Studies.. MRID 43035900.

Monsanto Co. 1994. Submittal of Product Chemistry Data in Support of Registration of MON-65005 Herbicide. Transmittal of 1 study.. MRID 43468600.

Monsanto Co. 1995a. Submission of Metabolism, Toxicology, and Hazard to Non-Target Organisms Data in Support of FIFRA 6(a)(2) for Roundup Herbicide. (Glyphosate). Transmittal of 4 Studies. MRID 43637300.

Monsanto Co. 1995b. Submission of Toxicology Data in Support of FIFRA 6(a)(2) for Roundup Herbicide. (Glyphosate IPA Salt). Transmittal of 1 Study. MRID 43731700.

Monsanto Co. 1997a. Submission of Metabolism and Residue Data in Support of the Registration of Roundup Herbicide. Transmittal of 1 Study.. MRID 44279300.

Monsanto Co. 1997b. Submission of Residue Data in Support of the Application for Registration of Roundup Ultra Herbicide for Treatment Over-the-Top of Roundup Ready Sugar Beets. Transmittal of 3 Studies.. MRID 44331600.

Monsanto Co. 1998a. Submission of Product Chemistry Data in Support of the Registration for Roundup Ultra Herbicide. Transmittal of 1 Study.. MRID 44465100.

Monsanto Co. 1998b. Submission of Toxicity Data in Support of the Registration for Roundup Ultra Herbicide. Transmittal of 3 Studies.. MRID 44465700.

Monsanto Co. 1998c. Submission of Toxicity Data in Support of the Registration of Roundup Ultra Herbicide. Transmittal of 3 Studies.. MRID 44538200.

Monsanto Company. 1998d. Submission of Product Chemistry and Toxicity Data in Support of the Application for Registration of Roundup UltraDry Herbicide. Transmittal of 7 Studies.. MRID 44615500.

Monsanto Company. 1998d. Submission of Residue Chemistry Data in Support of the Registration of Roundup Ultra and MON-65005 Herbicide. Transmittal of 1 Study.. MRID 44646600.

Monsanto Company. 1999a. Submission of Pesticide Use Data in Support of the Registration of Roundup Ultra Herbicide. Transmittal of 2 Studies.. MRID 44721200.

Monsanto Co. 1999b. Submission of Product Chemistry Data in Support of the Application for Registration of Roundup UltraDry Herbicide. Transmittal of 1 Study.. MRID 44757800.

Monsanto Company. 1999c. Submission of Product Chemistry Data in Support of the Registration of Roundup UltraDry and Roundup ProDry Herbicides. Transmittal of 1 Study.. MRID 44885500.

Monsanto Company. 2000a. Submission of Residue Chemistry Data in Support of the Registration of Roundup Ultra Herbicide, Roundup Ultra RT Herbicide, and Roundup UltraDry Herbicide. Transmittal of 1 Study.. MRID 45066900.

Monsanto Company. 2000b. Submission of Reduced Risk Data in Support of the Registration of Roundup Ultra Herbicide. Transmittal of 1 Study.. MRID 45290400.

Monsanto Co. 2001a. Submission of Residue and Toxicity Data in Support of the Registration of Roundup Ready Rice and the Petition for Tolerance of Glyphosate in/on Rice. Transmittal of 4 Studies.. MRID 45365000.

Monsanto Co. 2001b. Submission of Toxicity Data in Support of the Registration of Roundup Ultra Herbicide. Transmittal of 2 Studies.. MRID 45370300.

Monsanto Ag Co. 1991. Submission of an incident report accorded FIFRA 6(a)(2) involving Accord herbicide. (glyphosate). Transmittal of 1 study. MRID 42126700.

Monsanto Australia Limited. 2000. RounduprBiactive Herbicide. Monsanto Australia Limited, ACN 006 725 560, Melbourne, Victoria 3004..

Monsanto The Agricultural Group. 1995a. Toxicology and Metabolism Studies: Tier 1 Summaries. (Glyphosate). Unpublished study. 202 p. MRID 43637301.

Monsanto The Agricultural Group. 1995b. Ecotoxicological Studies: Tier 1 Summaries-Glyphosate. Unpublished study. 64 p.. MRID 43637302.

Monsanto The Agricultural Group. 1995c. Glyphosate: Mammalian Toxicology Summary. Unpublished study. 81 p.. MRID 43637303.

Monsanto The Agricultural Group. 1995d. Glyphosate Ecological Toxicology Summary. Unpublished study. 34 p.. MRID 43637304.

Moore JA. 1964. Physiology of the Amphibia. Academic Press, New York. 654 p. (Cited in USDA 1993).

Morgan MJ; Kiceniuk JW. 1992. Response of rainbow trout to a two month exposure to vision a glyphosate herbicide. Bull Environ Contam Toxicol; 48 (5) 1992 772-780.

Morgan JD, et al.. 1991. Acute avoidance reactions and behavioral responses of juvenile rainbow trout (Oncorhynchus mykiss) to Garlon 4, Garlon 3A and Vision Herbicides.. Environmental Toxicology and Chemistry. 10:73-79.

Morillo E; Undabeytia T; Maqueda C; Ramos A. 2000. Glyphosate adsorption on soils of different characteristics. influence of copper addition. Chemosphere 2000 Jan;40(1):103-7.

Morillo E; Undabeytia T; Maqueda C; Ramos A. 2002. The effect of dissolved glyphosate upon the sorption of copper by three selected soils. Chemosphere. 47(7):747-52.

Moriya M; Ohta T; Watanabe K; Miyazawa T; Kato K; Shirasu Y. 1983. Further mutagenicity studies on pesticides in bacterial reversion assay systems. Mutat Res 116:185-216.

Mortensen OS; Sorensen FW; Gregersen M; Jensen K. 2000. [poisonings with the herbicides glyphosate and glyphosate-trimesium]. Ugeskr Laeger 2000 Aug 28;162(35):4656-9.

Moxon M. 1996a. Glyphosate Acid: Developmental Toxicity Study in the Rat: Lab Project Number: CTL/P/4819: RR0690. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 367 p.. MRID 44320615.

Moxon M. 1996b. Glyphosate Acid: Developmental Toxicity Study in the Rabbit: Lab Project Number: CTL/P/5009: RB0709. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 349 p.. MRID 44320616.

Mueller MM; Rosenberg C; Siltanen H; Wartiovaara T. 1981. Fate of glyphosate and its influence on nitrogen cycling in 2 finnish agriculture soils. Bull Environ Contam Toxicol; 27 (5) 1981 724-730.

Muller MM; Rosenberg C; Siltanen H; Wartiovaara T. 1981. Fate of glyphosate and its influence on nitrogen-cycling in two finnish agriculture soils. Bull Environ Contam Toxicol. 27(5): 724-30.

Mumtaz MM; DeRosa C; Durkin PR.. 1996. Approaches and challenges in risk assessments of chemical mixtures. In: Toxicology of Chemical Mixtures. Yang, R. ed. Chicago, IL: Academic Press. pp. 565-597.

Mutter L. 1989. Acute Delayed Neurotoxicity of ICIA-0224: Lab Project Number: SA41/88: Y06380/001/001: T-12324. Unpublished study prepared by ICI Americas Inc. 109 p.. MRID 43151201.

NRC (National Research Council).. 1983. Risk assessment in the Federal government: managing the process. Washington, DC: National Academy Press; 176 p. + app..

NRC (National Research Council). 1983. Risk assessment in the Federal government: managing the process. Washington, DC: National Academy Press; 176 p. + app. .

NTP (National Toxicology Program). 1992. Toxicity studies of glyphosate. National Toxicology Program Toxicity Report Series Pg:39 P Yr:1992 Ip: Vi:16.

Nalewaja JD; Matysiak R. 1992. 2,4-D and salt combinations affect glyphosate phytotoxicity. Weed Technol; 6 (2) 1992 322-327.

Nalewaja JD; Matysiak R. 1993. Optimizing adjuvants to overcome glyphosate antagonistic salts. Weed Technol; 7 (2) 1993 337-342.

Nalewaja JD; Devilliers B; Matysiak R. 1996a. Surfactant and salt affect glyphosate retention and absorption. Weed Research; 36 (3) 1996 241-247.

Neal JC; Skroch WA. 1985. Effects of timing and rate of glyphosate application on toxicity to selected woody ornamentals. J Am Soc Hortic Sci; 110 (6) 1985 860-864.

Neary DG; Bush PB; Michaels JL.. 1993. Fate, dissipation and environmental effects of pesticides in southern forests: a review of a decade of research progress. Environ Toxicol Chem. 12: 411-28.

Nelson R. 2000a. Group B-Physical/Chemical Properties for GF-368; a Liquid End-Use Product Containing Glyphosate: Lab Project Number: NAFST326. Unpublished study prepared by Dow AgroSciences LLC. 5 p. {OPPTS 830.6302, 830.6303, 830.6304, 830.6314, 830.6315, 830.6316, 8. MRID 45278402.

Nelson R. 2000b. Group B-Physical/Chemical Properties for GF-369, a Liquid End-Use Product Containing Glyphosate: Lab Project Number: NAFST327. Unpublished study prepared by Dow AgroSciences LLC. 5 p. {OPPTS 830.6302, 830.6303, 830.6304, 830.6314, 830.6315, 830.6316, 8. MRID 45278502.

Neskovic NK; Poleksic V; Elezovic I; Karan V; Budimir M. 1996. Biochemical and histopathological effects of glyphosate on carp, cyprinus carpio l. Bull Environ Contam Toxicol 1996 Feb;56(2):295-302.

Newmaster SG; Bell FW; Vitt DH. 1999. The effects of glyphosate and triclopyr on common bryophytes and lichens in northwestern Ontario. Canadian Journal of Forest Research; 29 (7) 1999 1101-1111.

Newton M; Howard KM; Kelpsas BR; Danhaus R; Lottman CM; Dubelman S. 1984. Fate of glyphosate in an Oregon (USA) forest ecosystem. J Agric Food Chem. 32(5):1144-1151.

Newton M; Horner LM; Cowell JE; White DE; Cole EC. 1994. Dissipation of glyphosate and aminomethylphosphonic acid in north american forests. Journal of Agricultural and Food Chemistry; 42 (8) 1994 1795-1802.

Nicholson PS; Hirsch PR. 1998. The effects of pesticides on the diversity of culturable soil bacteria. Journal of Applied Microbiology; 84 (4) 1998 551-558.

Nomix-Chipman Ltd. 1999. Roundup Biactive Material Safety Data Sheet. Nomix-Chapman Ltd., Staple Hill, Bristol BS16 4PS, Australia.

Nosanchuk JD; Ovalle R; Casadevall A. 2001. Glyphosate inhibits melanization of cryptococcus neoformans and prolongs survival of mice after systemic infection. J Infect Dis 2001 Apr 1;183(7):1093-9.

Nufarm Inc. 2001. Submission of Toxicity Data in Support of the Registration of Nufarm Credit Herbicide. Transmittal of 2 Studies.. MRID 45362500.

O'Malley M. 1997. Clinical evaluation of pesticide exposure and poisonings. Lancet (north American Edition); 349 (9059) 1997 1161-1166.

O'Sullivan PA; O'Donovan JT; Hamman WM. 1981. Influence of nonionic surfactants, ammonium sulfate, water quality and spray volume on the phytotoxicity of glyphosate. Can J Plant Sci; 61 (2) 1981 391-400.

Obama K. 1996. Studies on allergic skin disease caused by pesticides in citrus growers: field survey study and animal experiments. Medical Journal of Kagoshima University; 48 (1) 1996 13-22.

Obojska A; Lejczak B; Kubrak M. 1999. Degradation of phosphonates by streptomycete isolates. Appl Microbiol Biotechnol 1999 Jun;51(6):872-6.

Obojska A; Ternan NG; Lejczak B; Kafarski P; Mcmullan G. 2002. Organophosphonate utilization by the thermophile geobacillus caldoxylosilyticus t20. Appl Environ Microbiol. 68(4):2081-4.

Ogner G. 1987a. Glyphosate application in forest ecological aspects IV. the water quality of forest brooks after routine application. Scand J Forest Res. 2(4): 499-508.

Ogner G. 1987b. Glyphosate application in forest-ecological aspects V. the water quality of forest brooks after manual clearing Or extreme glyphosate application. Scand J Forest Res. 2(4): 509-516.

Ogner G. 1987c. Glyphosate application in forest-ecological aspects. II. The quality of water leached from forest soil lysimeters. Scand J For Res. 2(4): 469-480..

Ogner G. 1987d. Glyphosate application in forest ecological aspects. II. The quality of water leached from forest soil lysimeters. Scand J Forest Res. 2(4): 469-480.

Olaleye VF; Akinyemiju OA. 1996. Effect of a glyphosate (N-(phosphonomethyl) glycine) application to control Eichhornia crassipes mart. on fish composition and abundance in Abiala creek, Niger delta, Nigeria. Journal of Environmental Management; 47 (2) 1996 115-122.

Olorunsogo OO. 1990. Modification of the transport of protons and Ca2+ ions across mitochondrial coupling membrane by n-(phosphonomethyl)glycine. Toxicology 1990 Apr 17;61(2):205-9.

Olorunsogo OO; Bababunmi EA. 1980. Inhibition of succinate-linking reduction of pyridine nucleotide in rat liver mitochondria 'in vivo' by n-(phosphonomethyl)glycine. Toxicol Lett 1980 Dec;7(2):149-52.

Olorunsogo OO; Bababunmi EA; Bassir O. 1978. Toxicity of n-(phosphonomethyl)glycine to chick embryo. Toxicology Letters. 2: 319-321.

Olorunsogo OO; Bababunmi EA; Bassir O. 1979. The inhibitory effect of N-(phosphonomethyl)glycine in vivo on energy-dependent, phosphate-induced swelling of isolated rat liver mitochondria. Tox. Letters. 4: 303-306.

Olorunsogo OO; Bababunmi EA; Bassir O. 1979a. The inhibitory effect of N-(phosphonomethyl)glycine in vivo on energy-dependent, phosphate-induced swelling of isolated rat liver mitochondria. Tox. Letters. 4: 303-306..

Olorunsogo OO; Bababunmi EA; Bassir O. 1979b. Effect of glyphosate on rat liver mitochondria in vivo. Bull Environ Contam Toxicol 1979 Jun;22(3):357-64.

Oppenhuizen M. 1993. The Terrestrial Field Dissipation of Glyphosate: Final Report: Lab Project Number: MSL-12651: 91-63-R-1: AL-91-121. Unpublished study prepared by The Agricultural Group of the Monsanto Co. and Pan-Agricultural Labs, Inc. 1244 p.. MRID 42765001.

Oppenhuizen ME; Cowell JE. 1991. Liquid chromatographic determination of glyphosate and aminomethylphosphonic acid (Ampa) in environmental water: collaborative study. J Assoc Off Anal Chem. 74(2):317-23.

Ovidi E; Gambellini G; Taddei AR; Cai G; Del Casino C; Ceci M; Rondini S; Tiezzi A. 2001. Herbicides and the microtubular apparatus of nicotiana tabacum pollen tube: immunofluorescence and immunogold labelling studies. Toxicol in Vitro 2001 Apr;15(2):143-51.

PHED Task Force. 1995. PHED: The Pesticide Handlers Exposure Database. Version 1.1. Health Canada, U.S. Environmental Protection Agecny, and American Crop Protection Association.

Palmer S; Beavers J. 1997a. MON 65005: A Dietary LC50 Study with the Mallard:. (Final Report): Lab Project Number: 1399: WL-95-259: 139-420. Unpublished study prepared by Wildlife International Ltd. 40 p. MRID 44465701.

Palmer S; Beavers J. 1997b. MON 65005: A Dietary LC50 Study with the Northern Bobwhite:. (Final Report): Lab Project Number: 139-419: WL-95-258: 1399. Unpublished study prepared by Wildlife International Ltd. 41 p. MRID 44465702.

Palmer S; Beavers J. 1997c. MON 65005: An Acute Contact Toxicity Study with the Honey Bee:. (Final Report): Lab Project Number: 139-415A: WL-95-238: 1399. Unpublished study prepared by Wildlife International Ltd. 35 p. MRID 44465703.

Palmer S; Krueger H. 2001a. MON 77360: An Acute Contact Toxicity Study with the Honey Bee: Lab Project Number: WL-97-099: 139-433. Unpublished study prepared by Wildlife International, Ltd. 20 p. MRID 45370301.

Palmer S; Krueger H. 2001b. MON 77360: An Acute Oral Toxicity Study with the Honey Bee: Lab Project Number: WL-97-100: 139-432. Unpublished study prepared by Wildlife International, Ltd. 19 p. MRID 45370302.

Parrot F; Bedry R; Favarel-Garrigues JC. 1995. Glyphosate herbicide poisoning: USE of a routine aminoacid analyzer appears to be a rapid method for determining glyphosate and its metabolite in biological fluids. J Toxicol Clin Toxicol 1995;33(6):695-8.

Patsias J; Papadopoulou A; Papadopoulou-Mourkidou E. 2001. Automated trace level determination of glyphosate and aminomethyl phosphonic acid in water by on-line anion-exchange solid-phase extraction followed by cation-exchange liquid chromatography and post-column derivatization. J Chromatogr A. 932(1-2):83-90.

Payne NJ. 1992. Off-target glyphosate from aerial silvicultural applications, and buffer zones required around sensitive areas. Pestic Sci. 34 (1): 1-8.

Payne NJ. 1993. Spray dispersal from aerial silvicultural glyphosphate applications. Crop Prot. 12 (6): 463-469.

Payne NJ; Thompson DG. 1992. Off-target glyphosate deposits from aerial silvicultural applications under various meteorological conditions. Pestic Sci. 34(1): 53-59.

Payne NJ; Feng JC; Reynolds PE. 1990. Off-target deposits and buffer zones required around water for aerial glyphosate applications. Pestic Sci. 30(2): 183-198.

Pell M; Stenberg B; Torstensson L. 1998. Potential denitrification and nitrification tests for evaluation of pesticide effects in soil. Ambio; 27 (1) 1998 24-28.

Peluso M; Merlo F; Munnia A; Bolognesi C; Puntoni R; Parodi S. 1996. 32P-postlabeling detection of DNA adducts in peripheral white blood cells of greenhouse floriculturists from Western Liguria, Italy. Cancer Epidemiology Biomarkers & Prevention; 5 (5) 1996 361-369.

Peluso M; Munnia A; Bolognesi C; Parodi S. 1998. 32P-postlabeling detection of DNA adducts in mice treated with the herbicide RoudUp. Environ Mol Mutagen 1998;31(1):55-9.

Penaloza-Vazquez A; Mena GL; Herrera-Estrella L; Bailey AM. 1995. Cloning and sequencing of the genes involved in glyphosate utilization by pseudomonas pseudomallei. Appl Environ Microbiol 1995 Feb;61(2):538-43.

Penaloza-Vazquez A; Mena GL; Herrera-Estrella L; Bailey AM. 1995. Cloning and sequencing of the genes involved in glyphosate utilization by pseudomonas pseudomallei. Applied and Environmental Microbiology; 61 (2) 1995 538-543.

Perkins PJ; Boermans HJ; Stephenson GR. 2000. Toxicity of glyphosate and triclopyr using the frog embryo teratogenesis assay--Xenopus. Environ Toxicol Chem 2000 Apr;19(4):940-5.

Perry MJ; Christiani DC. 1999. Herbicide and insecticide exposures among dairy farm pesticide applicators. American Journal of Public Health; 89 (7) 1999 1118-1119.

Perschbacher PW; Stone N; Ludwig GM; Guy C B JR. 1997. Evaluation of effects of common aerially-applied soybean herbicides and propanil on the plankton communities of aquaculture ponds. Aquaculture; 157 (1-2) 1997 117-122.

Peterson HG; Boutin C; Martin PA; Freemark KE; Ruecker NJ; Moody MJ. 1994. Aquatic phyto-toxicity of 23 pesticides applied a expected environmental concentrations. Aquatic Toxicology (Amsterdam). 28(3-4): 275-292.

Petit F; Le Goff P; Cravedi J-P; Valotaire Y; Pakdel F. 1997. Two complementary bioassays for screening the estrogenic potency of xenobiotics: recombinant yeast for trout estrogen receptor and trout hepatocyte cultures. Journal of Molecular Endocrinology; 19 (3) 1997 321-335.

Piccolo A; Celano G; Arienzo M; Mirabella A. 1994. Adsorption and desorption of glyphosate in some european soils. Journal of Environmental Science and Health Part B. 29(6): 1105-1115.

Piccolo A; Celano G; Conte P. 1996. Adsorption of glyphosate by humic substances. Journal of Agricultural and Food Chemistry; 44 (8) 1996 2442-2446.

Pinto P. 1993. Glyphosate Trimesium: 90 Day Oral Toxicity Study in Dogs: Lab Project Number: CTL/P/3962: PD0924. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 306 p.. MRID 44246704.

Pinto P. 1996. Glyphosate Acid: 21 Day Dermal Toxicity Study in Rats: Lab Project Number: CTL/P/4985: LR0574. Unpublished study prepared by Zeneca Central Toxicology Laboratory. 232 p.. MRID 44320611.

Pline WA; Wilcut JW; Duke SO; Edmisten KL; Wells R. 2002. Tolerance and accumulation of shikimic acid in response to glyphosate applications in glyphosate-resistant and nonglyphosate-resistant cotton (Gossypium hirsutum l.). J Agric Food Chem. 50(3):506-12.

Powell HA; Kerby NW; Rowell P. 1991. Natural tolerance of cyanobacteria to the herbicide glyphosate. New Phytol; 119 (3) 1991 421-426.

Pushnoy LA; Avnon LS; Carel RS. 1998. Herbicide (Roundup) pneumonitis. Chest 1998 Dec;114(6):1769-71.

Quellet M; Bonin J; Rodrigue J; Desgranges J-L; Lair S. 1997. Hindlimb deformities (Ectromelia, ectrodactyly) in free-living anurans from agricultural habitats. Journal of Wildlife Diseases; 33 (1) 1997 95-104.

Ralphs MH; Manners GD; Gardner DR. 1998. Toxic alkaloid response to herbicides used to control tall larkspur. Weed Science; 46 (1) 1998 116-119.

Rank J; Jensen AG; Skov B; Pedersen LH; Jensen K. 1993. Genotoxicity testing of the herbicide roundup and its active ingredient glyphosate isopropylamine using the mouse bone marrow micronucleus test, salmonella mutagenicity test, and allium anaphase-telophase test. Mutat Res. 300(1): 29-36.

Rank J; Jensen A-G; Skov B; Pedersen LH; Jensen K. 1993. Genotoxicity testing of the herbicide roundup and its active ingredient glyphosate isopropylamine USING the mouse bone marrow micronucleus test, salmonella mutagenicity test, and allium anaphase-telophase test. Mutation Research, Vol 300, No 1, Pages 29-36, 15 References, 1993.

Ravanel P; Liegeois MH; Chevallier D; Tissut M. 1999. Soil thin-layer chromatography and pesticide mobility through soil microstructures. New technical approach. J Chromatogr a 1999 Dec 9;864(1):145-54.

Reding M. 1995. Glyphosate and Aminomethylphosphonic Acid Residues in Cereal Grain and Straw Following Preharvest Application with Roundup Herbicide in Europe: Lab Project Number: R.D. 1326: MLL 30.177: GE 7901. Unpublished study prepared by Monsanto Co. 235 p.. MRID 43870203.

Reinbothe C; Ortel B; Parthier B; Reinbothe S. 1994. Cytosolic and plastid forms of 5enolpyruvylshikimate-3-phosphate synthase in euglena gracilis are differentially expressed during light-induced chloroplast development. Mol Gen Genet 1994 Dec 1;245(5):616-22.

Reinert KH; Rodgers JH. 1987. Fate and persistence of aquatic herbicides. Rev Environ Contam Toxicol 98: 86-98.

Reyna M. 1985. Twelve Month Study of Glyphosate Administered by Gelatin Capsule to Beagle Dogs: Project No. ML-83-137: Study No. 830116. Unpublished study prepared by Monsanto Company Environ- mental Health. 317 p. MRID 00153374.

Reynolds PE; Scrivener JC; Holtby LB; Kingsbury PD. 1993. Review and synthesis of Carnation Creek herbicide research. For Chron. 69(3): 323-330.

Riechers DE; Wax LM; Liebl RA; Bullock DG. 1995. Surfactant effects on glyphosate efficacy. Weed Technology; 9 (2) 1995 281-285.

Ritchie DC; Harestad AS; Archibald R. 1987. Glyphosate treatment and deer mice in clearcut and forest. Northwest Sci. 61(3): 199-202..

Roberts F; Roberts CW; Johnson JJ; Kyle DE; Krell T; Coggins JR; Coombs GH; Milhous WK; Tzipori S; Ferguson DJ; Chakrabarti D; Mcleod R. 1998. Evidence for the shikimate pathway in apicomplexan parasites. Nature 1998 Jun 25;393(6687):801-5.

Roberts CW; Roberts F; Lyons RE; Kirisits MJ; Mui EJ; Finnerty J; Johnson JJ; Ferguson DJ; Coggins JR; Krell T; Coombs GH; Milhous WK; Kyle DE; Tzipori S; Barnwell J; Dame JB; Carlton J; Mcleod R. 2002. The shikimate pathway and its branches in apicomplexan parasites. J Infect Dis. 185 Suppl 1:S25-36.

Robinson DE; Mansingh A. 1999. Insecticide contamination of Jamaican environment. IV. Transport of residues from coffee plantations in the blue mountains to coastal waters in eastern Jamaica. Environmental Monitoring and Assessment; 54 (2) 1999 125-141.

Rodriguez B. 1996. Environmental Fate of Glyphosate. Environmental Monitoring and Pest Management, Dept. Pesticide Regulation, Sacramento, CA.

Rodwell DE; Tasker EJ; Blair AM; et al. 1980a. Teratology Study in Rats: IRDC No. 401-054. (Unpublished study including IRDC no. 999-021; received May 23, 1980 under 524-308; prepared by International Research and Development Corp., submitted by Monsanto Co., Washington, D.C.; CDL:242516-A). (Unpublished study including IRDC no. 999-021; received May 23, 1980 under 524-308; prepared by International Research and Development Corp., submitted by Monsanto Corp., submitted by Monsanto Co., Washington, D.C.; CDL:242516-A). MRID 00046362.

Rodwell DE; Tasker EJ; Blair M; et al. 1980b. Teratology Study in Rabbits: IRDC No. 401-056. (Unpublished study received May 23, 1980 under 524-308; prepared by International Research and Development Corp., submitted by Monsanto Co., Washington, D.C.; CDL:242516-B). (Unpublished study received May 23, 1980 under 524-308; prepared by International Research and Development Corp., submitted by Monsanto Co., Washington, D.C.; CDL:242516-B). (Unpublished study received May 23, 1980 under 524-308; prepared by International Research and Development Corp., submitted by Monsanto Co., Washington, D.C.; CDL:242516-B). MRID 00046363.

Roshon RD; Mccann JH; Thompson DG; Stephenson GR. 1999. Effects of seven forestry management herbicides on myriophyllum sibiricum, as compared with other nontarget aquatic organisms. Canadian Journal of Forest Research; 29 (7) 1999 1158-1169.

Roy DN; Konar SK; Banerjee S; Charles DA; Thompson DG; Prasad R. 1989. Persistence movement and degradation of glyphosate in selected canadian boreal forest soils. J Agric Food Chem; 37 (2) 1989 437-440.

Rubin L. 2000. Memo to the file: Glyphosate. USDA, APHIS, Riverdale MD...

Rueppel ML; Brightwell BB; Schaefer J; Marvel JT. 1977. Metabolism and degradation of glyphosate in soil and water. J Agric Food Chem 25(3): 517-528; 1977(45 References).

Ruffle B; Burmaster DE; Anderson PD; Gordon HD. 1994. Lognormal distributions for fish consumption by the general U.S. population. Risk Analy. 14(4): 395-404.

Russo VM. 1990. Reaction of tomato cultivars to a sublethal dose of glyphosate. Hortscience; 25 (12) 1990 1662.

SERA (Syracuse Environmental Research Associates). 1996. SELECTED COMMERCIAL FORMULATIONS OF GLYPHOSATE - ACCORD, RODEO, ROUNDUP and ROUNDUP PRO. SERA TR 96-22-02-01c, Final Report dated June 30, 1996.

SERA (Syracuse Environmental Research Associates, Inc.). 2001a. Preparation of Environmental Documentation and Risk Assessments. SERA MD 2001-01a, draft dated July 2001. Syracuse Environmental Research Associates, Inc., Fayetteville, NY. Available at www.sera-inc.com.

SERA (Syracuse Environmental Research Associates, Inc.). 2001b. Documentation for Worksheets Version 2.01 - Human Health and Ecological Risk Assessments. SERA WSD 01-2.01, report dated October 13, 2001. Syracuse Environmental Research Associates, Inc., Fayetteville, NY. Available at www.sera-inc.com.

Saenz ME; Di Marzio WD; Alberdi JL; Del Carmen Tortorelli M. 1997. Effects of technical grade and a commercial formulation of glyphosate on algal population growth. Bull Environ Contam Toxicol. 59(4):638-44.

Samsoe-Petersen L. 1995. Effects of 67 herbicides and plant growth regulators on the rove beetle Aleochara bilineata (Col.: staphylinidae) in the laboratory. Entomophaga; 40 (1) 1995 95-104.

Sannino F; Gianfreda L. 2001. Pesticide influence on soil enzymatic activities. Chemosphere. 45(4-5):417-25.

Santillo DJ. 1994. Observations on moose, Alces alces, habitat and use on herbicide-treated clearcuts in Maine. Canadian Field-Naturalist. 108(1): 22-25.

Santillo DJ; Brown PW; Leslie D M JR. 1989a. Response of songbirds to glyphosate-induced habitat changes on clearcuts. J Wildl Manage. 53(1): 64-71.

Santillo DJ; Leslie D M JR; Brown PW. 1989b. Responses of small mammals and habitat to glyphosate application on clearcuts. J Wildl Manage; 53 (1) 1989 164-172.

Santos AM; Flores. 1995. Effects of glyphosate on nitrogen fixation of free-living heterotrophic bacteria.. Letters in Applied Microbiology. 20:349-352.

Savitz DA; Arbuckle T; Kaczor D; Curtis KM. 1997. Male pesticide exposure and pregnancy outcome. Am J Epidemiol 1997 Dec 15;146(12):1025-36.

Sawada Y; Nagai Y; Ueyama M; Yamamoto I. 1988. Probable toxicity of surface-active agent in commercial herbicide containing glyphosate. Lancet 1988 Feb 6;1(8580):299.

Schabenberger O; Tharp BE; Kells JJ; Penner D. 1999. Statistical tests for hormesis and effective dosages in herbicide dose response. Agronomy Journal; 91 (4) 1999 713-721.

Schiffman SS; Suggs MS; Abou Donia MB; Erickson RP; Nagle HT. 1995. Environmental pollutants alter taste responses in the gerbil. Pharmacol Biochem Behav 1995 Sep;52(1):189-94.

Schneider F; Edmiston S; Spencer J; Cowan C; Hernandez B; Orr M 1999. 1999. Exposure of hand applicators to glyphosate in forest settings, 1995. HS-1764. 28 p. Pesticide and Environmental Toxicology Section, Office of Environmental Health Hazard Assessment, CalEPA. December 1997. 19p..

Schonbrunn E; Eschenburg S; Shuttleworth WA; Schloss JV; Amrhein N; Evans JN; Kabsch W. 2001. Interaction of the herbicide glyphosate with its target enzyme 5-enolpyruvylshikimate 3-phosphate synthase in atomic detail. Proc Natl Acad Sci U S a 2001 Feb 13;98(4):1376-80.

Schonherr J. 2002. A mechanistic analysis of penetration of glyphosate salts across astomatous cuticular membranes. Pest Manag Sci. 58(4):343-51.

Schrader KK; De Regt MQ; Tidwell PD; Tucker CS; Duke SO. 1998. Compounds with selective toxicity towards the off-flavor metabolite-producing cyanobacterium oscillatoria cf. chalybea. Aquaculture; 163 (1-2) 1998 85-99.

Schroeder RE; Hogan GK. 1981. A Three-generation Reproduction Study with Glyphosate in Rats: Project No. 77-2063. (Unpublished study received Sep 22, 1981 under 524-308; prepared by Bio/dynamics, Inc., submitted by Monsanto Co., Washington, D.C.; CDL:245909-A). MRID 00081674.

Schuette J. 1998. Environmental fate of glyphosate. 13 pages.. California EPA, Environmental Monitoring and Pest Managment.

Schuytema GS; Nebeker AV; Griffis WL. 1994. Effects of dietary exposure to forest pesticides on the brown garden snail helix aspersa mueller. Archives Environmental Contamination Toxicology. 26(1): 23-28.

Schweinsberg F; Abke W; Rieth K; Rohmann U; Zullei-Seibert N. 1999. Herbicide use on railway tracks for safety reasons in germany?. Toxicol Lett 1999 Jun 30;107(1-3):201-5.

Segawa R; Bradley A; Lee P; Tran D; Hsu J; White J; Goh KS. 1997. Residues of forestry herbicides in plants of importance to California native americans. Bulletin of Environmental Contamination and Toxicology; 59 (4) 1997 556-563.

Selvapandiyan A; Majumder K; Fattah FA; Ahmad S; Arora N; Bhatnagar RK. 1995. Point mutation of a conserved arginine (104) to lysine introduces hypersensitivity to inhibition by glyphosate in the 5-enolpyruvylshikimate-3-phosphate synthase of bacillus subtilis. Febs Lett 1995 Oct 30;374(2):253-6.

Servizi JA; Gordon RW; Martens DW. 1987. Acute toxicity of Garlon 4 and Roundup herbicides to salmon, daphnia, and trout. Bull Environ Contam Toxicol. 39(1): 15-22.

Shinabarger DL; Braymer HD. 1986. Glyphosate catabolism by Pseudomonas sp. strain pg2982. J Bacteriol 1986 Nov;168(2):702-7.

Shirai N; Momma K; Ozawa S; Hashimoto W; Kito M; Utsumi S; Murata K. 1998. Safety assessment of genetically engineered food: detection and monitoring of glyphosate-tolerant soybeans. Biosci Biotechnol Biochem 1998 Jul;62(7):1461-4.

Shuma JM; Quick WA; Raju M VS; Hsiao AI. 1995. Germination of seeds from plants of avena fatua l. treated with glyphosate. Weed Research; 35 (4) 1995 249-255.

Sidhu SS; Chakravarty P. 1990. Effect of selected forestry herbicides on ectomycorrhizal development and seedling growth of lodgepole pine and white spruce under controlled and field environment. Eur J for Pathol; 20 (2) 1990 77-94.

Siltanen H; Rosenberg C; Raatikainen M; Raatikainen T. 1981. Triclopyr, glyphosate and phenoxyherbicide residues in cowberries, bilberries and lichen. Bull Environ Contam Toxicol. 27(5): 731-7.

Simenstad CA; Cordell JR; Tear L; Weitkamp LA; Paveglio FL; Kilbride KM; Fresh KL; Grue CE. 1996. Use of rodeo and X-77 spreader to control smooth cordgrass (Spartina alterniflora) in a southwestern washington estuary: 2. Effects on benthic microflora and invertebrates. Environmental Toxicology and Chemistry; 15 (6) 1996 969-978.

Singh BK; Shaner DL. 1998. Rapid determination of glyphosate injury to plants and identification of glyphosate-resistant plants. Weed Technology; 12 (3) 1998 527-530.

Singh S; Yadav N. 1978. Toxicity of Some Herbicides to Major Carp Fingerlings. Indian J. Ecol. 5(2):141-147. MRID 45088217.

Smith GR. 2001. Effects of acute exposure to a commercial formulation of glyphosate on the tadpoles of two species of anurans. Bull Environ Contam Toxicol. 67(4):483-8.

Smith AE; Aubin AJ. 1993. Degradation of 14C-glyphosate in Saskatchewan soils. Bull Environ Contam Toxicol 1993 Apr;50(4):499-505.

Smith EA; Oehme FW. 1992. The biological activity of glyphosate to plants and animals: a literature review. Vet Hum Toxicol 1992 Dec;34(6):531-43.

Smith NJ; Martin RC; St Croix RG. 1996. Levels of the herbicide glyphosate in well water. Bull Environ Contam Toxicol 1996 Nov;57(5):759-65.

Smith NJ; Martin RC; St Croix RG. 1996. Levels of the herbicide glyphosate in well water. Bulletin of Environmental Contamination and Toxicology; 57 (5) 1996 759-765.

Smith EA; Prues SL; Oehme FW. 1997. Environmental degradation of polyacrylamides. II. effects of environmental (Outdoor) exposure. Ecotoxicol Environ Saf 1997 Jun;37(1):76-91.

Smyth D; Kent S; Morris D; et al. 1995. Glyphosate Acid: Acute Toxicity to the Green Alga (Selenastrum capricornutum): Lab Project Number: BL5550/B: AB0503/B. Unpublished study prepared by Zeneca Limited. 24 p. (Selenastrum capricornutum): Lab Project Number: BL5550/B: AB0503/B. Unpublished study prepared by Zeneca Limited. 24 p. MRID 44320637.

Smyth D; Kent S; Morris D; et al. 1996a. Glyphosate Acid: Acute Toxicity to Duckweed (Lemna gibba): Lab Project Number: BL5662/B: AB0503/L. Unpublished study prepared by Zeneca Limited. 24 p. MRID 44320638.

Smyth D; Kent S; Morris D; et al. 1996b. Glyphosate Acid: Acute Toxicity to the Marine Alga (Sleletonema(sic) costatum): Lab Project Number: BL5684/B: AB0503/I. Unpublished study prepared by Zeneca Limited. 25 p. MRID 44320640.

Smyth D; Kent S; Morris D; et al. 1996c. Glyphosate Acid: Toxicity to the Freshwater Diatom (Navicula pelliculosa): Lab Project Number: BL5673/B: AB0503/K. Unpublished study prepared by Zeneca Limited. 25 p. MRID 44320641.

Smyth D; Shillabeer N; Morris D; et al. 1996d. Glyphosate Acid: Acute Toxicity to Blue-Green Alga (Anabaena flos-aquae): Lab Project Number: BL5698/B: AB0503/J. Unpublished study prepared by Zeneca Limited. 25 p. MRID 44320639.

Smyth D; Kent S; Shillabeer N. 1997a. Glyphosate Trimesium: Toxicity to Duckweed (Lemna Gibba): Lab Project Number: AD0240/E. Unpublished study prepared by Zeneca Ltd. 24 p.. MRID 44578303.

Smyth D; Kent S; Shillabeer N. 1997b. Glyphosate Trimesium: Toxicty to the Marine Alga: Skeletonema costatum: Lab Project Number: AD0240/C. Unpublished study prepared by Zeneca Ltd. 25 p.. MRID 44578304.

Smyth D; Kent S; Shillabeer N. 1997c. Glyphosate Trimesium: Toxicity to the Blue-Green Alga: Anabaena flos-aquae: Lab Project Number: AD0240/B. Unpublished study prepared by Zeneca Ltd. 25 p. MRID 44578305.

Smyth D; Kent S; Shillabeer N. 1997d. Glyphosate Trimesium: Toxicity to the freshwater diatom Navicula pelliculosa: Lab Project Number: AD0240/D. Unpublished study prepared by Zeneca Ltd. 25 p. MRID 44578306.

Solberg KL; Higgins KF. 1993. Effects of glyphosate herbicide on cattails, invertebrates, and waterfowl in south dakota wetlands. Wildlife Society Bulletin. 21(3): 299-307.

Sorensen FW; Gregersen M. 1999. Rapid lethal intoxication caused by the herbicide glyphosate-trimesium (Touchdown). Hum Exp Toxicol 1999 Dec;18(12):735-7.

Springett JA; Gray R AJ. 1992. Effect of repeated low doses of biocides on the earthworm aporrectodea-caliginosa in laboratory culture. Soil Biol Biochem. 24 (12): 1739-1744.

Stasiak MA; Hofstra G; Payne NJ; Prasad R; Fletcher RA. 1991. Alterations of growth and shikimic acid levels by sublethal glyphosate applications on pin cherry and trembling aspen. Can J for Res; 21 (7) 1991 1086-1090.

Stebbins K; Brooks K. 1999a. Glyphosate IPA Salt (NAF-552): Acute Oral Toxicity Study in Fischer 344 Rats: Lab Project Number: 991048. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 1100}. (NAF-552): Acute Oral Toxicity Study in Fischer 344 Rats: Lab Project Number: 991048. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 1100}. MRID 44863401.

Stebbins K; Brooks K. 1999b. Glyphosate IPA Salt (NAF-552): Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: 991049. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1200}. (NAF-552): Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: 991049. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1200}. MRID 44863402.

Stebbins K; Brooks K. 1999c. Glyphosate IPA Salt (NAF-552): Acute Eye Irritation Study in New Zealand White Rabbits: Lab Project Number: 991052. Unpublished study prepared by The Dow Chemical Company. 16 p. {OPPTS 870.2400}. (NAF-552): Acute Eye Irritation Study in New Zealand White Rabbits: Lab Project Number: 991052. Unpublished study prepared by The Dow Chemical Company. 16 p. {OPPTS 870.2400}. MRID 44863404.

Stebbins K; Brooks K. 1999d. Glyphosate IPA Salt (NAF-552): Acute Dermal Irritation Study in New Zealand White Rabbits: Lab Project Number: 991051. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2500}. (NAF-552): Acute Dermal Irritation Study in New Zealand White Rabbits: Lab Project Number: 991051. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2500}. MRID 44863405.

Stebbins K; Brooks K. 1999e. Glyphosate IPA Salt (NAF-552): Dermal Sensitization Study in Hartley Albino Guinea Pigs: Lab Project Number: 991050. Unpublished study prepared by The Dow Chemical Company. 37 p. {OPPTS 870.2600}. (NAF-552): Dermal Sensitization Study in Hartley Albino Guinea Pigs: Lab Project Number: 991050. Unpublished study prepared by The Dow Chemical Company. 37 p. {OPPTS 870.2600}. MRID 44863406.

Stebbins K; Brooks K. 1999f. Glyphosate IPA Salt (NAF-545): Dermal Sensitization Study in Hartley Albino Guinea Pigs: Lab Project Number: 991036. Unpublished study prepared by The Dow Chemical Company. 36 p. OPPTS 870.2600}. (NAF-545): Dermal Sensitization Study in Hartley Albino Guinea Pigs: Lab Project Number: 991036. Unpublished study prepared by The Dow Chemical Company. 36 p. OPPTS 870.2600}. MRID 44863806.

Stebbins K; Brooks K. 1999g. Glyphosate IPA Salt (NAF-546): Acute Oral Toxicity Study in Fischer 344 Rats: Lab Project Number: 991053. Unpublished study prepared by The Dow Chemical Company. 66 p. {OPPTS 870.1100}. (NAF-546): Acute Oral Toxicity Study in Fischer 344 Rats: Lab Project Number: 991053. Unpublished study prepared by The Dow Chemical Company. 66 p. {OPPTS 870.1100}. MRID 44918601.

Stebbins K; Brooks K. 1999h. Glyphosate IPA Salt (NAF-546): Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: 991054. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1200}. (NAF-546): Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: 991054. Unpublished study prepared by The Dow Chemical Company. 41 p. {OPPTS 870.1200}. MRID 44918602.

Stebbins K; Brooks K. 1999i. Glyphosate IPA Salt (NAF-546): Acute Dermal Irritation Study in New Zealand White Rabbits: Lab Project Number: 991056. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2500}. (NAF-546): Acute Dermal Irritation Study in New Zealand White Rabbits: Lab Project Number: 991056. Unpublished study prepared by The Dow Chemical Company. 15 p. {OPPTS 870.2500}. MRID 44918605.

Stebbins K; Brooks K. 1999j. Glyphosate IPA Salt (NAF-546): Dermal Sensitization Potential Study in Hartley Albino Guinea Pigs: Lab Project Number: 991055. Unpublished study prepared by The Dow Chemical Company. 37 p. {OPPTS 870.2600}. (NAF-546): Dermal Sensitization Potential Study in Hartley Albino Guinea Pigs: Lab Project Number: 991055. Unpublished study prepared by The Dow Chemical Company. 37 p. {OPPTS 870.2600}. MRID 44918606.

Steinmetz J. 1999. Magnitude of Glyphosate Residues in Alfalfa and Clover Raw Agricultural Commodities Following Broadcast Application of Roundup Ultra Herbicide: Lab Project Number: 97-63-R-3: MSL-14991: 1531. Unpublished study prepared by Monsanto Company. 510 p. {OPPT. MRID 45365401.

Stevens R. 1999. Summary of Physical/Chemical Properties: Self Certification Per EPA PR Notice 98-1: Glyfos AU Herbicide. Unpublished study prepared by Cheminova Agro A/S. 5 p. {OPPTS 830.6302, 830.6303, 830.6304, 830.6314, 830.63198, 830.7300}. MRID 45043201.

Stevens R. 2000a. GLYFOS AU Herbicide (CHA 4509) Summary of Physical/Chemical Properties Self-Certification Per EPA PR Notice 98-1. Unpublished study prepared by Cheminova A/S. 5 p. {OPPTS 830.6302, 830.6303, 830.6304, 830.6314, 830.6315, 830.6316, 830.6317, 830.6319, 83. (CHA 4509) Summary of Physical/Chemical Properties Self-Certification Per EPA PR Notice 98-1. Unpublished study prepared by Cheminova A/S. 5 p. {OPPTS 830.6302, 830.6303, 830.6304, 830.6314, 830.6315, 830.6316, 830.6317, 830.6319, 830.6320, 830.7000, 83.

Stevens R. 2000b. GLYFOS AU Herbicide (CHA 4509): Description of Formulation Process Discussion of the Formation of Impurities. Unpublished study prepared by Cheminova A/S. 7 p. {OPPTS 830.1650, 830.1670}. (CHA 4509): Description of Formulation Process Discussion of the Formation of Impurities. Unpublished study prepared by Cheminova A/S. 7 p. {OPPTS 830.1650, 830.1670}. MRID 45181303.

Stevens R. 2000c. GLYFOS AU Herbicide (CHA 4509): Description of Formulation Process Discussion of the Formation of Impurities. Unpublished study prepared by Cheminova A/S. 7 p. {OPPTS 830.1650, 830.1670}. (CHA 4509): Description of Formulation Process Discussion of the Formation of Impurities. Unpublished study prepared by Cheminova A/S. 7 p. {OPPTS 830.1650, 830.1670}. MRID 45181303.

Stone JF; Hanna M; Guo C; Imerman P. 2001. Protective headgear for midwestern agriculture: a limited wear study. J Environ Health 2001 Mar;63(7):13-9, 21.

Stout L. 1991. One Month Study of AMPA Administered by Capsule to Beagle Dogs: Lab Project Number: 90074: ML/90/186: 1251. Unpublished study prepared by Monsanto Environmental Health Lab. 154 p.. MRID 43334701.

Stratton GW; Stewart KE. 1991. Effects of the herbicide glyphosate on nitrogen cycling in an acid forest soil. Water Air Soil Pollut; 60 (3-4) 1991 231-248.

Stratton GW; Stewart KE. 1992. Glyphosate effects on microbial biomass in a coniferous forest soil. Environ Toxicol Water Qual. 7(3): 223-236.

Street R. 1982. Letter sent to R. Taylor dated Jul 6, 1982: Roundup herbicide: Addendum to pathology report for a three- generation reproduction study in rats with glyphosate.. (Unpublished study received Jul 7, 1982 under 524-308; submitted by Monsanto Co., Washington, DC; CDL:247793-A). MRID 00105995.

Strek G; Spaan WP. 1997. Wind erosion control with crop residues in the Sahel. Soil Sci Soc Am J. 61(3): 911-917.

Strek G; Stein A. 1997. Mapping wind-blown mass transport by modeling variability in space and time. Soil Sci Soc Am. J. 61(1): 232-239.

Sudo K; Kiuchi N; Mizuyama K; Tai T; Yamashite M. 1987. The corrosive effect of Roundup on the stomach and small intestine of dosg [abstract]. Gekkan Yakuji. 29:13. (Cited in Talbot et al. 1991)..

Sullivan TP. 1990. Influence of forest herbicide on deer mouse and Oregon vole population dynamics. J Wildl Manage. 54(4): 566-576.

Sullivan DS; Sullivan TP; Bisalputra T. 1981. Effects of Roundup herbicide on diatom populations in the aquatic environment of a coastal forest. Bull Environ Contam Toxicol. 26(1): 91-6.

Sullivan TP; Sullivan DS; Lautenschlager RA; Wagner RG. 1997. Long-term influence of glyphosate herbicide on demography and diversity of small mammal communities in coastal coniferous forest. Northwest Science; 71 (1) 1997 6-17.

Sullivan TP; Wagner RG; Pitt DG; Lautenschlager RA; Chen DG. 1998a. Changes in diversity of plant and small mammal communities after herbicide application in sub-boreal spruce forest. Canadian Journal of Forest Research; 28 (2) 1998 168-177.

Sullivan TP; Sulivan DS; Hogue EJ; Lautenschlager RA; Wagner RG. 1998b. Population dynamics of small mammals in relation to vegetation management in orchard agroecosystems: compensatory responses in abundance and biomass\*. Crop Protection; 17 (1) 1998 1-11.

Sumpter JP. 1998. Xenoendocrine disruptors--environmental impacts. Toxicol Lett. 102/103: 337-42.

Sun F; Wong SS; Li GC. 1998. Evaluating the effect of fentin acetate and glyphosate on reproduction of scapholeberis kingi sars. Plant Protection Bulletin (taichung); 40 (1) 1998 73-79.

Sundaram A. 1990. Effect of a nalco-trol II on bioavailability of glyphosate in laboratory trials. J Environ Sci Health Part B Pestic Food Contam Agric Wastes; 25 (3) 1990 309-332.

Sundaram A; Sundaram K MS. 1997. Solubility products of six metal-glyphosate complexes in water and forestry soils, and their influence on glyphosate toxicity to plants. Journal of Environmental Science and Health Part b Pesticides Food Contaminants and Agricultural Wastes; 32 (4) 1997 583-598.

Sundaram A; Leung JW; Webster G RB; Nott R; Curry J; Sloane L. 1996. Effect of glycerol on spreading and drying of vision droplets containing Silwet 1-77: Relevance to rainfastness and herbicidal activity of glyphosate on trembling aspen Populus tremuloides michx.). Journal of Environmental Science and Health Part b Pesticides Food Contaminants and Agricultural Wastes; 31 (4) 1996 901-912.

Szarek J; Siwicki A; Andrzejewska A; Terech-Majewska E; Banaszkiewicz T. 2000. Effects of the herbicide Roundup on the ultrastructural pattern of hepatocytes in carp (Cyprinus carpio). Mar Environ Res. 50(1-5):263-6.

Tai T; Yamishita M. 19??a. The Effect of Roundup on the Isolated Rat Cardiac Muscles. Unpublished study prepared by University of Tsukuba and The Ministry of Labour. (Japan). 17 p. MRID 43674201.

Tai T; Yamishita M. 19??b. The Effect of Roundup to the Coronary Artery on the Perfused Rat Heart. Unpublished study prepared by University of Tsukuba and Ministry of Labor. (Japan). 17 p. MRID 43674202.

Tai T; Yamashita M; Wakimori H. 1994a. The 5th World Congress of the World Federation of Associations of Clinical Toxicology Centers & Poison Control Centers: Abstract Book: The Effects of Roundup on the Isolated Rat Cardiac Muscles: (Summary): Lab Project Number: G11.. Unpublished study prep(Japan); University of Tsukuba; and Monsanto Japan Ltd. 2 p. MRID 43673901.

Tai T; Yamashita M; Wakimori H. 1994b. The 5th World Congress of the World Federation of Associations of Clinical Toxicology Centers & Poison Control Centers: Abstract Book: The Effects of Roundup to the Coronary Artery on the Perfused Rat Heart: (Summary): Lab Project Number: G12. Unpublish. (Japan); University of Tsukuba; and Monsanto Japan Ltd. 2 p. MRID 43673902.

Tai T; Tamashita M; Wakimori H. 1996. No title provided. Jpn. J. Toxicol. 3:63. (Cited in Martinez and Brown 1991).

Talbot AR; Shiaw M-H; Huang J-S; Yang S-F; Goo T-S; Wang S-H; Chen C-L; Sanford TR. 1991. Acute poisoning with a glyphosate-surfactant herbicide (Round-up): A review of 93 cases. Hum Exp Toxicol; 10 (1) 1991 1-8.

Tate TM; Spurlock JO; Christian FA. 1997. Effect of glyphosate on the development of Pseudosuccinea columella snails. Archives of Environmental Contamination and Toxicology; 33 (3) 1997 286-289.

Tate TM; Jackson RN; Christian FA. 2000. Effects of glyphosate and dalapon on total free amino acid profiles of Pseudosuccinea columella snails. Bull Environ Contam Toxicol 2000 Feb;64(2):258-62.

Taylor M. 1993. Product Chemistry Data to Support the Registration of MON-65005 Herbicide: Lab Project Number: 0469-93-5: MSL-12751: MSL-12911. Unpublished study prepared by Monsanto Co. 73 p. MRID 43020901.

Taylor M. 1994. Product Chemistry Data for MON 65005 Herbicide: Storage Stability and Corrosion Characteristics: Addendum: Lab Project Number: MSL-13618: 0469-93-5-D: MSL-12751.. Unpublished study prepared by The Agricultural Group of Monsanto Company. 26 p. MRID 43468601.

Temple WA; Smith NA. 1992. Glyphosate herbicide poisoning experience in New zealand. N Z Med J 1992 May 13;105(933):173-4.

Teske ME; Curbishley TB. 1990. Forest Service Aerial Spray Computer Model, FSCBF 4.0 User Manu. Continuum Dynamics, Inc, Princeton, NJ. CDI Report No. 90-06.

Teske ME; Bird SL; Esterly DM; Ray SL; Perry SG. 2001. A User's Guide for AgDRIFT 2.0: A Tiered Approach for the Assessment of Spray Drift. Continuum Dynamics, Inc. Public Use Version. C.D.I. Report No. 01-01. Available, with executable model at: http://www.agdrift.com/.

Texas Dept of Agriculture. 1992. Submission of a Report of the Investigation of a Complaint of Adverse Effects of Garlon 3A and Rodeo on Horses and Property in Accordance with FIFRA 6. 6(a)2 Requirements. Transmittal of 1 study. MRID 42305500.

Texas Dept of Agriculture. 1992. Report of an Investigation Concerning Alleged Adverse Effects of Garlon 3A and Rodeo on Horses and Property: Lab Project Number: 10-91-0075. Unpublished study. 31 p.. MRID 42305501.

Thelen KD; Jackson EP; Penner D. 1995a. The basis for the hard-water antagonism of glyphosate activity. Weed Science; 43 (4) 1995 541-548.

Thelen KD; Jackson EP; Penner D. 1995b. Utility of nuclear magnetic resonance for determining the molecular influence of citric acid and an organosilicone adjuvant on glyphosate activity. Weed Science; 43 (4) 1995 566-571.

Thelen KD; Jackson EP; Penner D. 1995c. 2,4-D interactions with glyphosate and sodium bicarbonate. Weed Technology; 9 (2) 1995 301-305.

Thompson DG; Pitt DG; Buscarini T; Staznik B; Thomas DR; Kettela EG. 1994. Initial deposits and persistence of forest herbicide residues in sugar maple (Acer saccharum) foliage. Canadian Journal of Forest Research. 24(11): 2251-2262.

Thompson DG: Pitt DG; Buscarini TM; Staznik B; Thomas DR. 2000. Comparative fate of glyphosate and triclopyr herbicides in the forest floor and mineral soil of an Acadian forest regeneration site. Canadian Journal of Forest Research. 30: 1808-1816.

Tominack RL; Yang GY; Tsai WJ; Chung HM; Deng JF. 1991. Taiwan national poison center survey of glyphosate--surfactant herbicide ingestions. J Toxicol Clin Toxicol 1991;29(1):91-109.

Tompkins E. 1991. 90-Day Oral (Capsule) Toxicity Study in Dogs with AMPA: Final Report: Lab Project Number: 50173: WI/90/354: 1251. Unpublished study prepared by WIL Research Labs., Inc. 670 p. (Capsule) Toxicity Study in Dogs with AMPA: Final Report: Lab Project Number: 50173: WI/90/354: 1251. Unpublished study prepared by WIL Research Labs., Inc. 670 p. MRID 43334702.

Tompkins J. 2000. Herbicide Branch, OPP/TS. U.S. EPA. Undated Letter to Leslie Rubin, USDA/APHIS...

Tooby TE; Lucey J; Stott B. 1980. The tolerance of grass carp, Ctenopharyngodon idella val. to aquatic herbicides. J Fish Biol. 16(5): 591-597.

Torstensson NT; Lundgren LN; Stenstrom J. 1989. Influence of climatic and edaphic factors on persistence of glyphosate and 2,4-D in forest soils. Ecotoxicol Environ Saf 1989 Oct;18(2):230-9.

Tria J. 1994. Vapor Pressure Determination for Isopropylamine-Glyphosate Salt: Lab Project Number: MSL/9762: 9762: 1276. Unpublished study prepared by Monsanto Agricultural Co. 27 p.. MRID 43434505.

Trumbo J. 2002. An assessment of the non-target aquatic impacts of the herbicide Rodeo and the surfactant R-11 when used to control purple loosestrife, Lythrum salicaria. California Department of Fish and Game, Office of Spill Prevention and Response, Administrative Report 02-011.

U.S. EPA, Integrated Risk Information System. 1990. IRIS Entry for Glyphosate. Originally posted on 09/01/1990. Available at www.epa.gov/iris/.

U.S. EPA, Integrated Risk Information System. 1992. IRISEntry for 1,4-Dioxane. IRIS Summary www.epa.gov/iris/.

U.S. EPA/ODW (U.S. Environmental Protection Agency, Office of Drinking Water). 1992. Drinking water criteria document for glyphosate. Us Epa, Office of Assistant Administrator, Washington, Dc Ntis/pb92-173392.

U.S. EPA/ODW (U.S. Environmental Protection Agency, Office of Drinking Water). 2002. Technical Factsheet on: GLYPHOSATE. Available at: www.epa.gov/OGWDR/dwh/t-soc/glyphosa.html.

U.S. EPA/OPP (U.S. Environmental Protection Agency/Office of Pesticide Programs). 1992. Pesticides in Ground Water Database - A compilation of Monitoring Studies: 1971-1981 National Summary.

U.S. EPA/OPP (U.S. Environmental Protection Agency, Office of Pesticide Programs). 1993a. Health Effects Division's Chapter of the Reregistration Eligibility Document (RED) for Glyphosate, Case #0178. Document dated Jan 15, 1993..

U.S. EPA/OPP (U.S. Environmental Protection Agency, Office of Pesticide Programs). 1993b. Environmental Fate and Effects Division's Chapter for the Reregistration Eligibility Document (RED) for Glyphosate, Case #0178. Document dated May 27, 1993..

U.S. EPA/OPP (U.S. Environmental Protection Agency, Office of Pesticide Programs). 1993c. Reregistration Eligibility Decision (RED) for Glyphosate. EPA 738-R-93-014, dated September 1993.

U.S. EPA/OPP (U.S. Environmental Protection Agency/Office of Pesticide Programs). 2001. Lists of Inert Pesticide Ingredients. Http://www.epa.gov/opprd001/inerts/. Updated June 12, 2001. .

U.S. EPA/OPP (U.S. Environmental Protection Agency, Office of Pesticide Programs). 2002. Glyphosate: Pesticide Tolerances, 40 CFR Part 180. Federal Register. 67(188): 60934-60950.

U.S. EPA/ORD (U.S. Environmental Protection Agency/Office of Research and Development). 1989. Recommendations for and Documentation of Biological Values for use in Risk Assessment. U.S. EPA, Environmental Criteria and Assessment Office, Office of Health and Environmental Assessment, Cincinnati, OH. ECAO-CIN-554. [pagination not continuous].

U.S. EPA/ORD (U.S. Environmental Protection Agency/Office of Research and Development). 1992. Dermal Exposure Assessment: Principles and Applications. EPA/600/8-91/011B. Interim Report. Exposure Assessment Group, Office of Health and Environmental Assessment, Washington, DC.

U.S. EPA/ORD (U.S. Environmental Protection Agency/Office of Research and Development). 1993. Wildlife Exposure Factors Handbook. Volumes 1 and 2. EPA/600/R-93/187a,b. Pagination not continuous. Available NTIS: PB94-174778 and PB94-174779.

U.S. EPA/ORD (U.S. Environmental Protection Agency/Office of Research and Development). 1996. Exposure Factors Handbook. National Center for Environmental Assessment, U.S. EPA, Washington, DC. EPA/600/P-95/002Ba-c. Avail. NTIS: PB97-117683, 97-117691, PB97-117709.

U.S. EPA/ORD (U.S. Environmental Protection Agency/Office of Research and Development. 2000. Supplementary Guidance for Conducting Health Risk Assessment of Chemical Mixtures. Office of Research and Development, U.S. EPA, Washington, DC. EPA/630/R-00/002. Report dated August2000. Available at www.epa.gov/ncea.

U.S. EPA/PPTS (U.S. Environmental Protection Agency/Prevention, Pesticides, and Toxic Substances). 1998. Health Effects Test Guidelines. Available at www.epa.gov/OPPTS\_Harmonized/.

U.S. Weather Service. 1998. NWS Climate Tables, 1961-1990. http://www.nws.noaa.gov/climatex.shtml. Files downloaded on January 14, 1999.

USDA (U.S. Department of Agriculture). 1998. Cropland acreage, soil erosion, and installation of conservation buffer strips: preliminary estimates of the 1997 National Resource Inventory. http://www.nhq.nrcs.usda.gov/land /pubs/buffer1.html.

USDA/APHIS (U.S. Department of Agriculture Animal and Plant Health Inspection Service). 1993. Nontarget Risk Assessment for the MEDFLY Cooperative Eradication Program. USDA Animal and Plant Health Inspection Service. February 1993.

USDA/FS (U.S. Department of Agriculture/Forest Service). 1989a. Final Environmental Impact Statement: Vegetation Management in the Coastal Plain/Piedmont. Management Bulletin R8-MB-23, dated January, 1989. 1213 pp.

USDA/FS (U.S. Department of Agriculture/Forest Service). 1989b. Draft Environmental Impact Statement: Vegetation Management in the Ozark/Ouachita Mountains. Management Bulletin R8-MB-23, dated June, 1989. 499 pp.

USDA/FS (U.S. Department of Agriculture/Forest Service). 1989c. Final Environmental Impact Statement: Vegetation Management in the Appalachian Mountains. Management Bulletin R8-MB-38, dated July, 1989. 1104 pp.

USDA/FS (United States Department of Agriculture/Forest Service). 2002. Regional Reports of Pesticide Use on National Forest System Lands: Fiscal Year 2001.. Available at: www.fs.fed.us/foresthealth/pesticide/pur/reports.html.

USDA/FS (U.S. Department of Agriculture/Forest Service). 2002. Regional Reports of Pesticide Use on National Forest System Lands: Fiscal Year 2001. Available at: www.fs.fed.us/foresthealth/pesticide/pur/reports.html.

USGS (U.S. Geological Survey). 1998. USGA Annual Use Maps for Pesticides for 1992. Revised Oct. 23, 1998.. http://wwwdwatcm.wr.usgs.gov/pnsp/use92.html.

USGA Annual Use Maps for Pesticides for 1992. 1998a. Revised Oct. 23, 1998. http://wwwdwatcm.wr.usgs.gov/pnsp/use92.html. .

USGS (U.S. Geological Survey). 1998b. Data on Pesticides in Surface and Ground Water of the United States. Results of the National Water Quality Assessment Program (NAWQA). Revised Oct. 23, 1998. http://wwwdwatcm.wr.usgs.gov/cppt/pns\_data/data.html.

Undabeytia T; Cheshire MV; Mcphail D. 1996. Interaction of the herbicide glyphosate with copper in humic complexes. Chemosphere; 32 (7) 1996 1245-1250.

Uotila M; Gullner G; Komives T. 1995. Induction of glutathione s-transferase activity and glutathione level in plants exposed to glyphosate. Physiologia Plantarum; 93 (4) 1995 689-694.

van Hemmen JJ. 1992. Agricultural pesticide exposure data bases for risk assessment. Rev Environ Contam Toxicol. 126:1-85.

Veiga F; Zapata JM; Fernandez Marcos ML; Alvarez E. 2001. Dynamics of glyphosate and aminomethylphosphonic acid in a forest soil in galicia, north-west spain. Sci Total Environ 2001 Apr 23;271(1-3):135-44.

Vigfusson NV; Vyse ER. 1980. The effect of the pesticides, dexon, captan and roundup, on sister-chromatid exchanges in human lymphocytes in vitro. Mutat Res. 79(1): 53-7.

Vigfusson NV; Vyse ER. 1980. The effect of the pesticides, dexon, captan, and roundup, on sister-chromatid exchanges in human lymphocytes in vitro. Mutat Res 79(1): 53-57 1980 (24 References).

Von Wiren-Lehr S; Komossa D; Glaessgen WE; Sandermann H JR; Scheunert I. 1997. Mineralization of (14c)glyphosate and its plant-associated residues in arable soils originating from different farming systems. Pesticide Science; 51 (4) 1997 436-442.

Vreeland JK; Servello FA; Griffith B. 1998. Effects of conifer release with glyphosate on summer forage abundance for deer in maine. Canadian Journal of Forest Research; 28 (10) 1998 1574-1578.

Vyse ER; Vigfusson NV. 1979. Pesticide induction of sister chromatid exchange in human lymphocytes in-vitro. Genetics; 92 (1 Suppl 1) 1979 S133-s134 (Cited in U.S. EPA 1992a).

WHO (World Health Organization). 1994. Environmental Health Criteria Environmental Health Criteria 159: Glyphosate, International Programme on Chemical Safety, Genevia, Switzerland. 177 pp. Available at http://www.inchem.org/pages/ehc.html.

WHO (World Health Organization). 1998. Aminomethylphosphonic acid (AMPA) Pesticide Residues in Food - 1997 Toxicological Evaluations. Available at http://www.inchem.org/documents/jmmpr/jmpmono/v097pr04.htm.

Wakasugi N; Kano S; Ikeda J. 1987. he toxicity of the herbicide Roundup [abstract]. Gekkan Yakuji. 29:13. (Cited in Talbot et al. 1991).

Walsh LP; Mccormick C; Martin C; Stocco DM. 2000. Roundup inhibits steroidogenesis by disrupting steroidogenic acute regulatory (Star) protein expression. Environ Health Perspect 2000 Aug;108(8):769-76.

Wan MT; Moul DJ; Watts RG. 1987. Acute Toxicity to juvenil pacific salomids of Garlon 3A, Garlon 4, triclopyr, triclopyr ester, and their transformation products: 3,5,6-tirchloro-2-pyridinol and 2-methoxy-2,5,6-trichloropyridine. Bull Environ Contam Toxicol. 39: 721-728.

Wan MT; Watts RG; Moul DJ. 1989. Effects of different dilution water types on the acute toxicity to juvenile pacific salmonids and rainbow trout of glyphosate and its formulated products. Bull Environ Contam Toxicol. 43(3):378-85.

Wan MT; Watts RG; Moul DJ. 1991. Acute toxicity to juvenile pacific northwest salmonids of basacid blue nb755 and its mixture with formulated products of 2,4-D, glyphosate, and triclopyr. Bull Environ Contam Toxicol. 47(3): 471-8.

Wan MT; Watts RG; Moul DJ. 1992. Effects of different dilution water types on the acute toxicity to juvenile pacific salmon and rainbow trout of fluroxypyr formulated product xrm-5084. Bull Environ Contam Toxicol. 49(6): 914-21.

Wan MT; Rahe JE; Watts RG. 1998. A New technique for determining the sublethal toxicity of pesticides to the vesicular-arbuscular mycorrhizal fungus glomus intraradices. Environmental Toxicology and Chemistry; 17 (7) 1998 1421-1428.

Wang Y-S; Jaw C-G; Chen Y-L. 1994a. Accumulation of 2,4-D and glyphosate in fish and water hyacinth. Water Air and Soil Pollution; 74 (3-4) 1994 397-403.

Wang Y-S; Yen J-H; Hsieh Y-N; Chen Y-L. 1994b. Dissipation of 2,4-D glyphosate and paraquat in river water. Water Air and Soil Pollution; 72 (1-4) 1994 1-7.

Wardle DA; Parkinson D. 1990a. Influence of the herbicide glyphosate on soil microbial community structure. Plant Soil. 122(1): 29-38.

Wardle DA; Parkinson D. 1990b. Effects of three herbicides on soil microbial biomass and activity. Plant Soil. 122(1): 21-28.

Wardle DA; Parkinson D. 1991. Relative importance of the effect of 2,4-D, glyphosate, and environmental variables on the soil microbial biomass. Plant Soil. 134(2): 209-220.

Wardle DA; Parkinson D. 1992a. The influence of the herbicide glyphosate on interspecific interactions between four soil fungal species. Mycol Res. 96(3): 180-186.

Wardle DA; Parkinson D. 1992b. Influence of the herbicides 2 4-D and glyphosate on soil microbial biomass and activity, a field experiment. Soil Biol Biochem. 24 (2): 185-186.

Wardle DA; Nicholson KS; Rahman A. 1994. Influence of herbicide applications on the decomposition, microbial biomass, and microbial activity of pasture shoot and root litter. New Zealand Journal of Agricultural Research. 37 (1): 29-39.

Weber JB. 1991. Fate and behavior of herbicides in soils. Appl Plant Sci. 5(1): 27-41.

Weppelman R. 1991. Letter Sent to Office of Pesticide Programs, DP Desk dated December 9, 1991: Human exposure report for Accord Herbicide . Prepared by Monsanto Agricultural Co. 1 p.. MRID 42126701.

Weppelman R. 1992. Letter Sent to OPP dated May 5, 1992 from Roger Weppelman providing summaries of twenty aerial drift incidents involving Roundup and Accord herbicides. 5 p. MRID 42317201.

Weppelman R. 1994. Letter Sent to Office of Pesticide Programs dated 5/9/94:. (English translation of doctoral thesis summarizing various human reactions to glyphosate). Prepared by Monsanto Agricultural Group. 14 p. MRID 43226801.

Weppelman R. 1998a. Testing Toxicity to Beneficial Arthoropods (sic) Predacious Mite-Typhlodromus pyri: (Roundup Ultra Herbicide):. Lab Project Number: 95 10 48 056. Unpublished study Prepared by Monsanto Company. 16p. MRID 44687501.

Weppleman R. 1998b. Testing Toxicity to Beneficial Arthoropods (sic) Cereal Aphid Parasitoid-Aphidius rhopalosiphi: (Roundup Ultra Herbicide):. Lab Project Number: 95 10 48 054: 080694. Unpublished study prepared by Monsanto Company. 17 p. MRID 44687401.

Wester RC; Melendres J; Sarason R; Mcmaster J; Maibach HI. 1991. Glyphosate skin binding, absorption, residual tissue distribution, and skin decontamination. Fundam Appl Toxicol 1991 May;16(4):725-32.

Wester RC; Melendres J; Serranzana S; Maibach HI. 1994. Time-response necessary in validation for extraction of pesticides from cloth patches used in field exposure studies. Arch Environ Contam Toxicol 1994 Aug;27(2):276-80.

Wester RC; Quan D; Maibach HI. 1996. In vitro percutaneous absorption of model compounds glyphosate and malathion from cotton fabric into and through human skin. Food Chem Toxicol 1996 Aug;34(8):731-5.

Who Working Group. 1994. Glyphosate. Ta:environmental Health Criteria Pg:177 P Yr:1994 Ip: Vi:159.

Wigfield YY; Deneault F; Fillion J. 1994. Residues of glyphosate and its principle metabolite in certain cereals, oilseeds, and pulses grown in canada, 1990-1992. Bull Environ Contam Toxicol 1994 Oct;53(4):543-7.

Willard T. 1996. Study of the Effects of Glyphosate Acid on Seedling Emergence: A Tier 1 Terrestrial Nontarget Plant Hazard Evaluation: Lab Project Number: AA950702: BA/950015. Unpublished study prepared by American Agricultural Services, Inc. 120 p. MRID 44125712.

Williams GM; Kroes R; Munro IC. 2000. Safety evaluation and risk assessment of the herbicide roundup and its active ingredient, glyphosate, for humans. Regul Toxicol Pharmacol 2000 Apr;31(2 Pt 1):117-65.

Winegardner DL. 1996. An Introduction to Soils for Environmental Professionals. CRC Press, Boca Raton, Florida. 270 pp.

Wong PK. 2000. Effects of 2,4-d, glyphosate and paraquat on growth, photosynthesis and chlorophyll-a synthesis of Scenedesmus quadricauda Berb 614. Chemosphere 2000 Jul;41(1-2):177-82.

Woo Y-T; Lai DY; Argus MF; Arcos JC. 1996. Carcinogenicity of organophosphorus pesticides-compounds an analysis of their structure-activity relationships. Journal of Environmental Science and Health Part C Environmental Carcinogenesis & Ecotoxicology Reviews; 14 (1) 1996 1-42.

Wratten S. 1998. Reduced Risk Rationale Roundup for Ultra Herbicide: Lab Project Number: RD 1416. Unpublished study prepared by Monsanto Company. 182 p. MRID 44560501.

Wratten S. 1999. Identity of Inert Ingredients to Support the Registration of MON 77063: Lab Project Number: 1424A. Unpublished study prepared by Monsanto Co. 53 p.. MRID 44757801.

Yang CC; Wu J-F; Ong H-C; Kuo Y-P; Deng J-F; Ger J. 1997. Children poisoning in Taiwan. Indian Journal of Pediatrics; 64 (4) 1997 469-483.

Ye GN; Hajdukiewicz PT; Broyles D; Rodriguez D; Xu CW; Nehra N; Staub JM. 2001. Plastidexpressed 5-enolpyruvylshikimate-3-phosphate synthase genes provide high level glyphosate tolerance in tobacco. Plant J 2001 Feb;25(3):261-70. Yokoyama VY; Pritchard J. 1984. Effect of pesticides on mortality, fecundity, and egg viability of Geocoris pallens (Hemiptera: Lygaeidae)... J Econ Entomol. 77(4): 876-879..

Yokoyama V; Pritchard J; Dowell RV. 1984. Laboratory toxicity of pesticides to geocoris pallens (Hemiptera: heteroptera: lygaeidae), a predator in California (Usa) cotton. J Econ Entomol; 77 (1) 1984 10-15.

Young JC; Khan SU. 1978. Kinetics of nitrosation of the herbicide glyphosate. J Environ Sci Health B. 13(1):59-72.

Yousef MI; Figenschau Y. 1997. A new sensitive cell culture test for the assessment of pesticide toxicity. Journal of Environmental Science and Health Part B. B32(2): 195-211.

Yousef MI; Salem MH; Ibrahim HZ; Helmi S; Seehy MA; Bertheussen K. 1995. Toxic effects of carbofuran and glyphosate on semen characteristics in rabbits. J Environ Sci Health B. 30(4): 513-34.

Yousef MI; Bertheussen K; Ibrahim HZ; Helmi S; Seehy MA; Salem MH. 1996. A sensitive sperm-motility test for the assessment of cytotoxic effect of pesticides. J Environ Sci Health b 1996 Jan;31(1):99-115.

Zaranyika MF; Nyandoro MG. 1993. Degradation of glyphosate in the aquatic environment: an enzymatic kinetic model that takes into account microbial degradation of both free and colloidal (or sediment) particle adsorbed glyphosate. J Agric Food Chem. 41(5): 838-842.

Zeneca Ag Products. 1994. Submittal of Neurotoxicity Data in Support of Registration of Touchdown Herbicides. (glyphosate-trimesium) and Tolerance Petition. Transmittal of 1 study. MRID 43132300.

Zeneca Ag Products. 1994. Submittal of Neurotoxicity Data in Support of Registration of Touchdown Herbicides. (glyphosate-trimesium) and Tolerance Petition. Transmittal of 1 study. MRID 43132300.

## APPENDICES

- Appendix 1: Commercial formulations of glyphosate
- Appendix 2: Glyphosate used by the Forest Service
- Appendix 3a: Summary of standard toxicity values in mammals for various formulations
- Appendix 3b: Summary of standard toxicity values in wildlife species for various formulations
- Appendix 3c: Summary of standard toxicity values in aquatic species for various formulations
- Appendix 4: Case reports of poisoning by glyphosate formulations
- Appendix 5: Effects on mammals of long-term exposure to glyphosate
- Appendix 6: Assays for reproductive/teratogenic effects in mammals after exposure to glyphosate
- Appendix 7: Studies assessing the mutagenicity of glyphosate
- Appendix 8: Summary of field or field simulation studies on glyphosate formulations
- Appendix 9: Toxicity of glyphosate and glyphosate formulations to fish
- Appendix 10: Acute toxicity of glyphosate to aquatic invertebrates
- Appendix 11: Toxicity of glyphosate and glyphosate formulations to aquatic plants

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
Accord SP Dow AGROSCIENCES 41.0% glyphosate (isopropylamine salt) 59.0% inert ingredients	Forestry Site Preparation: Broadcast: Aerial or Ground: 2-10 qt/acre Spray-to-Wet:Handgun of Backpack: 1-2% by volume Low Volume Directed Spray:Handgun of Backpack: 1-2% by volume	not specified	
	Woody Brush and Tree Rates: 2 qt/acre (Birch) to 4 qt/acre (Kudzu)		
4 lbs a.i./gallon 3 lbs a.e./gallon liquid formulation	PERENNIAL WEEDS: 0.5-5.0 qt/acre (Bindweed) to 4-5 qt/acre (Torpedograss)		
No additional surfactant needed or recommended	INJECT AND FRILL APPLICATION: 25-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches		
recommended for aerial application by helicopter only	Annual Weeds: 12 to 48 fl oz/acre		
<i>Aqua Neat</i> Riverdale	ANNUAL WEED CONTROL: BROADCAST APPLICATON: 1 <sup>1</sup> / <sub>2</sub> pints/acre to 3 pints/acre (Italian Ryegrass) HAND-HELD HIGH-VOLUME APPLICATION: 3/4 percent solution	not specified MSDS indicates no hazardous chemicals under OSHA Hazard Communication Standard	
<ul><li>53.8% glyphosate</li><li>(isopropylamine salt)</li><li>46.2% inert ingredients</li></ul>	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)		
5.4 lbs a.i./gallon 4 lbs a.e./gallon	WOODY BRUSH AND TREE RATES: 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)		
liquid formulation	AQUATIC AND OTHER NON-CROP SITES: Maximum application rate is 7 <sup>1</sup> / <sub>2</sub> pints/acre in any single broadcast application made over water		
Recommends use of nonionic surfactant labeled for herbicide use <i>labeled for aerial</i> <i>application (NOS)</i>	Weeds Controlled or Suppressed: 12 to 48 fl oz/acre		
	INJECTION AND FRILL APPLICATION: 25-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches		
	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
Aquamaster Monsanto	ANNUAL WEED CONTROL: BROADCAST APPLICATION: 1 <sup>1</sup> / <sub>2</sub> pints/acre to 3 pints/acre (Italian Ryegrass) HAND-HELD HIGH-VOLUME APPLICATION: 3/4 to 1 <sup>1</sup> / <sub>2</sub> percent solution	water CAS No. 7732-18-5	
<ul><li>53.8% glyphosate</li><li>(isopropylamine salt)</li><li>46.2% inert ingredients</li></ul>	AQUATIC AND OTHER NON-CROP SITES: Maximum application rate is 7 <sup>1</sup> / <sub>2</sub> pints/acre in any single broadcast application made over water	MSDS indicates no hazardous chemicals under OSHA Hazard	
5.4 lbs a.i./gallon 4 lbs a.e./gallons	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)	Communication Standard	
liquid formulation	WOODY BRUSH AND TREE RATES : 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)		
Requires use of nonionic surfactant labeled for herbicide use	Weeds Controlled or Suppressed: 12 to 48 fl oz/acre		
	INJECTION AND FRILL APPLICATION: 25-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches		
labeled for aerial application (NOS)	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
<i>Cornerstone</i> Agrilliance	ANNUAL WEED CONTROL: Low-Volume Broadcast Application: 8 oz/acre (Foxtail) to 48 oz/acre (Fliaree) HIGH-VOLUME BROADCAST APPLICATION: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre	not specified	
<ul><li>41.0% glyphosate</li><li>(isopropylamine salt)</li><li>59.0% inert ingredients</li></ul>	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)		
4 lbs a.i./gallon 3 lbs a.e./gallons	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
liquid formulation	WEEDS CONTROLLED OR SUPPRESSED: 12 to 64 fluid ounces/acre		
Nonionic surfactants labeled for use with herbicides may be used			
	INJECTION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
labeled for aerial application by fixed wing and helicopter			
	TREE AND VINE CROPS: 8 fl oz/acre (Barley) to 16 fl oz/acre (Crabgrass)		
Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
--	--	---	--
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Credit</i> Nufarm	ANNUAL WEED CONTROL: Low-Volume Broadcast Application: 8 oz/acre (Foxtail) to 48 oz/acre (Fliaree) HIGH-VOLUME BROADCAST APPLICATION: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre	exthoxylated tallowamines CAS No. 61791-83-6	
<ul><li>41.0% glyphosate</li><li>(isopropylamine salt)</li><li>59.0% inert ingredients</li></ul>	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	(SARA Title III) MSDS indicates that this	
4 lbs a.i./gallon 3 lbs a.e./gallons	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)	there is a hazardous chemical under OSHA Hazard Communication Standard	
liquid formulation	Conifer Release: 1 to 1 $\frac{1}{2}$ qt/acre	Standard	
Nonionic surfactants labeled for use with herbicides may be used	SILVICULTURAL SITES AND RIGHTS-OF-WAY: where repeat applications are necessary do not exceed 10.6 qt/acre per year		
labeled for aerial application by fixed wing and helicopter	Weeds Controlled or Suppressed: 12 to 64 fl oz/acre		
<i>Credit Systemic</i> NUFARM	ANNUAL WEED CONTROL: Low-Volume Broadcast Application: 8 oz/acre (Foxtail) to 48 oz/acre (Fliaree) HIGH-VOLUME BROADCAST APPLICATION: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre	exthoxylated tallowamines CAS No. 61791-83-6 (SARA Title III) MSDS indicates that this inert is a hazardous chemical under OSHA Hazard Communication Standard	
<ul><li>41.0% glyphosate</li><li>(isopropylamine salt)</li><li>59.0% inert ingredients</li></ul>	PERENNIAL WEEDS: 16 fl oz/acre (Leafy spurge) to 5 qt/acre (Bermudagrass)		
4 lbs a.i./gallon 3 lbs a.e./gallons	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
liquid formulation			
Nonionic surfactants labeled for use with herbicides may be used	Weeds Controlled or Suppressed: 12 to 64 fl oz/acre		
labeled for aerial application by fixed wing and helicopter	MIDDLES MANAGEMENT (ANNUAL WEEDS): 8 (Barley) to 32 fl oz/acre (Filaree)		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
Debit TMF NUFARM	ANNUAL WEED CONTROL: BROADCAST APPLICATION: 1 <sup>1</sup> / <sub>2</sub> pints/acre to 3 pints/acre (Italian Ryegrass) HAND-HELD HIGH-VOLUME APPLICATION: 3/4 percent solution	not specified MSDS indicates no hazardous chemicals under OSHA Hazard Communication Standard	
(isopropylamine salt) 46.2% inert ingredients	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)		
4 lbs a.e./gallons	WOODY BRUSH AND TREE RATES : 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)		
liquid formulation	roison Oak)		
Requires use of nonionic surfactant labeled for herbicide use	FARM DITCHES : 4 <sup>1</sup> / <sub>2</sub> to 6 fl oz/acre Dormant Rangeland : 6 to 12 fl oz/acre		
labeled for aerial application by fixed wing and helicopter	Weeds Controlled or Suppressed: 6 to 48 fl oz/acre		
<b>Eagre</b> Aquatic Herbicide GRIFFIN	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)	not specified	
53.8% glyphosate ( <b>isopropylamine salt</b> ) 46.2% inert ingredients	WOODY BRUSH AND TREE RATES: 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)		
5.4 lbs a.i./gallon 4 lbs a.e./gallons	AQUATIC AND OTHER NON-CROP SITES: Maximum application rate is 7 <sup>1</sup> / <sub>2</sub> pints/acre in any single broadcast application made over water		
liquid formulation	Weeds Controlled or Suppressed: 6 to 48 fl oz/acre		
Requires use of nonionic surfactant labeled for herbicide use	INJECTION AND FRILL APPLICATION: 25-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
labeled for aerial application (NOS)	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<b>Foresters'</b> Non-Selective Herbicide	ANNUAL WEED CONTROL: BROADCAST APPLICATION: 1 <sup>1</sup> / <sub>2</sub> pints/acre to 3 pints/acre (Italian Ryegrass) HAND-HELD HIGH-VOLUME APPLICATION: 3/4 percent solution	not specified MSDS indicates no	
RIVERDALE 53.8% glyphosate (isopropylamine salt)	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)	under OSHA Hazard Communication Standard	
46.2% inert ingredients 5.4 lbs a.i./gallon	WOODY BRUSH AND TREE RATES: 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)		
4 los a.e./gallons	INJECT AND FRILL APPLICATION: 25-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
Recommends use of nonionic surfactant labeled for herbicide use	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
labeled for aerial application by fixed wing and helicopter	WETLAND SITES: not to exceed maximum application rate of 3 3/4 qt/acre		
<i>Glyfos</i> Cheminova	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	exthoxylated tallowamines CAS No.	
41.0% glyphosate ( <b>isopropylamine salt</b> )	HIGH-VOLUME BROADCAST APPLICATION: 1 to 1 $\frac{1}{2}$ qt/acre Low-Volume Broadcast Application: 1 to 1 $\frac{1}{2}$ qt/acre	61791-83-6 (SARA Title III)	
59.0% inert ingredients	Perennial Weeds: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	MSDS indicates that this inert is a hazardous chemical under OSHA	
3 lbs a.e./gallon	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)	Hazard Communication Standard	
liquid formulation	MIDDLES MANAGEMENT (ANNUAL WEEDS): 8 (Barley) to 32 fl oz/acre (Filaree)		
Requires use of nonionic surfactant labeled for herbicide	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre		
labeled for aerial application by fixed	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
wing and helicopter	Conifer Release: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Glyfos Aquatic</i> CHEMINOVA	ANNUAL WEED CONTROL: BROADCAST APPLICATION: 1 <sup>1</sup> / <sub>2</sub> pints/acre to 3 pints/acre (Italian Ryegrass) HAND-HELD HIGH-VOLUME APPLICATION: 3/4 percent solution	water included in otherwise unidentified inert ingredients	
<ul><li>53.8% glyphosate</li><li>(isopropylamine salt)</li><li>46.2% inert ingredients</li></ul>	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)		
5.4 lbs a.i./gallon 4 lbs a.e./gallons	WOODY BRUSH AND TREE RATES: 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)		
liquid formulation	Weeds Controlled or Suppressed: 6 to 48 fl oz/acre		
Recommends use of			
nonionic surfactant labeled for herbicide use	INJECT AND FRILL APPLICATION: 25-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches		
labeled for aerial application by fixed wing and helicopter	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
<i>Glyfos Pro</i> (No Surfactant Needed)	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface	MSDS specifies surfactants and water @	
CHEMINOVA	CHEMICAL MOWING-PERENNIALS: 6 fl oz/acre	59% by weight and tallow alkylamine ethoxylate @	
41.0% glyphosate ( <b>isopropylamine salt</b> ) 59.0% inert ingredients	Chemical Mowing-Annuals: 4 to 5 fl oz/acre	3-7% by wt	
4 lbs a.i./gallon	Dormant Turfgrass: 8 to 64 fl oz/acre		
3 lbs a.e./gallon	INJECTION AND FRILL APPLICATION: 50-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches		
No additional surfactant needed or recommended	ANNUAL WEED CONTROL: 1 to 4 qt/acre BROADCAST APPLICATION: 1 pint/acre SPRAY-TO-WET APPLICATION: 0.5 to 2 percent solution		
labeled for aerial	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)		
application (NOS)	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
Glyfos X-TRA Cheminova	Conifer Release: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre	MSDS indicates that product contains surfactant @ 10-30% by	
<ul><li>41.0% glyphosate</li><li>(isopropylamine salt)</li><li>59.0% inert ingredients</li></ul>	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface	wt and that surfactant has EU classification: harmful is swallowed; irritating to eyes; toxic to	
4 lbs a.i./gallon 3 lbs a.e./gallon liquid formulation	INJECTION AND FRILL APPLICATION: 50-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches	aquatic organisms, may cause long-term adverse effects in aquatic environment.	
No additional surfactant needed or recommended	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre	Product contains tallow alkylamine exthoxylate.	
labeled for aerial application by fixed wing and helicopter	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
<i>Glyphomax</i> Dow AgroSciences	ANNUAL WEED CONTROL: LOW-VOLUME BROADCAST APPLICATON: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree) HIGH-VOLUME BROADCAST APPLICATON: 1 to 1.5 qt/acre	not specified	
(isopropylamine salt)			
58.7% inert ingredients	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)		
4 lbs a.i./gallon 3 lbs a.e./gallon			
liquid formulation	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
Recommends use of nonionic surfactant labeled for herbicide			
use	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
labeled for aerial application by fixed wing and helicopter			

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Glyphomax Plus</i> Dow AgroSciences	ANNUAL WEEDS: 12 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	not specified	
41.0% glyphosate ( <b>isopropylamine salt</b> ) 59.0% inert ingredients	PERENNIAL WEEDS: 0.5 at/acre (Johnsongrass) to 5 at/acre (Bermudagrass)		
4 lbs a.i./gallon 3 lbs a.e./gallon			
liquid formulation	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
No additional surfactant needed or			
recommended labeled for aerial application by fixed wing and helicopter	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
<i>Glyphosate</i> DuPont	ANNUAL WEEDS: 12 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	exthoxylated tallowamines (MSDS)	
41.0% glyphosate ( <b>isopropylamine salt</b> ) 59.0% inert ingredients	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)		
4 lbs a.i./gallon 3 lbs a.e./gallon	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre		
liquid formulation	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
Recommends use of nonionic surfactant			
labeled for herbicide use	Conifer Release: 1.5 to 2 qt/acre		
labeled for aerial application by fixed wing and helicopter (see additional	TREE AND VINE CROPS: 8 fl oz/acre (Barley) to 32 fl oz/acre (Filaree)		
instructions at end of label for California and Arkansas)	INJECTION AND FRILL APPLICATION: 50-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Glyphosate Original</i> Griffin	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	not specified	
41.0% glyphosate ( <b>isopropylamine salt</b> ) 59.0% inert ingredients	HIGH-VOLUME AND LOW-VOLUME BROADCAST APPLICATION: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre		
4 lbs a.i./gallon 3 lbs a.e./gallon	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)		
liquid formulation	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre		
Recommends use of nonionic surfactant labeled for herbicide	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
use	TREE AND VINE CROPS: 8 fl oz/acre (Barley) to 32 fl oz/acre (Filaree)		
labeled for aerial application by fixed wing and helicopter (see additional	INJECTION AND FRILL APPLICATION: 50-100% concentration to continuous frill around tree or as cuts evenly spaced around tree below all branches		
instructions at end of lable for California and Arkansas)	Conifer Release: $1\frac{1}{2}$ to 2 qt/acre		
Glyphosate VMF DuPont 53.8% glyphosate	GROUND BROADCAST EQUIPMENT: ANNUAL WEEDS: 3/4 to 1 ½ qt/acre Perennial Weeds: 1 ½ to 3 3/4 qt/acre Woody Brush and Trees: 3 3/4 to 7 ½ qt/acre	not specified	
( <b>isopropylamine salt</b> ) 46.2% inert ingredients	HANDHELD DIRECTED SPRAY EQUIPMENT: SPRAY-TO-WET (HANDGUN OR BACKPACK): 3/4% to 1 ½% by volume DIRECTED SPRAY (BACKPACK): 3 3/4% to 7 ½ % by volume		
4 lbs a.e./gallons	(Modified High Volume): 1 1/2 % to 3% by volume Weeds Controlled or Suppressed: 6 to 48 fl oz/acre		
Requires use of			
nonionic surfactant labeled for herbicide use	CONIFER RELEASE: $3/4$ to $1 \frac{1}{2}$ qt/acre as broadcast spray (outside Southeastern US) CONIFER RELEASE: $1 \frac{1}{8}$ to $1 \frac{7}{8}$ qt/acre as broadcast spray (in Southeastern US)		
recommended for aerial			
by helicopter. Fixed wing may be used for applications to	WETLAND SITES: not to exceed maximum application rate of 7 <sup>1</sup> / <sub>2</sub> pints/acre in a single over-water broadcast application (see exceptions on pg 5 of label)		
and pastures in CO, ID, IA, KS, MT, NE, ND, OR, SD, UT, WA, and WY	AQUATIC AND OTHER NON-CROP SITES: apply as directed under conditions described in "Weeds Controlled" section of label		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
Glypro Dow AgroSciences	ANNUAL WEED CONTROL: BROADCAST APPLICATION: 1 <sup>1</sup> / <sub>2</sub> pints/acre to 3 pints/acre (Italian Ryegrass) HAND-HELD HIGH-VOLUME APPLICATION: 3/4 percent solution	not specified	
( <b>isopropylamine salt</b> ) 46.2% inert ingredients	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)		
5.44 lbs a.i./gallon 4 lbs a.e./gallons			
liquid formulation	BRUSH OR TREE CONTROL: 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)		
<b>Recommends use of</b>			
nonionic surfactant labeled for herbicide use	AQUATIC AND OTHER NON-CROP SITES: Maximum application rate is 7 <sup>1</sup> / <sub>2</sub> pints/acre in any single broadcast application made over water		
recommended that Glypro be applied by helicopter only in forestry site and utility rights-of-way	<b>FORESTRY SITE PREPARATION:</b> BROADCAST: <b>AERIAL OR GROUND:</b> 1 <sup>1</sup> / <sub>2</sub> to 7 <sup>1</sup> / <sub>2</sub> qt/acre Spray-to-Wet: <b>Handgun, Backpack, or Mistblower:</b> 0.75-2% by volume Low Volume Directed Spray: <b>Handgun, Backpack, or Mistblower:</b> 5-10% by volume		
<i>Glypro Plus</i> Dow AgroSciences	ANNUAL WEEDS: 12 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	not specified	
41% glyphosate ( <b>isopropylamine salt</b> ) 59% inert ingredients	Perennial Weeds: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)		
4 lbs a.i./gallon 3 lbs a.e./gallons	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
liquid formulation			
No additional surfactant needed or recommended	<b>INJECT ION AND FRILL APPLICATION:</b> 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
labeled for aerial application by fixed wing and helicopter	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface immediately after cutting		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Honcho</i> Monsanto	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	Composition on MSDS: 41 % glyphosate	
41% glyphosate ( <b>isopropylamine salt</b> ) 59% inert ingredients 4 lbs a.i./gallon	HIGH-VOLUME BROADCAST APPLICATION: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre Low-Volume Broadcast Application: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre Perennial Weeds: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	tallow amine exthoxylate (CAS No. 61791-26-2) 51% water	
3 lbs a.e./gallons	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre	MSDS also states that this product is hazardous according to the OSHA	
Recommends use of nonionic surfactant	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)	Standard	
labeled for herbicide use labeled for aerial application by fixed wing and helicopter	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
<i>Mirage</i> UAP	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	exthoxylated tallowamines CAS No.	
41% glyphosate ( <b>isopropylamine salt</b> )	High-Volume and Low-VoluemBroadcast Application: 1 to 1 $\frac{1}{2}$ qt/acre	(SARA Title III)	
59% inert ingredients	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	MSDS indicates that this inert is a hazardous	
4 lbs a.i./gallon 3 lbs a.e./gallons	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre	chemical under OSHA Hazard Communication Standard	
liquid formulation	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
Recommends use of nonionic surfactant labeled for herbicide	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
use labeled for aerial application by fixed wing and helicopter	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
	Conifer Release: 1.5 to 2 qt/acre		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Prosecutor</i> Lesco	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	59% inerts including exthoxylated	
41% glyphosate (isopropylamine salt)	High-Volume and Low-Volume Broadcast Application: 1 to 1 $\frac{1}{2}$ qt/acre	<b>tallowamines</b> CAS No. 61791-83-6 (SARA Title III)	
59% inert ingredients	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	MSDS indicates that this	
4 lbs a.i./gallon 3 lbs a.e./gallons	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre	inert is a hazardous chemical under OSHA Hazard Communication	
liquid formulation	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)	Standard	
Recommends use of nonionic surfactant labeled for herbicide	Conifer Release: 1.5 to 2 qt/acre		
use labeled for aerial application by fixed wing and helicopter	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
Prosecutor Plus Tracker	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	59% inerts including exthoxylated	
Lesco 41% glyphosate	HIGH-VOLUMEAND LOW-VOLUME BROADCAST APPLICATION: 1 to 1 <sup>1</sup> / <sub>2</sub> qt/acre	tallowamines CAS No. 61791-83-6 (SARA Title III)	
( <b>isopropylamine salt</b> ) 59% inert ingredients	Perennial Weeds: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	MSDS indicates that this inert is a hazardous	
4 lbs a.i./gallon 3 lbs a.e./gallons	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre	chemical under OSHA Hazard Communication Standard	
liquid formulation	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
Recommends use of nonionic surfactant labeled for herbicide	Conifer Release: 1.5 to 2 qt/acre		
use labeled for aerial	<b>INJECT ION AND FRILL APPLICATION:</b> 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
application by fixed wing and helicopter	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Rattler</i> Helena Chemical Co.	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	not specified	
41% glyphosate	HIGH-VOLUME AND LOW-VOLUME BROADCAST APPLICATION: 1 to 1 1/2 qt/acre		
59% inert ingredients	Perennial Weeds: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)		
4 lbs a.i./gallon 3 lbs a.e./gallons	Weeds Controlled or Suppressed: 8 to 64 fl oz/acte		
liquid formulation	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
Recommends use of nonionic surfactant labeled for herbicide	Conifer Release: 1.5 to 2 qt/acre		
use labeled for aerial	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
wing and helicopter	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
Razor Riverdal F	Annual Weeds: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	59% inerts including exthoxylated tallowamines CAS No. 61791-83-6	
41% glyphosate	High-Volume and Low-Volume Broadcast Application: 1 to 1 $\frac{1}{2}$ qt/acte		
( <b>isopropylamine salt</b> ) 59% inert ingredients	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	(SARA Title III)	
4 lbs a.i./gallon	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre	MSDS indicates that this inert is a hazardous	
3 lbs a.e./gallons	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)	chemical under OSHA Hazard Communication	
liquid formulation	CHEMICAL MOWING-PERENNIALS: 6 fl oz/acre (Kentucky bluegrass) to 8 fl oz/acre (tall fescue, fine fescue, orchardgrass, or quackgrass covers)	Standard	
nonionic surfactant	CHEMICAL MOWING-ANNUALS: 4 to 5 fl oz/acre		
use	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
labeled for aerial application by fixed wing and helicopter	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
	Conifer Release: 1.5 to 2 qt/acre		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)			
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)	
<i>Razor SPI</i> Riverdale.	Annual Weeds: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	59% inerts including exthoxylated	
41% glyphosate (isonronylamine salt)	High-Volume and Low-Volume Broadcast Application: 1 to 1 $\frac{1}{2}$ qt/acre	<b>tallowamines</b> CAS No. 61791-83-6 (SARA Title III)	
59% inert ingredients	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)	MSDS indicates that this	
4 lbs a.i./gallon 3 lbs a.e./gallons	Weeds Controlled or Suppressed: 8 to 64 fl oz/acre	inert is a hazardous chemical under OSHA Hazard Communication	
liquid formulation	Conifer Release: 1.5 to 2 qt/acre	Standard	
Contains a temporary blue colorant to assist in accurate and uniform	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface		
applications. Recommends use of nonionic surfactant	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches		
labeled for herbicide use	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)		
labeled for aerial application by fixed	CHEMICAL MOWING-PERENNIALS: 6 fl oz/acre (Kentucky bluegrass) to 8 fl oz/acre (tall fescue, fine fescue, orchardgrass, or quackgrass covers)		
wing and helicopter	Chemical Mowing-Annuals: 4 to 5 fl oz/acre		

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)							
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)					
<i>Rodeo</i> Dow AgroSciences	ANNUAL WEED CONTROL: BROADCAST APPLICATION: 1 <sup>1</sup> / <sub>2</sub> pints/acre to 3 pints/acre (Italian Ryegrass) HAND-HELD HIGH-VOLUME APPLICATION: 3/4 percent solution	not specified					
<ul><li>53.8% glyphosate</li><li>(isopropylamine salt)</li><li>46.2% inert ingredients</li></ul>	PERENNIAL WEEDS: 3 pints/acre (Quackgrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)						
5.44 lbs a.i./gallon 4 lbs a.e./gallons							
liquid formulation	BRUSH OR TREE CONTROL: 3 pints/acre (Birch) to 7 ½ pints/acre (Poison Ivy or Poison Oak)						
Requires use of							
nonionic surfactant labeled for herbicide use	AQUATIC AND OTHER NON-CROP SITES: Maximum application rate is 7 <sup>1</sup> / <sub>2</sub> pints/acre in any single broadcast application made over water						
labeled for aerial							
application (NOS). In California, aerial application may be made in aquatic sites	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface						
and noncrop areas,							
including aquatic sites present in noncrop areas that are part of the intended treatment.	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches						

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)								
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)						
<i>Roundup CUSTOM</i> Monsanto	PERENNIAL WEEDS: 1.5 pints/acre (Paragrass) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Bermudagrass)	water CAS No. 7732-18-5						
53.8% glyphosate ( <b>isopropylamine salt</b> )		MSDS indicates no hazardous chemicals						
46.2% inert ingredients	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface	under OSHA Hazard Communication						
5.4 lbs a.i./gallon 4 lbs a.e./gallons		Standard						
liquid formulation	ANNUAL WEEDS: 12 fl oz/acre (Foxtail) to 36 fl oz/acre (Filaree)							
Requires use of nonionic surfactant labeled for herbicide								
use	WOODY BRUSH AND TREE RATES: 3 pints/acre (Birch) to 7 <sup>1</sup> / <sub>2</sub> pints/acre (Poison Ivy or Poison Oak)							
labeled for aerial application by fixed wing and helicopter								
<i>Roundup ORIGINAL</i> Monsanto	ANNUAL WEEDS: 8 fl oz/acre (Foxtail) to 48 fl oz/acre (Filaree)	Composition on MSDS: 41 % glyphosate 8% surfactant						
41% glyphosate		tallow amine						
( <b>isopropylamine salt</b> ) 59% inert ingredients	HIGH-VOLUME BROADCAST APPLICATION: 1 to 1 $\frac{1}{2}$ qt/acre Low-Volume Broadcast Application: 1 to 1 $\frac{1}{2}$ qt/acre	exthoxylate (CAS No. 61791-26-2) 51% water						
4 lbs a.i./gallon 3 lbs a.e./gallons	PERENNIAL WEEDS: 1 gt/acre (Alfalfa) to 5 gt/acre (Bermudagrass)	MSDS also states that						
liquid formulation		this product is hazardous according to the OSHA Hazard Communication						
Recommends use of nonionic surfactant labeled for herbicide use	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)	Standard						
labeled for aerial application by fixed wing and helicopter	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface							

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)								
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)						
<i>Roundup PRO</i> Monsanto	ANNUAL WEEDS: 1 qt/acre (<6" tall) and 1.5 to 4 qt/acre (>6" tall)	44.5% water 14.5% surfactant (NOS) (SARA Title III)						
<ul><li>41% glyphosate</li><li>(isopropylamine salt)</li><li>59% inert ingredients</li></ul>	HIGH-VOLUME BROADCAST APPLICATION: 1 to 1 $\frac{1}{2}$ qt/acre Low-Volume Broadcast Application: 1 to 1 $\frac{1}{2}$ qt/acre							
4 lbs a.i./gallon 3 lbs a.e./gallons	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 5 qt/acre (Bermudagrass)							
liquid formulation	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface							
Label indicates that product is formulated								
as a water soluble liquid containing surfactant and no	WOODY BRUSH AND TREES: 2 qt/acre (Birch) to 5 qt/acre (Poison Ivy or Oak)							
additional surfactant is needed	CHEMICAL MOWING-PERENNIALS: 6 fl oz/acre (Kentucky bluegrass) to 8 fl oz/acre (tall fescue, fine fescue, orchardgrass, or quackgrass covers)							
labeled for aerial application (NOS)	Chemical Mowing-Annuals: 4 to 5 fl oz/acre							
<i>Roundup PRO Concentrate</i> MONSANTO	ANNUAL WEEDS: 26 fl oz/acre (<6" tall) and 1.2 to 3.2 qt/acre (>6" tall)	specific identity withheld because it is a trade secret						
50.2% glyphosate ( <b>isopropylamine salt</b> )	PERENNIAL WEEDS: 0.8 qt/acre (Alfalfa) to 4 qt/acre (Bermudagrass)	MSDS indicates that the product is hazardous according to the OSHA						
49.8% inert ingredients 5 lbs a.i./gallon	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface	Hazard Communication Standard						
liquid formulation Label indicates that	Woody Brush and Trees: BROADCAST APPLICATION: 1.6 qt/acre (Birch) to 4 qt/acre (Poison Ivy or Oak) HAND-HELD SPRAY-TO-WET: 0.8% (Birch) to 1.6% (Poison Ivy or Poison Oak)							
product is formulated as a water soluble liquid containing surfactant	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches							
labeled for aerial application (NOS)	CHEMICAL MOWING-PERENNIALS: 5 fl oz/acre (Kentucky bluegrass) to 6.4 fl oz/acre (tall fescue, fine fescue, orchardgrass, or quackgrass covers)							
	CHEMICAL MOWING-ANNUALS: 3 to 4 fl oz/acre							

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)							
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)					
<i>Roundup ProDry</i> Monsanto	ANNUAL WEEDS: 1.2 lbs/acre (<6" tall) and 1.8 to 4.8 lbs/acre (>6" tall)	<0.5% sodium sulphite 28.6% trade secret					
71.4% glyphosate ( <b>monoammoniumsalt</b> ) 28.6% inert ingredients		MSDS indicates that the product is hazardous according to the OSHA					
71.4% of glyphosate in the form of its ammonium salt is eequal to 64.9% of the acid glyphosate	PERENNIAL WEEDS: 1.25 lbs/acre (Alfalfa) to 6.25 lbs/acre (Bermudagrass)	Hazard Communication Standard					
1.56 lbs of Roundup ProDry contains 0.75 lbs of glyphosate acid and is equivalent to 1 quart of Roundup Pro Herbicide	WOODY BRUSH AND TREES: BROADCAST APPLICATION: 2.5 lbs/acre (Birch) to 6.25 lbs/acre (Poison Ivy or Oak) HAND-HELD SPRAY-TO-WET: 0.64% (Birch) to 1.3% (Poison Ivy or Poison Oak)						
granular formulation	CHEMICAL MOWING-PERENNIALS: 3.6 oz/acre (Kentucky bluegrass) to 4.8 oz/acre (tall fescue, fine fescue, orchardgrass, or quackgrass covers)						
Label indicates that product is formulated as water soluble granules containing							
surfactant and no additional surfactant is needed	Chemical Mowing-Annuals: 2.4 oz/acre						
labeled for aerial application (NOS)							

Appendix 1: Glyphosate Formulations with Forestry Applications (C&P Press Database)							
Brand Name Company/Composition	Application Rate (Specified by Label)	Inerts (Specified)					
<i>Roundup UltraDry</i> Monsanto	ANNUAL WEEDS: 1.2 lbs/acre (<6" tall) and 1.8 lbs/acre (>6" tall)	<ul><li>25% surfactant/antifoam</li><li>4% trade secret</li></ul>					
<ul><li>71.4% glyphosate</li><li>(monoammoniumsalt)</li><li>28.6% inert ingredients</li></ul>		MSDS indicates that both inert ingredients are hazardous according to the OSHA Hazard					
71.4% of glyphosate in the form of its ammonium salt is equal		Communication Standard					
to 64.9% of the acid glyphosate	PERENNIAL WEEDS: 1.2 lbs/acre (Alfalfa) to 5.9 lbs/acre (Bermudagrass)						
1.2 lbs of Roundup ProDry contains 0.75 lbs of glyphosate acid and is equivalent to 1 quart of Roundup Pro Herbicide							
granular formulation							
Label indicates that product is formulated as water soluble granules containing surfactant and no additional surfactant is needed	WOODY BRUSH AND TREES: BROADCAST APPLICATION: 2.4 lbs/acre (Birch) to 5.9 lbs/acre (Poison Ivy or Oak) HAND-HELD SPRAY-TO-WET: 0.6% (Birch) to 1.1% (Poison Ivy or Poison Oak)						
labeled for aerial application by fixed wing and helicopter							
<i>Roundup ULTRA MAX</i> MONSANTO	Annual Weeds: 26 fl oz/acre (<6" tall) and 40 fl oz/acre (>6" tall)	specific identity withheld because it is a trade secret					
50.2% glyphosate ( <b>isopropylamine salt</b> )	PERENNIAL WEEDS: 1 qt/acre (Alfalfa) to 4 qt/acre (Bermudagrass)	MSDS indicates that the product is hazardous according to the OSHA					
49.8% inert ingredients	CUT STUMP APPLICATION: 50-100% solution to freshly cut surface	Hazard Communication Standard					
5 lbs a.i./gallon							
liquid formulation	WOODY BRUSH AND TREES: BROADCAST APPLICATION: 1.75 qt/acre (Birch) to 4 qt/acre (Poison Ivy or Oak) HAND-HELD SPRAY-TO-WET: 1% (Birch) to 2% (Poison Ivy or Poison Oak)						
surfactant needed or recommended	INJECT ION AND FRILL APPLICATION: 50-100% concentration of product either to continuous frill around the tree or as cuts evenly spaced around the tree below all branches						
application by fixed wing and helicopter	CHEMICAL MOWING-PERENNIALS: 5 fl oz/acre (Kentucky bluegrass) to 6.5 fl oz/acre (tall fescue, fine fescue, orchardgrass, or quackgrass covers)						

2002).						
Region	Forest	Use Classification	Acres		Pounds	lbs/acre
	1	10 Agricultural Weed Control		0.25	0.3	1.20
	1	11 Noxious Weed Control		6.5	6.5	1.00
	1	14 Noxious Weed Control		2	5.4	2.70
	1	16 Noxious Weed Control		32.5	43.9	1.35
	1	17 Noxious Weed Control		3	1.211	0.40
	1	18 Noxious Weed Control		25	9.5	0.38
	1	2 Noxious Weed Control		10	7	0.70
	1	4 Noxious Weed Control		2.25	1.14	0.51
	1	4 Nursery Weed Control		34	183	5.38
	1	4 ROW Veg Management		1.5	4	2.67
	1	4 Site Preparation		16	2.06	0.13
1 Total	Northe	rn		133	264	1.99
	2	3 Noxious Weed Control		2	0	0.02
	2	4 Noxious Weed Control		12	9	0.75
	2	7 Noxious Weed Control		11	37	3.35
	2	10 Noxious Weed Control		7	14	2.00
	2	13 Noxious Weed Control		163	48	0.29
	2	14 Noxious Weed Control		15	30	2.00
	2	7 Nursery Weed Control		54	44	0.82
2 Total	Rocky	Mountain		264	182	0.69
	3	3 Noxious Weed Control		3	3	1.00
3 Total	Southv	vestern		3	3	1.00
	4	1 Noxious Weed Control		15	22	1.47
	4	2 Noxious Weed Control		68	16	0.23
	4	3 Noxious Weed Control		0	1	3.01
	4	7 Noxious Weed Control		30	3	0.10
	4 10A	Noxious Weed Control		123	166	1.35
	4	13 Noxious Weed Control		0	1	1.67
	4	14 Noxious Weed Control		3	6	2.07
	4 15A	Noxious Weed Control		80	16	0.20
	4	17 Noxious Weed Control		75	15	0.20
	4	18 Noxious Weed Control		9	12	1.33
	4	19 Noxious Weed Control		2	4	2.00
4 Total	Interm	ountain		405	261	0.64
	5	8 Agricultural Weed Control		172	628	3.64
	5	15 Conifer and Hardwood Release		361	1,161	3.22
	5	3 Conifer Release		803	2,918	3.63
	5	13 Conifer Release		1,400	2,695	1.93
	5	16 Conifer Release		3,990	19,999	5.01
	5	17 Conifer Release		45	83	1.85
	5	1 Facilities Maintenance		5	2	0.40
	5	3 Facilities Maintenance		6	11	1.87
	5	13 Facilities Maintenance		2	4	2.05
	5	15 Facilities Maintenance		10	4	0.44
	5	1 Noxious Weed Control		93	29	0.31
	5	7 Noxious Weed Control		9	90	9.96

Appendix 2: Forest Service Use of Glyphosate in 2001 (sorted by region and use) (USDA/FS 2002).

	5 15	Noxious Weed Control	42	34	0.82
	5 16	Noxious Weed Control	0	0	0.32
	5 3	Nursery Weed Control	9	30	3.48
	5 10	Nursery Weed Control	83	192	2.30
	5 13	Recreation Improvement	8	23	2.88
	5 1	Right-of-Way	10	20	2.00
	5 16	Site Preparation	1,346	6,815	5.06
5 Total	Pacific So	uthwest	8,395	34,740	4.14
	6 16	Conifer Release	14	11	0.80
	6 15	Facilities Maintenance	1	1	1.25
	6 1	Noxious Weed Control	6	6	1.00
	6 2	Noxious Weed Control	3	3	1.00
	6 3	Noxious Weed Control	280	93	0.33
	6 4	Noxious Weed Control	0	0	2.13
	6 5	Noxious Weed Control	18	13	0.69
	6 7	Noxious Weed Control	11	25	2.33
	6 8	Noxious Weed Control	3	3	1.10
	6 10	Noxious Weed Control	1	1	0.83
	6 12	Noxious Weed Control	113	62	0.55
	6 14	Noxious Weed Control	0	2	6.67
	6 16	Noxious Weed Control	241	1,033	4.29
	6 17	Noxious Weed Control	62	121	1.96
	6 18	Noxious Weed Control	20	30	1.49
	6 20	Noxious Weed Control	129	21	0.16
	6 21	Noxious Weed Control	16	1	0.03
	6 10	Nursery Weed Control	67	268	4.00
	6 15	Nursery Weed Control	0	1	3.75
	6 1	Right-of-Way	1	1	1.00
	6 6	Seed Orchard Protection	16	10	0.63
6 Total	Pacific No	orthwest	1,003	1,706	1.70
	8 10	Aquatic Weed Control	3	2	0.67
	8 13	Aquatic Weed Control	2	3	1.80
	8 9	Conifer and Hardwood Release	288	284	0.99
	8 10	Conifer and Hardwood Release	305	56	0.18
	8 7	Conifer Release	1,089	324	0.30
	8 10	Conifer Release	514	2	0.00
	8 8	Facilities Maintenance	1	1	1.00
	8 4	Noxious Weed Control	93	150	1.61
	8 5	Noxious Weed Control	47	16	0.34
	8 7	Noxious Weed Control	30	292	9.75
	8 8	Noxious Weed Control	1	2	2.00
	8 9	Noxious Weed Control	2	6	2.82
	8 10	Noxious Weed Control	210	1,050	5.00
	8 12	Noxious Weed Control	9	48	5.33
	8 1	Recreation Improvement	35	1	0.03
	8 3	Recreation Improvement	2	2	1.00
	<b>8</b> 7	Recreation Improvement	30	8	0.27
	8 8	Recreation Improvement	3	5	1.80
	8 3	Right-of-Way	1	1	1.00

	8 4 Right-of-Way	9	26	2.97
	8 7 Right-of-Way	2	2	1.00
	8 7 Site preparation	161	296	1.84
	8 9 Site Preparation	40	35	0.88
	8 10 Site preparation	474	13	0.03
	8 12 Site Preparation	30	40	1.33
	8 7 Wildlife Habitat Improvement	211	298	1.41
	8 10 Wildlife Habitat Improvement	92	406	4.41
	8 12 Wildlife Habitat Improvement	205	50	0.24
8 Total	Southern	3,888	3,419	0.88
	9 5 Facilities Maintenance	25	25	1.00
	9 5 Noxious Weed Control	30	12	0.40
	9 15 Noxious Weed Control	4,199	3,149	0.75
	9 3 Recreation Improvement	1	0	0.20
	9 5 Right-of-Way	474	787	1.66
	9 19 Right-of-Way	79	54	0.69
	9 19 Site Preparation	122	118	0.97
9 Total	Eastern	4,930	4,146	0.84
Grand T	otal	19,021	44,721	2.35

Appendix 3a: Summary of standard toxicity values for glyphosate, AMPA, POEA surfactant and various formulations of glyphosate a						
Formulation	Rat Oral LD <sub>50</sub>	Rabbit Dermal LD <sub>50</sub>	4-Hour Rat Inhalation $LC_{50}$	Skin Irritation	Eye Irritation	Allergic Sensitization
	-	Gl	yphosate, AMPA, an	d POEA		-
GLYPHOSATE ACID	>5000 mg/kg	>5000 mg/kg		Mild irritation <sup>b</sup>	Severe <sup>b</sup>	Negative <sup>b</sup>
GLYPHOSATE IPA				Non-irritating <sup>b</sup>	Non-irritating <sup>b</sup>	
AMPA	8300 mg/kg					
POEA SURFACTANT	1200 mg/kg <sup>b</sup>	$>1260$ mg/kg $^{\rm b}$		Corrosive <sup>b</sup>	Severe <sup>b</sup>	
		-	Formulations	-	-	-
ACCORD SP	>5000 mg/kg	>5000 mg/kg	>5 mg/L	Non-irritating	Slight	Negative
AQUA NEAT	>5000 mg/kg	>5000 mg/kg	>1.3 mg/L	Non-irritating	Non-irritating	Negative
AQUAMASTER	>5000 mg/kg	>5000 mg/kg	>1.3 mg/L	Non-irritating	Category IV	Negative
CORNERSTONE	>5000 mg/kg	>5000 mg/kg (rat acute LD <sub>50</sub> )	>7.03 mg/L	Category IV	Irritating (Category III)	Negative
CREDIT SYSTEMIC	>5000 mg/kg	>5000 mg/kG	>2.05 mg/L	Slightly irritating (Category IV)	Moderately irritating to unrinsed eyes; mildly irritating to rinsed eyes (Category III)	Negative
Credit	>5000 mg/kg	>5000 mg/kg	2.6 mg/L (Category III)	Slight erythema (Category IV)	Slight to moderate irritation (Category II)	Negative
DEBIT TMF	>5000 mg/kg	>5000 mg/kg (rat acute LD <sub>50</sub> )	>2.07 mg/L (slightly toxic)	Non-irritating	Minimally irritating	Negative
EAGRE	>5000 mg/kg	>5000 mg/kg	>1.3 mg/L (slightly toxic)	Slightly irritating	Slightly irritating	Negative
FORESTERS' NON- Selective Herbicide	>5000 mg/kg	>5000 mg/kg	>1.3 mg/L (slightly toxic)	Practically non- irritating	Non-irritating	Negative
GLYFOS	>5000 mg/kg	>5000 mg/kg	>2.6 mg/L (slightly toxic) (Category III)	Slight erythema clearing within 24 hours (Category IV)	Slight to moderate irritation (Category II)	Negative

Appendix 3a: Summary of standard toxicity values for glyphosate, AMPA, POEA surfactant and various formulations of glyphosate a						
Formulation	Rat Oral LD <sub>50</sub>	Rabbit Dermal LD <sub>50</sub>	4-Hour Rat Inhalation LC <sub>50</sub>	Skin Irritation	Eye Irritation	Allergic Sensitization
GLYFOS X-TRA	>5000 mg/kg	>2000 mg/kg	>4.86 mg/L	Non-irritating	May cause substantial but temporary eye injury	Negative
GLYFOS PRO	>5000 mg/kg	>2000 mg/kg	4.24 mg/L	Non-irritating	Minimal	Negative
GLYFOS AQUATIC	>5000 mg/kg	>2000 mg/kg	>4.86 mg/L	Very slightly irritating	Slightly irritating	Negative
GLYPHOMAX	>5000 mg/kg	>5000 mg/kg	>5.00 mg/L	Non-irritating	Slightly irritating	Negative
GLYPHOMAX PLUS	>5000 mg/kg	>5000 mg/kg	>5.00 mg/L	Non-irritating	Slightly irritating	Negative
GLYPHOSATE	>5000 mg/kg	>5000 mg/kg	2.6 mg/L (slightly toxic)	Non-irritating	Slight to moderate	Negative
GLYPHOSATE ORIGINAL	5000 mg/kg (slightly toxic)	>5000 mg/kg	3.18 mg/L (slightly toxic)	Slightly irritating	Moderately irritating	Negative
GLYPHOSATE VMF	>5000 mg/kg	>5000 mg/kg	1.3 mg/L (Slightly toxic)	Non-irritating	Non-irritating	Negative
Glypro	>5000 mg/kg	>5000 mg/kg	>6.37	Non-irritating	Slight irritation	Negative
GLYPRO PLUS	>5000 mg/kg	>5000 mg/kg	>5.00 mg/L	Non-irritating	Slight irritation	Negative
Нолсно	>5000 mg/kg	>5000 mg/kg (rat LD <sub>50</sub> )	2.6 mg/L (Category IV)	Non-irritating	Moderate irritation (Category II)	Negative
MIRAGE	>5000 mg/kg	>5000 mg/kg	2.6 mg/L	Non-irritating	Slight to moderate irritation (Category II)	Negative
PROSECUTOR PLUS TRACKER	>5000 mg/kg	>5000 mg/kg	>2.05 mg/L	Slightly irritating	Moderately irritating	Negative
PROSECUTOR	>5000 mg/kg	>5000 mg/kg	>2.05 mg/L	Slightly irritating	Moderately irritating	Negative
RATTLER	>5000 mg/kg (mouse LD <sub>50</sub> )	>5000 mg/kg	>3.18 mg/L	Slightly irritating	Non-irritating	NS

Appendix 3a: Summary of standard toxicity values for glyphosate, AMPA, POEA surfactant and various formulations of glyphosate a							
Formulation	Rat Oral LD <sub>50</sub>	Rabbit Dermal LD <sub>50</sub>	4-Hour Rat Inhalation LC <sub>50</sub>	Skin Irritation	Eye Irritation	Allergic Sensitization	
RAZOR	>5000 mg/kg	>5000 mg/kg	>2.05 mg/L	Slightly irritating	Moderately irritating to unrinsed eyes; mildly irritating to rinsed eyes (Category III)	Negative	
RAZOR SPI	>5000 mg/kg	>5000 mg/kg	>2.05 mg/L	Slightly irritating	Moderately irritating to unrinsed eyes; mildly irritating to rinsed eyes (Category III)	Negative	
Rodeo	>5000 mg/kg	>5000 mg/kg	>6.37 mg/L	Non-irritating	Slight irritation	Negative	
ROUNDUP ULTRAMAX	>5000 mg/kg	>5000 mg/kg (rat LD <sub>50</sub> )	>2.01 mg/L (Category IV)	Slight irritation (Category IV)	Moderate irritation (Category III)	Negative	
ROUNDUP PRO	5108 mg/kg	>5000 mg/kg (rat LD <sub>50</sub> )	2.9 mg/L	Non-irritating	Slight irritation (Category III)	Negative	
ROUNDUP CUSTOM	>5000 mg/kg	>5000 mg/kg	>1.3 mg/L (Category III)	Non-irritating	Category IV	Negative	
ROUNDUP ORIGINAL	>5000 mg/kg	>5000 mg/kg (rat LD <sub>50</sub> )	2.6 mg/L	Non-irritating	Moderate irritation (Category II)	Negative	
ROUNDUP PRODRY	3794 mg/kg (female rat) (Category III)	>5000 mg/kg (rat LD <sub>50</sub> )	2.6 mg/L	Slight irritation	Slight irritation (Category III)	Negative	
ROUNDUP ULTRADRY	5827 mg/kg (slightly toxic) (Category III) ( <i>Female rat LD</i> <sub>50</sub> = 3700 mg/kg)	>5000 mg/kg (rat LD <sub>50</sub> )	granular formulation	Slightly irritating (Category IV)	Moderately irritating (Category III)	Negative	
ROUNDUP PRO CONCENTRATE	>5000 mg/kg	>5000 mg/kg (rat LD <sub>50</sub> )	>2.01 mg/L (Category IV)	Slight irritation (Category IV)	Moderate irritation (Category III)	Negative	
<sup>a</sup> Taken from Material Safet <sup>b</sup> Williams et al. 2000	<sup>4</sup> Taken from Material Safety Data Sheets in C&P Press (2002) unless otherwise specified. <sup>b</sup> Williams et al. 2000						

Appendix 3b: Nontarget toxicity values for various formulations of glyphosate <sup>a</sup>							
Formulation	Bobwhite quail (Colinus virginianus)	Mallard duck (Anas platyrhynchos)	Dog	Goat	Earthworm (Eisenia foetida)	Honeybee (Apis mellifera)	
ACCORD SP	LD <sub>50</sub> >2000 mg/kg				LC <sub>50</sub> >1000 mg/kg		
AQUA NEAT				$LD_{50} = 5700 \text{ mg/kg}$			
AQUAMASTER	$LD_{50}>3851 mg/kg$ $LC_{50}>4640 mg/kg$ $(5 days, dietary)$ $NOEC>1000 mg/kg$ $(17 weeks, diet)$	LC <sub>50</sub> >4640 mg/kg (5 days, dietary) NOEC>1000 mg/kg (16 weeks, diet)			LC <sub>50</sub> >5000 mg/kg (14 days, dry soil)	LD <sub>50</sub> >100 µg/bee (48 hours, oral/contact)	
CORNERSTONE	NS	NS	NS	NS	NS	NS	
CREDIT SYSTEMIC	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)					
CREDIT	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)					
DEBIT TMF				$LD_{50} = 5700 \text{ mg/kg}$			
EAGRE				$LD_{50} = 5700 \text{ mg/kg}$			
Foresters' Non-selective Herbicide				$LD_{50} = 5700 \text{ mg/kg}$			
GLYFOS	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)					
GLYFOS PRO	LD <sub>50</sub> >2000 mg/kg	LD <sub>50</sub> >2000 mg/kg					
GLYFOS X-TRA	LD <sub>50</sub> >2000 mg/kg <sup>b</sup>	LD <sub>50</sub> >2000 mg/kg <sup>b</sup>					
GLYFOS AQUATIC	NS	NS	NS	NS	NS	NS	
GLYPHOMAX PLUS	LD <sub>50</sub> >2000 mg/kg				LC <sub>50</sub> >1000 mg/kg	$LD_{50}>100 \ \mu g/bee$ (acute contact) $LD_{50}>100 \ \mu g/bee$ (acute oral)	

Appendix 3b: Nontarget toxicity values for various formulations of glyphosate <sup>a</sup>										
Formulation	Bobwhite quail (Colinus virginianus)	Mallard duck (Anas platyrhynchos)	Dog	Goat	Earthworm (Eisenia foetida)	Honeybee (Apis mellifera)				
GLYPHOMAX	LD <sub>50</sub> = 2000 mg/kg				LC <sub>50</sub> = 1000 mg/kg	$\begin{array}{l} LD_{50} > 100 \ \mu g/bee \\ (acute \ contact) \\ LD_{50} > 100 \ \mu g/bee \\ (acute \ oral) \end{array}$				
GLYPHOSATE	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)v								
GLYPHOSATE ORIGINAL			LD <sub>50</sub> >5.0 mg/kg	LD <sub>50</sub> = 4660 mg/kg		$\begin{array}{l} LD_{50} > 100 \ \mu g/bee \\ (48-hour \ oral) \\ LD_{50} > 100 \ \mu g/bee \\ (48-hour \ dermal) \end{array}$				
GLYPHOSATE VMF				$LD_{50} = 5700 \text{ mg/kg}$						
GLYPRO PLUS	LD <sub>50</sub> >2000 mg/kg				LC <sub>50</sub> >1000 mg/kg	$LD_{50}>100 \ \mu g/bee$ (acute contact) $LD_{50}>100 \ \mu g/bee$ (acute oral)				
Glypro	LD <sub>50</sub> >2000 mg/kg				LC <sub>50</sub> >1000 mg/kg					
Нопсно	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)			LC <sub>50</sub> >5000 mg/kg (14 days, dry soil)	LD <sub>50</sub> >100 µg/bee (48 hours, oral/contact)				
MIRAGE	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)								
PROSECUTOR PLUS TRACKER	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)								
PROSECUTOR	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)								
RATTLER	NS	NS	NS	NS	NS	NS				
RAZOR	LC <sub>50</sub> >6300 mg/kg (8 days)	LC <sub>50</sub> >6300 mg/kg (8 days)								

Appendix 3b: Nontarget toxicity values for various formulations of glyphosate <sup>a</sup>										
Formulation	Bobwhite quail (Colinus virginianus)	Mallard duck (Anas platyrhynchos)	Dog	Goat	Earthworm (Eisenia foetida)	Honeybee (Apis mellifera)				
RAZOR SPI										
RODEO	LD <sub>50</sub> >2000 mg/kg				LC <sub>50</sub> >1000 mg/kg					
ROUNDUP PRO CONCENTRATE	$LC_{50}$ >5620 mg/kg (5 days, dietary)	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)			LC <sub>50</sub> >1250 mg/kg (14 days, soil)	LD <sub>50</sub> >100 µg/bee (48 hours, oral/contact)				
ROUNDUP PRODRY	LD <sub>50</sub> = 1651 mg/kg body weight				LC <sub>50</sub> >1250 mg/kg (14 days, dry soil)	LD <sub>50</sub> >100 µg/bee (48 hours, oral/contact)				
ROUNDUP ULTRADRY	Not tested		Not tested	Not tested	Not tested	Not tested				
ROUNDUP CUSTOM	$LD_{50}>3851 \text{ mg/kg}$ $LC_{50}>4640 \text{ mg/kg}$ (5 days, dietary) NOEC>1000 mg/kg (17 weeks, diet)	LC <sub>50</sub> >4640 mg/kg (5 days, dietary) NOEC>1000 mg/kg (16 weeks, diet)			LC <sub>50</sub> >5000 mg/kg (14 days, dry soil)	LD <sub>50</sub> >100 μg/bee (48 hours, oral/contact)				
ROUNDUP PRO BIACTIVE					LC <sub>50</sub> >1250 mg/kg (14 days, soil)	$\begin{array}{c} LD_{50} > 1000 \ \mu\text{g/bee} \\ (48-\text{hour oral}) \end{array}$				
ROUNDUP PRO	$LC_{50}$ >5620 mg/kg (5 days, dietary)	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)			LC <sub>50</sub> >1250 mg/kg (14 days, soil)	LD <sub>50</sub> >100 μg/bee (48 hours, oral/contact)				
ROUNDUP ORIGINAL	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)			LC <sub>50</sub> >5000 mg/kg (14 days, dry soil)	LD <sub>50</sub> >100 µg/bee (48 hours, oral/contact)				
ROUNDUP ULTRAMAX	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)	LC <sub>50</sub> >5620 mg/kg (5 days, dietary)			LC <sub>50</sub> >1250 mg/kg (14 days, soil)	LD <sub>50</sub> >100 μg/bee (48 hours, oral/contact)				
<sup>a</sup> Taken from Material Safety Data <sup>b</sup> MSDS specifies glyphosate acid NS = Not specified	Sheets in C&P Press (20	02) unless otherwise sp	pecified.		-	·				

Appendix 3c:	Appendix 3c: Standard aquatic toxicity values for various formulations of glyphosate <sup>a</sup>										
Formulation	Rainbow trout LC <sub>50</sub>	Bluegill sunfish LC <sub>50</sub>	Fathead minnow LC <sub>50</sub>	Channel catfish LC <sub>50</sub>	Chinook salmon LC <sub>50</sub>	Coho salmon LC <sub>50</sub>	Carp LC <sub>50</sub>	Crustacea LC <sub>50</sub> /EC <sub>50</sub>	Daphnia LC <sub>50</sub> /EC <sub>50</sub>	Algae EC <sub>50</sub>	
ACCORD SP	60 mg/L (acute)										
AQUA NEAT	>1000 mg/L (96-hr)	>1000 mg/L (96-hr)							930 mg/L (48-hr)		
AQUAMASTER	>1000 mg/L <sup>b</sup> (96-hr)	>1000 mg/L <sup>b</sup> (96-hr)							930 mg/L <sup>b</sup> (48-hr)	72.9 mg/L <sup>b</sup> (72-hr) (Scenedesmus subspicatus)	
CORNERSTON E	NS	NS							NS	NS	
CREDIT Systemic	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)	5.8 mg/L (96-hr) (dynamic) 14 mg/L (96-hr) (static)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	2.1 mg/L (96-hr)		42 mg/L (48-hr)	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	2.1 mg/L (72-hr) (S. Capricornutum)	
CREDIT	22 mg/L (96- hr) (static) 8.2 mg/L (96-hr) (dynamic)	5.8 mg/L (96-hr) (dynamic) 14 mg/L (96-hr) (static)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	2.1 mg/L (96-hr)		42 mg/L (48-hr)	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	2.1 mg/L (72-hr) (S. Capricornutum)	
DEBIT TMF	>1000 mg/L (96-hr)	>1000 mg/L (96-hr)							930 mg/L (48-hr)		
EAGRE	>1000 mg/L (96-hr)	>1000 mg/L (96-hr)							930 mg/L (48-hr)		
FORESTER'S	>1000 mg/L (96-hr)	>1000 mg/L (96-hr)							930 mg/L (48-hr)		

Appendix 3c:	Appendix 3c: Standard aquatic toxicity values for various formulations of glyphosate <sup>a</sup>										
Formulation	Rainbow trout LC <sub>50</sub>	Bluegill sunfish LC <sub>50</sub>	Fathead minnow LC <sub>50</sub>	Channel catfish LC <sub>50</sub>	Chinook salmon LC <sub>50</sub>	Coho salmon LC <sub>50</sub>	Carp LC <sub>50</sub>	Crustacea LC <sub>50</sub> /EC <sub>50</sub>	Daphnia LC <sub>50</sub> /EC <sub>50</sub>	Algae EC <sub>50</sub>	
GLYFOS PRO	95-171 mg/L <sup>e</sup> (96-hr)								87 mg/L (48-hr)	118 mg/L (96-hr) (Scenedesmus subspicatus)	
GLYFOS X- TRA	18.6 mg/L (96-hr) (static)	11.9 mg/L (96-hr) (static)							21.6 mg/L (48-hr)	17.4 mg/L (72-hr)	
GLYFOS AQUATIC	95-171mg/L <sup>d</sup> (96-hr)								87 mg/L <sup>d</sup> (48-hr)	118 mg/L <sup>d</sup> (96-hr) (Scenedesmus subspicatus)	
GLYFOS	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)	5.8 mg/L (96-hr) (dynamic) 14 mg/L (96-hr) (static)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	22 mg/L (96-hr)		42 mg/L (48-hr)	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	2.1 mg/L (72-hr) (S. <i>Capricornutum</i> )	
GLYPHOMAX Plus	109 mg/L (acute)								105 mg/L (acute immobiliza -tion)	2.50 mg/L° (Scenedesmus subspicatus) 48.4 mg/L° (duckweed)	
GLYPHOMAX	109 mg/L (acute)								105 mg/L (acute immobiliza -tion)	2.50 mg/L ( <i>Scenedesmus</i> <i>subspicatus</i> ) 48.4 (duckweed)	

Appendix 3c:	Appendix 3c: Standard aquatic toxicity values for various formulations of glyphosate <sup>a</sup>										
Formulation	Rainbow trout LC <sub>50</sub>	Bluegill sunfish LC <sub>50</sub>	Fathead minnow LC <sub>50</sub>	Channel catfish LC <sub>50</sub>	Chinook salmon LC <sub>50</sub>	Coho salmon LC <sub>50</sub>	Carp LC <sub>50</sub>	Crustacea LC <sub>50</sub> /EC <sub>50</sub>	Daphnia LC <sub>50</sub> /EC <sub>50</sub>	Algae EC <sub>50</sub>	
GLYPHOSATE	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)	5.8 mg/L (96-hr) (dynamic) 14 mg/L (96-hr) (static)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	22 mg/L (96-hr)		42 mg/L (48-hr)	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	2.1 mg/L (72-hr) (S. <i>Capricornutum</i> )	
GLYPHOSATE VMF	>1000 mg/L (96-hr)	>1000 mg/L (96-hr)							930 mg/L (48-hr)		
GLYPHOSATE ORIGINAL	15.26 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (flow- through)	14 mg/L (96-hr) (static) 5.8 mg/L (96-hr) (flow- through)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	22 mg/L (96-hr)	19.7 ppm (96-hr TL <sub>50</sub> )	42 mg/L (48-hr) (Gammaru s pseudolim naeus) >1000 ppm (96-hr) (crawfish)	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)		
Glypro	60 mg/L (acute)										
GLYPRO PLUS	109 mg/L (acute)								105 mg/L (acute immobiliza -tion)	2.50 mg/L <sup>c</sup> ( <i>Scenedesmus</i> <i>subspicatus</i> ) 48.4 mg/L <sup>c</sup> (duckweed)	
Нолсно	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	

Appendix 3c: Standard aquatic toxicity values for various formulations of glyphosate <sup>a</sup>										
Formulation	Rainbow trout LC <sub>50</sub>	Bluegill sunfish LC <sub>50</sub>	Fathead minnow LC <sub>50</sub>	Channel catfish LC <sub>50</sub>	Chinook salmon LC <sub>50</sub>	Coho salmon LC <sub>50</sub>	Carp LC <sub>50</sub>	Crustacea LC <sub>50</sub> /EC <sub>50</sub>	Daphnia LC <sub>50</sub> /EC <sub>50</sub>	Algae EC <sub>50</sub>
MIRAGE	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)	14 mg/L (96-hr) (static)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	22 mg/L (96-hr)		42 mg/L (48-hr)	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	2.1 mg/L (72-hr) (S. <i>Capricornutum</i> )
PROSECUTOR Plus Tracker	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)								37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	
PROSECUTOR	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)								37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	
RATTLER	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS
RAZOR SPI	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)	5.8 mg/L (96-hr) (dynamic) 14 mg/L (96-hr) (static)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	2.1 mg/L (96-hr)		42 mg/L (48-hr	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	

Appendix 3c:	Appendix 3c: Standard aquatic toxicity values for various formulations of glyphosate a										
Formulation	Rainbow trout LC <sub>50</sub>	Bluegill sunfish LC <sub>50</sub>	Fathead minnow LC <sub>50</sub>	Channel catfish LC <sub>50</sub>	Chinook salmon LC <sub>50</sub>	Coho salmon LC <sub>50</sub>	Carp LC <sub>50</sub>	Crustacea LC <sub>50</sub> /EC <sub>50</sub>	Daphnia LC <sub>50</sub> /EC <sub>50</sub>	Algae EC <sub>50</sub>	
RAZOR	22 mg/L (96-hr) (static) 8.2 mg/L (96-hr) (dynamic)	5.8 mg/L (96-hr) (dynamic) 14 mg/L (96-hr) (static)	9.4 mg/L (96-hr)	16 mg/L (96-hr)	20 mg/L (96-hr)	2.1 mg/L (96-hr)		42 mg/L (48-hr)	37 mg/L (48-hr) (aeration) 24 mg/L (48-hr) (without aeration)	2.1 mg/L (72-hr) (S. Capricornutum)	
Rodeo	60 mg/L (acute)										
ROUNDUP ProDry	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	
Roundup UltraDry	Not tested	Not tested	Not tested	Not tested	Not tested	Not tested	Not tested	Not tested	Not tested	Not tested	
ROUNDUP PRO	5.4 mg/L (96-hr) (static)	7.3 mg/L (96- hr) (static)							11 mg/L (48-hr) (static)		
ROUNDUP Original	8.2 mg/L (96-hr) (flow- through)	5.8 mg/L (96-hr) (flow- through)							12.9 mg/L (48-hr) (static)	2.6 mg/L (96-hr) (static) ( <i>S</i> . <i>Capricornutum</i> )	
ROUNDUP PRO CONCENTRAT E	5.4 mg/L (96-hr) (static)	7.3 mg/L (96- hr) (static)							11 mg/L (48-hr) (static)		
ROUNDUP CUSTOM	>1000 mg/L (96-hr) (static)	>1000 mg/L (96-hr) (static)							930 mg/L (48-hr) (static)	72.9 mg/L (72-hr) (Scenedesmus subspicatus)	

Appendix 3c:	Appendix 3c: Standard aquatic toxicity values for various formulations of glyphosate <sup>a</sup>										
Formulation	Rainbow trout LC <sub>50</sub>	Bluegill sunfish LC <sub>50</sub>	Fathead minnow LC <sub>50</sub>	Channel catfish LC <sub>50</sub>	Chinook salmon LC <sub>50</sub>	Coho salmon LC <sub>50</sub>	Carp LC <sub>50</sub>	Crustacea LC <sub>50</sub> /EC <sub>50</sub>	Daphnia LC <sub>50</sub> /EC <sub>50</sub>	Algae EC <sub>50</sub>	
ROUNDUP PRO BIACTIVE	>989 mg/L (96-hr)						>895 mg/L (96-hr)		676 mg/L° (48-hr)	150 mg/L (EbC <sub>50</sub> ) (growth total) 393 mg/L (ErC <sub>50</sub> ) (growth rate) ( <i>Selenastrum</i> <i>capricornutum</i> )	
ROUNDUP UltraMax	5.4 mg/L (96-hr) (static)	7.3 mg/L (96- hr) (static)							11 mg/L (48-hr) (static)	72.9 mg/L <sup>b</sup> (72-hr) (Scenedesmus subspicatus)	
<ul> <li><sup>a</sup> Taken from Material Safety Data Sheets in C&amp;P Press (2002) unless otherwise specified.</li> <li><sup>b</sup> Isopropylamine salt of glyphosate (62%)</li> <li><sup>c</sup>Growth inhibition</li> <li><sup>d</sup> MSDS specifies that value represents the acute toxicity of glyphosate free acid</li> <li><sup>e</sup> MSDS specifies that value represents the acute toxicity of the active ingredient glyphosate</li> </ul>											

Number of individuals, formulation, [Location]	Average Dose	Symptoms, Outcome, and post mortem pathology	Reference
50, Roundup [Japan]	181±201 mL	Esophageal injury observed in 68% of patients; gastric injury in 72%; and duodenal injury in 16%. One patient died on the second hospital day due to refractory shock and aspiration pneumonia.	Chang et al. 1999
2, Roundup [New Zealand]	200-250 ml [fatal]	Vomiting and acidosis. Both individuals died. Ulcerated oropharynx, congested lungs and airway mucosa, petechial submucosal hemorrhages and gastric funduc, acute pulmonary edema, and acute tubular necrosis of the lungs in on individual. Edema of the bronchi and lungs in the other individual.	Dickson et al. 1988
2, glyphosate [Spain]	NS [fatal]	Concentrations in blood (1.64-892.27 ppm) and gastric content (0.08-11.06 ppm)	Garcia- Repetto et al. 1998
53, Roundup [Taiwan]	258±347 mL (range 15- 2000 mL)	Blood WBC counts significantly higher and hospital stays significantly longer in patients with laryngeal injury (p<0.005); laryngeal injury stronly correlated with aspiration pneumonitis ( $X^2$ =4.449, p<0.05)	Hung et al. 1997
1, Roundup [U.S.]	N.S.	A self report of "nervous system and immune system problems" that "no doctor has been able to accurately diagnose and treat"	Jensen 1989
1, Roundup [Japan]	N.S.	Foam and fluid in the trachea and bronchi. Death attributed to infalation of vomitus into the lungs	Kageura et al. 1988
N.S., Roundup and others, [France]	N.S.	Estimated lethal dose of about 1 g/kg.	Kammerer 1995
131, GlySH (glyphosate- surfactant herbicide), [Taiwan]	330±42 mL [fatal]	11 fatalities (mortality rate of 8.4%); most common presentations included sore throat, nausea (with or without vomiting) and ferver; most common laboratory abnormalities included leukocytosis (68%), decreased bicarbonate (48.1%); acidosis (35.8%), elevated AST (33.6%), hypoxemia (28.4%), and elevated BUN (17.1%).	Lee et al. 2000
1, Chun-Dou- Dou (41% isopropylamine salt of glyphosate, 15% polyoxy- ethylene-amine) [China]	~150 mL	Cardiogenic shock with accelerated idio-ventricular rhythm	Lin et al. 1999
4, Roundup [New Zealand]	50 -1,000 ml [non-fatal] 200-250 ml [fatal]	Abdominal pain, diarrhea and vomitting. Decreased urinary output. Estimates of non-fatal doses: 85 g for 27 year old male, 18-36 g for 15 year old female, " <i>up to 1 liter</i> " for a 38 year old male. About 72-91 g for a 43 year old woman.	Menkes et al. 1991

## Appendix 4. Case reports of poisoning by glyphosate formulations

Number of individuals, formulation, [Location]	Average Dose	Symptoms, Outcome, and post mortem pathology	Reference
1, Roundup, [Israel]	NS [non fatal]	Shortness of breath, irritative cough, dizziness, throat discomfort, episodes of hemoptysis, temperature of 38.40 °C (101.12 °F), mild to moderate respiratory distress, diffuse rales and crackles heard over the lungs. 42-year-old, male mechanic.	Pushnoy et al. 1998
56, Roundup [Japan]	104 ml [non- fatal] 206 ml [fatal]	Hypovolemic shock. Sore throat, abdominal pain, and vomiting. Pulmonary edema (3 cases) and severe pneumonia (2 cases). Oliguria, anuria, and hypotension in all fatal cases. Increases serum amylase and WBC count, some with increased bilirubin and LDH activity, probably attributable to hemolysis.	Sawada et al. 1988
93, Roundup [Taiwan]	184±70 mL (range 85- 200 mL) [fatal]; however, ingestion of much higher amounts (500 mL) only resulted in mild to moderate signs and symptoms	<ul> <li>Mild: mainly GIT symptoms (nausea, vomiting, diarrhea, abdominal pain, mouth and throat pain) that resolved in 24 hours</li> <li>Moderate: GII symptoms lasting longer than 24 hours, GIT hemorrhage, endoscopically verified oesophagitis or gastritis, oral ulceration, hypotension responsive to IV fluids, pulmonary dysfunction not requiring intubation, acid-base disturbance, evidence of transient hepatic or neal damage, or temporary oliguira.</li> <li>Severe: pulmonary dysfunction requiring intubation, renal failure requiring dialysis, hypotension requiring treatment with pressor amines, cardiac arrest, coma, repeated seizures, or death.</li> </ul>	Talbot et al. 1991
1, Roundup [New Zealand]	200-250 ml [fatal]	Hypotension, metabolic acidosis, and vomiting, and hyperkalemia. Death due to respiratory and cardiac arrest. Pulmonary edema and acute renal tubular necrosis.	Temple and Smith 1992
92, Roundup [Taiwan]	120 ml (range of 5- 500 ml) [non-fatal] 263 ml (range of 150-500 ml) [fatal]	Irritation and pain in the throat and mouth, some with oral mucosal ulceration. Gastritis, esophagitis, and mucosal edema. Vomiting and diarrhea. Abdominal or epigastric pain. Diffuse pulmonary damage, non- cardiogenic pulmonary edema. Intensive therapy failed to reverse hypoxemia in fatal cases. Oliguria or anuria in 10 patients, perhaps related to hypotension. Metabolic acidosis. Mild temperature elevations in 7 patients.	Tominack et al. 1991
74, Glyphosate (NOS) [Taiwan]	NS	Glyphosate was among the 28 substances (1.27%) most frequently involved in pediatric poisoning exposures reported to NPC Taiwan 1985-1993. Details regarding the signs and symptoms of poisoning are not provided.	Yang et al. 1997
83 [France]	NS	34 cases were associated with Roundup. 40 individuals evidenced no signs of toxicity. 3 individuals died. The most frequent clinical signs involved gastrointestinal irritation.	Weppelman 1994

## Appendix 4. Case reports of poisoning by glyphosate formulations

Species/ Strain/ Sex/No.	Route/ Exposure Level (Estimated Dose) and Duration	Effects	Reference
Rats/ F344/N 10/sex/ dose	3125, 6250, 12500, 25000, 50000 ppm in diet for 13 weeks.	Decrease in body weight in males (20%) and females (5%) at the highest dose level. In males, small increases in relative liver, kidney, and testicle weights and a decrease in relative thymus weight. No significant organ weight changes in females.	NCI 1992
	(205, 410, 811, 1678, 3393 mg/kg/day for males) (212, 421, 844	Hematologic changes (increased hematocrit, RBC) at the three higher dose levels and increased hemoglobin at the two higher dose levels in males. The hematologic effects are unremarkable and attributed to mild dehydration. Treatment related increases in alkaline phosphatase in both sexes at all time points suggestive of mild liver toxicity.	
	(213, 421, 644, 1690, 3393 mg/kg/day for females)	In males at the two higher dose levels, a 20% decrease in sperm counts. In females, a longer estrous cycle at the highest dose.	
		Salivary gland lesions in both sexes at all dose levels with increasing incidence and severity with increasing dose. The effect could be blocked by isoproterenal, indicating an adrenergic mechanism.	
Rat/ Sprague Dawley6 0/sex/ group	2000, 8000, or 20,000 ppm in diet for 24 months (89, 362, or 940 mg/kg/day for males) (113–45	Significant decrease in body weight gain in high-dose females (day 51-month 20); significant increases in cataracts and lens abnormalities in high-dose males; significant decrease in urinary tract pH in high-dose males; increased relative liver weights; significantly in-creased incidence of inflammation of the gastric mucosa in mid-dose females.	Stout and Ruecker 1990
	or 1183 mg/kg/day for females)	This study reports a NOAEL of 8000 ppm based on decreased body weight data. Increased incidence of pancreatic islet cell adenomas (low-dose males) and C-cell adenomas in the thyroid of mid- and high-dose males and females; slight increase in hepatocellular adenomas in males.	
		Due to the high incidence of pancreatic cell adenomas, the EPA recommended that the carcinogenic potential of glyphosate be evaluated by the Peer Review Committee.	
Rat/ Sprague	0, 30, 100, or 300 ppm in diet for 26 months	No significant changes in body weight gain, organ weights, organ/body weight ratios, or hematological and clinical chemistry parameters.	Lankas and Hogan
Dawley5 0/sex/ group	(3.1, 10.3, or 31.5 mg/kg/day for	Increased rate of interstitial cell tumors of the testes in high-dose males.	Bio/ dynamics,
	males) (3.4, 11.3, or 34.0 mg/kg/day	Systemic NOAEL for nonneoplastic effects = 31 mg/kg/day.	Inc. 1981a
	tor temales)	Authors concluded that HDT not carcinogenic to rats. EPA concluded that since the HTD was not an MTD, study was not a valid carcinogenicity study under EPA guidelines.	

## Appendix 5. Effects on mammals of long-term exposure to glyphosate<sup>a</sup>
Species/ Strain/ Sex/No.	Route/ Exposure Level (Estimated Dose) and Duration	Effects	Reference
Mouse/C D/50/sex/ group	1000, 5000, or 30,000 ppm in diet for 24 months (111–250, 519–1264, or 3465–7220 mg/kg/day for males) (129–288, 690–1322, or 4232–9859 mg/kg/day for females)	Lower mean body weights (as much as 11% at week 102) among high-dose males; elevated mean absolute and relative weights of testes in high-dose males. Histo- pathological changes included hepatic centrilobular hypertrophy and necrosis of hepatocytes in high-dose males and chronic interstitial necrosis and proximal tubule epithelial cell basophilia and hypertrophy of the kidneys in high-dose females. Sporadic occurrence (not dose related) of lymphoreticular tumors in treated females and renal tubular adenomas in males. The NOAEL for non-neoplastic chronic effects from this study is 5000 ppm, which corresponds to a dose of 750 mg/kg/day. The oncogenic response in this study (occurrence of renal adenomas in male mice) is considered equivocal.	U.S. EPA 1986
		NOTE: U.S. EPA 1995 [Federal Register July 7, Vol 60, No. 130 indicates that the exposure duration was 18 months, not 24 ( <i>cf</i> #5, pg. 35366)].	
Dog/ NS/ 6/sex/ group <sup>b</sup>	20, 100, or 500 mg/kg/day in gelatin capsules for 1 year	At 3 months, slight but toxicologically important decrease in serum sodium and potassium concentrations in males at mid- and high-dose levels and in females at high-dose level. Apparent decreases in absolute and relative weights of pituitaries in mid- and high-dose males not correlated with histopathological effects. Systemic NOAEL >500 mg/kg/day.	Reyna, 1985 Monsanto Co. 1985
Mice/ B6C3F <sub>1</sub> / 10/sex/ dose	3125, 6250, 12500, 25000, 50000 ppm in diet for 13 weeks. (507, 1065, 2273, 4776, 10780 mg/kg/day for males) (753, 1411, 2707, 5846, 11977 mg/kg/day for females)	Body weight depression at the two highest dose levels for both sexes. Increases in relative heart, kidney, liver, lung, thymus, and testis for male mice. No differences in food consumption between the dosed and control groups. No effects on sperm motility or estrous cycle length. Salivary gland lesions.	NCI 1992

Appendix 5. Effects on mammals of long-term exposure to glyphosate<sup>a</sup>

<sup>a</sup> Adapted from U.S. EPA 1992, except for NCI 1992.

<sup>b</sup> Although U.S. EPA 1992 indicates that the strain of dogs is not specified, the title of the study title indicates that Beagle dogs were used.

NS = Not specified; M = male; F = female; NOAEL = no-observed-adverse-effect level; MTD = maximum tolerated dose

Species/ Strain/Sex/ No	Exposure	Effects	Reference
		TERATOLOGY STUDIES	
Rats/CD/females/ 25/dose group	0, 300, 1000, or 3500 mg/kg/day glyphosate (98.7% pure) by gavage on days 6-19 of gestation	At 3500 mg/kg/day, severe maternal toxicity, including decreased weight gain and mortality in 6/25 animals was accompanied by decreases in fetal weights, viability, and ossification of sternebrae. NOEL = 1000 mg/kg/day for maternal and developmental toxicity	Farmer et al. 2000b
Rats/CD/ females/ 25/dose group	0, 15, 100, or 300 mg/kg/day POEA by gavage on days 6-15 of gestation	No developmental toxicity at any doses. At 100 mg/kg/day, slight maternal toxicity – i.e., decreased food consumption and mild clinical signs. At 300 mg/kg/day, mortality as well as decreases in food consumption and body weight gain. NOEL = 15 mg/kg/day for maternal toxicity	Farmer et al. 2000b
Rats/CD/ females/ 25/dose group	0, 15, 50, or 150 mg/kg/day phosphate ester neutralized POEA by gavage on days 6-15 of gestation	No developmental toxicity. At 150 mg/kg/day, mortality as well as decreases in food consumption and body weight gain. NOEL = 50 mg/kg/day for maternal toxicity Note: Farmer et al. 2000b state that the NOEL for maternal toxicity was 150 mg/kg/day. This is not consistent with the reported data and appears to be a typographical error.	Farmer et al. 2000b
Rats/Wistar/F/24	0, 250, 500 or 1000 mg/kg/day by gavage on Days 7-16 of gestation	No signs of maternal or developmental toxicity.	Moxon 1996a
Rat/CD/F/NS	0, 300, 1000, or 3500 mg/kg/day by gavage	Breathing difficulty, reduced activity, diarrhea, stomach hemorrhages, weight gain deficits, altered physical appearance, and mortality during treatment in high-dose dams; unossified sternebrae in fetuses from high-dose dams. The NOAELs for fetotoxicity and maternal toxicity are each 1000 mg/kg/day and the NOAEL for teratogenicity is 3500 mg/kg/day (HDT).	Rodwell et al. 1980a; Cited as Monsanto Co. 1980 in U.S. EPA 1986
Rabbits/New Zealand White/F/20	0, 100, 175, and 300 mg/kg/day by gavage on Days 8-20 of gestation	Maternal toxicity – observed at 175 and 300 mg/kg/day – diarrhea, reduced fecal output, reduced food intake and body weight. Fetal toxicity – observed at 300 mg/kg/day – reduced body weight and delayed ossification. No effects on survival and no signs of teratogenicity.	Moxon 1996b

Appendix 6. Assays for reproductive/teratogenic effects in mammals after exposure to glyphosate, POEA, or neutralized POEA.

Species/ Strain/Sex/ No	Exposure	Effects	Reference
Rabbits/Dutch Belted/F/16	0, 75, 175, or 350 mg/kg/day by gavage on Days 6-27 of gestation	Maternal toxicity – 175 mg/kg/day – diarrhea. 350 mg/kg/day – diarrhea and nasal discharge. Some animals in both controls and dosed groups died from causes unrelated to glyphosate treatment. No developmental effects.	Rodwell et al. 1980b
		<b>REPRODUCTION STUDIES</b>	
CD Rats/12 M, 24F	0, 3, 10, or 30 mg/kg bw/day in diet for 3 generations	No effects on any reproductive parameters. Increase in unilateral renal tubule dilation at 30 mg/kg/day in F3b pups. Note: This appears to be a synopsis of Schroeder and Hogan (1981), summarized below.	Farmer et al. 2000a
CD Rats/30 per sex per group	0, 2000, 10000, 30000 ppm (97.7%) in the diet.	Decreased body weight in parents and pups and equivocal decrease in average litter size at 30000 ppm. No effects at lower doses. NOAEL for systemic and reproductive effects: 10000 ppm (equivalent to 740 mg/kg/day). LOAEL for reproductive effects: 30000 ppm (equivalent to 2268 mg/kg/day).	Farmer et al. 2000a
Charles River CD Rats/12 M, 24F	0, 3, 10, or 30 mg/kg bw/day in diet for 60 days	An increase in unilateral focal tubular dilation of the kidney in the male $F_{3b}$ pups (7/10 in treated animals compared with 2/10 in concurrent controls) of dams treated with 30 mg/kg/day. No compound-related effects were observed on fetal, pup, and adult survival; mean parental and pup body weight and food consumption; and mating, pregnancy, fertility, and gestation length. The authors of this study noted that the historical control indices of tubular lesions varied markedly in male weanling rat, and on the basis of the data from this 3-generation study concluded that the highest dose tested (30 mg/kg/day) had no adverse reproductive effects. Nonetheless, in view of the observed kidney lesions in the male $F_{3b}$ pups of dams treated with the highest dose, U.S. EPA 1992 concludes that a more appropriate systemic NOAEL for this study is 10 mg/kg/day, and that the LOAEL is 30 mg/kg/day based on renal effects observed in male $F_{3b}$ weanlings.	Schroeder and Hogan 1981
		OTHER	
Rabbits/ New Zealand white/ male/ 4/dose	$1/10^{\text{th}}$ and $1/100^{\text{th}}$ of the $\text{LD}_{50}$ orally in geletin capsul for 6 weeks with an additional 6 week recovery period.	Decreased body weight, libido, ejaculate volume, sperm concentrations, semen initial fructose and semen osmolality. Increases in abnormal and dead sperm.	Yousef et al. 1995

Appendix 6. Assays for reproductive/teratogenic effects in mammals after exposure to glyphosate, POEA, or neutralized POEA.

Formulation	Organism	Exposure Level	Nature of Exposure	Effects	Reference
Glyphosate	bovine	17-70 μM	in vitro lymphocyte cultures	statistically significant increase of structural aberrations, sister chromatid exchanges, and G6PD activity	Lioi et al. 1998a
Glyphosate	human	5.0, 8.5, 17.0, or 51.0 μM	in vitro lymphocyte cultures	dose-related incrase in the percent of abberant cells and an increse of SCE/cell	Lioi et al. 1998b
Glyphosate	Vicia faba	35, 70, 105, 140, 350, 700, 1050, 1400 μg/g soil <sup>c</sup>	frequency of micronucleated cells	no genotoxicity	De Marco et al. 1992
Glyphosate	Allium	1440, 2880 μg/L	Allium anaphase- telophase assay	no effect	Rank et al. 1993
Roundup	Salmonella typhimurium	360, 720, 1081, 1440 μg/plate	plate incorporation assay in the absence or presence of Aroclor induced S9 mix	slight but significant number of revertants at 360 µg/plate for TA98 (without S9) and at 720 µg/plate for TA100 (with S9)	Rank et al. 1993
Roundup	Allium	1440, 2880 μg/L	Allium anaphase- telophase assay	statistically significant increase in chromosome aberrations	Rank et al. 1993
Roundup	Tadpole ( <i>Rana</i> catesbeiana)	1.69, 6.75, or 27 mg/L	Alkaline SCG assay (24- hour exposure).	No significant increase (p>0.05) in DNA damage, compared with control at 1.69 mg/L; significant increases in DNA damage at 6.75 mg/L (p<0.05) and 27 mg/L (p<0.001), compared with controls	Clements et al. 1997
Roundup	Drosophila larvae	1 ppm	sex-linked recessive lethal (SLRL)	high frequency of lethals in laraval spermatocytes and in spermatogonia	Kale et al. 1995
Pondmaster	Drosophila larvae	0.1 ppm	sex-linked recessive lethal (SLRL)	high frequency of lethals in laraval spermatocytes and in spermatogonia	Kale et al. 1995
Roundup	mouse	133, 200 mg/kg bw	mice bone marrow micronucleus assay	no clastogenicity	Rank et al. 1993

## Appendix 7. Studies assessing the mutagenicity of glyphosate

Appendix 7.	Studies as	sessing the	mutagenicity	of glyphosate
-------------	------------	-------------	--------------	---------------

Formulation	Organism	Exposure Level	Nature of Exposure	Effects	Reference
Roundup	Tilapia rendalli	50.0, 100.0, or 200.0 mg/kg	erythrocyte micronuclei (MN) assay	statistically significant induction of MN frequencies at all doses	Grisolia 2002
Roundup	mouse	two injections of 0.5 mL within 24- hour interval	erythrocyte micronuclei (MN) assay	no MN induction	Grisolia 2002
Roundup	human	0.25, 2.5, 25 mg/mL	SCE in human lymphocytes <i>in vitro</i>	statistically significant increase (p<0.001) in SCE at 0.25 and 2.5 mg/mL; no lymphocyte growth at highest dose	Vyse and Vigfusson 1979, Vigfusson and Vyse 1980

<sup>a</sup>This study not considered adequate for assessing endpoint of concern.

<sup>b</sup>Highest nontoxic concentration.

°Used as an emulsifiable liquid in Solado trading formulation (SIAPA) containing 21% active ingredient

CHO = Chinese hamster ovary; SCE = sister-chromatid exchange, NOS = not otherwise specified

Application	Observations	Reference
Glyphosate (NOS), 0.75 lbs/acre, aerial application. Less than 7 year post cutting clear cut. Comparable are uses as control.	<b>Vegetation:</b> Mortality in only about 5% of shrubs (primarily salmonberry and thimpleberry). Defoliation in about 50% of shrubs one year postspray with increase in herbaceous (grass) cover.	Anthony and Morrison 1985
	<b>Small Mammals:</b> No marked changes in diversity and evenness of small-mammal communities over two year post-application observation period. Transient increase in <i>Microtus oregoni</i> associated with increase in grasses.	
Roundup at 2.5 to 5 kg a.e./ha, two sites in British Columbia	Either no significant differences in plant community or an increase in diversity and species richness after 10 to 12 years.	Baoteng et al. 2000
Roundup, 1 ml applied in drilled holes around root collar of treated pine trees. Untreated trees served as controls.	Increased attack success as well as egg and larval development of mountain pine beetle (MPB). Corresponding increases observed in MPB predators and parasites.	Bergvinson and Borden 1991
Roundup, applied in drilled holes around root collar at doses ranging from about 0.006 to 0.6 g/tree.	Increased predation by woodpeckers on mountain pine beetles (MPB) over a 1 year observation period.	Bergvinson and Borden 1992
Roundup, 1.7 kg a.e./ha, in summer of 1985 using a spray system mounted on a crawler- tractor. Site Description: Central Georgia, herbaceous and woody species. 0.6-0.8 ha. Woody plants removed prior to treatment. Loblolly pine seedlings planted in 1982.	Observations made in 1992-1993. No significant differences in species richness for any plant groups [Arborescents, nonarborescents, legume and nonlegume forbs, grasses, and woody vines]. No effect on plant species diversity. The only effect compared to controls was a reduction in nonarborescent species <i>Vaccinium stamineum</i> and all <i>Vaccinium</i> species combined.	Boyd et al. 1995
Roundup	ED50 of 0.7-93 $\mu$ g/plant for 14 non-target plant species. Dispersion model indicated that glyphosate could damage non-target plant species when aerially applied at concentrations of 6.4 g/L.	Breeze et al. 1992
Roundup	Applications of 3 kg a.i./ha to ponderosa pine plantations over a period of 7 to 13 years. No substantial effect on soil microorganisms based on basal respiration, metabolic quotient, total bacteria, or mineralizable nitrogen.	Busse et al., 2001

Application	Observations	Reference
Herbicidal glyphosate spray formulated as the isopropylamine salt applied at the rate of 1.4 kg a.i./ha by tractor mounted sprayer to silty clay loam soil from 1980 to 1983.	No pesticide residues detected in the soil 17 months after the last experimental treatment; no deleterious effects on crop productivity; and no differences noted in microbial processes in soils sampled in April 1992	Bromilow et al. 1996
Glyphosate (NOS)	In laboratory toxicity tests using adult carabids, no signs of toxicity at exposures equivalent to an application rate of 1.57 kg/ha. No repellant effects under laboratory conditions. In field studies, no toxic or repellent effects. Decreased numbers of carabids in field plots were secondary to effects on vegetation.	Brust 1990
Glyphosate (NOS), 3.4 kg a.i/ha.	Effects on soil invertebrates were secondary to effects on alfalfa density.	Byers and Bierlein 1984
Roundup, 2 lbs/acre by tractor mounted pump and hand-held sprayer in pine release.	Significant increase (38%) in mortality of pine seedlings after 1 year. Increased mortality also apparent after 5 years. There was, however, an increase in the number of free-to-grow survivors after 5 years.	Cain 1991
Roundup, 1.4 kg a.i./ha by hand held controlled drop band applicators in a six year old spruce plantation (North Wales)	An initial decrease in Calluna and increased amount of bare ground. After 2 years, no difference in the abundance of <i>Vaccinium</i> and <i>Empetrum</i> species. Black grouse evidenced a preference for treated areas, probably because of increased accessibility or fruiting quality.	Cayford 1988
Roundup, 0.54-3.23 kg a.i./ha	At 0.54 kg/ha, a decrease in soil fungi and bacterial populations after 2 months. No effect after 6 months. At 3.23 kg/ha, no effect on soil fungi and bacteria after 10-14 months.	Chakravarty and Chatarpaul 1990
Glyphosate applied at 2.2.kg a.i./ha via spray application to 0.75x40 m strips of crested wheatgrass (height: 20-30 cm) in June 1989 and same application repeated in May 1991(height of wheatgrass 10- 15 cm) in Swift Current Saskatchewan.	Glyphosate residues in treated foliage decreased to <50 mg/kg (international MRL for fodder of grasses) within 2 weeks of application. The major route of dissipation appeared to be washoff by rainfall. AMPA residues were generally about one order of magnitude less than the corresponding glyphosate residues.	Cessna and Waddington 1995

Appendix 8: Summary of field or field simulation studies on glyphosate formulations

Application	Observations	Reference
Roundup, 2.25 kg/ha applied aerially to field to suppress angiosperms competing with conifer regeneration.	Herbicide treatment had no effect on captures of most small mammal species over a one year observation period [Masked shrew, deer mouse, pygmy shrew, short-tailed shrew, southern bog lemming, or meadow jumping mouse]. Southern Red-backed voles were more numerous in control than in treated sites. This effect was attributed to defoliation of overhead cover.	D'Anieri et al. 1987
Roundup, 6 L/ha (about 2.1 kg/ha)	Assays for the degradation of leaf litter by isopods. There was an increased decomposition of birch and a decreased decomposition of black cherry. Possible signs of toxicity but not statistically significant.	Eijsackers 1992
Roundup	Inhibition of growth in three species of ectomycorrhizal fungi in laboratory cultures at concentrations of over 10 mg/L.	Estok et al. 1989
Roundup applied in September 1984 at a rate of 2.0 kg a.e./ha via helicopter equipped with boom to 45 ha of coastal British Columbia watershed.	Maximum glyphosate residues in two intentionally oversprayed tributaries were: <b>stream water</b> : 162 $\mu$ g/L; <b>sediments</b> : 6.80 $\mu$ g/g dry mass <b>suspended sediments</b> : <0.03 $\mu$ g/L these levels dissipated to <1 $\mu$ g/L within 96 hours Residue levels of glyphosate in buffered streams were very low (2.4-3.2 $\mu$ g/L). In the off-target assessment, <0.1% of applied glyphosate was detected at 8 m from the spray boundary.	Feng et al. 1990
Glyphosate applied aerially at rate of 1.3 kg/ha in 92 L/ha water to 40- to 60-year-old hardwood stands in Oregon during the slate summer	No direct effect on the six sampled species of amphibians	Cole et al. 1997
Glyphosate (NOS), 2.2-3 kg/ha aerial over pine forest.	Glyphosate applications had a greater impact on stream water quality than clearcutting. Effects were evident over a 5-year period. Changes in water quality would not impact the suitability of the water for human consumption.	Feller 1989
Glyphosate (NOS) applied to litter.	Concentrations of 5,000 to 10,000 ppm in litter caused a significant decrease in decomposition.	Fletcher and Freedman 1986
Glyphosate, 2 lb/acre, broadcast ground application		Haywood 1994

Application	Observations	Reference
Glyphosate (NOS), 1 kg/ha in clearcut area.	Substantial decrease utilization by mountain hare one year after spraying. A lesser decrease, not statistically significant, after 2 years.	Hjeljord et al. 1988
Roundup, 4 L/ha (1.4 kg/ha), pre- harvest treatment of pasture.	No significant effects on the consumption of treated hay by sheep.	Jones and Forbes 1984
Rodeo repeated applications via hand sprayer to control smooth cordgrass in July 1997 and July 1998. Application rates on mudflat plots during 1997 ranged from 59.5 to 67.4 L/ha, while rates during 1998 ranged from 31.5 to 34.3 L/ha. Application rates for	<ul><li>Glyphosate concentrations in sediment from mudflat plots decreased 88-96% by day 1 after treatment in 1997 to 1 year after the second Rodeo application.</li><li>Glyphosate concentrations in <i>Spartina</i> plots increased 231-591% from 1997 to 1999 because <i>Spartina</i> rhizomes did not readily metabolize or exude the compound.</li></ul>	Kilbride and Paveglio 2001
<i>Spartina</i> plots ranged from 34.1 to 39.3 L/ha in 1997 and from 39.5 to 43.0 L/ha in 1998	Comparison between the results of the study and toxicity values for marine biota suggests that under worst-case conditions, detrimental effects to aquatic biota are highly unlikely to result from repeated application of Rodeo to control <i>Spartina</i>	
Roundup, 2.52 kg/ha on pasture	Cattle preferred grazing on treated pasture over first 5-7 days post-treatment. There was an aversion to the treated area 15-21 days post treatment. Reasons for the preference and aversion were not apparent.	Kisseberth et al. 1986
Roundup, 2.0 kg a.i./ha by Microfil boom w/1.5 mm hayrack nozzles mounted on Bell helicopter. Applied to 11 spray blocks at various times over 4 days to watershed on west coast of Vancouver Island in September 1984	No undue disturbance of stream invertebrates; no increase in drift densities of most aquatic invertebrates; no significant increase in total invertebrate catches.	Kreutzweiser et al. 1989
Rodeo, 5.8 kg a.i./ha with a surfactant and drift retardant over a wetland areas.	An increase or no significant change in the usage of treated wetlands by black terns over a two year observation period. The increased usage was associated with an increase in open water and newly formed mats of dead emergent vegetation.	Linz et al. 1994

Application	Observations	Reference
Rodeo, aerially applied to a designated pool of 23 cattail- dominated wetlands in N. Dakota at 50, 70, or 90% coverages in 1990 and 1991 to assess the effects of herbicide treatments on the densities of territorial male Red- wing Blackbirds, Yellow-headed Blackbirds, and Marsh Wrens.	Two years after treatment, the densities of all three species of birds were greater in the control plots than in the treated plots. There was a positive correlation between the percent coverage of live emergent vegetation (mostly cattails) and the numbers of blackbirds and wrens. The results suggest that the numbers of the wetland dwelling birds were limited by the alteration of the cattail density, due to herbicide treatment. The investigators recommend staggering vegetation management treatments on large wetland complexes in order to help diversify the stages of cattail regeneration.	Linz et al. 1996
Rodeo, aerially applied to a designated pool of 17 cattail- dominated wetlands in N. Dakota at 50, 70, or 90% coverages in 1990 and 1991 to assess the effects of herbicide treatments on the densities of American Coots and Soras.	American Coot densities were lower in the control wetlands than in the glyphosate treated wetlands 1 year ( $p=0.04$ ) and 2 years ( $p=0.09$ ) after treatment. There was a positive correlation between the numbers of American Coots and the coverages of water and dead vegetation; however, there was a negative correlation between the numbers of American Coots and live vegetation ( $p<0.1$ ).	Linz et al. 1997
	One year after treatment, Soras were more numerous in the control wetlands ( $p=0.08$ ) than in the treated wetlands, but by 2 years after treatment, the numbers of Soras were similar among treatments. There was a positive correlation between the numbers of Soras and the coverage of live vegetation.	
Rodeo, aerially applied to a designated pool of 24 cattail- dominated wetlands in N. Dakota at 50, 70, or 90% coverages in 1990 and 1991to assess the influence of habitat changes on birds.	Positive correlation between the Black Terns and selected duck species and open water and dead cattails; positive correlation between blackbird numbers and live cattails; positive correlation between the numbers of Black Terns and the numbers of Mallards, Blue-winged Teals, Redheads, and Yellow-headed Blackbird, which suggests some common habitat requirements among these species.	Linz and Blixt 1997
	Investigators conclude that cattail management programs designed to specifically enhance duck use and decrease Red-winged Blackbird numbers may be benefit Black Terns.	
Glyphosate (NOS), 0.75-1.0 kg/ha.	Reduction of plant coverage by brush species by about 60%. Vegetation recovered after 3 years. No effect on plant species diversity. A substantial increase in the number of Norway spruce over 50 cm in height on treated vs untreated plots.	Lund-Hoie and Gronvold 1987

Application	Observations	Reference
Glyphosate (NOS), 1 kg/ha by portable mist blower.	Glyphosate used for comparison to imazapyr.	Lund-Hoie and Rognstad 1990
Glyphosate (NOS), 2.3 kg/ha aerial over clearcut.	Heavy defoliation of ferns, birch, raspberry, maple, and other taxa. No difference in abundance of breeding birds in first-post spray season. A decrease in abundance of breeding birds was noted in the second post-spray season. Changes in bird density were associated with changes in vegetation.	MacKinnon and Freedman 1993
Glyphosate (NOS), 2.2 kg/ha. Tractor-mounted team sprayer.	Bioassay of drift using five species of plants in pots. Plants were placed in greenhouse after spraying. Most species evidenced no effect when placed 4 meters downwind and no plants exposed to glyphosate drift evidenced a decrease in yield at the end of the season.	Marrs et al. 1991
Glyhosate applied at simulated drift rates (4, 14, 43, or 143 g/ha) via CO <sub>2</sub> pressurized backpack sprayer to grapevines ( <i>Vitis</i> <i>vinifera</i> )	Leaf area was reduced only by the highest application rate.	Bhatti et al. 1997
see Powers 1995	No effect on soil arthropods.	Moldenke 1992
Glyphosate (NOS), 2.6 kg/ha.	Initial glyphosate residues of 17 ppm in loam and 3.8 ppm in silt. No effect on soil nitrification or denitrification.	Mueller et al. 1981
Glyphosate (NOS), 0.8-3.0 kg/ha,	Three dose levels assayed at five different application times during the year to 13 species of wood ornamentals. The most sensitive species, damaged at all times and exposure levels, were ajuga, azalea, and variegated liriope. Other species, such as juniper, evidenced only minor and transient damage.	Neal and Skroch 1985
Glyphosate (NOS), 3.3 kg/ha.	Levels in wildlife monitored over a 55 day period. No residues exceeded 2 mg/kg in viscera and 0.5 mg/kg in whole body [shrews, deermice, woodrats, squirrel, voles, and chipmunks]. Body residues were consistently less than residues on vegetation.	Newton et al. 1984
Roundup, 1.7 and 3.3 kg/ha.	Vegetative hardwood and shrub cover over 1.5 meters in height virtually eliminated. Differences in height and cover were apparent at 9 years after application.	Newton et al. 1992a [NJAF 9:126]
Roundup, 1.7 and 3.3 kg/ha.	Conifers dominated over hardwoods. Some injury to conifers at the higher application rate.	Newton et al. 1992b [NJAF, 9:130]

Appendix 8:	Summary of	field or field	simulation	studies or	ı gly	phosate	formulations
-------------	------------	----------------	------------	------------	-------	---------	--------------

Application	Observations	Reference
Glyphosate (NOS), 0.1 g/m <sup>2</sup> in lysimeters (30 cm x 45 cm). [1 kg/ha]	Death of vegetation in lysimeters associated with increased leaching of nitrates and cations from soil. Reestablishment of vegetation over 28 month observation period retarded leaching.	Ogner 1987a,b
Glyphosate (NOS), 0.72 kg/ha, mechanical ground application in forest	Increase in nitrogen levels in streams, consistent with lysimeter studies. Increases were small and did not significantly affect water quality. Similar effects were observed after manual clearing and were judged to be secondary to changes in vegetation.	Ogner 1987c,d
Roundup, 2.88 kg a.e./ha, ground application made by 5 men in dogout canoes using knapsack sprayers in Abiala creek, Nigeria to control hyacinth infestation.	Significant (p<0.5) increase in fish population after treatment.	Olaleye and Akinyemiju 1996
Glyphosate (NOS), 2.2 kg/ha applied to 20 randomly selected larkspur plants. Direct application by single cone nozzle on CO <sub>2</sub> - pressurized backpack. Each plant was sprayed to wetness and analyzed for alkaloid concentration	No effect on the absolute amount of toxic alkaloids, compared with controls. Nonetheless, gyphosate treatment did not decrease the larkspur toxicity. Consequently, the risk of poisoning (to cattle) remains until the plants dessicate. The investigators did not examine how the herbicide used in the study affect larkspur palatability	Ralphs et al. 1998
Roundup, simulated direct spraying of pond at application rate of 0.43 kg/ha or about 0.4 lbs/acre	No effects on plankton productivity, zooplankton populations, or water quality.	Perschbacher et al. 1997
Glyphosate (NOS), 1.2 kg/ha aerial or 1.1 kg/ha manual, 54 ha clearcut and surrounding old growth forest.	No effect on body size and apparent reproductive capacity [assayed as number of placental scars and foeti] of deer mice. Deer mice were more abundant in untreated clearcut probably due to changes in food abundance and quality secondary to changes in vegetation.	Ritchie et al. 1987
Roundup, aerial application at 4.7 L a.i./42.1 L water/ha. [ $\approx$ 1.7 kg/ha?] on 4-5 year old clearcuts in North Maine.	Decrease in available browse plants on 2-year post- treatment clearcuts. Moose used treated areas less than untreated areas.	Santillo 1994
Roundup, aerial application at 4.7 L a.i./42.1 L water/ha. [ $\approx$ 1.7 kg/ha?] on 4-5 year old clearcuts in North Maine.	Total shrub, forb, and grass cover was diminished 1- 3 years post treatment. Decrease in species richness of shrubs and forbs on treated clearcuts. Decrease in numbers of invertebrates. Fewer small herbivorous mammals at 1-3 years post-treatment. No effect on carnivorous mammals. Effects attributable to changes in cover, food resources, and microclimate.	Santillo et al. 1989a,b

Application	Observations	Reference
Rodeo (4.7 L/ha) and X-77 Speader (1 L/ha) aerially applied to three mudflat sites in Willapa Bay, WA with invasive <i>Spartina</i> <i>alterniflora</i> in August 1992.	No direct or indirect short term (28 days post treatment) or long term (119 days post treatment) effects on mudflat biota.	Simenstad et al. 1996
Rodeo, 2.8 L/ha [1.3 kg/ha] in wetlands to control cattails.	Effective control of cattails. Breeding ducks and over-water duck nest densities greater on treated areas because of increase wetland opening. Decrease in aquatic invertebrates in treated areas. Could not determine if this was due to toxicity or habitat changes.	Solberg and Higgins 1993
Glyphosate (NOS), 0.7, 1.4, and 2.8 g/ha, sprayed twice weekly on to culture dishes.	Earthworms evidenced decreased growth over 100 day exposure period with an uneven dose-effect relationship. Mortality observed in some worms after about 80 days. Co-exposure to Captan appeared to reduce the response. Co-exposure to azinphos-methyl and Captan had no more effect than exposure to glyphosate alone.	Springett and Gray 1992
Roundup, aerial application to conifer forest at 1.7 kg a.i./ha.	No significant impact on numbers of bacteria, fungi, and actinomycetes in litter or soil. In laboratory bioassays, no effects are rates up to 100 times field application rates.	Stratton and Stewart 1992.
Roundup, aerial application to 2- year clearcut at 3.0 kg/ha.	Little difference in recruitment of voles between control and treated areas. Decline in deer mice during first post-spray summer and winter only. Population of deer mice increased in subsequent years. Significantly (p<0.05) better survival of female voles on treated sites.	Sullivan 1990
Roundup aerially applied at a rate of 3.0 kg/ha of active ingredient on June 18, 1982 or Roundup aerially applied at a rate of 2.2 kg/ha of active ingredient on September 12, 1979. Purpose of study is to investigate the effects of forest herbicide use on demography and diversity of small mammal communities extending to a	No adverse affect on reproduction, survival. or growth of deer mice and Oregon voles in a coastal forest one decade after application. Little change noted in species richness or diversity of small mammal communities.	Sullivan et al. 1997

decade.

<b>Appendix 8:</b>	Summary	of field or field	simulation :	studies on g	glyphosate	formulations
--------------------	---------	-------------------	--------------	--------------	------------	--------------

Application	Observations	Reference
Vision, aerial application of 2.14 kg a.i./ha during August 17-28, 1987; one treated site was retreated in 1988 due to poor application. Study area included 8 sites in the sub-boreal spruce forest in British Columbia.	Decrease in species richness of shrubs in year 1 after treatment and remained lower on treated sites throughout the 5-year period; initial decrease in crown volume index of herbaceous vegetation with quick recovery to untreated levels by year 2; no adverse effect on herbaceous species diversity or on small mammal communities.	Sullivan et al. 1998a
Roundup, 1.5 kg/ha a.i. to total orchard floor on two contiguous treatment blocks in British Columbia in July and September 1983, May, July, and September 1984 and 1985.	Vole populations consistently reduced in response to treatment, with average abundance ranging from 2.8 to 28.0 times higher on control plots, compared with treated plots. Voles declined to or near extirpation in all orchards during the winer of 1985-1986. No differences in the abundance of deer mice or northwestern chipmunks after treatment. The average abundance of deer mice ranged from 1.3 to 11.1 times higher and that of chipmunks ranged from 1.8 to 13.3 time higher on treated blocks, compared with control blocks. The large numbers of deer mice and chipmunks on treated blocks were composed mainly of resident animals.	Sullivan et al. 1998b
Roundup, 5 and 8 L/ha randomly applied to tree rows in an umbric Regosol located in northwest Spain in spring of 1996. Residues of glyphosate and AMPA were monitored in the solid and liquid phases for 8 weeks after treatment	Treated soil peaked at 6.9 $\mu$ g/g glyphosate; soil water samples peaked at 0.74 $\mu$ g/mL glyphosate. Glyphosate and AMPA concentrations in soil and water samples were almost negligible 1 month after treatment. AMPA peaked at 0.77 $\mu$ g/mL in soil water samples. Both Glyphosate and AMPA exhibited high vertical mobility in the treated soil, quickly reaching high concentrations in subsurface horizons where degradation is slower.	Veiga et al. 2001
Roundup, applied aerially in August 1991 to six clearcuts harvested between 1983 and 1985. Deciduous tree cover dominated the clearcuts and was approximately 1-2 m high.	Abundance of leaves of deciduous trees was greater on untreated sites (38 vs 11%) 1 year after treatment, but the difference was less (18 vs 12%) 7-10 years after treatment. A similar pattern was observed for deciduous shrubs. The abundance of forbs was similar (13-14%) 1 year after treatment but great on tread sties (29 vs 15%) 7-10 year after treatment. Grasses and ferns were less abundant than other forage classes. Overall, glyphosate application initially decreased the abundance of leaves of deciduous trees and shrubs used as food in summer by white-tailed deer.	Vreeland et al. 1998
Roundup, aerial application at 4 kg/ha on farmland planted for hay in previous 5 years.	No effect on any microbial soil variables tested: biomass, substrate-induced respiration, basal respiration, bacterial:fungal ratio.	Wardle and Parkinson 1991

<b>Appendix 8:</b>	Summary	of field or field	simulation	studies on	glyphosate	formulations
--------------------	---------	-------------------	------------	------------	------------	--------------

Application	Observations	Reference
Glyphosate (NOS), 5 kg/ha directly incorporated into soil of barley or weed plots.	No direct effect on basal soil respiration, microbial activity, or microbial biomass. Transient decrease in biomass on some plots secondary to toxic effects on weeds.	Wardle and Parkinson 1992
Glyphosate (NOS), 1.1 and 6.7 kg/ha, on cotton leaves.	Bioassay using Western bigeyed bug, <i>Geocoris pallens</i> . Females exposed to glyphosate laid slightly more viable eggs than matched controls. A slight dose/response related improvement in survival is also apparent over a 192 day observation period.	Yokoyama and Pritchard 1984

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Roundup Pro	rainbow trout	NS	96 hours	$LC_{50} = 8.3 \text{ ppm}$		Matura 1996a
Roundup Pro	bluegill sunfish	NS	96 hours	$LC_{50} = 6.5 \text{ ppm}$		Matura 1996a
Roundup	rainbow trout	static bioassay	24 hours 96 hours	$LC_{50} = 8.3 \text{ ppm}$ $LC_{50} = 8.3 \text{ ppm}$	(7.0-9.9 ppm) 12°C (54°F) (7.0-9.9 ppm) 12°C (54°F)	Folmar et al. 1979
Roundup	fathead minnow	static bioassay	24 hours 96 hours	$LC_{50} = 2.4 \text{ ppm}$ $LC_{50} = 2.3 \text{ ppm}$	(2.0-2.9 ppm) 22°C (72°F) (1.9-2.8 ppm) 22°C (72°F)	Folmar et al. 1979
Roundup	channel catfish	static bioassay	24 hours 96 hours	$LC_{50} = 13 \text{ ppm}$ $LC_{50} = 13 \text{ ppm}$	(11-16 ppm) 22°C (72°F) (11-16 ppm) 22°C (72°F)	Folmar et al. 1979
Roundup	bluegill	static bioassay	24 hours 96 hours	$LC_{50} = 6.4 \text{ ppm}$ $LC_{50} = 5.0 \text{ ppm}$	(4.8-8.6 ppm) 22°C (72°F) (3.8-6.6 ppm) 22°C (72°F)	Folmar et al. 1979
Roundup	rainbow trout eyed eggs	static bioassay	24 hours 96 hours	$LC_{50} = 46 \text{ ppm}$ $LC_{50} = 16 \text{ ppm}$	(35-61 ppm) (13-19 ppm)	Folmar et al. 1979
Roundup	rainbow trout sac fry	static bioassay	24 hours 96 hours	$LC_{50} = 11 \text{ ppm}$ $LC_{50} = 3.4 \text{ ppm}$	(8.8-13 ppm) (2.2-5.3 ppm)	Folmar et al. 1979
Roundup	rainbow trout swim- up fry	static bioassay	24 hours 96 hours	$LC_{50} = 2.4 \text{ ppm}$ $LC_{50} = 2.4 \text{ ppm}$	(2.0-2.9 ppm) (2.0-2.9 ppm)	Folmar et al. 1979

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Roundup	rainbow trout fingerling (1.0 g)	static bioassay	24 hours 96 hours	$LC_{50} = 2.2 \text{ ppm}$ $LC_{50} = 1.3 \text{ ppm}$	(0.93-5.2 ppm) (1.1-1.6 ppm)	Folmar et al. 1979
Roundup	rainbow trout fingerling (2.0 g)	static bioassay	24 hours 96 hours	$LC_{50} = 8.3 \text{ ppm}$ $LC_{50} = 8.3 \text{ ppm}$	(7.0-9.9 ppm) (7.0-9.9 ppm)	Folmar et al. 1979
Roundup	channel catfish eyed eggs	static bioassay	24 hours 96 hours	$LC_{50} = 43 \text{ ppm}$ $LC_{50} = \text{ND}$	(36-51 ppm) ND	Folmar et al. 1979
Roundup	channel catfish sac fry	static bioassay	24 hours 96 hours	$LC_{50} = 4.3 \text{ ppm}$ $LC_{50} = 4.3 \text{ ppm}$	(3.6-5.1 ppm) (3.6-5.1 ppm)	Folmar et al. 1979
Roundup	channel catfish swim-up fry	static bioassay	24 hours 96 hours	$LC_{50} = 3.7 \text{ ppm}$ $LC_{50} = 3.3 \text{ ppm}$	(3.4-4.1 ppm) (2.8-3.9 ppm)	Folmar et al. 1979
Roundup	channel catfish fingerling (2.2 g)	static bioassay	24 hours 96 hours	$LC_{50} = 13 \text{ ppm}$ $LC_{50} = 13 \text{ ppm}$	(11-16 ppm) (11-16 ppm)	Folmar et al. 1979
Roundup	rainbow trout fingerling (1.4 g)	static bioassay	96 hours	LC <sub>50</sub> = 54.8 ppm	(50-60 ppm)	Hildebrand et al. 1982
Roundup	rainbow trout fingerling (1.6 g)	static bioassay (field)	96 hours	LC <sub>50</sub> = 52 ppm	not reported	Hildebrand et al. 1982
Roundup	rainbow trout fingerling (2.1 g)	manual application	1 hour	100% survival; short period (15 minutes) of increased swimming activity during and shortly after application; no acute manifestations of physical discomfort such as coughing or loss of equilibrium	indigenous cutthroat trout and caddis fly larae in pools along the stream course did not show signs of stress during the period of spraying	Hildebrand et al. 1982

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Roundup	rainbow trout fingerling (2.3 g)	aerial application	NS	100% survival; no obvious signs of physical stress or discomfort from the time of spraying to conclusion of study (17 days)	no indication of stressful behavior by fish after first rainfall	Hildebrand et al. 1982
Roundup	rainbow trout	static bioassay	96 hours	$LC_{50} = 26 \text{ ppm}$	(12-38 ppm) 11°C	Mitchell et al. 1987a
Roundup	chinook salmon	static bioassay	96 hours	$LC_{50} = 20 \text{ ppm}$	(17-27 ppm) 11°C	Mitchell et al. 1987a
Roundup	coho salmon	static bioassay	96 hours	$LC_{50} = 22 \text{ ppm}$	(12-38 ppm) 11°C	Mitchell et al. 1987a
Roundup	bluegill	not reported	96 hours	$TL_{50} = 14 \text{ ppm}$	none	Monsanto Co. 1982b
Roundup	carp	not reported	96 hours	$TL_{50} = 3.9 \text{ ppm}$	none	Monsanto Co. 1982b
Roundup	trout	not reported	96 hours	$TL_{50} = 11 \text{ ppm}$	none	Monsanto Co. 1982b
Roundup	catfish	not reported	96 hours	$LC_{50} = 16 \text{ ppm}$	none	Monsanto Co. 1982b
Roundup	fathead minnow	not reported	96 hours	$LC_{50} = 9.4 \text{ ppm}$	none	Monsanto Co. 1982b
Roundup	rainbow trout	not reported	96 hours	$TL_{50} = 48 \text{ ppm}$	none	USDA 1981
Roundup	bluegill	not reported	96 hours	$TL_{50} = 24 \text{ ppm}$	none	USDA 1981
Roundup	rainbow trout	static bioassay	24 hours 96 hours	$LC_{50} = 8.3 \text{ mg/L}$ $LC_{50} = 8.3 \text{ mg/L}$	none	Folmar et al. 1979
Roundup	channel catfish	static bioassay	24 hours 96 hours	$LC_{50} = 13 \text{ mg/L}$ $LC_{50} = 13 \text{ mg/L}$	none	Folmar et al. 1979
Roundup	bluegill	static bioassay	24 hours 96 hours	$LC_{50} = 6.4 \text{ mg/L}$ $LC_{50} = 5.0 \text{ mg/L}$	none	Folmar et al. 1979
Roundup	bleak	static bioassay	96 hours	$LC_{50} = 16 \text{ ppm}$	(15-18 ppm)	Linden et al. 1979
Roundup	harpacticoid	static bioassay	96 hours	$LC_{50} = 22 \text{ ppm}$	(17-29 ppm)	Linden et al. 1979

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Roundup	coho salmon smolts	sublethal exposure	10 days	no affect on seawater adaptation: plasma sodium values not significantly different from control	there was no effect on growth and several sublethal parameters at exposure concentrations up to 2.78 ppm	Mitchell et al. 1987b
Roundup	carp (Cyprinus caripio)	ermersion in 205 mg/L glyphosate f or 410 mg/L glyphosate rs	1 hour or 0.5 hours	significant mortality (p=0.01); treated caused the appearanc eof myelin-like structures in carp hepatocytes, swelling of mitochondria, and disappearance of internal membrane of mitochondria at both exposure concentrations	none	Szarek et al. 2000
Roundup	grass carp	intermittent dosing	24 hours	$LC_{50} = 26 \text{ ppm}$	(22-30 ppm) 18-21°C; pH 8.1; hardness 270 mg/L	Tooby et al. 1980
			48 hours	$LC_{50} = 24 \text{ ppm}$	(21-28 ppm) 18-21°C; pH 8.1; hardness 270	
			96 hours	$LC_{50} = 15 \text{ ppm}$	mg/L	
					(13-18 ppm) 18-21°C; pH 8.1; hardness 270 mg/L	
Roundup	sockeye (fingerling)	static bioassay	96 hours	$LC_{50} = 26.7 \text{ ppm}$	4.2°C; pH 7.95; average weight 3.8 g	Servizi et al. 1987
Roundup	sockeye (fingerling)	static bioassay	96 hours	$LC_{50} = 27.7 \text{ ppm}$	4.2°C; pH 8.0; average weight 3.7 g	Servizi et al. 1987
Roundup	sockeye (fry)	static bioassay	96 hours	$LC_{50} = 28.8 \text{ ppm}$	4.5°C; pH 7.7; average weight 0.25 g	Servizi et al. 1987
Roundup	rainbow trout (fry)	static bioassay	96 hours	$LC_{50} = 28.0 \text{ ppm}$	15°C; pH <6.3; average weight 0.33 g	Servizi et al. 1987
Roundup	rainbow trout (fry)	static bioassay	96 hours	$LC_{50} = 25.5 \text{ ppm}$	14.5°C; pH <6.3; average weight 0.60 g	Servizi et al. 1987

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Roundup	coho salmon (fry)	static bioassay	96 hours	$LC_{50} = 42.0 \text{ ppm}$	15°C; pH <6.3; average weight 0.30 g	Servizi et al. 1987
Roundup	coho salmon (juvenile)	static bioassay	96 hours	$LC_{50} = 31 \text{ ppm}$	14°C; intermediate pH	Wan et al. 1991
Roundup	pink salmon (juvenile)	static bioassay	96 hours	$LC_{50} = 10 \text{ ppm}$	14°C; intermediate pH	Wan et al. 1991
Roundup	rainbow trout (juvenile)	static bioassay	96 hours	LC <sub>50</sub> = 31 ppm	14°C; intermediate pH	Wan et al. 1991
Roundup	Channel catfish	static bioassay	48 hours 96 hours	$LC_{50} = 16.2 \text{ ppm}$ $LC_{50} = 14.5 \text{ ppm}$		Abdelghani et al. 1997
Roundup	Bluegill sunfish	static bioassay	48 hours 96 hours	$LC_{50} = 13.2 \text{ ppm}$ $LC_{50} = 13.0 \text{ ppm}$		Abdelghani et al. 1997
Rodeo	carp	not reported	96 hours	TL <sub>50</sub> >10,000	none	Monsanto Co. 1982d
Rodeo	trout	not reported	96 hours	$TL_{50} > 1000$	none	Monsanto Co. 1982d
Rodeo	bluegill	not reported	96 hours	TL <sub>50</sub> >1000	none	Monsanto Co. 1982d
Rodeo	plains minnow	renewal	96 hours	NOAEC = 1000 mg/L	none	Beyers 1995
Rodeo	fathead minnow	renewal	96 hours	NOAEC = 1000 mg/L	none	Beyers 1995
Rodeo	rainbow trout (0.52 g)	static bioassay	96 hours	$LC_{50} = 1100$	(850-1300 ppm) 11°C; pH 6.0; hardness 5.0 mg/L	Mitchell et al. 1987a
Rodeo/X-77 <sup>b</sup> )	rainbow trout (0.52 g)	static bioassay	96 hours	$LC_{50} = 680 \text{ ppm}$	(600-820 ppm) 11°C; pH 6.0; hardness 5.0 mg/L	Mitchell et al. 1987a
Rodeo/X-77 <sup>b</sup> )	rainbow trout (0.21 g)	static bioassay	96 hours	$LC_{50} = 1070 \text{ ppm}$	(600-1920 ppm) 11°C; pH 7.8; hardness 75 mg/L	Mitchell et al. 1987a
Rodeo/X-77 <sup>b</sup> )	chinook salmon (4.2 g)	static bioassay	96 hours	$LC_{50} = 750 \text{ ppm}$	(600-1100 ppm) 11°C; pH 5.8; hardness 5.0 mg/L	Mitchell et al. 1987a

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Rodeo/X-77 <sup>b</sup> )	chinook salmon (5.9 g)	static bioassay	96 hours	LC <sub>50</sub> = 1440 ppm	(1070-1920 ppm) 11°C; pH 7.4; hardness 77 mg/L	Mitchell et al. 1987a
Rodeo/X-77 <sup>b</sup> )	coho salmon (17.9 g)	static bioassay	96 hours	$LC_{50} = 1000 \text{ ppm}$	(600-1900 ppm) 11°C; pH 5.8; hardness 5.0 mg/L	Mitchell et al. 1987a
Rodeo/X-77 <sup>b</sup> )	coho salmon (11.8 g)	static bioassay	96 hours	$LC_{50} = 600 \text{ ppm}$	(340-1100 ppm) 11°C; pH 6.2; hardness 4.5 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Rodeo)	rainbow trout (0.52 g)	static bioassay	96 hours	$LC_{50} = 580$	(460-730 ppm) 11°C; pH 6.0; hardness 5.0 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Roundup)	rainbow trout (0.37 g)	static bioassay	96 hours	$LC_{50} = 12 \text{ ppm}$	(5.7-18 ppm) 11°C; pH 6.1; hardness 4.5 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Roundup)	rainbow trout (0.37 g)	static bioassay	96 hours	LC <sub>50</sub> = 11 ppm	(5.7-18 ppm) 11°C; pH 7.6; hardness 85 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Roundup)	rainbow trout (0.37 g)	static bioassay	96 hours	$LC_{50} = 7.4 \text{ ppm}$	(5.7-10 ppm) 11°C; pH 7.7; hardness 81 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Roundup)	chinook salmon (4.6 g)	static bioassay	96 hours	$LC_{50} = 9.6 \text{ ppm}$	(7.9-13 ppm) 11°C; pH 6.1; hardness 4.5 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Roundup)	coho salmon (11.8 g)	static bioassay	96 hours	$LC_{50} = 11 \text{ ppm}$	(5.7-18 ppm) 11°C; pH 6.2; hardness 4.5 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Rodeo/X-77 <sup>b</sup> )	rainbow trout (0.52 g)	static bioassay	96 hours	$LC_{50} = 130 \text{ ppm}$	(120-160 ppm) 11°C; pH 6.0; hardness 5.0 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Rodeo/X-77 <sup>b</sup> )	rainbow trout (0.21 g)	static bioassay	96 hours	$LC_{50} = 210 \text{ ppm}$	(120-380 ppm) 11°C; pH 7.8; hardness 75 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Rodeo/X-77 <sup>b</sup> )	chinook salmon (4.2 g)	static bioassay	96 hours	$LC_{50} = 140 \text{ ppm}$	(120-220 ppm) 11°C; pH 5.8; hardness 5.0 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Rodeo/X-77 <sup>b</sup> )	chinook salmon (5.9 g)	static bioassay	96 hours	LC <sub>50</sub> = 290 ppm	(210-380 ppm) 11°C; pH 7.4; hardness 77 mg/L	Mitchell et al. 1987a

Appendix 9:	Toxicity of glyphosate	e and glyphosate	formulations to fish
-------------	------------------------	------------------	----------------------

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Glyphosate (IPA salt in Rodeo/X-77 <sup>b</sup> )	coho salmon (17.9 g)	static bioassay	96 hours	$LC_{50} = 200 \text{ ppm}$	(120-370 ppm) 11°C; pH 5.8; hardness 5.0 mg/L	Mitchell et al. 1987a
Glyphosate (IPA salt in Rodeo/X-77 <sup>b</sup> )	coho salmon (11.8 g)	static bioassay	96 hours	$LC_{50} = 120 \text{ ppm}$	(68-220 ppm) 11°C; pH 6.2; hardness 4.5 mg/L	Mitchell et al. 1987a
Glyphosate	sockeye (fingerling)	static bioassay	96 hours	$LC_{50} = 8.1 \text{ ppm}$	4.2°C; pH 7.95; average weight 3.8 g	Servizi et al. 1987
Glyphosate	sockeye (fingerling)	static bioassay	96 hours	$LC_{50} = 8.4 \text{ ppm}$	4.2°C; pH 8.0; average weight 3.7 g	Servizi et al. 1987
Glyphosate	sockeye (fry)	static bioassay	96 hours	$LC_{50} = 8.7 \text{ ppm}$	4.5°C; pH 7.7; average weight 0.25 g	Servizi et al. 1987
Glyphosate	rainbow trout (fry)	static bioassay	96 hours	$LC_{50} = 8.5 \text{ ppm}$	15°C; pH <6.3; average weight 0.33 g	Servizi et al. 1987
Glyphosate	rainbow trout (fry)	static bioassay	96 hours	$LC_{50} = 7.8 \text{ ppm}$	14.5°C; pH <6.3; average weight 0.60 g	Servizi et al. 1987
Glyphosate	coho salmon (fry)	static bioassay	96 hours	$LC_{50} = 12.8 \text{ ppm}$	15°C; pH <6.3; average weight 0.30 g	Servizi et al. 1987
Glyphosate	rainbow trout	static bioassay	96 hours	$LC_{50} = 10.42 \text{ ppm}$	(9.37-11.67) 12°C; pH 6.01; hardness 9.6 mg/L	Morgan and Kiceniuk 1992
Glyphosate	rainbow trout	static bioassay	24 hours 96 hours	$LC_{50} = 140 \text{ ppm}$ $LC_{50} = 140 \text{ ppm}$	(120-170 ppm) 12°C (54°F) (120-170 ppm) 12°C (54°F)	Folmar et al. 1979
Glyphosate	fathead minnow	static bioassay	24 hours 96 hours	$LC_{50} = 97 \text{ ppm}$ $LC_{50} = 97 \text{ ppm}$	(79-120 ppm) 22°C (72°F) (79-120 ppm) 22°C (72°F)	Folmar et al. 1979
Glyphosate	channel catfish	static bioassay	24 hours 96 hours	$LC_{50} = 130 \text{ ppm}$ $LC_{50} = 130 \text{ ppm}$	(110-160 ppm) 22°C (72°F) (110-160 ppm) 22°C (72°F)	Folmar et al. 1979
Glyphosate	bluegill	static bioassay	24 hours 96 hours	$LC_{50} = 150 \text{ ppm}$ $LC_{50} = 140 \text{ ppm}$	(120-190 ppm) 22°C (72°F) (120-190 ppm) 22°C (72°F)	Folmar et al. 1979
Glyphosate	rainbow trout fry	not reported	96 hours	$LC_{50} = 50 \text{ ppm}$	3 lbs a.e./gallon	Folmar 1976

Appendix 9:	Toxicity of	glyphosate and	glyphosate	formulations to fish
-------------	-------------	----------------	------------	----------------------

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Glyphosate	bleak	not reported	96 hours	$LC_{50} = 16 \text{ ppm}$	(15-18 ppm) 10°C (50°F)	Linden et al. 1979
Glyphosate (95% pure)	flagfish	pulse exposure	96 hours	$LC_{20} = 29.6 \text{ ppm}$	fed 8-day-old flagfish	Holdway and Dixon 1988
Glyphosate, technical	rainbow trout	not reported	96 hours	$TL_{50} = 38 \text{ ppm}$	none	USDA 1981
Glyphosate, technical	bluegill	not reported	96 hours	$TL_{50} = 78 \text{ ppm}$	none	USDA 1981
Glyphosate, technical	bluegill	dynamic test	96 hours	$TL_{50} = 24 \text{ ppm}$	none	USDA 1981
Glyphosate, technical	bluegill	not reported	96 hours	$LC_{50} = 120 \text{ ppm}$	none	Monsanto Co. 1982a
Glyphosate, technical	trout	not reported	96 hours	LC <sub>50</sub> = 86 ppm	none	Monsanto Co. 1982a
Glyphosate, technical (62% pure)	carp	semi-static	48 hours	$LC_{50} = 645 \text{ ppm}$	(400, 500, 600, 700, or 800 mg/L) 20.0±1.0°C	Neskovic et al. 1996b
Glyphosate, technical (62% pure)	carp	semi-static	96 hours	$LC_{50} = 620 \text{ ppm}$	(400, 500, 600, 700, or 800 mg/L) 20.0±1.0°C	Neskovic et al. 1996b
Glyphosate, technical	carp	not reported	96 hours	$LC_{50} = 115 \text{ ppm}$	none	Monsanto Co. 1982a
Glyphosate, technical	harlequin fish	not reported	96 hours	$LC_{50} = 168 \text{ ppm}$	none	Monsanto Co. 1982a
Glyphosate, technical	carp	static bioassay	48 hours	TL <sub>50</sub> =119 ppm TL <sub>1</sub> =146 ppm TL <sub>99</sub> =96.7 ppm	none	USDA 1981
Glyphosate, technical	carp	static bioassay	96 hours	$TL_{50} = 115 \text{ ppm}$ $TL_{1} = 125 \text{ ppm}$ $TL_{99} = 105 \text{ ppm}$	none	USDA 1981
Glyphosate, technical	rainbow trout	static bioassay	96 hours	LC <sub>50</sub> = 25,605 mg/L NOEC = 8,000 mg/L	36% active ingredient	Anton et al. 1994
Glyphosate, technical	rainbow trout	static bioassay	96 hours	$LC_{50} = 25,657 \text{ mg/L}$ NOEC = NR	38% active ingredient	Anton et al. 1994

Appendix 9: Toxicity of glyphosate and glyphosate formulations to fish

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Glyphosate, technical	rainbow trout	static bioassay	96 hours	LC <sub>50</sub> = 7,620 mg/L NOEC = 6,250 mg/L	54.9% active ingredient	Anton et al. 1994
Glyphosate, technical	goldfish	static bioassay	96 hours	LC <sub>50</sub> = 7,816 mg/L NOEC = 1,500 mg/L	54.9% active ingredient	Anton et al. 1994
Vision-10% surfactant	coho salmon	closed system respirometer	4 hours	hematocrit significantly increased over controls at lowest(3.75 and 60 ppm) concentrations (p<0.05) but expected to decrease as a result of stress; no significant increases in plasma lactate or plasma glucose	data suggest that a staress threshold was not reached for Vision-10% surfactant at concentrations up to 80% of the 96-hour $LC_{50}$	Janz et al. 1991
Vision	rainbow trout	sublethal exposure	1 month	fish in highest concentration (45.75 µg/L) had significantly higher frequency of wigwags	little overall effect of exposure to Vision on rainbow trout	Morgan and Kiceniuk 1992
Vision	rainbow trout	sublethal exposure	2 months	fish in lowest concentration (4.25 µg/L) performed significantly fewer wigwags	little overall effect of exposure to Vision on rainbow trout; it is not clear what the implications of a change in one agonistic activity in the repetoire of aggressive behavior would be in terms of fish's ability to hold a feeding station	Morgan and Kiceniuk 1992
MONO818	sockeye (fingerling)	static bioassay	96 hours	$LC_{50} = 4.0 \text{ ppm}$	4.2°C; pH 7.95; average weight 3.8 g	Servizi et al. 1987
MONO818	sockeye (fingerling)	static bioassay	96 hours	$LC_{50} = 4.2 \text{ ppm}$	4.2°C; pH 8.0; average weight 3.7 g	Servizi et al. 1987
MONO818	sockeye (fry)	static bioassay	96 hours	$LC_{50} = 4.3 \text{ ppm}$	4.5°C; pH 7.7; average weight 0.25 g	Servizi et al. 1987

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
MONO818	rainbow trout (fry)	static bioassay	96 hours	$LC_{50} = 4.2 \text{ ppm}$	15°C; pH <6.3; average weight 0.33 g	Servizi et al. 1987
MONO818	rainbow trout (fry)	static bioassay	96 hours	$LC_{50} = 3.8 \text{ ppm}$	14.5°C; pH <6.3; average weight 0.60 g	Servizi et al. 1987
MONO818	coho salmon (fry)	static bioassay	96 hours	$LC_{50} = 6.3 \text{ ppm}$	15°C; pH <6.3; average weight 0.30 g	Servizi et al. 1987
Glyphosate	rainbow trout	static bioassay	96 hours	$LC_{50} = 10.42 \text{ ppm}$	(9.37-11.67) 12°C; pH 6.01; hardness 9.6 mg/L	Morgan and Kiceniuk 1992
Rodeo/X-77	rainbow trout	static bioassay	96 hours	$LC_{50} = 880$ ppm for Rodeo alone $LC_{50} = 640$ ppm with X-77	10±2°C	Mitchell and Chapman 1985a
Rodeo/X-77	coho salmon	static bioassay	96 hours	$LC_{50} = 930 \text{ ppm}$	10±2°C	Mitchell and Chapman 1985b
Rodeo/X-77	chinook salmon	static bioassay	96 hours	LC <sub>50</sub> = 700 ppm	10±2°C	Mitchell and Chapman 1985c
Rodeo/X-77	chinook salmon	static bioassay	96 hours	$LC_{50} = 1440 \text{ ppm}$ NOEC = 597 ppm	10±2°C	Mitchell and Chapman 1987a
Rodeo/X-77	rainbow trout	static bioassay	96 hours	$LC_{50} = 1070 \text{ ppm}$ NOEC = 340 ppm	10±2°C	Mitchell and Chapman 1987b
Rodeo/X-77	chinook salmon smolt	static bioassay	10 days	Concentrations of 0.71 to 60.4 ppm. No effect on adaptation from fresh to salt water	11±2°C fresh water for 10 days followed by transfer to salt water	Mitchell and Chapman 1987a
Surfactant used in Roundup	rainbow trout	static bioassay	24 hours 96 hours	$LC_{50} = 2.1 \text{ ppm}$ $LC_{50} = 2.0 \text{ ppm}$	(1.6-2.7 ppm) 12°C (54°F) (1.5-2.7 ppm) 12°C (54°F)	Folmar et al. 1979
Surfactant used in Roundup	fathead minnow	static bioassay	24 hours 96 hours	$LC_{50} = 1.4 \text{ ppm}$ $LC_{50} = 1.0 \text{ ppm}$	(1.2-1.7 ppm) 22°C (72°F) (1.2-1.7 ppm) 22°C (72°F)	Folmar et al. 1979

Appendix 9: Toxicity of glyphosate and glyphosate formulations to fish

Appendix 9-10

Formula-tion	Species	Nature of Exposure	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Surfactant used in Roundup	channel catfish	static bioassay	24 hours 96 hours	$LC_{50} = 18 \text{ ppm}$ $LC_{50} = 13 \text{ ppm}$	(8.5-38 ppm) 22°C (72°F) (10-17 ppm) 22°C (72°F)	Folmar et al. 1979
Surfactant used in Roundup	bluegill	static bioassay	24 hours 96 hours	$LC_{50} = 3.0 \text{ ppm}$ $LC_{50} = 3.0 \text{ ppm}$	(2.5-3.7 ppm) 22°C (72°F) (2.5-3.7 ppm) 22°C (72°F)	Folmar et al. 1979
Syndets surfactant	channel catfish	static bioassay	48 hours 96 hours	$LC_{50} = 3.8 \text{ ppm}$ $LC_{50} = 3.6 \text{ ppm}$		Abdelghani et al. 1997
Syndets surfactant	bluegill sunfish	static bioassay	48 hours 96 hours	$LC_{50} = 3.1 \text{ ppm}$ $LC_{50} = 3.1 \text{ ppm}$		Abdelghani et al. 1997

<sup>a</sup>Values in parentheses are the 95% confidence limits.

<sup>b</sup>Rodeo /X-77 consists of 312 mL Rodeo mixed with 699 mL water and 4 mL X-77 surfactant.

NOEC = No-observed-effect concentration; NOAEC = No-observed-acute-effect concentration; ND = not determined

Formulation	Species	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Roundup	Carwfish (Proambarus)	48 hours 96 hours	$LC_{50} = 96597 \text{ ppm}$ $LC_{50} = 64002 \text{ ppm}$		Abdelghani et al. 1997
Syndets Surfactant	Carwfish ( <i>Proambarus</i> )	48 hours 96 hours	$LC_{50} = 27.9 \text{ ppm}$ $LC_{50} = 19.0 \text{ ppm}$		Abdelghani et al. 1997
Glyphosate	midge larvae ( <i>Chironomus</i> <i>plumosus</i> ; insecta)	48 hours	$EC_{50} = 55 \text{ ppm}$	(31-97 ppm) 22°C (72°F)	Folmar et al. 1979
Roundup surfactant	midge larvae ( <i>Chironomus</i> <i>plumosus</i> ; insecta)	48 hours	EC <sub>50</sub> = 13 ppm	(7.1-24 ppm) 22°C (72°F)	Folmar et al. 1979
Roundup	red swamp crawfish (Procambarus clarkii)	96	$LC_{50} = 47.31 \text{ ppm}$	(41.06-51.69)	Holck and Meek 1987
Roundup	fourth instar Anopheles quadrimaculatus larvae	24	LC <sub>50</sub> = 673.43 ppm	(572.57-770.17)	Holck and Meek 1987
Roundup	fourth instar Psurophora columbiae larvae	24	LC <sub>50</sub> = 940.84 ppm	(823.08-1067.12)	Holck and Meek 1987
Roundup	fourth instar Culex salinarius larvae	24	$LC_{50} = 1563.69$ ppm	(1262.00-2214.54)	Holck and Meek 1987
Roundup	cladoceran ( <i>Daphnia magna</i> ; crustacea)	48 hours	$EC_{50} = 3.0 \text{ ppm}$	(2.6-3.4 ppm) 22°C (72°F)	Folmar et al. 1979
Roundup	cladoceran ( <i>Daphnia pulex</i> ; crustacea)	48 hours	$EC_{50} = 3.2 \text{ ppm}$	(3.0-3.4 ppm) 22°C (72°F)	Hartman and Martin 1984
Roundup	cladoceran ( <i>Daphnia pulex</i> ; crustacea)	48 hours	$EC_{50} = 7.9 \text{ ppm}$	(7.2-8.6 ppm) 22°C (72°F)	Hartman and Martin 1984
Roundup	cladoceran ( <i>Daphnia</i> sp.; crustacea)	48 hours	$LC_{50} = 192 \text{ ppm}$	(181-205 ppm)	USDA 1981
Roundup	cladoceran ( <i>Daphnia</i> sp.; crustacea)	48 hours	$LC_{50} = 5.3 \text{ ppm}$	NS	Monsanto Co. 1982b
Roundup Pro	cladoceran ( <i>Daphnia magna</i> ; crustacea)	48 hours	$LC_{50} = 8.9 \text{ ppm}$		Matura 1996a
Rodeo	cladoceran ( <i>Daphnia</i> sp.; crustacea)	48 hours	$LC_{50} = 930 \text{ ppm}$	NS	Monsanto Co. 1982d

### Appendix 10: Acute toxicity of glyphosate to aquatic invertebrates

Appendix 10-1

Formulation	Species	Exposure Time	Effects	Comments <sup>a</sup>	Reference
Rodeo	Daphnia magna	48 hours	$LC_{50} = 218 \text{ ppm}$	(150-287 ppm)	Henry et al. 1994
Rodeo	Hyalella azteca	96 hours	$LC_{50}=720\ ppm^b$	(399-1076 ppm)	Henry et al. 1994
Rodeo	Chironomus riparius	48 hours	$LC_{50} = 1216 \text{ ppm}^{b}$	(996-1566 ppm)	Henry et al. 1994
Rodeo	Nephelopsis obscura	96 hours	$LC_{50} = 1177 \text{ ppm}^{b}$	(941-1415 ppm)	Henry et al. 1994
Glyphosate	snails (Pseudosuccinea columella)	4 weeks	biochemical alteration	increased protein concentration of snails reared in 1.0 mg/L compared with those reared in 0.1 mg/L; exact mechanism for response not determined	Christian et al. 1993
Roundup	snails (Pseudosuccinea columella)	3 generations	delayed effect on growth and development, egg- laying capacity, and hatching	0.1-10 mg/L	Tate et al. 1997
Glyphosate (97%)	snails (Pseudosuccinea columella)	4 weeks	increased quantity of free amino acid pool	1-10 mg/L	Tate et al. 2000
Roundup	amphipod ( <i>Gammarus pseudolimnaeus</i> ; crustacea)	48 hours 96 hours	$LC_{50} = 62 \text{ ppm}$ $LC_{50} = 43 \text{ ppm}$	(40-98 ppm) 12°C (54°F) (28-66 ppm) 12°C (54°F)	Folmar et al. 1979
Roundup	Harpacticoid ( <i>Nitocra spinipes</i> ; crustacea)	96 hours	$LC_{50} = 22 \text{ ppm}$	(17-29 ppm) 21.1°C (70 ± 2°F)	Linden et al. 1979

### Appendix 10: Acute toxicity of glyphosate to aquatic invertebrates

<sup>a</sup> Values in parentheses are the 95% confidence limits.

<sup>b</sup>Only 50% of the test organisms were killed in the highest concentration tested.

NS = Not specified.

Appendix 11: Toxicity of glyphosate and glyphosate formulations to a	quatic j	plants
	,	,

Species	Endpoint	Reference
	<u>Glyphosate</u>	
Selenastrum capricornutum, green algae	4 day $EC_{50} = 12.5 \text{ mg/L}$	U.S. EPA 1993b
Navicula pelliculosa, diatom	4 day $EC_{50} = 39.9 \text{ mg/L}$	
Skeletonema costatum,	4 day $EC_{50} = 0.85 \text{ mg/L}$	
Anabaena flosaquae, cyanobacter	4 day $EC_{50} = 11.7 \text{ mg/L}$	
Lemna gibba, duckweed	7 day $EC_{50} = 11.7 \text{ mg/L}$	
Chlorella fusca	1-generation cycle (24 hours) $EC_{50} = 377 \text{ mg/L}$	Faust et al. 1994
Chlorella pyrenoidosa, green algae	4 day $EC_{50} = 590 \text{ mg/L}$	Maule and Wright 1984
Chlorococcum hypnosporum, green algae	4 day $EC_{50} = 68 \text{ mg/L}$	
Zygnema cllindricum, green algae	4 day $EC_{50} = 88 \text{ mg/L}$	
Anabaena flosaquae, cyanobacter	4 day $EC_{50} = 304 \text{ mg/L}$	
Cyclotella meneghiana, green algae	73% inhibition at 2.8 mg/L	Peterson et al. 1994
Nitzschia sp., green algae	77% inhibition at 2.8 mg/L	[Inhibition of carbon fixation after 24 hours. Negative values indicate
Scenedesmus quadricauda, green algae	3% inhibition at 2.8 mg/L	stimulation.]
Selenastrum capricornutum, green algae	18% inhibition at 2.8 mg/L	
Microcystis aeruginosa, cyanobacter	-41% inhibition at 2.8 mg/L	
Microcystis aeruginosa, cyanobacter	16% inhibition at 2.8 mg/L	
Oscillatoria sp., cyanobacter	-12% inhibition at 2.8 mg/L	
Pseudoanabaena sp, cyanobacter	12% inhibition at 2.8 mg/L	
Anabaena inaequalis, cyanobacter	11% inhibition at 2.8 mg/L	
Aphanizomenon flos-aquae, cyanobacter	74% inhibition at 2.8 mg/L	
Lemna minor, duckweed	no inhibition at 2.8 mg/L over 5 days	

Appendix 11: Toxicity of glyphosate and	d glyphosate formulations to aquatic plan	nts
---	---	-----

Species	Endpoint	Reference
Scenedesmus quadricauda, green algae	growth stimulation at 0.02 mg/L; stimulation of photosynthesis at 0.2 mg/L, and stimulation of chlorophyll-a synthesis at 0.02 mg/L glyphosate; at 2 mg/L glyphosate; at $\geq$ 20 mg/L glyphosate, algal growth, photosynthesis, and chlorophyll-a synthesis was completely inhibited.	Wong 2000
Scenedesmus acutus	NOEC = 2 mg/L LOEC = 4 mg/L 96-hr EC <sub>50</sub> = 10.2 mg/L	Saenz et al. 1997
Scenedesmus quadricauda	NOEC = 3.2 mg/L LOEC = 4.08 mg/L 96-hr EC <sub>50</sub> = 9.08 mg/L	Saenz et al. 1997
Scenedesmus acutus	NOEC = 3.2 mg/L LOEC = 4.08 mg/L 96-hr EC <sub>50</sub> = 9.08 mg/L	Saenz et al. 1997
Scenedesmus acutus	NOEC = 1.25 mg/L LOEC = 2.5 mg/L 96-hr EC <sub>50</sub> = 9.09 mg/L	Saenz et al. 1997
	<u>ROUNDUP</u>	
Mixed colonies of periphytic aglae	4 hour $EC_{50}s = 35.4$ - 44.4 mg/L for inhibition of photosynthesis. NOEC = 0.89 mg/L.	Goldsborough and Brown 1988